National Library of Canada du Canada	e i i i i i i i i i i i i i i i i i i i
Canadian Theses Division Division des thèses ca	anadiennes
Ottawa, Canada K1A 0N4 49040	
PERMISSION TO MICROFILM - AUTO	RISATION DE MICROFILMER
• Please print or type — Écrire en lettres moulées ou dactylograp	hier
Full Name of Author — Nom complet de l'auteur	
LAURIE JANE M	IREAU
Date of Birth – Date de naissance $P \in C$ 7 – 1944	Country of Birth – Lieu de naissance CA N A D A
Permanent Address - Résidence fixe 3922 - 76 STREET EDMONTON, ALBERTA	
$\frac{1}{\text{Title of Thesis} - \text{Titre de la thèse}}$	
TEACHER EXPECTANC AND STUDENT AT	
University — Université	·
UNIVERSITY OF ALBERTA	- EDMONTON
Degree for which thesis was presented — Grade pour lequel cette t Ph. D. IN. EDUCATTONAL	
Year this degree conferred — Année d'obtention de ce grade	Name of Supervisor - Nom du directeur de these
FALL 1980	DR. AL. MACKAY
Permission is hereby granted to the NATIONAL LIBRARY OF CANADA to microfilm this thesis and to lend or sell copies of the film.	L'autorisation est, par la présente, accordée à la BIBLIOTHÈ- QUE NATIONALE DU CANADA de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film.
The author reserves other publication rights, and neither the thesis nor extensive extracts from it may be printed or otherwise reproduced without the author's written permission.	L'auteur se réserve les autres droits de publication; ni la thèse ni de longs extraits de cellé-ci ne doivent être imprimés ou autrement reproduits sans l'autorisation écrite de l'auteur.
Date (Signature
Sept 30 / 80	Laurie Direan

ζ.

Ì,

NL-91 (4/77)

National Library of Canada Collections Development Branch

Canadian Theses on Microfiche Service Bibliothèque nationale du Canada Direction du développement des collections

Service des thèses canadiennes sur microfiche

AVIS

The quality of this microfiche is heavily dependent upon the quality of the original thesis submitted for microfilming. Every effort has been made to ensure the highest quality of reproduction possible.

NOTICE

If pages are missing, contact the university which granted the degree.

Some pages may have indistinct print especially if the original pages were typed with a poor typewriter ribbon or if the university sent us a poor photocopy.

Previously copyrighted materials (journal articles, published tests, etc.) are not filmed.

Reproduction in full or in part of this film is governed by the Canadian Copyright Act, R.S.C. 1970, c. C-30. Please read the authorization forms which accompany this thesis.

THIS DISSERTATION HAS BEEN MICROFILMED EXACTLY AS RECEIVED

La qualité de cette microfiche dépend grandement de la qualité de la thèse soumise au microfilmage. Nous avons tout fait pour assurer une qualité supérieure de reproduction.

S'il manque des pages, veuillez communiquer avec l'université qui a conféré le grade.

La qualité d'impression de certaines pages peut laisser à désirer, surtout si les pages originales ont été dactylographiées à l'aide d'un ruban usé ou si l'université nous a fait parvenir une photocopie de mauvaise qualité.

Les documents qui font déjà l'objet d'un droit d'auteur (articles de revue, examens publiés, etc.) ne sont pas microfilmés.

La reproduction, même partielle, de ce microfilm est soumise à la Loi canadienne sur le droit d'auteur, SRC 1970, c. C-30. Veuillez prendre connaissance des formules d'autorisation qui accompagnent cette thèse.

LA THÈSE A ÉTÉ MICROFILMÉE TELLE QUE NOUS L'AVONS REÇUE

Ottawa, Canada K1A 0N4

FALL, 1980

EDMONTON, ALBERTA

DEPARTMENT OF EDUCATIONAL ADMINISTRATION

OF DOCTOR OF PHILOSOPHY

SUBMITTED TO THE FACULTY OF GRADUATE STUDIES AND RESEARCH. IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE

A THESIS



1

THE UNIVERSITY OF ALBERTA

TEACHER EXPECTANCY EFFECTS AND

by

LAURIE JANE MIREAU

THE UNIVERSITY OF ALBERTA

FACULTY OF GRADUATE STUDIES AND RESEARCH

The undersigned certify that they have read, and recommend to the Faculty of Graduate Studies and Research, for acceptance, a thesis entitled Teacher Expectancy Effects and Student Attributes submitted by Laurie Jane Mireau in partial fulfilment of the requirements for the degree of Doctor of Philosophy in Educational Administration.

Thesis Supervisor

Exter Examiner

Date 1980

ABSTRACT.

This study investigated the interaction of one grade three teacher and his 26 students using a case study methodology in conjunction with a variety of data collection techniques. The study was based on the assumption that the information, perceptions, and attitudes held by a teacher affect the nature and quality of interaction with students and of teacher behavior in general. The major purposes of the study were to examine the extent to which either process or product expectancy effects could be detected in the behavior of this teacher and to describe his teaching behaviors and the relationships existing between the teacher and his students.

Numerous data collection procedures were adopted in order to gain an understanding of the factors affecting instructional strategies employed by this teacher and how he coped with student-imposed demands during teaching. The attempt was made to triangulate data sources on teacher-exhibited behavior with teacher-expressed thoughts and with teacher verbal behavior.

Individualized student profiles were developed for all 26 student subjects. Verbal interaction received by the student in relation to the teacher's affective reaction to him and in relation to teacher-held expectancy for his achievement were examined in detail.

Self reports of the teacher's interactive thoughts about students during eight videotaped lessons were gathered using stimulated recall procedures. Additional introspective information about teaching performance and about students was obtained during regular interviews.

All process data were recorded over a three month period, March to June, 1979, during language arts and mathematics instruction. Subject area comparisons were made for the various factors under study and

iv

differential verbal treatment of individual students and of groups of students was noted. In the affective realm, the most verbal contacts occurred with "concern" students and "rejection" students and the least occurred with "indifference" students. "Attachment" students received more qualitative interaction in the form of higher level questions. The teacher had assigned a high, middle, or low expectancy rating to his students for the subject areas of language arts and math. Although high expectancy students in language arts received the highest percentage of total interactions and substantially more qualitative interactions, the reverse trend was in evidence during math lessons when more total interaction and more qualitative interaction was accorded to the lowest expectancy group. When all studente were regrouped into the two reading groups, total amounts of interaction were nearly equal but more qualitative interaction was received by the better reading group and more managerial and feedback contacts were directed toward the poorer reading group. The 13 boys were awarded a slightly higher average expectancy score in both subject areas and in fact received marginally higher academic results in the end of the year achievement tests when compared with all 13 girls.

Averaged scores tended to mask the individual differences to a great degree. Detailed examination of the 26 individualized profiles confirmed that differentiated treatment was provided for each child on the basis of the teacher's assessment of his relative strengths and weaknesses as a student. Traces of the expectancy effect were found, as evidenced in the verbal interaction exchanged with students, and the introspective information from the teacher helped to present the rationale behind the differential treatment of individual students and of groups of

v

students.

ACKNOWLEDGEMENTS

Numerous people assisted me in the preparation of this dissertation by providing the hecessary support and encouragement which was most sincerely appreciated. In particular V wish to thank

Dr. Al MacKay, thesis supervisor, whose good humor, advice, and guidance were unfailing during all stages of the project.

٦.

Dr. Myer Horowitz, a member of my examining committee, whose falued and perceptive comments helped to define and clarify the research in its various stages. Dr. Max van Manen, Dr. T.E. Kieren, and Dr. Murray Smith who as members of my examining committee offered constructive suggestions and moral support as the work progressed.

Dr. Naomi Hersom, of the University of British Columbia, who served as the external examiner and who provided insightful reactions to the completed dissertation.

The teacher and students in the pilot study school, and most particularly, the teacher and his 26 students who served as the subjects in the main study whose professional cooperation and generous assistance made the study itself possible. Assurance of anonymity precludes a more specific acknowledgement of their considerable contribution.

Chris Prokop and Ron Richmond who assisted with data preparation. and computer analysis.

All those associated with the Centre for Research in Teaching. whose comaraderie and shared ideas were a constant source of inspiration and encouragement.

My husband, Gregory, who so willingly did double duty on the homefront, and my young sons, Jamie and Robbie, who managed to cope with a somewhat preoccupied mother until her "report" was completed.

V1

TABLE OF CONTENTS

6

CHAPTER •	PAGE
I. INTRODUCTION AND STATEMENT OF THE PROBLEM	1
Introduction	1
Purpose of the Study	3
Statement of the Problem	5
Definition of Operational Terms	
Significance of the Study	8
Organization of the Study	
IT. REVIEW OF RELATED LITERATURE AND RESEARCH,	13
Recent Trends in Research in Teaching	13
The Case Study	19
The Case Study in Conjunction with Other Methodologies	20
Summary and Implications for the Present Study	27
Teacher Thought Processes	. 28
Stimulated Recall	. 34
Summary and Implications for the Present Study	35
Teacher Expectany Effect	. 36
Attribution Theory	. 36
L Summary	. 39
Teacher Expectancy	. 39
Formation and Maintenance of Expectancy for Students.	
Summary and Implications for the Present Study	
Teacher Attitudes to Students	
Summary and Implications for the Present Study	
Classroom Interaction	
Summary and Implications for the Present Study,	a de la composition d
	. 20

vii

CHAPTER	PAGE
Summary	56
III. DESIGN AND PROCEDURES	59
Overview	
Features of the Design	
stand sta	
The Subjects	
Instrumentation	
Dyadic Interaction Observation System (DIOS)	
Reliability for the Verbal Interaction Instrument	67
Six Rankings	. 68
The 13 Attribute Scales	
Teacher Attitude to Student Assignment	
Conners' Micro Content Analysis System	
Reliability of Coding Using Conners' Micro Cont	an a
Analysis System	71
High Inference Scales Measuring Teacher Behavior	. 73
Reliability for High Inference Scales Measuring	
	• 73
General Interviews and Stimulated Recall Interviews	
Specific Research Questions	
Research Question 1, Teacher Attitude to Students	• 75
Research Question 2, Expectancy Effects	• 76
Research Question 3, Student Profiles	• 77
Research Question 4, Interview Data	
Research Question 5, Teacher Behaviors	
Assumptions	
Limitations	80
Procedures	81
	81 ,
The Preparatory Phase	82

ņ

and the second

viii

IV.

The Pilot Stúdy	•••• 83
Data Collection Plan	84
> Lesson Context	84
Familiarization Period	•••• 85
Data Collection Procedures	•••• 86
Data Preparation and Analysis	
Verbal Data	
Attitude Groups	
Ranhings	e de la companya de l
Additional Data	
Interview Data	A 1
Correlations	92
Analysis of Variance	93
Summary	••• 93
TEACHER ATTITUDES TO STUDENTS	94
Initial Selection of Students into the Four Attitude to Student Groups	••• 94
Attachment Group	95
Indifference Group	••• 96
Concern Group	
Rejection Group	98
Final Selection of Students into the Four Attitude to Student Groups	99
Characteristics of the Groups	103
Corroboration of Teacher Affect toward the Four Attitude to Student Groups	104
Attachment	
ix	

PAGE

ŝ

CHAPTER	PAGE
Concern	105
Personality Attributes of the Four Attitude to Student Groups	109¥
Work Habits and Effort	110
Age and Maturity	113
Cooperation	115
Intelligence and I.Q	116
Creativity	119
Calm/Good Self Control	120
Noticeable	121
Eye Contact	123
Attractiveness	123
Нарру	124
Summary	125
V. EXPECTANCY EFFECTS	128
Product Expectancy Effects	128
Research Question 2.1	128
The Product Measures	128
Expectancy Measures	129
General Academic Performance Expectancy and Product Expectancy Effects	130
Language Arts Expectancy and Product Expectancy Effects	131
Reading Groups Expectancy and Product Expectancy Effects	132
Math Expectancy and Product Expectancy Effects	133
Attitude to Student Groups and Product Expectancy Effects	, 135
Summary of Product Expectancy Effects	139

۱

ι,

t

СНАРТ	TER	S PAGE
v.	Process Expectancy Effects	. 141
	Process Expectancy Measures	. ,145
	Research Question 2.2	. 147
	Sex Differences and the Process Expectancy Effect	. 148
	Attitude to Student Groups and the Process Expectancy Effect	. 149
R	General Academic Performance Expectancy and Process Expectancy Effects	. 151
	Language Arts Expectancy and Process Expectancy Effects	154
	Reading Groups and Process Expectancy Effects	157
nten 1944 - Anie 1944 - Anie	Math Expectancy and Process Expectancy Effects	160
*	Subject Area Interaction Differences	163
	A Contingency Model for Expectancy Effects	165
.0	Summary	170
VI.	THE STUDENT PROFILES	171
	Case #1, Sharon	178
	Case #2, Keith	184
	Case #3, Geoffrey	189
	Case #4, Edward	196
	Case #5, Lena	199
	Case #6, Ellen	203
	Case #7, Brenda	207
	Case #8, Anna	211
	Case #9, Emily	215
	Case #10, Brent	218
	Case #11, Simon	222
	Case #12, Marilyn	225 、

6

		DAOT
CHAPTE	승규가 중에 가지 않는 것 같아? 이 동안은 것 같아? 생활한 것 같아? 영화 문서 문	PAGĚ
	Саве #13, Grant,	•• 230
	Саве #14, Ian	•• 234
	Case #15, Tom	•• 241
	Case #16, Matthew	•• 246
	Case #17, John	•• 249
•	. Case #18, Bonnie	254
	Case #19, Nicholas	262
	Case #20, Mason	268
	Case #21, Trisha	273
	Case #22, Merrill	. 277
\sim	Case #23, Pamela	281
	Case #24, Maureen	. 288
	Case #25, Joanne	
	Case #26, Sonia	
	Summary	
17 7 7	EACHER THOUGHT PROCESSES	312
VII. T		
	Micro-Analysis of Interactive Thoughts	
	Teacher Perceptions	
	Academic Performance	• 316
	Student Verbal Behavior	• 316
	Student Movement	. 316.
	Noise/Unacceptable Behavior	. 317
	Student Expression	
4. j.	Equipment and Materials	. 318
6	Student Presence or Absence	. 318
	Miscellaneous	. 318

•

IAPTER /		
VII. · · · · · · · · · · · · · · · · · ·	TAGE)	۰. ۲
Instructional Moves	318	
Control/Discipline	319	, •
Greedback-Teacher	•• 319	t
Structuring	320 🔽	1
Involving		•
Organizing	· · ·	
•	c	
Feedback-Pupil	· .	·,
Reinforcing	•	
Reviewing	. 322	
Presenting	• 322	•
Repetition	. 322	
Motivating,	. 323	•
Transfer of Learning	. 323	с., у
Learning Process		
Miscellaneous	. 324	
Interpretations	. .	
Student Attention and Motivation		
Student Cognitive Processing.		
Feelings	. 326	
Academic Performance	. 326	
Student Movement	. 326	4
Verbal Behavior and Noise	. 327.	.*
Materials/Content	. 327	
Miscellaneous	. 327	
Information-Pupil		الأهر ا
	e	
Academic Performancexiii	. 328	r

СНЛ	APTER	PAGE
v	II. Classroom Behavior	320
	Personality	
	Academic Ability	
	Social Behavior	
	Home Background	
·	Physical Characteristics	330
	General Information	
	Self Awareness	
	Instructional Evaluation	
	Instructional Reflection	
	Instructional Awareness	
	Personal Reflection	333
	Content Reflection	34
	Affective Dissonance 3	34
	Feelings	34
•	Beliefs	35
	General Pedagogical Beliefs	35
	Developmental Beliefs 33	» 36
-	General Beliefs Concerning Learning Principles 33	36 ·
	Beliefs Concerning Specific Learning Principles 33	16
	Beliefs Concerning Psychological Principles 33	7
	Beliefs Concerning Memory 33	
	Meditation Pupil 33	7
o	Lesson Content	B, i
	Objectives	3
-	General Objectives)

ə .

xiv

.

Ą

VII.	Lesson Specific Objectives	339
	Expectations	339
	Task Performance Expectations	339
	Cognitive Performance Expectations	340
	Affective Expectations	340
	Information-Other	340
•	Summary of the Micro-Analysis of Interactive Thoughts	341
	Macro-Analysis of Interview Data	344
VIII.	TEACHER BEHAVIOR	
	High Inference Variables	
	Teacher Verbal Behavior	
	Verbal Interaction Data	:
	Sinit Interactions	
	Tinit Interactions	
,	Tinit Types	371
	Effects of Grouping on Product and Process Questions	374
	Tinit Comments	379
	Evaluative Tinits	379
	Summary of Verbal Behavior	382
	Summary	383
IX.	SUMMARY, CONCLUSIONS, IMPLICATIONS, AND RECOMMENDATIONS FOR FURTHER RESEARCH	384
	Summary of the Study	384
•	Purpose of the Study	384
	Methodology	384
	Analysis	386

IX.

۰,

8

X.	
Conclusions	386
Research Question 1.0	387
Research Question 1.1	388
Research Question 2.1	389
Research Question 2.2	391
Research Question 3.0	392
Research Question 4.0 and 4.1	392
Research Question 4.3	393
Research Question 4.2	393
Research Question 5.0 and 5 1	394
Research Question 5.2	395
Implications and Recommendations	395
Approaches to Research Methodology	395
Focus on the Student	396
Thought Processes	397
Classroom Interaction	397
Verbal Interaction	398
Expectancy Research	399
Teacher Attitudes to Students	399
Effects of Participating in Introspective Research	399
Reflections on Using an Eclectic Approach	400
BIBLIOGRAPHY	401
APPENDIX	412

PAGE

xvi

ç.

σ

LIST OF TABLES

TABLE		PAGE
1	Attitude Group Scores on the Attachment Attribute	105
2	Attitude Group Scores on the Concern Attribute	106
3	Attitude Group Scores on Work Habits and Effort	110
4	Attachment Group Sex Differences for Work Habits and Effort	111
5	Indifference Group Sex Differences for Work Habits and Effort	112
6	Attitude Group Scores on Combined Work Habits and Effort Measures.	112
7	Attitude Group Scores for Age and Maturity	114
8	Attitude Group Scores for Cooperation	116
9	Attitude Group Scores on Intellectual Ability and I.Q. Measures	11°7
10	Sex Differences in Attitude Groups for General Intellectual Ability and Total I.Q. Scores	118
11	Attitude Group Scores for Creativity and Non-Verbal I.Q. Scores	。 119
12	Attitude Group Scores on the Calm Dimension	121
13	Attitude Group Scores for the Noticeable Attribute	122
14	Attitude Group Scores for the Eye Contact Attribute	123
15	Attitude Group Scores on the Attractiveness Attribute	124
16	Attitude Group Scores on the Happy Attribute	125 ·
17 .	Probable Highest Achiever Groups and Product Scores	130
18	Language Arts Groups and Product Scores	131
19	Reading Groups' Expectancy Scores and Achievement Scores	132

xvii

TABLE		PAGE
20	Math Achievement Scores of Math Expectancy Groups	134
21	Attitude Groups' Language Arts Expectancy and Achievement Scores	136
22	Attitude Groups' Math Expectancy and Achievement Scores	138
23	Sex Differences and the Process Expectancy Effect	148
24	Attitude Groups and the Amount and Kind of Verbal Interaction Received	149
25	General Achievement Expectancy and Process Expectancy Effects	152
26	Language Arts Expectancy and Process Expectancy Effects	154
27	Language Arts Expectancy and Process Expectancy Effects	155
28	Reading Groups and Process Expectancy Effects	158
29	Math Expectancy Groups and Process Expectancy Effects	161
30	Math Expectancy Groups and Process Expectancy Effects	162
31	Expectancy Group Scores for Language Arts and Mathematics	164
32	The Six Ranking Scales	172
33	The Thirteen Attribute Scales	173,
34	Dyadic Verbal Interaction Variables	174
35	Percentages of Types of Verbal Interaction	371
36	Differences Among High, Middle, and Low Expectancy Tinit Ratios	372
37	Percentages of Total Interaction	373

•

2

p inter

xviii

TABLE

. 38	Language Arts Interaction Ratios and Achievement for "Good" and "Poor" Reading Groups
39	Praise and Criticism Received by Expectancy Groups 374
40	Praise and Criticism Received by Attitude Groups

xix

LIST OF FIGURES

ŕ20

FIGURE ,

1

Contingency Model	of Expectancy	Effects.	• • • • • • • •	••••	.167
Subject Area Conti	Dency Modele	of		U.	
Expectancy Effects			•••••	• • • • •	.168
Summary of Categor	ies in Conners	' Micro	a fuatta construit. Maria de la construit	an ar agairtí a	

PAGE

CHAPTER I

INTRODUCTION AND STATEMENT OF THE PROBLEM

INTRODUCTION

Classroom-based research often generates more questions than it answers. This study addressed some of the questions that continue to intrique educational researchers and to guide their research efforts.

With a growing interest in the interactive nature of the teachinglearning process it is no longer sufficient to focus exclusively onidentifying effective teaching strategies, even though it is widely accepted that "teachers make a difference" (McDonald, 1976). Therefore, in conjunction with curriculum concerns and teaching behaviors, the pupil's perspective (King, 1979) must be considered in future research studies. Research based on the analysis of group scores, while valuable, needs to be complemented by in-depth studies of observed behaviors of individual students (as in Smyth's, 1979, examination of four students' use of academic learning time) and of various groupings of students. Brophy et al. (1976:1) claimed that statements about students in general have very limited meaning and Eggert (1977:8) concurred that since it is the individual who does the learning, the student should be the unit of analysis in classroom-based studies about teachers and students. Viewing the student in isolation and as a member of various groups in the classroom will help to provide this needed information.

The ways' in which a teacher perceives individual students and subsequently makes judgments about them determines in large measure the pedagogical strategies selected for them. Studies have indicated that the teacher's perceptions of individual students do lead to differential

treatment of students as evidenced in verbal interaction patterns (Brophy and Good, 1974).

The recognition that teachers are individuals with beliefs about teaching which may influence their teaching behavior has led to the investigation of the mental processes that underlie teaching performance. It is believed that how teachers think will determine how they will act. If such a direct connection exists then investigating teacher thought processes and the perceptions that govern teaching behaviors would seem to be a worthwhile pursuit. Thus a focus on teacher thought processes is considered to yield a potentially rich source of data. However, it may prove to be too individualized a process which may not answer the broader questions or help to explain the instructional process in a generalizable way.

The National Institute of Education's Panel 6 Report (1975:1) concerning investigations of the teacher's "mental life", such as the information about students which is processed covertly, indicated that students do influence teacher behavior and decision-making. The realization that a classroom of students is comprised of individual students who hold differing attitudes toward school and who possess unique attributes and levels of ability has led to more serious consideration of the individual student in recent research studies. Brophy and Good (1974:3) claimed that relatively little educational research had focussed on the individual student: "We believe that this is a major reason why educational research has contributed relatively little knowledge that teachers can apply in classrooms."

The two-way interaction which occurs in the classroom is as much due to student effects on teachers as to the better known teacher effects on students . Researchers (Brophy, Evertson, Anderson, Baum, and Crawford, 1976) tried to ascertain to what extent certain personality types and certain student behaviors predictably influence teacher behavior toward a specific student and, in turn, whether the resultant (fairly predictable) teacher treatment of this particular student influences his individual academic achievement. Hence it might become possible to estimate a student's achievement level simply by describing his salient personality characteristics. Predictions of great accuracy are unlikely but general trends, true of certain types of students, may enable future researchers to develop useful guidelines for effective teacher interaction with specific types of students.

Teachers act on the basis of the perceptions they hold about students. The degree to which they are aware of these tendencies determines the extent of influence these perceptions will have . Also the degree of accuracy of these perceptions and the extent of flexibility exercised in their maintenance affects the suitability of teacher treatment of students.

PURPOSE OF THE STUDY

In the attempt to investigate the "bidirectionality" (Fiedler, 1975:735) of the instructional process and the effects of student characteristics and behavior upon a teacher, the present study examined the ongoing instruction in one classroom over a three month period.

A major focus of the study was an exploration of the teacher's cognitive processes such as the ways in which he had formed expectancies for student performance (Finn, 1972) and his prescriptions, diagnostic judgment and decision making (Shavelson, 1973, 1977) on the basis of his revealed perception of the individual students in his class. The data collected took the form of teacher-provided background information about pupils, teacher rankings of all pupils, and ratings using numerical scores on such varied measures as personality characteristics, usual work habits, projected achievement scales, and through teacher-expressed attitudes and feelings toward the students which were gathered using general interviews and stimulated recall procedures. It was anticipated that an examination of this teacher-provided information might help to explain how and why this teacher's behavior toward his students had become differentiated, if indeed it <u>had</u>, and might help to describe the links between achievement expectancy held for students, teacher treatment of students, and student achievement. In other words, in what ways did this teacher's expressed perceptions about his students appear to influence his treatment of them and, in turn, their achievement.

Dyadic verbal interaction data were recorded. The amount and kind of verbal interaction occurring between a teacher and individual students within the context of classroom instruction have been found to vary considerably in other studies (Dusek, 1975 and Firestone and Broudy 1975). The differences that occurred were analyzed and teacher-provided data about the individual student were used in the attempt to describe how differential verbal treatment was effected.

Various data sources for the teacher's introspective thoughts and perceptions about students were complemented by other data sources in the form of dyadic verbal interaction data, end of the year wachievement data, and data on the overt teaching behaviors exhibited during classroom instruction.

The information is presented in narrative, descriptive form reporting teacher-expressed thoughts and perceptions about students. The results of analyses of scores received by the different groupings of students and by each individual student are presented and discussed. 5

STATEMENT OF THE PROBLEM

The major overarching question addressed in this study was:

To what extent is it possible to explain how or why differential treatment of students occurs by examining the teacher's perceptions of his students?

The study was further guided by the following, more specific,

research questions:

- 1.0 What are the common characteristics of students assigned by the teacher to each of four attitude-to-student groups (attachment, concern, indifference, and rejection)?
 - 1.1 What were the perceived personality characteristics which might serve to explain the teacher-expressed attitudes and affective feelings toward these groups of students?
- 2.0 What evidence is there, if any, of expectancy effects?
 - 2.1 What evidence is there, if any, of product expectancy effects? What is the relationship of teacher-held expectancy to end of the year achievement results?
 - 2.11 for students assigned to high, middle, and low "probable highest achiever" levels (for general achievement in all subjects)?
 - 2.12 for students assigned to high, middle, and low "probable achievement in language arts" levels?
 - 2.13 for "good" and "poor" reading groups?
 - 2.14 for students assigned to high, middle, and low "probable achievement in mathematics" levels?
 - 2.15 for boys and for girls?
 - 2.16 for the four attitude-to-student groups?

- 2.2 What evidence is there, if any, of process expectancy effects? What is the relationship of teacher-expressed expectations to the amount and kind of verbal interaction exchanged?
 - 2.21 for students assigned to high, middle, and low "probable highest achiever" levels (for general achievement in all subjects)?
 - 2.22 for students assigned to high, middle, and low "probable achievement in language arts" levels?
 - 2.23 for "good" and "poor" reading groups?
 - 2.24 for students assigned to high, middle, and low "probable achievement in math" levels"?
 - 2.25 for boys and for girls?
 - 2.26 for the four attitude-to-student groups?

3.0 Using all available information, to what extent is it possible to construct a comprehensive student profile which would describe the student's behavior, his achievement, his membership in various grouping patterns, and the relationship he had with this teacher?

- 4.0 What information is obtainable from the interviews held with the teacher?
 - 4.1 Which kinds of thoughts about students in particular were reported during stimulated recall interviews?
 - 4.2 What teacher attitudes and priorities are revealed in all interview data?
 - 4.3 To what extent do teacher-reported thoughts clarify/corroborate numerical scores assigned on the different student ratings?
- 5.0 How could the teaching behavior of the teacher subject be described?
 - 5.1 What are the predominant teaching behaviors of this teacher?
 - 5.2 What are the predominant verbal interaction behaviors of this teacher?

DEFINITION OF OPERATIONAL TERMS

<u>Expectations</u>: are those inferences and predictions that teachers make about the present and future academic achievement and general classroom behavior of students (Brophy and Good, 1974:32). These expectations are closely tied to academic achievement and refer to the teacher-projected estimates of student behavior and achievement.

<u>Attitude-to-Student</u>: are affective responses of teachers to students (Crano and Mellon, 1978:47). These are closely related to personal qualities of the student and his reactions to the teacher. These teacher <u>attitudes</u> correlate with differential teacher behavior (Willis and Brophy, 1974:529).

<u>Pupil Attributes</u>: are personality characteristics and qualities which are used by the teacher to describe the pupil and they form the basis on which teachers develop attitudes and expectations.

<u>Self-Fulfilling Prophecy</u>: is an expectation or prediction, which may be false initially, which initiates a series of events that cause the original expectation to become true (Brophy and Good, 1974:35).

Dyadic Verbal Interaction: refers to the verbal interaction exchanged by the teacher and an individual statement in an instructional setting.

<u>Expectation Effects</u>: refer to the communication of teacher expectancy to children. When the teacher's behavior does this, expectation effects are said to be in operation (Muttart, 1977:7). These expectations could be communicated directly (told to the student) or indirectly by the amount and quality of attention given to individual students by the teacher. Interactive Thoughts and Decisions: are those thoughts and decisions which occur only within the stream of conscious thoughts the teacher is having during actual instruction.

Stimulated Recall: is a branch of introspective methodology in which audio and/or visual records of a subject's past behavior are used to facilitate that subject's recall of the covert mental activity which was occurring simultaneously with the recorded overt behavior (Marland, 1977:11). It is a method of obtaining from the subject a retrospective account of his covert mental activity which co-occurred with the subject's overt behavior.

Content Analysis: is a technique for making inferences by systematically and objectively identifying specified characteristics of messages (Holsti, 1969:14). The technique involves segmentation and coding' of the qualitative (written transcript) data thus transforming it so it becomes susceptible to quantitative treatment and permits precise description of relevant content characteristics.

SIGNIFICANCE OF THE STUDY

If the routine events that occur in elementary classrooms are to be fully comprehended, the day-to-day activity of teachers and pupils must become the focus of serious and intensive investigation in future

research studies in education. This study attempted to examine and describe such events and interpersonal relationships that operate in the learning environment by using a phenomenological, case study perspective.

The teacher processes countless pieces of information during the teaching day. It appears that certain kinds of information (for example, perceptions of student ability or student attention) are particularly salient for interactive thinking and decision making in which teachers are engaged. By investigating this teacher's interactive thought processes using stimulated recall methodology and general interviewing techniques, this study adds to the findings on interactive thoughts and decisions by narrowing the focus to those particular thoughts which occur as a reaction to perceived student behavior and attributes during eight instructional periods chosen for examination.

Although general interviewing techniques can reveal global teacher thoughts and beliefs as well as specific thoughts about students, an insightful and promising method of obtaining the teacher's interactive thoughts is to present the teacher with issue-specific stimuli from a videotaped lesson in progress and thereby elicit the covert mental thoughts which were associated with these specific incidents. This study adds to the research on stimulated recall as a technique for obtaining such information and also adds to the findings concerning the kinds of interactive thoughts about students and other instructional concerns that the teacher processes during the teaching of the lesson.

Although extensive research has been conducted on the identification of teacher expectation effects, the investigation of the process by which

the effect operates in the classroom is a relatively recent research area. Acting on the basis of our expectations and affect is a necessary and unavoidable human tendency. The actual effect of holding performance expectations in the classroom will not be in operation unless the teacher communicates these expectations and affective feelings to students who become influenced by them reciprocally. Differential teaching behaviors, as evidenced in verbal interaction patterns, can communicate this information to students as can assignment to different grouping patterns such as reading groups. The researcher has obtained one teacher's expectations for and attitudinal feelings about the students in the class and has gathered extensive verbal interaction data in the attempt to identify and describe any relationships which seem to indicate an expectancy effect is in operation. Thus the study helps to extend the findings in expectancy research.

The overt behaviors of teachers are subject to potential misinterpretation if classroom research is limited to observational measures strictly. By delving into the mental life of teachers and their perceptions of students, researchers may begin to explain some of the reasoning behind the differential treatment of students exhibited by teachers. This study has combined observable behavioral data on exhibited teacher behaviors with a complementary, introspective investigation of the teacher's thought processes during instructional time and of teacher-held expectations and affective feelings concerning students in the class. One of the main contributions of this study is the triangulation or corroboration of varying sources of data about the teacher (thoughts, classroom behavior, verbal interaction) and the resultant description of selected aspects of classroom life during the three month duration of the study.

A final contribution of this study is the recommendation that students must be viewed not as a "class" but as individuals in future research studies. It is this focus on the child that will result in specific information about teacher-student relationships which collectively may provide the factual basis necessary for more generalizable theory on teacher-student relationships.

ORGANIZATION OF THE STUDY

This chapter has outlined the main purposes of the study, the research problems, definition of terms, and significance of the study.

In Chapter II, The Review of Related Literature and Research, the first section presents the background of research in teaching from which the present study is drawn. Strengths and weaknesses of case study methodology are reviewed next followed by the third section on teacher thought processes. Research on the teacher expectancy effect is then discussed as well as a survey of research conducted on teacher attitudes to students. The final section of the chapter reviews studies concerning interaction in the classroom and ends with a brief summary of the various research areas réviewed.

In Chapter III the design and procedures used in this study are described and the subjects, instrumentation, phases in the study, data preparation and data analysis are presented.

Chapter IV describes how the teacher assigned the student subjects to one of four attitude-to-student groups. The general characteristics of each of the four attitude groups are presented and compared to findings from previous research on teacher attitudes to students. Chapter V presents the data obtained on teacher-assigned expectancy measures for the students. Findings concerning the extent to which product expectancy effects and process expectancy effects seemed to be in operation in this classroom are presented.

Chapter VI presents all 26 student profiles as brief case studies in themselves. Each profile was created by using teacher-provided data on the student's personality characteristics, work habits, and ability which were related to achievement, I.Q., and dyadic verbal interaction exchanged between the particular student and the teacher.

In Chapter VII, concerning teacher thought processes, the results of a micro content analysis of the stimulated recall transcripts determined the substance of his expressed thoughts about students during teaching. The results of a macro-analysis of all interview data which indicate the priorities and attitudes held by the teacher are presented.

Ç.,

Chapter VIII presents findings concerning the teacher's behavior. Findings were derived from both the high inference measures of overt teaching strategies used and from the low inference measures of dyadic verbal interaction behavior in the attempt to describe the predominant aspects of the teacher's behavior.

In Chapter IX, a summary and discussion of the findings as they relate to the research questions are presented. Recommendations for future research are suggested in this final chapter as well as implications for teaching, teacher education, and educational administration.

CHAPTER II

REVIEW OF RELATED LITERATURE AND RESEARCH

In this review of related literature and research, the pertinent , antecedent research areas from which the present study evolved are identified. These seemingly disparate areas have made both theoretical and methodological contributions which have been incorporated into the research design of the present study.

This chapter focuses on recent trends in research in teaching with particular attention given to the following areas: the case, study method in conjunction with other methodologies; teacher thought processes; the expectancy effect; teacher attitude to student; and the reciprocal influences of teacher and students in the classroom.

RECENT TRENDS IN RESEARCH IN TEACHING

In reviewing studies conducted prior to the early 1970's, Rosenshine and Furst (1971 and 1973) and Dunkin and Biddle (1974) concluded that educational research findings to that date had been "tentative" at best. In future they recommended a concentration on the identification and investigation of mediating processes which were operating in classrooms and on the development and testing of more suitable observation and measurement techniques.

Active research in teaching in the years following generated a plethora of classroom observation instruments designed to record and measure specific classroom behaviors. If the collection presented in Simon and Boyer's (1974) <u>Mirrors for Behavior III</u> is any indication, a great proliferation of such classroom observation instrumentation abounds.

A second result of large scale experimental and correlational studies in teaching was the identification of specific teaching behaviors which appeared to correlate with gains in student achievement (Gage, 1978 and Stallings, 1978). The temptation has been to infer causality when high correlations were discovered. Exhaustive prescriptive lists of effective teaching behaviors required for competency were developed. However, the predominant emphasis had been on the teacher's side of the instructional process based on McDonald's (1976) assertion that "teachers do make a difference." The mediating processes and the "presumed links" (Magoon, 1977:653) accounting for effective teaching remained unexplained.

Ecological variables such as class size, arrangement of furniture (Adams and Biddle, 1970) exert influences on ongoing instruction but do not, in themselves, account for success or failure on achievement tests. Time of day, week, and year influence both the quality of instruction provided by teachers and the receptivity of students. Classroom "climate" had been investigated as early as 1949 by Withall. Dunkin and Biddle (1974), Marliave (1976), and Good and Grouws (1977) agreed that a "warm" climate with a relaxed atmosphere was one sign of an effective teacher. All of the foregoing environmental factors and many more exert influences that affect ongoing instruction in classrooms.

Several other research emphases have yielded promising results and are briefly outlined. Studies of teacher thought processes have provided valuable information about how teachers plan (Zahoric, 1975, and Oberg, 1975), how they make decisions (Shavelson, 1973-1978, and Wodlinger, in progress), how they think and react interactively while

teaching (Marland, 1977; Conners, 1978; Tuckwell, 1979), and how they think and feel about their students (Brophy, Evertson, Anderson, Baum, and Crawford; 1976 and King, 1979) and about the constraints such as time, curriculum concerns, school organizational features which exert specific effects on classroom instruction.

Interest in achievement motivation (Weiner, 1972-1978 and Bar-Tal, 1978), personality attributes of students (Brophy and Good, 1974 and Brophy et al, 1976), pupils' use of academic engaged time (Marliave, Fisher, and Dishaw, 1977 and Smyth, 1979), pupils' classroom behavior styles (Spaulding, 1973 and Mahen, 1977), and pupils' interactive thoughts during instruction (King, 1979), for example, have begun to shift the focus from the teacher onto the learner and have resulted in promising findings which highlight the pupil's active role in attaining his own success.

Teachers and students are involved in a unique sort of social interaction. Ostensibly the teacher is the controlling force but students do exert an influence on interaction patterns by virtue of their personal characteristics and behavior (Brophy et al, 1976). The recognition that it is neither teacher variables nor student variables that completely determine student achievement but rather the interaction between the two has prompted investigation of this mutual interdependence in classroom settings (Winne and Marx, 1977:669; Brophy and Good, 1974; and Fiedler, 1975:736).

Educational researchers have begun to undertake intensified study of these variables in the context of ongoing classroom life where they can be viewed in routine interaction with equally important curriculum concerns and teacher behavior variables.
A growing realization that naturalistic classroom observation was the most appropriate direction for the study of teaching was stated by Brophy and Good (1974, Preface ix):

> Data have convinced us that the complexities of classroom life can be profitably investigated only in real-world settings. The simplicity and control achievable in simulation and laboratory settings come at too high a price: lack of generalizability to the classroom.

However, even with ample instrumentation and much information about classroom life, definitive explanations and conceptualizations of the teaching-learning process remain elusive and unformulated. Winne and Marx (1977:671) felt that a different approach is needed. Such a reconceptualization of educational research will necessitate the development of different research techniques, revamping of previous ones, or a recombination of ones used successfully in the past. New approaches and methodologies that are being developed make the researcher's job more difficult, more complex, and more interpretive. A composite and eclectic set of skills is needed by today's classroom researcher.

The varying approaches to educational research need not be regarded as mutually exclusive as has been the tendency in the past. In fact, it is by adopting and using different methodologies in inventive combinations that complementarity and triangulation will result and may help to bring about a better understanding of life in classrooms.

MacKay and Osoba (1978:2) reviewed the three major thrusts in educational research currently being pursued; the experimental, the correlational, and the ethnographic. They suggested that researchers are reaching consensus that educational research is still in an "evolutionary" state and that a wide variety of data collection devices and methods is called for in order to develop the broader base needed for conceptualization and understanding of the teaching-learning process.

Wilson (1977:255) noted that "multi-modal" data collection techniques are commonly used in ethnography and he was convinced that "educational research will be considerably enriched as qualitative and quantitative researchers learn to integrate their approaches."

Perhaps the ideal type of classroom-based research study would involve the collaborative efforts and talents of several researchers based in one classroom over a long period of time at great expense of time, effort, and money and using a variety of observation techniques and recording devices. However, such an intrusive, but comprehensive approach would almost surely destroy the forces operating in a naturalistic setting.

The foregoing concerns might be summarized in the following way: Many valuable contributions have been made in research in teaching to date using methodologies which can only achieve a limited amount on their own. Researchers are only just developing an awareness of the overwhelming complexity of the elements at work in a classroom, of the various facets of teacher behavior and student behavior, and of the interaction between them. These forces are interrelated and extremely complex. It no longer seems fruitful or appropriate to conduct large scale studies when we lack a satisfactory theoretical base as an underpinning for the research. It is time to take stock and assess the positive contributions to date and to concentrate on the smaller units of analysis such as one class or one student. It is, in short, a time to revert to "fieldwork as a basis for theory building in research on teaching" (Yinger, 1978).

Valuable contributions in other fields of study have been made because of the singular and intensive study of one cell, bird, animal, or human which in turn led to hypothesis generating and experimentation involving larger numbers of subjects. Such an approach has been avoided in the past in educational research perhaps because of preoccupation with and faith in alternate forms of research, perhaps because it was not considered "respectable" and did not yield generalizable findings or more likely because educational researchers then, as now, were unsure of appropriate methodology, data gathering, and data analysis techniques.

Yinger (1978:8) proposed two reasons for our hesitancy to build theory in educational research. A "fear of the unknown" accounts for

part and the belief that only the "gifted few" should attempt and the belief that only the "gifted few" should attempt to generate cory prevents most of us from taking any responsibility of this In addition Yinger felt that the role of theory in science is uncluic to educational researchers for it not, as yet, seen as an ongoing tess while active research is being conducted but is still viewed as "product" that is developed after years of research and deliberation.

> Good theory in education, as in art, helps us see more. It helps us think about more of the qualities that constitute a set of phenomena; theory does not replace intelligence and perception and action, it provides some of the windows through which intelligence can look out into the world.

> > (Eisner, 1976:7)

Certainly the need to establish a firm base on which to build theory seems to be a pervasive priority with researchers in education.

> The preoccupation with data gathering has been recently reinforced in research on teaching. As the anthropologists, sociologists, and ecological psychologists have gradually convinced us of the complexity of the teaching endeavor, we have realized the necessity for a lot of work in describing and analyzing the classroom and its activities. Any discomfort that we may have in venturing out of the lab and into the field has been somewhat moderated by a feeling that all we need to do for a while is to merely describe what we see. (Yinger, 1978:9)

While this may be so, Yinger cautioned that after years of compiling "rich" descriptions we might become "thick" on description but "thin" on analysis and theorizing.

Yinger (1978:13) continued by emphasizing that the need is now recognized that in order

for conceptual categories and hypotheses that have meaning for real situations to emerge, they must be based on qualitative descriptions of behavior in natural situations ... our central concern should be that our theorizing is grounded in situations that are representative of real settings in which teaching and learning occur. Better yet, theorizing should be grounded in actual teaching/learning situations.

These descriptions do not preclude the use of other types of data collection which can be used to great advantage in conjunction with qualitative description. The case study would seem to be an appropriate approach for conducting such a study.

THE CASE STUDY

MacDonald and Walker (1975:3) described the case study method as a branch of exploratory, ethnographic research. Methods are rarely specifically stated in advance but the goals should be to increase the understanding of the variables, parameters, and dynamics of

the case under study.

Denton (1979:11) pointed out there is an important distinction between providing an "explanation" which attempts to answer the "why" questions in research, and facilitating the reader's "understanding" by presenting a context for interpreting the phenomenon under study.

Stake (1978:7) claimed that the case study is an appropriate method for educational research for in education few laws have been validated. Denton (1979:11) appeared to agree when he stated that in educational research there are few, if any, general laws which are genuinely "causal". Rather, an 'if-then' conditional form of reasoning seemed to be involved.

Stake (1976:129) cautioned against adopting a preordinate stance which depends on the capability of prestating the purposes of education concisely. By adopting such an approach the researcher cannot remain "sensitive" to ongoing changes and uniqueness; hence his bid for "responsive evaluation". Wilson (1977:255) concurred that the adoption of a particular framework is arbitrary for interpreting and coding behavior. Wilson went on to say that the case study researcher is constantly making strategic decisions as the study progresses and the theories that emerge must be used to direct subsequent data collection.

THE CASE STUDY METHOD IN CONJUNCTION WITH OTHER METHODOLOGIES

Wilson explained that the ethnographic and phenomenological methodologies often used in case studies come from the anthropological tradition. The (sometimes participant) observer collects information about "human behavior impossible to obtain by more quantitative methods (p.254)." The guiding principle of ethnographic and phenomenological research is the assumption that individuals' "meaning structures" determine much of their behavior. van Manen (1978:3) described phenomenology as "having as its primary objective the direct investigation and description of phenomena as consciously (and pretheoretically) experienced." The investigator engaged in phenomenological research is simply obliged to describe in "life world" language without making causal interpretations.

Denton (1979:6) claimed that phenomenological research is atheoretical in the beginning. He described the key concept of "bracketing" as the basic research tool:

> With bracketing the researcher is able to see and describe a phenomenon from a variety of perspectives and to interpret it against many backgrounds (or horizons) and to locate it in some particular conceptual field. The procedure itself is simple enough; one puts aside accustomed perceptual sets in order for the phenomenon to appear as itself...

> Bracketing is also the method for multiplying perspectives on the phenomenon in question, observing from as many angles as possible, placing against a variety of backgrounds, locating in different histories and futures, listening and participating with the people involved.

Wilson (p.251) recommended such "bracketing" until one's experience with the research begins to suggest the relevance of one's own thoughts. Denton (p.9) maintained it was not a question of objectivity versus subjectivity for the objectivity claimed is that which results from multiple perspectives.

Some of the events under study will be "patterned, recurring, and frequent", Denton claimed. As well, quantification is not prohibited for many "event structures" can be enumerated, classified, and measured. For example, the descriptive account itself can be subjected to quantitative analysis after a content analysis has been completed. While recognizing the limitations of this style of research in education Denton felt it was a potentially valuable direction for educational research.

> Phenomenological research will not provide propositions supposedly having implications for the teacher and the administrator. It will, however, provide a many-layered narrative against which the teacher and the administrator can draw parallels with their worlds of practice. (Denton, 1979:14)

The onus of drawing such parallels is placed squarely on the consumer of research. Kennedy (1978:16) agreed that this responsibility is indeed the reader's. A satisfactory set of rules for drawing inferences from single-case study findings does not exist at present. She argued that although they have been precisely stated and widely accepted for decades, statistically-based inferences are themselves still "tentative", never "conclusive" for "the strength of evidence is a matter of judgment."

Kennedy pointed out that the "advantage of the single-case methodology is that it forces the evaluator to look at the functional relationships between the treatment and the subject(s)(p.14)" and she emphasized that the "relationship" is what is generalized.

This generalization is not apt to be from a case to a population but rather from a case to another case. This procedure, she reminded the reader, is well respected in both the medical and legal professions.

Much medical "knowledge develops from the accumulation of findings about treatment effects in individuals (p.21)". Similarly, in law, the court decides whether a particular decision generalizes to the case being considered. "Thus it is the receiver of the information who determines the applicability of a finding to a new situation (p.17)". The duty of the researcher/evaluator is simply to produce and share the information.

Eisner appears to agree in part but assigned more responsibility for the interpretation and evaluation of findings to the researcher. Eisner (1976, 1977) maintained that the educational researcher must develop and use the skills of both the "educational connoisseur" and the "educational critic". The primary aim of educational connoisseurship is to develop the art of "appreciation" which Eisner (1977:8) described as the "awareness and understanding" of both the "character and quality" of what one has encountered. It is the development of educational connoisseurship, as the art of perception, that would make the appreciation of complexity, such as that found in today's classroom, possible. The development of such a heightened awareness provides the basis for judgment.

Eisner contended that one could be an accomplished educational connoisseur without going on to become a competent educational critic but the reverse could not be true for educational connoisseurship provides educational criticism with its subject matter. In the sense Eisner uses it, "educational criticism" is not concerned with negative evaluation but rather has as its primary aim the re-education of perception and is the "art of disclosure". The duty of the educational

critic is to <u>reveal</u> the character and quality of what one has encountered. An educational critic may be merely <u>descriptive</u> "in the effort to characterize or render the pervasive and sheerly descriptive aspects of the phenomena(p. 10)". By attempting to be <u>interpretative</u> the educational critic tries to reveal the meaning and significance that various forms of action have for those in a social setting. Eisner maintained that in order to interpret effectively the educational critic must "empathically participate in the life of another (p.14)". This endeavor required the "judicious and informed use of a variety of social sciences and the practical wisdom form of experience in schools (1976:14)". The educational critic who became <u>evaluative</u> would be making value judgments about teaching with respect to its educational significance.

> It is this aspect of educational criticism that sharply differentiates the work of the educational critic from the work of an ethnographer, psychologist, or sociologist...Educational critics ultimately appraise what they encounter with a set of educational criteria; they judge the educational value of what they see. (p. 14)

It would be required that the researcher presuming to take on this role would "possess far more than the writing skills possessed by a good novelist or journalist. It requires a broad grasp of educational theory, educational history, and it would be a distinct advantage for critics to have had experience as classroom teachers (1977:8)." Otherwise, as Eisner pointed out, without a sense of the practical realities of classroom life, subtleties observed in the classroom escape notice, the things that were overlooked or neglected by the teacher are not noted, and the values embraced or rejected by the

teacher are not comprehended.

Eisner (1977:8) explained the task of the educational critic was to provide a "vivid rendering" so that others might learn to see what transpires in a busy classroom.

Kennedy (1978:4) was concerned about the talent needed by the writer of a descriptive case study.

Arguments have been made for the validity of qualitative data, subjective impressions, as descriptions of naturally occurring events. While these arguments may be valid, they suggest that the evaluator needs the talent of Tolstoi to be able to describe these events in ways that allow the reader to draw the appropriate inference.

Eisner agreed that what was needed for such a "rendering" was a "form of linguistic artistry replete with metaphor, contrast, redundancy and emphasis that captures some aspect of the quality and character of educational life (p.8)". He claimed that "good critics use language in a way that requires a certain poetic and fluid range of words and phrases."

With regard to writing style in educational research Eisner (1976:5) had previously noted that "somehow if 'the author "r'we' conclude something it is more objective than if 'I'do." In the attempt to increase objectivity the shift in language that results acts to "de-emotionalize expression" and he felt "the opportunity to understand empathically and to communicate the quality of human experience diminishes." He claimed we must go beyond the description of behavior in order to know what the behavior <u>means</u>.

Eisner (1977:5) cautioned that "sheer description unguided by value considerations is rudderless." He seemed to be implying that it is not enough to describe and allow the reader to draw the parallels but as a qualified educator, the responsibility for the

interpretative and evaluative roles in educational criticism should be assumed.

Thus the ultimate consequence of educational criticism is evaluative in the sense that something must be made of what has been described and interpreted. The task of the critic is not simply one of neutral observer (an impossible position in any case) nor is it one of disinterested interpretation. The critic uses what he or she sees and interprets in order to arrive at some conclusion about the character of educational practice and to its improvement.

The rules for validity of educational criticism are complex but Eisner suggested that several procedures would help to dispel doubt about the procedures used. By attempting to effect "structural corroboration" the researcher demonstrates how a variety of facts or conditions within the phenomena support the conclusions drawn. "It is a process of demonstrating how the story hangs together, that the pieces fit (1976:16)". It sounds very much like effecting the "triangulation" of data as originally described by Webb(1966) and more recently elaborated by Jick (1979).

A second method of "determining the reliability of the critic's language is by judging the referential adequacy of what he has to say (1977:9)" that is by seeing if its referents can be found in the work or the event itself. This is more possible when judging the referential adequacy of a critic of art who evaluates a more stable object that can likewise be viewed by the reader. But Eisner pointed out that it is possible to do the same with more fluid forms of art such as symphony concerts. "Classroom activities flow into one another" and seldom present a complete whole and what is more they change over time. Eisner recommended (p. 10) that the classroom be "visited with sufficient persistency to enable the critic to locate its pervasive qualities". It is these qualities that become the objects of educational criticism. In addition, Eisner stated that by capturing segments of ongoing classroom life on videotape the referential adequacy of criticism created might be determined.

Eisner has faith that effective connoisseurship and criticism will result in a

... content within a particular classroom and a refined sensitivity concerning classrooms that is useful for studying other educational situations... and may provide cues useful for locating phenomena that might subsequently be pursued through conventional educational research... Creative scientific work in any field depends on new realizations, new models, or new methods to guide inquiry. (p.11)

Summary and Implications for the Present Study

This section has recognized that the case study method has its origins in anthropological, ethnographic, and phenomenological traditions. It is suitable for use in an exploratory, hypothesisgenerating study as well as for hypothesis testing. The ultimate purpose of conducting a case study may be to produce a description, an explanation, an interpretation, or an evaluation. By selective use of methodologies, it becomes a vehicle which is flexible, adaptable. and perhaps best suited to classroom-based research at the theory building stage. Since its methods may be unconventional and inventive, it allows the researcher freedom to approach the problem from different perspectives using eclectic data collection techniques. The present research study addressed rather open ended questions. A desirable way to answer these was to determine "what is" and then to describe what seemed to be. As Yinger stated, one hesitates to do much more than describe and therefore leaves the rest to the gifted few. As Eisner portrays them, the "gifted few" should be sufficiently

qualified to take an evaluative role in educational criticism.

Where possible, where enough "structural corroboration" was available, the present researcher ventured past description and possibly may have overstepped safer bounds. After three months of data collection and a year of data analysis and rumination, the "enduring and pervasive qualities" of this classroom in the present study as "understood" by the researcher are presented as the findings in this dissertation.

While the text of the description itself is largely teacherprovided data about students and not the researcher's description of ongoing instruction, the choice of data to be presented, the sequence of presentation and the method of presentation, however, reflect the perspectives of the researcher. Thus the format differs from that proposed by Eisner in many respects but the ideas of connoisseurship and criticism have been kept in mind constantly and used when deemed appropriate.

TEACHER THOUGHT PROCESSES

Another area of research which played a role in the present study is that concerning teacher thought processes. The N.I.E. Panel 6 Report (1975:1) confirmed the need to know more about the

"mental life" of teachers. "Research typically slights the problem of how teachers <u>think</u> about their pupils and instructional problems. It concentrates instead on how teachers <u>act</u> or perform in the classroom " (Shulman and Elstein, 1975:3). It was speculated that part of the delay in investigating cognitive processes of teaching stemmed from the lack of confidence in introspective techniques.

Magoon (1977:660) confirmed there was traditional objection to self reports. Radford (1974:250) had conceded that "introspection gives us information about experience. It yields some data otherwise inaccessible." Radford claimed that if verbal reports, or introspectionas-data, were subjected to the usual checks used for all experimental data and yielded satisfactory levels of significance they can be used with confidence. As long as one did not "take the response at face value but established empirically its relationship to some other variables" (p.246) it was reasonably credible. Harré and Secord (1972:84) supported the legitimacy of "self-reporting" techniques. Their anthropomorphic model of man presumes he is a self-monitoring individual with the ability to verbalize conscious thoughts which guide his actions.

It is possible to examine the covert behavior operant in planning instruction and in active teaching situations by asking the teacher to articulate the thinking behind the actions. Some initial training is usually necessary for the teacher subject because thinking aloud while performing a cognitive task, or re-experiencing it, is not a frequently practiced skill but Shulman and Elstein claimed it

is possible to verbalize these inner thoughts with the help of probing encouragement, Wodlinger (in progress) took particular care in the initial training of his teacher subject to help her identify and articulate the decision processes she was using during instruction. The teacher reported making several times as many decisions during a lesson than teacher subjects in other research studies have reported.

Shavelson (1973:146) described teacher decision making as the most important teaching skill and the fact that teachers do have information processing skills and decision making skills provides an important rationale for having teachers in the classroom (Berliner, 1976) as opposed to teaching machines.

Research has been conducted on both preactive decision making and planning (Oberg, 1975 and 1978) and interactive decision making (Clark and Peterson, 1976 and Wodlinger, in progress). Oberg (1975) concluded that in addition to many other factors that teachers did give consideration to student characteristics and ability when making preactive decisions about instruction. Shavelson and Atwood (1977:55) reported that on the basis of their "estimates of student states of mind", teachers decide which instructional approaches are suitable for certain students in preactive planning.

Marx and Peterson (1975:17) concluded that investigating interactive decisions was not as fruitful a pursuit as investigating preactive decisions Marland (1977) found there were fewer than 10 interactive decisions reported per lesson. Wodlinger's contradictory findings may be due to the training in articulation of the processes involved which

his teacher received.

Clark and Yinger (1977:292) concluded that interactive

decision making rarely resulted in an immediate change in the course of instruction but acted as more of a "process of fine tuning and adapting to the aspects of a situation that were unpredictable in principle" such as specific student responses. Shavelson (1976) found that most reported interactive decisions amounted to modifications of those made in planning. But such decisions are made on the basis of teacher-held expectations for and knowledge of the particular student (MacKay and Marland, 1978; Shavelson and Atwood, 1977; and Russo, 1978).

In contrast to preactive decision making where estimates of student states of mind appear to be more salient, overt studentcentered cues like bewilderment are apt to precipitate interactive decisions. Clark and Peterson (1976:6) found that during interactive decision making, student participation and involvement were the main cues used by teachers, particularly the "mood" of the group. However, in his study of 12 teachers, MacKay (1978:19) found that the absolute references to student attention during stimulated recall of the lesson in progress were few in number.

In comparing preactive and interactive decision makers, Zahoric (1970) found that teachers who planned extensively were less flexible during actual instruction. Marx and Peterson (1975:16) found that highly preactive teachers had students with more negative attitudes toward school than teachers who made fewer preactive decisions and who, therefore, made more interactive decisions which tended to involve students (by accepting their initiated questions and comments as the lesson progressed). They also reported that teachers who "are concerned about students in their planning tend to be concerned about students in their **eaching** (p.13)".

"In the decision making which appears to encompass the best skills of the teacher must make contingency decisions which suit the moment in time by integrating information from 1 sources (for example: pre-planning, information about students, past machine experience) yet remaining "aware" of situational information as welcome des, 1973 and Shavelson, 1978).

apidity and pace of classroom life appear to preclude the rational ode of decision making used in preactive planning. Researchers have sugnited that teachers' interactive decision making might be more "intuitive" than rational (Jackson, 1968, and Yinger, 1978). Wodlinger's teacher was able to relate antecedents for, rationale behind, and whether alternatives had been considered for her interactive decisions; aninding which implies that interactive decision making may be more rate al and logical than previously concluded.

In searching for a model for research on teacher thinking, Clark (1978) considered the teacher as a decision-makef (Shavelson, 1973; and Clark and Yinger, 1977) and the teacher as an information processor (Marland, 1977 and Conners, 1978). Clark judged the decision making type of study as a highly controlled inquiry focusing on decision making behavior, whereas the broader information processing model is a more inclusive approach through which the researcher can explore the ways that teachers define situations and cope with the task environment which is overwhelmingly rich in information to be processed.

Ryans (1963:274) described the teacher as an "information

processor" and Shulman and Elstein (1975) saw the teacher more precisely as an active "clinical" information processor who was involved in planning, anticipating, judging, diagnosing, prescribing, and problem solving.

Adams and Biddle (1970:85) portrayed the teacher as continually accommodating surrounding stimuli and manipulating the environment for others.

Consequently she has to act as a kind of feedback mechanism for the classroom. She receives stimuli from her pupils and other elements in the environment, she has to process this information in relation to her educational purposes, then she has to emit behavior, which, in her opinion, will bring about the desired pupil responses.

Marland's (1977) study on information processing of teachers examined teachers' cognitive functioning during instruction and the ways teachers coped with interactive demands. In addition to other solicited information, the teachers were requested to disclose the kinds of information about students they had brought to the lesson situation and which impinged upon or influenced their interactive thoughts and decisions.

Shulman and Elstein (1975:35) claimed that introspective techniques could be useful in the "study of sources, character, modifiability, and consequences of teacher expectations". Such procedures may help to identify how teachers interpret and evaluate "meaning and sufficiency of new information, which information sources are used by teachers to make pup11 judgments, and which are ignored."

Clark (1977) investigated how teachers gather, organize and evaluate information and felt that continued study would lead to an understanding of the uniquely human processes which determine and guide teacher behavior."

Clark and Yinger (1977:301) talked of the

exciting possibility that research on teacher thinking may unite the concerns of researchers in instruction and teacher behavior with those of researchers on curriculum and materials. All of these concerns come together in the minds of teachers as they make the plans, judgments, and decisions that guide their behavior. Indeed, the thinking of teachers may be the strategic research site that yields the first practical theory of instruction.

Stimulated Recall

Stimulated recall, used as a research device for investigating covert thought processes has yielded promising results. Bloom (1953 and 1954) felt that recall conducted within a 48 hour period produced accurate accounts of simultaneous covert thought processes that accompanied observed actions. He felt the teacher could relive an original situation with vividness and accuracy if presented with a large number of cues. He used both audio and video cues in his research. Gaier (1954) cautioned that the subject would tend to censor what he would reveal. Shulman and Elstein (1971) used multi-level protocols (video and think aloud protocols in conjunction with stimulated recall protocols) to ascertain the consistency and accuracy provided by this introspective technique. This tri-level data collection permitted cross referencing on the recall items provided by the physicians in the study. Such "triangulation" or corroboration of stimulated recall data was urged by Radford (1974:246).

Kagan et al. (1963:239) emphasized the importance of the role of the interviewer during stimulated recall. Unless a rapport, characterized by "communicated authenticity, positive regard for the other person, and empathy" (Fuller and Manning, 1973:499) were developed, the teacher subject would not be willing to verbalize covert thoughts. Other procedures such as assuring anonymity, allowing the subject selfviewing opportunities before actual recall, and informing the subject of the purpose of the recall help to ensure that the subject will cooperate fully (Marland, 1977).

Tuckwell (1980:282) suggested that filming from the same vantage point as the teacher's, instead of from the back of the room which is less obtrusive, would result in cues more like the original ones influencing the teacher during the lesson and this might facilitate more accurate recall.

Stimulated recall has been successfully used in researching covert behavior of teachers by Bloom (1953), Shavelson (1973), Clark and Peterson (1976), Shavelson and Atwood (1977), Marland (1977), Conners (1978), Cooper (1979), King (1979), and Tuckwell (1980) to date.

Summary and Implications for the Present Study

Increasing interest in the covert mental activity of both teachers and students has resulted in growing support for introspective studies in education. One research technique which has been found particularly useful in this pursuit is video playback stimulated recall during which the subject reveals his thought processes with respect to issue-specific stimuli. A comprehensive review of both teacher thought processes and stimulated recall methodology can be found in Tuckwell's (1980) dissertation.

The present study utilized the teacher-as-information-processor approach but purposely narrowed the focus of teacher thinking to information about students. Data concerning the teacher's expectations for student achievement, the teacher's affective reaction to students, the teacher's perception of various student personality attributes, and the teacher's perception of his students' classroom behavior and work habits were gathered during general interviews and by conducting eight stimulated recall interviews focusing on the teacher's thoughts about students during instruction. These expressed teacher thoughts were clarified and corroborated by other data sources in order to substantiate the authenticity of the teacher's self reports.

TEACHER EXPECTANCY EFFECT

Teacher-held expectancy is an area that has been shown to be investigated profitably through the use of introspective techniques. Attribution Theory

A knowledge of attribution theory helps to explain how the expectancy effect functions. Attribution theory, from the cognitive phase of social psychology, provides a general framework for interpreting actions, for identifying causes, and for predicting future actions. Bar-Tal (1978:259) defined an attribution as the "inference a person makes about the causes of behavior, either his own or another person's."

Jones and Nisbett (1972:80) outlined the discrepancies in interpretation that occur between the "actor's" (eg., a student) perception of an event and the "observer's" (eg., a teacher) perception of the same event. Teachers act on the basis of these perceptions,

however, and use them in forming expectations about students. The classroom observer, in turn, may interpret the same event involving both the student and the teacher in yet another way.

Weiner, (1972, 1974, 1976;184) in his work on causal attribution made a connecting link between attribution processes and expectancy theory. If the responsibility for an action observed is felt to be due to "stable" causes (eg., the student's lack of intelligence, or the difficulty of the task), a teacher would be apt to help the student overcome the difficulty encountered. In this example, one of the stable causes (ability) is said to be an "internal" cause while the other (task difficulty) is "external" to the student. In a second example another student might be perceived to fail on a test due to bad luck or a lack of effort on his part (both of which are seen to be "unstable" or changeable causes). The bad luck is external and can not be controlled but the effort exerted is internal and seen to be something a student can control. The teacher's attitude toward helping the student in the second example would differ from the first wherein the student was seen more as a victim of circumstances beyond his control. Weiner conceptualized that most people would attribute their success or failure on a task to one or more of the four causes mentioned above.

		Locus of concrot	
		Internal	External
Stability	stable	ability	task difficulty
	unstable	effort	luck

Locus of Control

Weiner (1972:213) claimed that it had been

convincingly demonstrated that causal attribution influenced the likelihood of undertaking achievement activities, the

intensity of work at these activities, and the degree of persistence in the face of failure.

Teachers form their own attributions to account for the successes and failures experienced by their students. These attributions contribute to their performance expectations for students and may account for differential teacher behavior toward students. Hence, a student viewed as "lazy, does not exert effort" may not receive the teacher-provided challenges that the "tries hard, persistent" student does in the classroom situation. The unchallenged student will not produce as much. If the teacher's attribution of "lazy" happened to be inaccurate, the self-fulfilling prophecy over time could eventually result in the student exhibiting the very characteristic which would confirm the teacher's attribution.

Varying factors such as the age and sex of the student, and personal achievement needs, would tend to govern what happens on an individual basis but it seems obvious that self-attribution and the attribution made by others would help to shape and determine performance in the classroom.

Smith (1977:7) claimed that the role of the teacher was crucial for manipulating the felt achievement needs of the pupil. The "expectancy" of success and the "value" of the outcome to the student (Weiner, 1976:184) have to be adjusted to suit and challenge each individual student. Confidence may result in greater effort and persistence and the student may begin to make more favorable selfattributions about his performance. Attribution <u>retraining</u> procedures have been successfully demonstrated by Entwisle and Webster (1970), de Charms (1972), Kulka (1972) and Andrews and Debus (1978). Kulka (p.168) stated that effort is the most salient determinant of outcome. Attributions to effort, then, are to be encouraged in the learning situation. King (1979) substantiated the importance of perceived effort on the part of students in the formation and maintenance of teacher expectations.

<u>Summary</u>. Attribution for causes of performance underlies both teacher expectancy and student self-expectancy for future performance. The student's causal perception of his own successes and failures may determine his motivation and persistence at academic tasks. The teacher's causal perception of the student's behavior influences her performance expectancy for that student and her behavior towards him during instruction.

Teacher Expectancy

The belief that teacher expectations could function to the detriment or advantage of student performance is becoming generally accepted. Winne and Marx (1977:672) concluded that relationships exist between expectations and student learning based on evidence found by many studies to date. Whether these relationships are correlational or causal in nature is the subject of much interest and research currently.

Researchers believing there was a causal relationship between teacher expectations and achievement (West and Anderson, 1976:615; Lockheed, 1976:16; and Dusek and O'Connell, 1973:375) raised the question of the "preponderance of causation" and suggested the view that expectations are formed on the basis of the child's past academic performance in opposition to the view that teacher-held expectations

influence future achievement. West and Anderson (1976:624) stated that future student achievement is best predicted by present student achievement. Lockheed (1976:16) suggested that student achievement is the strongest single predictor of teacher expectation. By contrast, Crano and Mellon (1978:39) concluded that teacher expectations affect students' achievement and further concluded that the teacher's expectations for the child's <u>social</u> development instead of achievement development exert the stronger influence on achievement.

Brophy and Good (1974:34) after many studies involving teacher expectations felt that teacher expectations usually <u>result</u> from observed student performance rather than <u>cause</u> it. However, they pointed out that teacher-held expectancies are capable of affecting student performance if they are inappropriate and are kept inflexibly rigid.

Formation and Maintenance of Expectancy for Students

The information sources that give rise to the formation of expectations are many and varied. Sheer personal reactions and feelings toward pupils (Willis and Brophy, 1974:527) can result in expectation for the pupil. Usually the information sources used in forming expectations are the attributions made about student performance, work habits, and personality (Shavelson, Atwood, and Borko, 1977:51). These sources of information used in the formation of expectations are often referred to as "cues". The following studies have dealt with the effects of specific, naturally occurring cues and their influence upon the formation of teacher expectations:

> divorce (Santrock and Tracy, 1978) teacher experience with older sibling (Seaver, 1973) unusual name (Harari; 1973)

physical physique (Stafferi, 1967 and Lerner, 1969) handwriting (Briggs, 1970)

sex (Palardy 1968, and Adams and Looie, 1974) facial attractiveness (Clifford and Walster, 1973) conduct (Adams and Lavoie, 1974) labeling (Yoshida and Meyers, 1975)

Lockheed (1976:16ff) and Braun (1976) presented a comprehensive list of such cues and the names and dates of research studies that confirm or question the relative strengths of these cues on the formation of teacher expectancies for students. However, the most salient and relevant cues would appear to be the child's actual classroom behavior and academic performance. The use of cues may depend on the time of the school year, other available information, or certain teacher characteristics.

Each teacher is idiosyncratic in her susceptibility to the expectancy effect due to factors such as years of classroom experience, her belief system, and her awareness of the effect (West and Anderson 1976:621; and Cornbleth, Davis and Button, 1974:58). Dusek (1975:666) claimed that in comparison with student teachers who had been used in expectancy research studies, that "real", or more experienced teachers treated students equally. Muttart (1977) who investigated presage and process teacher characteristics in relation to differential treatment of students, concluded that the phenomenon was not universal with teachers.

Good and Brophy (1975:115) distinguished among three general types of teachers who varied in their susceptibility to the expectancy effect. The "proactive" teacher uses his expectations for students in planning suitable activities designed to individualize and optimize learning. The "reactive" teacher has generally accurate and flexible expectations but he allows the students to control or condition the patterns of student-teacher interaction. The third type of teacher, the "overreactor", provides most evidence of expectancy effects. By adhering to simplistic and rigid stereotypes he allows himself to be conditioned by student differences but in addition, he exacerbates these differences by treating students as even more different than they really are, engaging in kinds of behavior that produce selffulfilling prophecy effects. Brophy and Good (1974:viii) stated, "Few teachers are able" to consciously monitor each student on a continuing basis and treat him in a deliberate, proactive manner... ..Most teachers are primarily <u>reactive</u> in responses to students... showing evidence that students have shaped their behavior."

Clark and Yinger (1977:295) stated that the "open minded" teacher was more flexible and made adjustments in her expectations. Haigh (1974:110) and Smith and Luginbuhl (1976:267) confirmed that the degree of "awareness" of the expectancy effect determined its exhibited strength in a teacher's behavior: Nash (1976:40) felt that once the teacher became cognizant of how her attitudes influenced her actions, she was more likely to attempt to give all pupils equal opportunities within the class. Nash believed that

> the teacher would take an active role in counteracting her natural biases by deliberately acting in positive ways towards the pupils- to make an anti-self-fulfilling prophecy of her expectancy- and make of it a virtue.

> > (p.40)

It is difficult to estimate the extent to which the expectancy effect is in operation in a classroom for it is contingent on many factors. Rosenthal (1973:59) found that the effect occurred between 34 to 37 percent of the time in his study. Good (1970:190) noted that teachers "expect one-third of the students to learn much, and onethird to just get by." This expectation tended to be supported by the schools' grading policies and practices. If the above information is any indication of the prevalence of expectancy effects in classrooms it might be safe to assume that depending on certain conditions, the lowest third in the class could be subject to negative expectancy effects.

Shavelson, Atwood, and Borko (1977:55) stated that "teachers constantly receive additional information about students. Many additional cues come from the teacher's observation of day to day interaction in the classroom." These cues and resulting attributions concerning probable causes for observed student behavior are part of the teacher's continual information-processing activities. The possibility therefore exists that negative expectancies may be revised on the basis of positive information received. However, the "anchoring " heuristic (Shavelson et al, 1977:55), or the strength of the "first impression", is difficult to overcome and may be the mechanism underlying the expectancy phenomenon for Shavelson et al felt that "adjustments" following the "anchoring" were typically insufficient for discomfirmatory evidence tends to be rejected. Fortunately, most teacher expectancies are reasonably accurate in the first place.

Clark and Yinger (1977:300) stated that

The evidence is mixed on the extent to which teachers' judgments are flexible and responsive to new information. It is clear that teachers vary in the accuracy of their predictions of student achievement and the weights that they assign to factors that influence their judgment.

Summary and Implications for the Present Study

The formation of performance expectancies for students varies in accuracy and strength with individual teachers. The relative importance they assign to these expectations is a function of their belief system and their ability to process and evaluate new information which may serve to maintain or revise their impressions of a student's capabilities. In particular an awareness of the phenomenon of the expectancy effect usually serves to reduce its strength and influence.

A comprehensive review of both attribution theory and expectation theory can be found in King's (1979) dissertation.

By March, when the present study began, performance expectations held for the students would have been formed, adjusted, and would then have become relatively stable after seven months of ongoing teacher-student interaction. If process expectancy effects existed, it was anticipated that they would be detectable because patterns of interactions based on mutual expectations would have become habitual. By delving into teacher thoughts and feelings about the students it was considered possible that the salient cues which had been used to form these teacher impressions of and expectations for individual students and for groups of students might be identified.

TEACHER ATTITUDES TO STUDENTS

By analyzing Grade 3 teachers' descriptions of their students, Silberman (1969) identified four basic attitudes (attachment, concern, indifference, and rejection) that teachers seemed to express toward students in their classes. Teachers were asked to assign the one most representative student in the class to each of the four attitude groups.

44

\$ 0

Silberman examined the teacher's interaction with the class and identified which specific teaching behaviors appeared to communicate these feelings to students. Silberman recorded and analyzed the dyadic verbal interaction which was exchanged between the teacher and the student subjects in the attitude categories. Differential verbal treatment of students was noted and teacher's affective reactions appeared to account for this occurrence. Due to the rapid pace of classroom life, and particularly during verbal exchanges, teachers are forced to react spontaneously.

> They are required to make many required decisions quickly by acting on their feelings towards the students involved in certain situations than by weighing the pros and cons of alternative actions. (Silberman, 1969:406)

The most common characteristics of students belonging to the four attitude groups have been distilled from several research studies and are presented in brief summary form:

Attachment: high achievers who responded warmly to the teacher and who were generally conforming and compliant, happy, attractive, mature.

The verbal interaction which was characteristic of this group was positive in nature with a high percentage of praise which was in acknowledgement of their excellent achievement. They initiated a high percentage of contacts with the teacher. Teachers seemed to keep overall contact with this group minimal, partly to avoid showing the favoritism felt, and partly because these students did not seem to need help and could be counted on to ask for assistance when they needed it.

<u>Concern</u>: Low achievers who were dependent on the teacher and who exhibited acceptable behavior, restless, immature, in need of encouragement, notably lacking in self confidence. They made appropriate but extensive demands on the teacher. Concern students received the highest amount of verbal contact from the teacher. Interactions were characterized by longer contacts, of both student-initiated and teacher-initiated types. They were praised for good behavior and criticized for poor behavior. It seemed as if teachers went out of their way to encourage and praise these students.

<u>Indifference</u>: had few identifying characteristics (blank expression was frequently mentioned), were middle achievers, exhibited some misbehavior and some negative attitudes toward school, not noticeable, unhappy, unattractive, avoided eye contact.

In contrast to Silberman's suspicion that indifference students were accidentally overlooked because they were not all that salient or noticeable, Good and Brophy (1974) suggested that teachers' indifference toward these students was a defence mechanism to protect teachers from the continued frustration and rejection experienced when working with indifference students. Their unresponsive and sullen behavior turned off teachers and conditioned them to minimize interaction with them. It was a matter of mutual avoidance.

The total number of interactions exchanged with this group was lower than for all other attitude groups. Infrequent student-initiated contacts indicated a student avoidance of the teacher. Reciprocally, a low percentage of teacher-initiated interactions occurred with this group.

<u>Rejection</u>: low achievers, frequently disruptive and defiant. Low ability was attributed to this group, sometimes inaccurately.

Interior with this group were characterized by frequent

havior, a high percentage of student-initiated

contacts. A higher percentage of interaction with these students occurred in private settings. Toward rejection students teachers tended to exhibit hostility and held them "continually under surveillance" (Silberman, p.406)." Teachers appeared to experience conflict in dealing with these students and appeared to praise them whenever possible. It seemed to Silberman that they were held up to the class as both good and bad examples.

Willis and Brophy (1974:520ff) defined "attitudes" as affective reactions which were tied to the personal qualities of the student and his reactions to the teacher , not to student achievement.

Good and Brophy (1974:192) concluded from reviewing their own and others' findings that the two main variables influencing the assignment of students into attitude groups were (1) the student's general level of success in school and (2) the student's tendency to reward teachers in personal contacts with them. Willis and Brophy (1974:529) added (3) the degree to which these children conformed to classroom rules. A general summary of the four groups would conclude

that;

Attachment students are high achievers, who reward teachers during personal contacts and who conform to classroom rules. Concern students have trouble in school but are generally conforming and rewarding.

Indifference students do not provide personally rewarding experiences for teachers although their achievement is sometimes better than average and although they basically conform to classroom rules.

Rejection students satisfy none of these criteria to any great degree.

In the <u>Student Attribute Study</u> (Brophy et al, 1976) the researchers set out to provide more objective descriptions of students and so

conducted a large scale study in order to "broaden and deepen knowledge concerning the relationship between teachers' perceptions of students and the actual attributes of these students (p.444)." The study addressed the questions "Why do teachers develop certain

attitudes toward some students but not to others; especially others who are similar?" and "What kinds of students provoke strong attitudinal

responses in teachers?(p.118)"

Studies involving small numbers of teachers are very sensitive to the unique attitudes and preferences of the teachers 4 involved so that their data may not be generalizable to other teachers. (p.126)

The researchers wished to clarify puzzling and contraditory findings that had emerged for the above reason from the attitude studies to date. Twenty-seven female elementary teachers and 362 of their students served as the subjects in the study. The findings were meant to generalize across many students in order to describe the "typical" attitude group student who would evoke a predictable and specific teacher attitudinal response. With this generalizable information, the dynamics of classroom life might be more satisfactorily explained.

Brophy et al. departed from the tradition of assigning students into the four attitude-to-student groups. Their information concerning attitudes was taken completely from two items on the 13 Attributes Scale (Brophy, Evertson, Anderson, Baum, and Crawford, 1976) namely;

> Item #10:Attachment (Would like to have removed from class versus would like to keep for another year for the sheer joy of it).

and

Item #11: <u>Concern</u> (Doesn't require special attention versus concerns me a great deal. I would like to be able to devote much more attention).

The students received a high, medium, or low score.

However, by using this procedure the same student appeared somewhere on both attitude measures; eg., a "high attachment" but "low concern" student. For each of the 13 attributes Brophy et al. analyzed the data for the strongest, most consistently rated students. Even so, it is difficult to be surprised at their conclusions:

Some unexpected and interesting findings appeared for the students ranked consistently low on the concern scale. Just as was the case with students ranked high on attachment, the teachers were unlikely to respond negativistically to misbehavior by the low concern students. No doubt, part of this was because the students misbehaved less often and less intensively. Even so, this pattern was unusual. It suggests that relationships between the teachers and the students consistently ranked low on concern might have been particularly positive, rather than neutral, as the term "low concern" connotes. (p.426)

Brophy et al. investigated the relationships among all the attributes and noted what types and amounts of verbal interactions were exchanged with students rated as high, middle, and low on specific attributes. These teacher perceptions and ratings were compared to observer perceptions and ratings for the same students and any discrepancies were noted. Comprehensive profiles of students in the different attitude categories were extended. Brophy et al. (p.125) commented on the accuracy and dependability of teacher-provided data about students.

There are even more questions about self-report data from teachers. Taken at face value, they seem to add rich detail and elaboration to teacher expectation and attitude data. They help indicate other characteristics that teachers notice about students in particular expectation and attitude groups as well as some of the reasons why teachers hold the expectations or attitudes they hold. However, like any self-report data, teacher self-report data are distorted by halo effects, logical errors and other sources of bias or inaccuracy.....Thus teacher perceptions hang together nicely to make a consistent picture but at least some of them were incorrect. This raises questions about the rest

Accuracy and bias aside, it is on the basis of their perceptions and assessments of students that teachers <u>do</u> operate. However, Brophy et al.were attempting to identify those specific characteristics of students which presumably would affect all teachers in similar ways. This was a difficult task, and, as Brophy et al. (p.449) implied, they found

a certain degree of unpredictability due to the "chemistry" of relationships between individual teachers and individual students for occasionally the right combination of teacher and student will produce relationships which defy all theory and previous research.

Crano and Mellon (1978:47) stated that teacher evaluation of and expectation for the social development of a child might exert an effect on student performance. Presumably a teacher who finds a child obedient and a pleasure to have in class would spend more time and energy helping and monitoring his academic progress. Especially in the early grades, they stated that "the affective response of the teacher to the student may have far more impact on the child's performance than the teacher's assessment of academic ability." The resultant self concept has a profound shaping effect on the child. Crano and Mellon concluded that the effects of these social expectations "based on aspects of the child totally incidental to his academic development, represent a potential educational inequity (p.48)."

Summary and Implications for the Present Study

Attitudes toward students contribute to the formation of performance expectations for students. Although based more on the personal qualities of the student, these attitudes tend to predispose

the teacher to assign certain attributes to the student and to affect his personal interaction with the student during instruction.

In the present study, the researcher obtained teacher-assigned scores for all 26 student subjects on Item #10-Attachment and Item #11-Concern, as in the Brophy et al. study (1976), but, in addition, had the teacher subject place all 26 students into one of the four attitude to student groups. Assigning all students to groups would weaken characteristics which are more apparent with the three most (or one most) representative student of each attitude group but all 26 students were subjects in the study and if any trends emerged for larger numbers of students, more confidence might be engendered in the data.

The achievement expectations of the teacher for each attitude group could then be examined as well as the personality and behavior attributes assigned (The present study used the 13 Attribute scales from the Brophy et al study). In addition the dyadic verbal interaction contacts received by individual students could be averaged into attitude group scores which could be compared with findings from previous research studies. No attempt was made to prove how accurate the teacher was in these expressed attitudes and perceptions but the strength and consistency of these teacher ratings were of major interest.

CLASSROOM INTERACTION

In order for the expectancy effect to operate, the performance expectations and affective reactions held by the teacher have to be communicated and understood by the student. Perhaps the most common vehicle for conveying this information to the student in the classroom-
is the dyadic verbal interaction he exchanges with the teacher. Indirect information is relayed to the student by virtue of the quality and quantity of interaction which he personally receives (Dusek, 1975). More direct communication of expectations is also possible.

Rosenthal (1973:60) proposed a four factor theory to explain how teachers communicate their expectations to students. First the teacher creates a "climate" (eg., a warm, socio-emotional mood). According to Rosenthal, the teacher monitors "input" (or the type of verbal interaction) addressed to the student on the basis of his perceived achievement and motivation levels. Beez (1970) noted that his teachers (student teachers) taught more content to "bright" students. He concluded that teacher expectations had been translated into overt alterations in teaching style and substance. The "output", as Rosenthal termed the response opportunities afforded the children usually result in the high achievers receiving more opportunities for verbal interaction with the teacher. Teachers send "clearer strong evaluation", or "feedback", to bright students. Use of praise is certainly a way of communicating these expectations. Firestone and Broudy (1975:549) noticed a feedback difference in that the low students were praised less when they required more supportive teacher behavior and received more criticism than the situation warranted.

That the resultant interaction patterns can be connected with teacher expectations has been well attested by numerous research studies to date (Brophy and Good, 1970; Kester and Letchworth, 1977;

Rothbart et al, 1971; Rubovits and Maehre, 1971, and Nash, 1976. Dusek (1975:681) stated that although differences in the amount and kind of interaction may be noted, if teacher expectancy is based on objective data then no "bias" is present in resultant verbal interaction. In fact the differences in interaction might be the teacher's attempt to provide teaching style differences to suit children with differing needs. But if differences are based on irrelevant information, then bias may be said to exist.

Ż

The student's reaction to this information may depend on his personal characteristics. Pupils with high achievement needs may exert more effort to counteract the sentiments the teacher seems to be communicating. Teacher use of criticism with more able pupils sometimes functions to encourage better effort on their part. However, low achievers tend to give up and behave in accordance with the low expectations expressed by the teacher.

The obvious function of verbal interaction is the exchange of information. Feedback to the student is important for such things as time on task (Bloom, 1974:686) and for providing the student with information essential to purposive behavior (Smith and Luginbuhl, 1976:270).

How much and what kind of verbal interaction occurs in the classroom Δ is very much controlled by the teacher. Her teaching style for example, may encourage or discourage pupil input during lessons. One would expect to find far more student-initiated interaction with an "indirect" style of teaching. Smith and Luginbuhl (p.270) stated that the amount and kind of interaction directed at the individual

student was due more to the degree of teacher awareness for the "aware teachers gave more equal verbal attention to both bright and dull" in their study.

Classroom interaction studies involve the observation of overt behavior in naturalistic classroom settings. Data are collected either in checklist form such as the Dyadic Interaction Observation System, derived from Brophy and Good(1969) for recording verbal exchanges, or by concurrently recording the observer's classification into a microphone (Haigh, 1975:100) for later decoding.

The "quantity" of verbal interaction refers to the frequency and amount that takes place between a student and the teacher. Some researchers have found no quantitative differences in the way the teachers interacted with high and low achieving students (Rubovits and Maehre, 1971 and Smith and Luginbuhl, 1976), but more frequently researchers have found both quantitative and qualitative differences in the way the teacher engaged the children in verbal interactions (Haigh, 1974; Jeter, 1972; Rothbart et al, 1972; and Bloom, 1980). Haigh found up to three times more interaction with perceived high-achieving students. "Qualitative" differences in interaction are evidenced by the use of higher order questions, probing questions to correct inaccurate answers, sustaining, evaluative remarks, and process feedback which are commonly addressed to the abler student and to the student for whom the teacher has concern.

Varying reasons may explain why differential interaction patterns are allowed to exist. For example, for motivational reasons the teacher may choose to call on the "bright" child on whom he can depend to provide

consistent and appropriate responses (Good, 1970:193). Perhaps in realizing ability differences the teacher is attempting to protect feelings of the student and in the hope of reducing anxiety purposely may not call on him in public settings. The net result, however, is that children note the amount and type of verbal interaction they experience with the teacher and develop a self concept partially based on this information.

Cooper (1977) suggested that either a more selfish desire to control personal rewards (Interacting with bright children is more satisfying to the teacher) or the desire to retain personal control over the "content, duration, and timing" of the interaction with "low" achieving students (particularly in student-initiated interactions) determined the patterns of interaction. The teacher often chooses the situation over which she has the most control. Cooper (1977:421) found that his teachers expressed higher expectations for and perceptions of greater control over the academic performance of females. Cooper felt that teachers' expectations for and sense of control over students were positively related. As a result the teacher will administer more negative feedback to low students in the hope of dissuading them from becoming involved in public interaction situations over which she perceives she has the least control.

Firestone and Broudy (1975:544) claimed that the knowledge of interaction patterns in the classroom enable one to predict academic performance independently of I.Q. They argued that teacher expectation, whether experimentally induced or naturally present, has an effect on the interaction that takes place in the classroom with "high students

having an interactional advantage (p.545)." Firestone and Broudy saw the danger of the self-fulfilling prophecy occurring for students repeatedly treated in a manner indicating that their verbal contributions were relatively worthless. Eventually students internalize these expectations.

Summary and Implications for the Present Study

Verbal interaction patterns indirectly communicate teacher performance expectations to students and hence can function as differential teacher treatment. The differing patterns frequently noted in classroom studies serve to reinforce existing differences in students' ability thus acting as a mediating agent for the expectancy effect.

A prime purpose of the present study was to examine teacherreported perceptions and feelings toward students with some attempt to compare other available student information with the set of teacher-expressed data and to attempt to interpret existing dyadic verbal interaction patterns using this particular information.

Verbal interaction data pertaining to high, middle, and low expectancy groups, the four attitude-to-student groups, and for the 26 individual students involved in this study were examined to determine and describe the predominant characteristics of the verbal exchanges and the quality and quantity of verbal contacts received.

SUMMARY .

This chapter has briefly outlined several research areas which gave rise to the present research project. Individuals in a classroom

make up a unique group of people who interact on the basis of perceptions, attributions, and expectations held for one another and by "observing the norms of the setting" (Wilson, 1977:246). It is only by getting to know these particular people, their characteristics and the ways in which they work together that researchers can attempt to develop causal explanations about instructional processes in specific classrooms. This would argue in favor of naturalistic observation. The case study method seems to be particularly well suited for exploratory, hypothesis-generating studies in naturalistic classrobm settings.

It was the increasing awareness that students' characteristics and behavior tended to exert influences on teachers' behavior, rather than completely vice versa, that prompted Brophy et al. (1976) to undertake their Student Attribute Study to try and determine which student characteristics affected teachers in predictable ways. Fiedler (1975:736) acknowledged this "bidirectionality" of influence and claimed that all classrooms lie somewhere along the continuum from "assymmetrical" to "reciprocal" control. A student senses how much control he is able to exert and uses it. More subtle control is gained by virtue of the personality characteristics he possesses.

Teacher attitudes to students and attributions for students' ability and effort contribute to the formation of teacher-held performance expectations for students. Resultant differences in the relative assessments of students are frequently disclosed through variation in the amount and kind of verbal interaction that is exchanged between the teacher and individual children in the classroom.

57 🗇

Student characteristics and classroom performance as evidenced by time spent working on task provide a daily information source for the teacher who monitors the student's progress and makes adjustments in, her assessments of his ability and effort. These expectations are cyclically formed, maintained, and communicated through the interpersonal relations and verbal interaction occurring during routine instruction in the classroom. The factors contributing to the formation of expectations may be partially explained by attribution theory, the affective reaction of the teacher to particular students, and partly by drawing interpretative heuristic inferences once the dynamics of the classroom interaction are intensively observed and analyzed.

CHAPTER III

DESIGN AND PROCEDURES

In this chapter, the design, the methodology adopted and the data collection procedures are presented. The subjects and the data sources are described and the subsequent treatment of the data is outlined.

THE DESIGN

Overview

In the design of this study the following recommendations which were suggested in background literature and related research were incorporated where possible.

(1) Research should use multi-faceted data collection instruments and procedures. This should include a combination of subjective and objective data, high and low inference data, and a mixture of qualitative and quantitative data which is mainly descriptive in nature. Such an eclectic approach might afford the opportunity to "triangulate" (Miles, 1979:590) or corroborate the same information gathered through various data collection techniques. MacKay (1978:1) recommended the incorporation of complementary data collection procedures into the design of a study in order to reduce the oversights and

limitations inherent in the use of a single instrument.

(2) Research should extend the data collection over a period of time in order to study the ongoing instructional processes more efficiently, accurately and naturalistically.

(3.) Research should single out the numerous elements that appear to affect the teaching-learning process and to study them both separately and in combination with one another in order to see what interactional effects these elements exert on each other (Bennett, 1978:127). (4.)Research should focus on different aspects in the classroom such as the teacher subject(the behavior he exhibits, his thought processes- particularly those thoughts concerning students, his perceptions of the individual students), the individual student (his characteristics, and his work habits), and on groupings made up of these students (such as expectancy groups, reading groups, affective attitudeto-student groups, and boy/girl groups).

(5.) Research should emphasize that the study of individual students is a worthwhile new direction for educational research (Brophy and Geod, 1974 ; Mahen, 1977) for each student possesses a unique fet of personality characteristics and work habits. These characteristics exert an influence on the teacher who subsequently provides each student with treatment based on his reaction to these perceived student qualities. The growing recognition of the "bidirectionality" (Fiedler , 1975:735 ; and Brophy et al, 1976: 128) of the classroom instructional process, or the reciprocal student effect on teacher behavior may provide information which researchers have overlooked formerly by concentrating on the one-way instructional process of teacher effects on students. In addition, to isolate for consideration the student's work lepits and use of academic learning time is to study a mediating process in itself which affects subsequent achievement (Bloom, 1974:684).

In summary an integrated approach was adopted which would replicate portions of many other studies and in so doing would retest the instruments and methodologies used therein in order to corroborate findings from these individual studies. In addition by looking at these varied forces in interaction with each other, a way of reducing the complexity of processes

at work in the classroom might be found and tentative explanations might be formulated which would simplify the description of these

processes. It was anticipated that questions raised during this endeavour might serve as topics for further research.

FEATURES OF THE DESIGN

Representative portions of the Dunkin and Biddle (1974) model were incorporated in the design (student <u>presage</u> data- as perceived by the teacher and teacher <u>presage</u> data as stated by the teacher in the form of attitudes and priorities in teaching; language arts and math <u>contextual</u> data, dyadic verbal interaction <u>process</u> data as well as student work habits and effort <u>process</u> data, and achievement <u>product</u> data). In addition the covert <u>thought processes</u> of the teacher were examined as recommended by Winne and Marx (1977) and the N.I.E. Panel 6 Report (1975) and were examined in conjunction with these more overt data sources.

Briefly the main instruments and methodologies of the present study were taken from the following sources: rankings of the students using instruments from Brophy and Good (1970 and 1974), Brophy et al. (1976) and Luce and Hoge (1978); the teacher's perception of student attributes as measured on the instrument (13 Attribute Scales) used in the Brophy et al. (1976) <u>Student Attribute Study</u>; the teacher's attitude to students from Silberman (1969), Willis and Brophy (1974) and Brophy et al. (1976); and the thought processes of the teacher from Shavelson (1977), Clark and Yinger (1977) and Clark and Peterson (1978). This study liberally borrowed aspects of 12 studies recently conducted at the University of Alberta, Edmonton, in four of which the present researcher played an active, though minor, role. Mahen (1977) by focusing on individual students studied the relationship of individual students' classroom behavior to process (verbal interaction) and product (achievement) measures. Muttart (1977) studied the relationship of teacher-held expectations for students

concerning their general level of achievement across subjects and the resultant amount and kind of dyadic verbal interaction received by these students from the teacher. Fasano (1977) studied how certain presage student characteristics appeared to relate to the amount and kind of dyadic verbal interaction the students received. Eggert (1977)

investigated the behaviors exhibited by six teachers during instruction (as recorded using high inference instruments) in relation to student achievement. Marland (1977) investigated the covert thought processes of six teachers during instruction by using videotaped lessons to stimulate recall of their interactive thoughts. Conners (1978) examined interactive thought processes of nine teachers in order to identify the underlying "principles of teaching" or belief systems that appeared to influence the teaching behaviors of these teachers. Smyth (1979) focused on four target students and studied their individual use of academic learning time. The present researcher acted as the reliability check during the data collection for the Smyth study. Among other things, King (1979) looked at the importance of effort attribution, the value the teacher placed upon individual student effort, and the ways in which the perception of student-exerted effort influenced teacher perception and treatment of students. The role of teacher expectations played a major role in his study as well. Tuckwell (1980) focused on the interactive thoughts of two teachers and in particular looked for evidence of specific thoughts

about his area of interest; the impact of an in-service program. (The present researcher was interested in isolating only those specific thoughts about particular students and their characteristics during instruction). The present researcher acted as the reliability check for Tuckwell's content analysis categorization of these interactive thoughts. One aspect of Project Quest (MacKay, 1979) concerned the teaching behaviors of the 61 teacher subjects (one of whom was the teacher subject in the present study) and noted whether these observed teaching behaviors changed over time and/or related to levels of achievement in each classroom. The present researcher assisted in the preparation of the data collection instruments for Project Quest, the training of the coders who gathered the data, and acted as the reliability check in several classrooms including the particular classroom under study in her own research. Moody (1980) developed detailed profiles on one target i student from each of six classrooms and described how the many influences in a classroom appeared to affect these students. (The present researcher developed less exhaustive but comprehensive profiles on all of the 26 students in the one classroom under investigation). Wodlinger (in progress) examined the interactive decision-making processes of one teacher over an extended period of time and attempted to trace the "antecedents" and "rationales" for these interactive decisions. (The present researcher acted as the reliability check for the content analysis categorization system developed and used by Wodlinger in his study).

The foregoing are the most influential sources from which the elements in the present study were selected. The purpose of drawing together so many elements was twofold:

(1) To answer more comprehensively the specific research questions which guided the data collection for the present study. By collecting many kinds of data using varied data sources and by approaching the questions from different angles it was hoped that a more plausible description of the ongoing classroom instruction would result.

(2) A second purpose of adopting a multi-faceted approach was a purposeful melding together of many elements from other studies which independently contributed to a partial explanation of the teachinglearning process. It is anticipated that the combination and interaction of the elements selected might serve to explain more fully some of the complexities of classroom life which previously have been masked by adopting too narrow a focus for investigation. A sub-purpose was to test and re-evaluate the instrumentation used in these studies.

Thus a case study approach was used in a naturalistic setting in a single classroom. The researcher acted as a non-participant observer and did not attempt to induce any experimental effects. Naturalistic observation and quantitative data collection techniques were combined with both general interviews and stimulated recall interviews conducted with the teacher to ascertain his perceptions of the students. Perceptual, qualitative, and quantitative data were gathered using varied instrumentation. The overarching purpose of the study was to describe, from the teacher's perspective, the teacher-student relationships and the ongoing instruction in the class over the three month period. Using the teacherprovided data, a description and explanation of how the teacher's perception of each individual student, and of groupings comprised of these students might relate to the teacher's treatment of these individual

students and groupings of students and how, in turn, this treatment might influence subsequent achievement levels was attempted.

Although research questions were posed to guide the observations and data collection a series of hypotheses was not proposed and tested. It is anticipated that hypotheses for future research can be generated from the findings. The main purpose of the present study was to explore and to describe the instructional processes in the classroom under study.

The Subjects

The present study was linked to a larger system-wide study, Project Quest, involving 61 experienced Grade 3 and Grade 6 teachers. The study was conducted jointly by the Centre for Research in Teaching and the Edmonton Public School Board during the winter of 1979. One of these Grade 3 teachers agreed to allow more extensive investigation of the teaching-learning process in his classroom and hence became the teacher subject for this study. The teacher, in his late twenties, had four years of teaching experience in the elementary grades prior to the 1978-1979 school year. He was informed that the general intent of the study was to explore and examine the instruction in his classroom during language arts and math periods, that observational data would be collected during instruction, that selected lessons would be videotaped, and he was advised that additional time would be required of him for both general interview and stimulated recall purposes. He was told that he would be de-briefed at the conclusion of the data collection, would be assured of anonymity, and would receive a copy of the dissertation in its final form.

The school was located in a generally high socio-economic area and all except three students in this class came from two-parent families.

The 26 student subjects (13 girls and 13 boys) ranged in age from 8 years 1 month to 9 years 5 months. The mean I.Q. for all 26 students was 112 and academic achievement was valued by the students and an interested set of parents.

Instrumentation

Although 14 data collection instruments were used in this study, no new instruments were developed. Minor adaptations were made to three of these instruments. Also both general interviews and stimulated recall interviews were conducted with the teacher to elicit teacher general thoughts and interactive thoughts about students.

The Dyadic Interaction Observation System. The dyadic interaction observation system, or DIOS, is directly descended from the Brophy and Evertson (1973) modification of the original Brophy and Good (1969) dyadic verbal interaction instrument. With some of the items deleted, the original two page instrument was collapsed into a one page format. This comprehensive low inference instrument is designed to capture a variety of dimensions of both student and teacher verbal behavior. These verbal interchanges which can be either teacher-initiated (Tinits) or student-. initiated (Sinits) and may occur in private, small group, or whole class settings, are recorded in the sequence in which they occur. The particular student involved in the interaction is identified. Thus it is possible to determine the amount and kind of verbal interaction specific to individual students in comparison with other class members or by different grouping patterns (as described before) or by the entire class using frequency and/or percentage scores. Provision for recording the lesson context enabled later subject area comparisons. A copy of this

instrument appears in Appendix A.

Reliability for the verbal interaction instrument. The instrument for recording verbal exchanges between the teacher and individual students, the dyadic interaction observation system (DIOS), is an adaptation of the 1969 and 1973 versions developed and used by Good and Brophy, and Brophy and Evertson respectively. Brophy and Evertson (1973) reported that it was possible, using their two-page, rather complicated system, to train coders to reach a criterion of 80% agreement for reliability purposes. The formula they proposed for the purpose of conducting a reliability check was:

Percentage agreement= number of coding decisions made by both coders and agreed upon, divided by itself plus the number of codings made by the first coder but not the second, plus the number of codings made by the second coder but not the first.

Project Quest used a simplified version of this low inference verbal interaction instrument. Although the individual student was not identified, 49 of the 60 categories on the DIOS used in the present study were identical. The present researcher acted as the reliability check for Project Quest in several classrooms, one of which was the classroom used in her own research. Using the above formula a reliability coefficent of 86.0% resulted with the Project Quest coder in the classroom under study in the present research

project.

In the reliability check for the present study, a fellow Ph.D. student who had collected verbal interaction data for his own doctoral dissertation research acted as the reliability check using the DIOS instrument. Using the above formula a reliability coefficient of 85.4% desulted for the verbal interaction instrument in this study.

Six Rankings. In the Brophy and Good (1973:93) study, the teachers were asked to rank all the children in the classroom in the order of expected achievement. "Instructions were kept deliberately vague to encourage teachers to use their own complex subjective criteria in making judgments." In the present study the six different instrument rankings of all 26 students by the teacher were administered using the specific instructions (not vague) as outlined in the Brophy et al (1976:469) study. The teacher was asked to assign all the students into seven weighted categories. The ranking list was divided into seven "clusters" (within which all students were considered "equal"): "the three highest, the four next highest, the five above the middle, the five middle, the five below the middle, the four next lowest, and the three lowest". The teacher was asked to determine the highest, then the lowest, then the next highest, then the next lowest, thereby approaching the middle. In this study separate class rankings were made by the teacher for:

- (1) Projected Achievement in Math.
- (2) Projected Achievement in Language Arts
- (3) Maturity
- (4) Cooperative/Compliant Behavior
- as in the above-mentioned studies. Two additional rankings for
 - (5) Motivation to Do School Work
 - (6) General Intellectual Ability

were included from the Luce and Hoge (1978:491) study.

The 13 Attribute Scales. The 13 Attribute Scales were used in the same form as they appeared in the Brophy, Evertson, Baum, Anderson and Crawford (1976:450) <u>Student Attribute Study</u>. The teacher rated each of his.26 students on a line continuum reflecting a "low", "medium", and "high" assessment of the child's possession of the following 13 attributes:

- (1) calm/ good self control
- (2) careful/deliberate worker
- (3) happy
- (4) probable highest achiever,
- (5) mature
- (6) cooperative/compliant behavior
- (7) creative
- (8) attractive
- (9) persistent worker
- (10) likeable/attachment
- (11) of concern to the teacher
- (12) noticeable
- (13) looks you in the eye

Brophy et al. (1976) used the values of "high", "medium" and "low" in their analysis. In the present study finer distinctions were made which enabled a score of one to seven to be derived from this instrument which would complement the ranking scores of one to seven for comparative purposes.

<u>Teacher Attitude-To-Student Assignments</u>. The teacher attitude-tostudent questionnaire was administered by asking the teacher to assign students into the affective categories of "attachment","indifference", "concern", or "rejection". (The specific definitions provided for each category appear in AppendixA). Silberman (1969) had asked for one student in each category. Willis and Brophy (1974) and Brophy and Good (1974) asked for three students and Evertson, Brophy and Good (1973; as reported in Brophy and Good 1974:137) asked for as many as five students in each of the attitude groups. Since the present researcher gathered data on all 26 students the teacher was requested to place all 26

students into the four attitude-to-student groups.

<u>Conners' Micro Analysis of Teachers' Interactive Thoughts.</u> Although it is customary and usually necessary to devise a unique content analysis system for stimulated recall data, Conners' Micro Analysis was used in the present study for two reasons:

(1) Conners had developed these detailed content analysis systems to unitize and categorize stimulated recall data from the nine teachers representing three elementary grades. The systems were global enough to accommodate the data from these varied sources yet exhaustive enough to include the thoughts unique to each teacher. Upon close examination of the systems it appeared as though the stimulated recall interview data from the present study could be analyzed very satisfactorily using these existing systems.

(2) A sub-purpose of this study was to replicate portions of other studies and to re-evaluate the instrumentation used therein. This occasion was opportune for determining how generalizable and applicable Conners' systems were to the stimulated recall data collected in the present study.

The interview data are first screened to include only "interactive" data (thoughts the teacher was having during actual instruction). These data are then unitized or separated into discrete units and subsequently filed into the appropriate categories which describe the types of thoughts. This procedure converts the qualitative interview data into units that can be counted and analyzed using quantitative analyses methods. Conners' Micro Analysis System divides the expressed interactive thoughts into 12 broad categories most of which, in turn, are sub-categorized.

Conners' Macro Analysis System re-analyzes the interactive data and "sieve" codes it (Guetzkow, 1950) in order to "investigate teacherheld beliefs, principles, rules and other factors that influence teacher

behavior (Conners, 1978:179)."

Reliability of coding using Conners' Micro Content Analysis

<u>System</u>. The intercoder reliability check was conducted by a fellow Ph.D. student similarly engaged in a content analysis of stimulated recall data in his own doctoral research. The same extracts from three different stimulated recall interviews were used for both the intercoder and the intracoder reliability checks. A period of about three months intervened between the intracoder reliability check and the original coding of the data by the researcher.

With the use of a content analysis system the reliability of the coding processes of unitizing and categorizing is a usual concern. In addition, however, the decision to include portions of the data as "interactive", and therefore codable, or to exclude them as "non-interactive" data was subjected to a reliability check as well because of the obvious importance of this initial coding decision.

Conners (1978:107) had conducted reliability checks at all three coding stages using three different coefficients of reliability. As these procedures had been followed insturn by King (1979), Tuckwell (1980), and Wodlinger (in progress, 1980), they were adopted by the present researcher as well and are now outlined.

Holsti (1969:138) had used the following formula to establish a reliability coefficient for codine "dichotomous decisions", such as whether or not a portion of data could be classified as "interactive" or "non-interactive":

Coefficient of reliability=

2M

+ N.

M represents those decisions agreed upon by the two coders and N_1 and N_2 refer to those decisions made by each coder respectively. The use of this formula resulted in a reliability coefficient of 88% for intercoder reliability and a coefficient of 93% for intracoder reliability for coding "interactive" versus "non-interactive" data.

The segmentation, or unitization of data, was examined in the second reliability check for which Conners (1978:107) cited Guetzkow's claim that for this purpose, it was appropriate to "express the difference between two coders as a percentage of the sum of the units obtained by each coder." Thus for unitizing, the formula used in the reliability checks was:

$$U = \frac{O_1 - O_2}{O_1 + O_2}$$

A ratio of \hat{O} (zero) would indicate that the coders were in perfect agreement. O_1 and O_2 refer to the number of units established by the two coders. Using this formula the intercoder reliability coefficient was .06 and the intracoder reliability coefficient was .02 for unitizing the data.

For the coding process of categorization, use was made of Scott's formula which makes adjustments for the probable frequency with which each of the numerous categories are used by chance alone. Scott's formula is:

No represents the agreed-upon decisions and Pe represents the chance occurrence of such agreements. The use of this formula for the intercoder reliability check resulted in a coefficient of 81% and for the intracoder reliability check a level of 85% was reached.

An acceptable level of reliability is determined by considering

the complexity of the content analysis system used. Marland (1977), Conners (1978), King (1979), and Tuckwell(1980) had defined a reliability coefficient of .70 as an acceptable level of reliability in their respective studies. In the present study both the intercoder and the intracoder reliability coefficients were above this established level of acceptability.

High inference scales measuring Teacher Behavior. Project Quest's data collected on high inference rating scales were used in the present study to describe predominant teaching behaviors exhibited by the teacher subject. Thirty^{Θ} one high inference variables were worded to reflect observable teaching strategies which were deemed to be desirable teaching behaviors from previous research in teaching. The high inference variables were coded as "agreements" as were ratings of 4 and 5 for the purpose of determining intercoder reliability.

Reliability for high inference scales measuring Teacher Behavior. The following formula was used to calculate the inter-rater reliability for this high inference scale:

Percentage agreement = $\frac{T_A}{T_A + T_D} \times 100$ percent

 T_A and T_D represent the total numbers of agreements and disagreements. The present researcher acted as the reliability check in several classrooms under study in Project Quest. one of which was the classroom used in the present research project. The level of agreement reached with the Project Quest coder was 80.6% for this intercoder reliability check.

. 73

General Interviews and Stimulated Recall Interviews. Eight stimulated recall interviews (completed within three weeks in April) and five long general interviews (ranging from March 6 to June 7th) plus many shorter recorded conversations were conducted with the teacher. The general purpose of all these interviews was to collect the teacher's thoughts, perceptions, and feelings about his students and their academic progress.

74

The stimulated recall interviews elicited those teacher thoughts about particular students during actual instruction which may have influenced teacher treatment of these students. Hence in this study the content analysis of teacher interactive thoughts was not the only information of value. In all, three types of analyses were applied to the interview data. A micro Content Analysis (Conners, 1978) of teacher interactive thoughts (was contained in the stimulated recall data) determined the nature of thoughts about students in particular during instruction. A macro analysis of interactive thoughts helped to identify which expressed attitudes and priorities may have influenced his treatment of particular students in specific instructional situations. A perusal of all non-interactive data from both stimulated recall interviews and general interviews helped to determine what general beliefs about teaching this teacher appeared to hold and finally, perhaps the most important analysis of all, a search of all interview data to collect all comments about particular students in order to clarify and

corroborate numerical scores assigned to these students by the teacher on the many different data collection instruments.

SPECIFIC RESEARCH QUESTIONS

In this study five major areas of interest gave rise to the

general, overarching research question addressed in this study;

To what extent is it possible to explain how or why differential treatment of students occurs by examining the teacher's perceptions of his students?

... and to the following, more specific research questions;

- 1.0 What are the common characteristics of students assigned by the teacher to each of four attitude-to-student groups (attachment, concern, indifference, and rejection)?
 - 1.1 What were the perceived personality characteristics which might serve to explain the teacher-expressed attitudes and affective feelings toward these groups of students?

The required information concerning each of the 26 students in the classroom was obtained by having the teacher assign a numerical rating specifically for each child on 19 attribute measures which dealt with student characteristics such as personality traits, work habits and effort, teacher-assessed ability levels, and the expectancy held for each child for academic performance. In addition verbal interaction information, age, I.Q., achievement scores were collected for each child as well as information about each child from the series of interviews held with the teacher.

The teacher assigned all 26 students into one of four attitude to student groups which reflected the teacher's affective reaction to each individual student. The scores received on the teacher-assigned measures, and on the other data sources, were averaged for the four groups in the collective attempt to describe all children belonging to a specific attitude-to-student group. While this masks individual differences to a certain extent, some characteristics emerged which may help to describe commonalities shared by members of the four groups and these results were compared to findings from previous research on teacher attitudes to

students

-75

2.0 What evidence is there, if any, of expectancy effects?

2.1 What evidence is there, if any, of product expectancy effects? What is the relationship of teacher-held expectancy to end of the year achievement results?

76

- 2.11 for students assigned to high, middle, and low "probable highest achiever" levels (for general achievement in all subjects)?
 - 2.12 for students assigned to high, middle, and low "probable achievement in language arts" levels?____
 - 2.13 for "good" and "poor" reading groups?
 - 2.14 for boys and for girls?
 - 2.15 for the four attitude-to-student groups?
- 2.2 What evidence is there, if any, of process expectancy effects? What is the relationship of teacher-expressed expectations to the amount and kind of verbal interaction exchanged?
 - 2.21 for students assigned to high, middle, and low "probable highest achiever" levels (for general achievement in all subjects)?
 - 2.22 for students assigned to high, middle, and low "probable achievement in language arts" levels?
 - 2.23 for "good" and "poor" reading groups?
 - -2.24 for students assigned to high, middle; and low "probable achievement in math" levels?
 - 2.25 for boys and for girls?
 - 2.26 for the four attitude-to-student groups?

The expectancy information required in order to answer both parts

of Research Question II was obtained in the form of the her-assigned numerical score (from 1 to 7; 1 being, "high") on three of the 19 attribute measures which were concerned with teacher-held expectancy for students.

The three expectancy measures used in the present study were:

probable highest achiever (for general, overall achievement in all academic subject areas).

probable achievement in language arts

probable achievement in mathematics.

Students who were assigned expectancy scores of 1 or 2 were classified as "high" expectancy students. Student receiving scores of 3, 4, or 5 were considered to be "middle" expectancy group students and in the same manner, students with expectancy scores of 6 and 7 were considered to be in the "low" expectancy group for each of the three expectancy measures. For the two reading groups, the average expectancy score for "probable" achievement in language arts" only was used. Expectancy scores assigned to boys and girls were averaged as were the expectancy scores for the four attitude-tostudent groups for all three expectancy measures.

The <u>product</u> measures used were the end of the year, system-wide Grade 3 language arts achievement test and the Grade 3 mathematics achievement test. The <u>process</u> measures used in this study were the percentage ratios of specific types of dyadic verbal interactions which were exchanged between the teacher and a specific student. The various groups' "product" and "process" scores were derived by averaging their individual member's scores for these product and process measures.

3.0

Using all available information, to what extent is it possible to construct a comprehensive student profile which would describe the student's behavior, his achievement, his membership in various grouping patterns, and the relationship he had with this teacher?

Since most of the data included in the student profiles were provided by the teacher, each profile tends to present the student as the teacher perceived him (which is what was desired). In part, Research Question 4.3 (To what extent do these teacher-reported thoughts about students clarify/corroborate numerical scores assigned to the student by the teacher?) is similarly addressed in the composition of these student profiles.

78

The presentation format in each of these 26 student profiles, or mini-case studies, is meant to be uniform in that the numerical scores on the 19 attributes are presented in narrative form and interwoven with direct teacher quotations taken from the transcribed interview texts which relate to the same attributes for which a numerical rating exists. Other teacher-provided information about the student, gleaned from the interviews, is presented in these mini-case study descriptions as well. In short, a presentation of the student as the teacher claimed to have perceived and assessed him was attempted (not a description of the individual student as the researcher viewed him). Additional information concerning age, I.Q., achievement and time on task where available (as not all students had been chosen for engaged time observation) followed this description. A personalized dyadic verbal interaction set of data were included in each student's profile. A summary statement, which did incorporate views held by the researcher, as well as a review of the more salient student characteristics previously presented ended each student profile.

4.0 What information is obtainable from the interviews held with the teacher?

4.1 Which kinds of thoughts about students in particular were reported during stimulated recall interviews?

ŝ

4.2 What teacher attitudes and priorities are revealed in all interview data?

4.3 To what extent do teacher-reported thoughts clarify/corroborate numerical scores assigned on the different student ratings?

Eight videotaped lessons during both language arts and math instruction were shown to the teacher in order to elicit the interactive thoughts he had been considering concurrently during instruction. Of particular interest to the researcher, were those interactive thoughts about students which the teacher reported during stimulated recall interviews for these expressed thoughts were presumed to indicate the kinds of student information which was processed by him during his actual teaching.

9

An examination of the eight stimulated recall interviews and the more general interviews held with the teacher over the three month period was undertaken in the attempt to identify and reveal the more global teacherexpressed priorities and attitudes which appeared to relate to observed teaching behaviors and which appeared to exert an influence on his teaching style.

The extent to which comments from the many interviews supported the numerical data was explored more fully during the compilation of all teacher-provided information about each student when the case study profiles were being assembled. Apparent discrepancies in the two types of information were clarified by asking the teacher for information which would resolve any misunderstanding on the part of the researcher. How could the teaching behavior of the teacher subject be described

5.0

- 5.1 What are the predominant teaching behaviors of this teacher?
- 5.2 What are the predominant verbal interaction behaviors • of this teacher?

This teacher was observed by a coder from Project Quest on ten occasions during the teaching of language arts and math and was rated on his observed performance on 31 teaching variables. The average of these scores has been used to describe his predominant teaching behaviors during the same time period as the present study and to compare his performance with that of 60 teachers who participated in Project Quest.

Data obtained on the DIOS in the present study have been averaged to indicate this teacher's usual verbal behavior, across the three month period, which was directed to the class as a whole and toward various student groupings which existed in the classroom.

ASSUMPTIONS

A major assumption established in previous research (Good and Brophy, 1975, and King, 1979) is that both teachers and students hold performance expectations which are formed and exchanged reciprocally through classroom interaction. These teacher-held expectations exert an influence on the way the teacher reacts to and treats the students.

It is assumed that "self reporting" techniques are valid ones for collecting introspective data (Harré and Secord, 1972:154).

In particular, it is assumed that the technique of stimulated recall is especially suited for this purpose. It is further assumed that the researcher was able to develop and maintain a rapport with the teacher and that the teacher was able and willing to recall and articulate his interactive thoughts and decisions to an accurate degree.

The researcher assumed that by observing the familiarization procedures (suggester by Marland, 1977, Conners, 1978, and King, 1979) that the presence of the researcher and the video equipment was of minimal distraction to the subjects in the study.

It is assumed that the samplings of verbal interaction patterns teacher perceptions and ratings and the interview data accurately represent the ongoing instructional processes in the classroom.

LIMITATIONS

As this study was conducted in one classroom, the generalizability of of results is somewhat limited.

There was a possible observer bias in the data collection because, of necessity, the researcher was aware of the teacher's perception of and feelings toward the students.

There is always the possibility that the intrusion of the researcher may have distorted some of the behaviors in this naturalistic setting.

This study was not able to investigate the students' thoughts and perceptions and focused only on the teacher's perceptions of ongoing classroom instructional processes.

PROCEDURES

In this section the different phases of the study are described and the procedures followed during each phase are outlined.

The Preparatory Phase

Prior to the pilot study the instruments for use in the study were selected and their formats were adapted where necessary to suit the purposes of the research.

With two fellow doctoral researchers, Tuckwell(1980) and Wodlinger (in progress), who were also involved in stimulated recall research, the researcher spent several sessions in the Audio Visual Laboratory in the Faculty of Education learning to operate the video recording equipment. Helpful hints and suggestions were shared during the data collection for all three studies. King (1979) gave advice concerning familiarization procedures and stimulated recall techniques.

All required video recording equipment was made available to the researcher on long-term loan for the duration of the study from the Audio Visual Media Centre of the Education Faculty. This equipment consisted of: a small-format television camera (Sony 3200, B/W), a video tape recorder (Sony AV3600 Solid State), a television monitor (Panasonic 9", B/W receiver), a wireless microphone (Lectrosonics M30R/R31) worn by the teacher and its receiver, a set of headphones (Sony DR-9), a portable two-shelf metal cart, accompanying cord, plugs and extensions and 10 video tapes (half inch, open reel, B/W). The researcher used an audio cassette recorder (Sears 19298) and remote microphone (Sony one point stereo microphone F-99B) to audio-tape actual lessons on occasion and to tape the general interviews as well as the stimulated recall interviews conducted with the teacher.

Training of the second coder who would act as the reliability check for the study commenced at this time. King (1979), experienced in using varied data collection techniques and classroom observation procedures, had

used an adaptation of the Brophy and Evertson (1973) dyadic verbal interaction system in his own study and had assisted in Project Quest with coder training and the development of the simplified verbal interaction instrument used therein. Hence training on the DIOS (dyadic interaction observation system) was readily accomplished.

The selection of the two cooperating Grade 3 classrooms (for the pilot study and for the main study) was facilitated by the Research and Development Office of the Edmonton Public School Board and the Field Services Office in the Education Faculty on campus.

The Pilot Study

The pilot study which spanned a one month period (February 1-27th, 1979) was conducted in a different school than the one used in the main study. A female teacher and her 24 students in a self-contained Grade 3 classroom agreed to participate as subjects for this phase of the research.

This stage of the study was essential so that the researcher could become skilled in using the coding instruments, gain confidence and expertise in using the video and audio equipment, and could perfect questioning and interviewing skills for conducting the stimulated recall interviews.

As a result of the pilot study minor alterations were made on the DIOS(verbal) instrument resulting in more efficient coding. Having experimented from various locations in the classroom, the researcher decided to video record from the back of the room keeping the teacher in view. During the three stimulated recall interviews that were conducted it was found that the stimulated recall methodology had to be supplemented with probing questions which helped the teacher to focus on thoughts about students and not on the broader spectrum of interactive thoughts. Ground rules were developed to guide these question probes. The transcripts of these stimulated recall interviews were scrutinized by a fellow researcher to see that protocols were being followed and that additional questioning techniques were appropriate. During the course of the pilot study ideas for collecting the data became the bases for the data collection procedures used in the main study. All intended purposes were accomplished in the month-long pilot study phase of the research.

Data Collection Plan

The main study extended over the three month period, March 1 to June 7, 1979. The bulk of the data were collected in March and April. <u>Lesson Context</u>. Observations were confined to language arts and math periods for several reasons. The teacher did not teach all subjects to his own class. It was in these lesson contexts that the teacher behavior data (for Project Quest) were being collected. The choice of these basic areas acknowledges their major importance in the elementary school curriculum. Also it was felt that sufficiently different instructional strategies and settings would be employed by the teacher in these subject areas.

Mathematics, with its relatively more formal structure would provide the opportunity of observing specific teaching behaviors. In the present study language arts settings were expanded to include the following activities: spelling drill, spelling workbook activities, handwriting instruction, student free-reading periods, teacher oral reading a novel to students, reading group instruction, creative writing, and general discussions. Since language arts in this classroom involved a high percentage of individualized or group work, certain social studies lessons

which provided suitable opportunities for observing the teacher and students participating in general discussions were included under the language arts general discussion category.

<u>Familiarization period</u>. The classroom subjects already were accustomed to the presence of the Quest observer who had begun collecting data for the larger study one month previous to the beginning of the present study. The researcher acted as the reliability check for this Quest coder. The percentage of agreement for coding on the Quest verbal interaction instrument was 86.0% and for coding on the high inference scales used in Project Quest the agreement percentage was 80.6%.

A period of about ten school days was allowed in order to learn the students' names (and identification numbers for coding purposes), to determine the scheduling of classes in language arts and math instruction, and to become knowledgeable about the teacher's operational routines. Eye contact with the student subjects was avoided in order to discourage their friendly attempts at interaction and to present a business-like appearance. This was difficult to do but was continued for the duration of the study. Some time was spent practice, coding using the DIOS instrument to build up speed and accuracy before actual data collection began.

Beginning the first day, the researcher video-recorded lessons in order to allow subjects to become accustomed to the presence of the observer using the video equipment in the classroom. In both classrooms (in the pilot study and in the main study) the researcher was convinced that the subjects virtually ignored her presence and the video recording equipment after the ten day familiarization period was over.

Data Collection Procedures

Verbal interaction data were recorded in a variety of instructional settings. Several lessons were coded before any interviews were conducted with the teacher of before any teacher-perceived data were obtained. This procedure was followed to ensure that at least some of the verbal interaction data could not have been affected by alerting the teacher to the researcher's purposes and concerns.

Over the three month period, dyadic verbal interactions for the present study were gathered during math (1191 interactions over 10.5 hours of math instruction) and language arts (1524 interactions over 14.9 hours of language arts instruction) periods. A total of 2715 separate dyadic verbal interactions were recorded in all, representing a total of 25.4 hours of instructional time sampled over the three month period. This allowed for the possibility that each of the 26 students could be involved in an average of 100 or more interactions gathered-during a variety of instructional settings.

After about one month the teacher was asked to rank order the class members on the six ranking measures. These forms were completed on three different occasions to allow the teacher ample time for consideration. Names of all of the children were printed on small cards which could be arranged and rearranged before final decisions were made and the names copied onto the forms. The ranking lists were divided into seven "clusters" (described above). Within each cluster spaces were provided for three to five names and all students were considered equal within each of the elusters.

At this time the students were assigned by the teacher into the four attitude-to-student groups. Into the <u>attachment</u> category ("If you could keep one student for another year for the sheer joy of it whom would you pick?) the teacher placed 15 students and one additional student, a boy, Nicholas. "Someone like Nicholas... He almost doesn't fit ..." He gave him a "conditional" placement in the attachment group. Because of the large number of students in this group the teacher was asked to identify his <u>first three choices</u> for the attachment group. Separate analyses for these three students were carried out and compared with analyses for the entire group of 16 attachment students(to which they belonged) and compared with the other attitude-to-student groups as well.

The definition provided for the <u>concern</u> group was "If you could devote all your attention to a child who concerned you a great deal, whom would you pick?" Into this group the teacher placed Ian (Case 14) and Tom (Case 15) who experienced academic problems and Bonnie (Case 18) who had emotional, medical, and academic problems.

The teacher hesitated to place any student into the <u>rejection</u> group ("If your class was to be reduced by one child whom would you be relieved to have removed?"). The teacher claimed that in other years he did have "rejection" students but he did not feel strongly negative about any student in this year's class. He eventually gave Pamela (Case 23) a "conditional" placement and Sonia (Case 26) who moved into the classroom in early April eventually received a "conditional" placement in this group as well. For the purposes of analyses these two girls were considered to belong in the rejection group.

The provided definition for the <u>indifference</u> category ("If a parent were to drop in unannounced for a conference whose child would you be least prepared totalk about?") seemed to fit Marilyn (Case 12) and John (Case 17). Three other students, Sharon (Case 1), Keith (Case 2),
and Mason (Case 20) were not all that noticeable and the teacher had neutral or "blah" feelings about them. The literal definition of <u>indifference</u> seemed to apply to these children who were placed in this group as a result.

The teacher was given a separate copy of the instrument, 13 Attribute Scales (Brophy et al, 1976), for every student in the class and was asked to consider the student as an individual when assigning a "low", "medium", or "high" rating along the line continuum. Brophy et al had used the values of "low", "medium", and "high" in their analysis of the 13 attributes. However the teacher in the present study chose a position along the line continuum thereby disregarding the specific "low", "medium", and "high" markings. This occurrence had not been anticipated by the researcher but it was mutually decided that this method yielded more precise information. Subsequently the researcher was confronted with a conversion problem. The line was segmented into seven clusters and a numerical score (of 1 to 7) was assigned to each student for his score on each of the 13 attributes. With recognized limitations, this procedure enabled cross comparisons with the class rankings for now all teacher-perceived measures (six rankings and 13 attributes) for each of the 26 students ranged from one (high) to seven (low) in possible numerical value. Within these 19 items there were two "maturity" measures and two "cooperative/compliant" measures. It was of interest to compare the two separate ratings for each child on these two particular characteristics for these ratings had been completed with at least two weeks intervening and by using two relatively different rating methods.

Spontaneous conversations with the teacher were audio-recorded when possible as well as the more formal, six general interviews. These taped conversations were transcribed and typed into 133 pages. The eight stimulated recall interviews averaged about 50 minutes each and were audio-taped and later transcribed into 113 typed pages. In all 246 typewritten pages of combined interview data were prepared for content analyses.

Although not included in the present report, data on engaged time (time on task) were collected for selected students in the class. These data appear, where available, in the Case Study Profiles in Chapter VI. Continual observation of the student under surveillance resulted in a minute-by-minute account of his use of time. Resultant percentages of time on task for teacher-directed settings, for self-paced-independent work settings, and for combined settings are interesting additions to the profiles of these selected students. An account of these findings for engaged time will appear separately in a later report.

In June, data from cumulative cards were gathered for each student. Precisely this information included the age of the student! three I.Q. (Verbal, Quantitative, and Non-Verbal) scores for each student, and the end of the school year (1979) achievement marks (with total scores and sub-category scores) for language arts and math.

Data Preparation and Analyses

Varied forms of data resulted from the data collection methods utilized in this study. Each type of data is described and the methods of analyses are outlined.

Verbal Data. Data for each of the 2715 dyadic verbal interactions we're punched onto separate IBM computer data cards to retain both the frequency counts for each of the 60 possible coding categories for <u>each</u> interaction and to preserve the sequence of occurrence of these interactions. Verbal data for eight types of language arts lesson contexts and five types of math lesson contexts were separately computed. This enabled the researcher to summarize for each student his specific verbal interaction occuring within certain kinds of lesson contexts subsumed under the broader subject area, or within the whole subject area, 'or within the total instruction observed. This procedure also provided teacher data on such verbal behaviors as the number of questions posed (eg., process questions or self-reference questions) in each lesson context, in each subject area, or for the total instructional time that was coded. This information was stored in a 48 page computer fife entitled "Verbal".

Percentages or "ratios" for each student on each measure were computed by dividing the student's specific total by the total delivered by the teacher to all students in the class. A separate computer file was created for <u>each</u> of the 26 students in which these ratios were stored. These 26 verbal student files had to be "tailor-made" for 13 of the students as they had been absent periodically during the verbal data coffection (resource room appointments, medical appointments, etc.) and their "possible" total had to be altered by deleting data from the time periods for which they had not been in attendance. This procedure was complicated but was eventually accomplished and thought to represent a more accurate account of the amount and kind of interaction received by

these students. If adjustments had not been made for these absences an average score of 3.8% should have resulted for all verbal interaction behaviors. As a result of the adjustments, this average ratio rose to the 4.1 - 4.2% range for all verbal behavior.

Student ratios could be used singly (as for developing the Case Study profiles which appear in Chapter VI) or combined with other students' data into the various grouping patterns under study. (Two examples of the individualized verbal data files have been included for perusal in the Case Study Chapter VI, specifically for Case Study #1-Sharon, and Case Study #3- Geoffrey.

<u>Attitude Groups</u>. Each student had been place into one of four attitude groups by the teacher and was thereby assigned a 1, 2, 3, or 4 designation which was entered into a column in the computer file called "Students". Whenever computations concerning attitude groups were needed, data for members of the groups were easily retrieved.

<u>Rankings</u>. Data obtained on the 13 Attribute Scales had been converted from a three point scale to a seven point scale because the teacher provided more discriminatory information than had been anticipated. Certain limitations of this procedure are acknowledged but this conversion enabled a number (from 1 to 7) to be assigned to each student for each of the 13 attributes. This information on attributes of students was entered into 13 columns in the computer "Students" file and was used in the same manner as data collected on the six class rankings which also resulted in scores from 1 to 7. All 19 measures were#treated similarly in the analyses that were conducted.

Additional Data. I.Q. scores, age and achievement scores were entered into the computer "Students" file and were used to compute means for the different group memberships and were used to conduct

Interview Data. Interview data were subjected to several analyses. Conners' Micro Content Analysis and Conners' Macro Content Analysis systems were both used on interactive data and a "sieve" code (Guetzkow, 1950) was conducted on all data which included both interactive and non-interactive data. The fourth analysis technique used with the interview data was a comb through of all 246 typewritten pages in order to retrieve all references to a particular student and to combine all such references together into a student file. These teacher comments were compared with the teacher-assigned scores on the 19 dimensions, with the various scores achieved by the student, and with the amount and kind of verbal interaction received by the student in order to develop a composite profile of all available information which could be obtained about each particular student. Such a

profile was developed for each of the 26 students in the class. (These student profiles appear as the "Case Studies" in Chapter VI and are to be used as a glossary when additional information is desired about certain group members).

Correlations

Pearson Product Moment Correlations were conducted in order to ascertain the relationships between presage variables of teacherperceived student attributes, process variables of student verbal

interaction and product variables of achievement and I.Q. tests. Analysis of Variance

The extent of the differences in the various scores received by , boys and girls and by the two reading groups was determined using t tests.

Differences in the various scores assigned to the three general achievement expectancy groups, the three language arts expectancy groups, the three math expectancy groups, and the four attitude-to-student groups were identified by f tests. Scheffé procedures pointed out the extent of these differences between specific groups.

SUMMARY

In this chapter the design of the study and the instrumentation used therein-were described. The phases in the research were outlined as well as the procedures used during each phase. The ways in which the various 'data sources were prepared and analyzed were also presented.

a

CHAPTER IV

TEACHER ATTITUDE TO STUDENTS

In this chapter the information presented is in response to Research Question I-What are the common characteristics of the students assigned to the four attitude to student groups? What factors (perceived personality characteristics) might serve to explain the teacher-expressed attitudes (affective feelings) towards these groups of students?

INITIAL SELECTION OF STUDENTS INTO THE FOUR ATTITUDE-TO-STUDENT GROUPS

Several weeks after data collection began, the teacher was requested to select several students for placement in each of four attitude-tostudent groups. In his original study, Silberman (1969) had requested that only one child be placed in each group. Willis and Brophy (1974) had requested three, and Brophy and Good (1974) obtained as many as five names from the teacher for each attitude group. In this study the teacher was presented with the instrument used by Silberman (1969, 1971) and Willis and Brophy (1974) which asked four questions:

- (1) If you could keep one student for another year for the sheer joy of it, whom would you pick?
- (2) If a parent were to drop in unannounced for a conference whose child would you be least prepared to talk about?
- (3) If you could devote all your attention to a child who concerned you a great deal, whom would you pick?
- (4) If your class was to be reduced by one child whom would you be relieved to have removed?

There was no hesitation in providing three <u>attachment</u> choices or <u>concern</u> choices. There was some confusion about the <u>indifference</u> category (question #2) but eventually he provided three names. Although one student was seriously considered as a possible candidate, no name was provided for membership in the rejection group. At a later date in May when data collection was ending, the teacher was questioned

about indicating possible rejection group students, At this point the student previously considered (case #23) was given "conditional" placement in the group. In June during a follow-up interview the teacher was requested to place all 26 students into one of the four attitude-tostudent groups which he was eventually able to do by assigning three "conditional" placements in all. All of the original placements were reassigned to the same groups. The different stages in the selection of students into the attitude groups and the teacher's comments while performing this task are now presented.

Attachment Group

The difficulty the teacher had with choosing three attachment students was to limit his favorites to just three. "Someone like Joanne (case 25) ...I <u>really</u> like Joanne you know and the thing is we have a lot of interaction after school and so on and she's enthusiastic about everything. ...She's into this and into that and a kid who really has her head together, knows where she's going, what she's doing and yet she's still a kid. She's not a pseudo-adult."

The second choice was Geoffrey (Case 3). "Good old Geoff. I like Geoff because I admire idiosyncratic behavior and Geoff calls his own too. He's <u>bright</u>.... He stands out a little bit and has ...well he's not one of the sheep."

"There are lots of them I'd like to keep for another year... I should put <u>others</u> down because there's a whole mess of them that I would like to keep but anyway Edward (Case 4) is a kid who....You know he's <u>really</u> bright.I shouldn't say this right after saying he's bright but he's the sort of a kid that <u>I</u> was. He's into sports. He's a bit \mathcal{F} a leader⁵ in class...Actually he's younger than a lot of the kids too, by months, not by a long time, but he's younger than most of them... He's quite cooperative."

Indifference Group

The teacher had considerable trouble placing students into the indifference category. He attempted to interpret the literal definition and encountered difficulty. "A lot of parents drop in and come in and so I've had to do it...a possibility?... Well I'm prepared to say a lot of things about Sharon (Case 1) but I don't have a lot of answers. Is that the kind of thing? I could describe a lot of behavior for her mother that I have questions about but...

"Well I'd have to say that Marilyn (Case 12) would be the first choice and I could typify her as being a nice little girl but one of the ones you can... Its partly because of her location in the class. She's at the back. It's partly because she's not...you know someone like Nicholas (Case 19) <u>demands</u> attention. She doesn't. She's not that kind of individual ... I have a general idea of where her strengths and weaknesses are generally but if somebody wanted to pin me down about something specific I'd have to say 'Hey, let me check my records."

"Someone like John (Case 17). I had had more contact with his parents and have talked to them but well, you know, some kids are not leaders and I'm not saying he <u>isn't</u>, or won't be, but he's younger... He's not the most visible person in class but he's somebody I would typify as being a reasonable, good average student; nothing outstanding but nothing I need to worry about a lot. The only thing I might say about him is with respect to the maturity thing...but in terms of "academics, you know , if I wanted to put my fingers on where he's at I might have a little more trouble doing that." He considered Sharon again. "Sharon. I know pretty well where she's at in terms of her academic achievement but the kinds of questions that remain unanswered... Well I find a lot of her behaviors a little unusual, shall we say. She's a little...uh, sneaky...at times really negative about things, at times really turned off on friends, on things that are happening, and yet at other times the opposite end of the stick so that she's a little "enigmatic" I guess is the word. And I guess I haven't got a real "fix" on her and that's why I would bring her up."

Concern Group

υč.

The teacher had three ready choices for the concern group. "I guess my first choice , the person in class who in terms of personality...Well I'll just put down Bonnie (Case 18) and tell you why. She's a bright child. She's hyperkinetic and on medication for that...She's had terrible eye/hand coordination difficulties. She's a frustrated girl in a lot of ways because she knows that she's bright but at the same time the prospect of keeping up with the brighter members of the class is something that she just has to say 'Well mentally I can do that but I'm unable to... He sometimes allowed her to do less work volume. "She knows. She knows exactly what's going on and she finds that quite frustrating... We've had some frank discussions about some of the problems she has... Recently she's been really upset with some peer group things as well. Kids are not as kind to her as they might be. At the same time it's a two-way street, it always is with kids. But anyway she's a kid that I think for example, if I could spend all my time with her that she could really move along. "Another kid that I could spend a lot of time with... going through

my mind here, someone like Ian (Case 14)...mostly cause I like the kid I guess and because he has some problems. (He had repeated Grade One and one month prior to the beginning of the study his mother had died following a lengthy illness). "Ian isn't the brightest kid across the board, he <u>isn't</u>. Not withstanding that he's a nice lad to work with. He's a kid who I think with a lot of attention, a lot of drill, well a lot of work...well he's not going to be a 'scholar' but obviously I think a one on one situation would improve whatever skill level, obtain a somewhat higher skill level, whether it would be retained I don't know but anyway he's the one I would try."

"Let's see...I'm sitting here thinking I've got kids like Tom (Case 15) who really needs a lot of work and tries pretty hard. He's willing to, at times intermittently at least, he's willing to really force himself to concentrate." (The teacher indicated that there was a possibility that Tom might have to repeat the grade.) " I think that both these Kids, Ian and Tom, are really approachable young people who are...well I've taught in other schools where kids have learning problems but they also have really severe personality problems. These kids have learning problems but they're 'fairly open about them...The important thing I think is to say 'Well just cause you don't know math doesn't mean you're dumb, it means you don't know <u>math</u>,' this kind of approach. And well if I had lots of time those are the kinds of people that I would spend a lot of time working with."

Rejection Group

In March the teacher provided no names for the rejection group. "Geez, you know another year I would have people for you. I'm not sure

that I can honestly say... There isn't a person there that I would like to see leave the class..."(Concerning Pamela, Case 23, as a possibility) ... "Okay if you pressed me to choose someone today I would say 'no'. Yesterday I would have said 'yes'- well not yesterday but last week I would have said 'Yes'... I don't feel terribly negative about <u>any</u> of them. I really don't. That's a genuine thing. In previous years I could have given you a list of five."

FINAL SELECTION OF STUDENTS INTO THE FOUR ATTITUDE-TO-STUDENT GROUPS

In May the topic was reopened. "Who would I like to 'unload'? Um... you know there are so many ways of looking at that. In terms of the amount of time that I'm required to spend with an individual I could say 'Well this could lighten my load. Like I could think of two. Someone like Ian (Case 14) and someone like Bonnie (Case 18). Those two for that reason ... I suppose the one who would be most marginal in terms of personality traits would be Pamela (Case 23).

Because Sharon (Case 1) and Keith (Case 2) had received low scores on the "attachment" attribute (Item 10 on the 13 Attribute Scales) their names were mentioned by the researcher. "Well the thing is with ones like Sharon and Keith they're a kind of fuiet people. I don't find their personalities as attractive as some of the other children but they're not disruptive, that sort of thing."

In May the indifference choices were discussed again. "Of all three...you know they don't stand out in class. They sort of, in each case, John, Marilyn, and Sharon... I guess in a sense their demands on me are not as great. They don't stand out positively and they don't stand out negatively either. I'm just sort of ... 'blah' about them. That's my perception and that's my feeling... During the May interview the teacher was informed of the general characteristics of indifference students from previous research and of the theory (Brophy and Good, 1974) that the negative attitudes of indifference students often "conditioned" teachers to become "turned off" and to avoid them. "You know I can almost buy that. In a sense I think of someone like Sharon (Case 1). She's somebody who, when you ask kids to get out books... 'groan', ask kids to do this or do that and it's never approached with enthusiasm. I shouldn't say <u>never</u> but frequently enough that it can be 'off-putting' to the teacher. 'If that's how much you care then'......"

You know I could buy it for Marilyn and Sharon but John... I don't know whether my perceptions change or what happens. There can be that as well you know. He's occasionally almost in a purposeful way, he's negative and thinks it's cute to tease a little bit. But he's not in quite the same category as the other two who tend to have somewhat negative approaches to lots of their tasks, or appear to have... You know the kinds of demands he (John) makes when he wants to volunteer. It's not sufficient to put up a hand it has to be 'oh, oh, oh', this sort of thing... Sort of lacking in self control."

Keith (Case 2) received a low score on the "attachment" attribute and the teacher was asked if Keith was viewed in this manner. "Someone like Keith, a quiet boy, but he's not a Kid who completely lacks enthusiasm for anything... I don't see him that way." Thus in the May interview the three first choices for <u>attachment</u>, <u>concern</u>, and <u>indifference</u> remained the same and Pamela had been "conditionally" designated as the only <u>rejection</u> student.

In June the teacher was requested to attempt to place all 26 students into one of the four categories. He was given the class list to work from so that no student would be overlooked. He was informed that if any student would fit into two categories he could place the student in both. "They won't... Gosh it's hard you know... I find this really difficult in a sense. I have one possibility here of keeping someone because I think he's super but no possibility for someone I feel 'blah' about. Like I know a lot about them. I'm not 'concerned' about them. I wouldn't 'reject' them but at the same time I don't feel really 'attached' to them. Do you see what I'm saying? I've got several of those. Like one I see is Keith. That's why I stopped, you see. So what do I do with him?" (Rejection?) No reply. (Indifference?) "but I know a lot about him... If you took this out, the definition, and just had 'indifferent' I could put him in <u>indifference</u> because 'indifference' that's neither positive or...well it's sort of negative, but it's a kind of a default."

"Sharon, if we can reject the definition, belongs in here. Marilyn would certainly be put in this category." (with the actual definition?) "with the definition although now she <u>doesn't</u> really any more because since it was brought to my attention, you know... you start doing some work that way. "

"I'd sort of put John into this list without the definition. Mason (Case 20) was added to the group. Keith's name remained as one of the few unfiled names. It appeared that the teacher felt rejection too strong a reaction to him. "He and I... It's kind of funny and it's probably a genuine change in my attitude toward him too. We've talked to each other a bit more. He's been working a little harder. He's done very well. He seems, I don't know, a little happier with the way things are going... something along these lines. That can change my approach to things."

The teacher placed Keith into the indifference group but without the definition. Thus the only indifference student for whom the provided definition seemed to fit at all was Marilyn and as the teacher indicated, he had begun to find out more about her after he had realized he lacked such information. This resulted in five students being placed into the indifference group in all.

A total of 16 students were placed into the attachment group. "Its easy to put these ones down. There's a whole crew of them." In all there were eight boys and eight girls. Nicholas (Case 19) was the last student placed in this group. "Someone like Nicholas... Where will I put him? I'm not indifferent about him. I'm not sure I'd want him for another year. I don't reject him. He almost doesn't fit. Like he's not one of <u>these</u> kids (indifference and rejection) who are sort of ones that I see without marvellous personalities and that sort of thing.., I guess I'd be closer to attached to him more than anything else- but not as close as some of these others. Maybe I'll do this to him...*..(conditional placement in the attachment group).

The teacher assigned Pamela (Case 23) to the rejection group with a "not-definite, conditional" rider attached. As well..."I'd put Sonia (Case 26), the new girl, in that sort of category... She's sort of a funny kid... I guess the 'chemistry's ' wrong or something... "

No more children were considered for or added to the concern list. In fact by this time in June, Ian had just moved to British Columbia. The teacher noted the great improvements that Tom (Case 15) had made recently and that his achievement marks were fairly good. "I wouldn't

fail him now because his math scores are high enough. He's got a lot of problems and he's going to need understanding teachers but I can't fail a'kid and hope that... I wouldn't do it anyway because he's been working very hard. I've had quite a lot of contact with his parents and they've really been working with him hard at home. He's been getting some positive results and feels a lot better about what he's doing, of course. If you know your basics and drill, drill, 'drill...That has made all the difference in the world."

"Bonnie, I'm really concerned about her. She's a little different than with Tom. I'm really concerned that she's emotionally disturbed... I wish I could put my finger on it but she's just so up and down and such wide swings in mood and so on. I've had a fair amount of contact with Mom and she seems quite supportive , and you never know... You can't get inside a person. There's no lack of money in the family. They have just put in a swimming pool in their place and we are going over there for the year end party which is all very nice and lovely but the girl is very unhappy quite frankly. I'm really concerned about her but I don't know if all my attention would do any good...She's different than Tom, you see. Tom is kind of a rewarding experience right now. I'm really concerned about her state of mind?"

CHARACTERISTICS OF THE GROUPS

In the attachment group the first three choices who were originally made in March and verified as the first three choices in June, were noted. In the following discussion they are sometimes isolated for comparison with the total group of 16 attachment students (which includes them) and with the other attitude groups as well. The 16th assignment to this group was "conditional" because the teacher could not decide on a better

fit in another group.

The indifference group was comprised of three boys and two

girls.

Membership in the concern group totalled three students. A girl, Bannie, who was hyperactive and who had emotional as well as medical problems which led, in turn, to academic problems in some subject areas and two boys who were low achievers. One of these boys, Ian, had repeated Grade One and was experiencing emotional and adjustment problems directly due to a recent death in his immediate family.

Only two students were placed into the rejection group. Both were girls and both were "conditional" placements for the teacher wondered whether this might be too negative a reaction on his part. One of these students was a new girl, Sonia (Case 26) who joined the class in early April and who, by June, had earned a negative group castigation from the teacher. From time to time both girls would cooperate and put forth a good effort but most of the time they carned rejection group status.

CORROBORATION OF TEACHER AFFECT TOWARD THE FOUR ATTITUDE-TO-STUDENT GROUPS

Substantiation of attitude group placements is plentiful in the many general and stimulated recall interviews held with the teacher. In addition, two of the traits measured on the 13 Attribute cales were really measures of teacher affect toward students. Brophy et al (1976) had relied solely on these measures for obtaining their measures of teacher affect toward students since they did not ask teachers to assign students into the four attitude-to-student groups. The scores received by the existing attitude groups in the present study should help to corroborate the teacher's affect toward them that was responsible for their original placement into the four attitude groups.

Attachment

Item #10-Attachment read as follows: "Would like to have removed from class versus would like to keep for another year for the sheer joy of it. The scores received for the "attachment" attribute ranged from one to six. (A score of one was high). These scores are presented as group scores and appear in Table 1.

	Ta	able l	¥	
ATTITUDE GROUP	SCORES	ON THE	ATTACHMENT	ATTRIBUTE

			· · · · · · · · · · · · · · · · · · ·
first 3 choices	1.00	Girls	Poys
attachment .	1.687	2.0	1.37
indifference	4.60	4.0	5.0
rejection	4.50	4.50	-
concern	3.66	4.0	3.5
whole class	2.69	2.84	2.53

The whole class average was 2.69 and it appeared that he was slightly more attached to the 13 boys than to the 13 girls. Not unexpectedly the first three attachment students received an average score of 1. By definition the students in the rejection group would not be recipients of teacher attachment but in the present study indifference students received even lower scores than rejection students. Perhaps this was due to the "conditional" assignment of students to⁶ this rejection group.

In general the information gained from this "attachment, like to keep" measure provided support for the teacher's placement of students into the four attitude-to-student groups.

Concern

The second measure dealing with teacher affect was Item #11 from the 13 Attribute Scales. This measure dealt with the amount of concern the teacher felt about the child. Specifically Item #11 was worded: "Doesn't require special attention versus concerns me a great deal. I would like to be able to devote more attention." A score of one was high. The average scores received by the four attitude groups are presented in Table 2.

Table 2	
---------	--

ATTITUDE GROUP SCORES ON THE CONCERN ATTRIBUTE

first 3 choices	6.66	Girls	Boys
attachment	6.14	6.0	6.25
indifference	3.40	2.0	4.3
rejection	1.50,	1.50	-
concern	1.00	1.00	1.00
whole class	4.66	4.30 [%]	5.00

As expected the teacher was highly concerned about the three students he assigned to the <u>concern</u> group. He was also very concerned about the rejection group students. This concern is not restricted to academic matters and in the case of these two rejection girls his concern appeared to extend to their home background situations as well as to the academic difficulties they experienced. Coincidentally, perhaps; these were the only two students in this classroom to come from a broken home situation. In one case, Pamela (Case 23) was forbidden by her mother to see her father who did not have legal visiting rights and privileges but with the aid of an older sister she did manage to see him every week. A fair amount of tension and apprehension was apparent in this Grade Three student's behavior. In the other case, Sonia (Case 26), a series of moves every year seemed to be part of the normal lifestyle. She joined the class in early April and subsequently, as traced by the researcher in an attempt to get a missing piece of information, had changed schools in September, 1979 and again in January, 1980. That was a total of four schools in two years. Different coping skills are required of a child who has to make a fresh start every six to eight months.

The following exerpt from the stimulated recall interview number six indicates that the teacher was beginning to form an impression of Sonia some two weeks after she had joined the class... "Well okay, Here's this little interaction I have with Sonia. I guess I sort of think...Iguess it maybe irks me a little bit to have someone tell me 'I hate doing that'...that sort of thing. Just challenging... The kids in this class are largely from stable home situations where respect is an important thing and there's lots of money and they're kind of status quo people by and large and she may be somewhat outside that norm in this class... but she <u>seems</u> like a nice enough kid...I'm not concerned that I'm going to have any real problem with her I just think that her approach and attitude are somewhat different than many of the other kids in the class." Researcher- Would you draw any similarities between Pamela and Sonia in that way?

"Well to the extent...I mean they're much different children, but to the extent that there are things that I can notice about their <u>behavior</u> that is in some ways different from most of the other kids and both of them are from broken home situations. I suppose there's <u>that</u> similarity..."

đ.

Researcher- Thinking specifically of the amount of respect.'..would the rest question your authority on occasion just as much? " I think this far, maybe it's early to say. I would say that I haven't run into that at <u>all</u> in this class until now. Either that or it's been in a friendlier manner or something, I don't know. It hasn't been as noticeable or maybe as overt or something... It's unlikely that she and I will have difficulties that reach crisis proportions or anything like that."

Willis and Brophy (1974:526) stated that the rejection boys "were likely to come from intact families in which both parents were living..." and that "concerning girls in the rejection group the teachers frequently mentioned negative family patterns (broken home or poor parental cooperation)." Although no rejection boys exist to make the full ______ comparison these are the only two students in the class (excepting Ian, whose mother had recently died) who came from single parent homes.

The two attitude-like measures for which numerical scores existed corroborated the memberships in the attitude groups to a satisfactory degree. The amount of attachment and concern the teacher expressed toward each student supported the teacher placement of these same students into the attitude groups. The rejection students were "conditionally" placed as they were not strongly rejected and they were of high concern to the teacher. Their scores on attachment and concern measures are less conclusive for these reasons.

The concern measure then confirmed the highest degree of concern was for <u>concern</u> students. A great deal of concern was felt for indifference students as a group but indifference girls in particular appeared to concern the teacher a great deal. Very little concern was expressed toward attachment group members.

PERSONALITY ATTRIBUTES OF THE FOUR ATTITUDE-TO-STUDENT GROUPS

In this section the average scores for the teacher-assigned attributes are presented for the four attitude-to-student groups. Originally these scores were assigned by the teacher to individual students and these scores were combined to provide an average attitude group score. The teacher rank ordered the entire class on some of these student characteristics and was asked to assign a score ranging from one to seven (low) while thinking only of a particular student for the remainder of these attributes. By combining these two forms of data, a general indication of the teacher's assessment of students on these measures is provided.

Two of these measures, "cooperation" and "maturity", were duplicated in that they appeared both as a class ranking scale and also as items #5 and 6 on the 13 Attribute Scales. Comparing these two sets of scores gave some indication of the consistency of the measures for a period of two weeks intervened and two slightly different methods were used to collect both sets of data. The teacher was asked in May to clarify any discrepancies between the sets of scores. The correlations using Pearson Product Moment Correlations to compare scores for these two measures of "cooperation" and "maturity" were .817 p<.001 and .885 p<.001 respectively.

All teacher assigned scores for the various student characteristics were obtained from the middle to the end of March (excepting those for Sonia who joined the class later on). These scores are assumed to represent the teacher's assessment at that time. In addition these scores are assumed to have remained relatively stable across the three months of the study. The four attitude-to-student groups' average scores on all personality and academic attributes as provided by the teacher in this study are compared with similar information and findings from previous research on attitude groups. For each student descriptor the attitudeto-student group averages are noted, any within-group sex differences are presented, and any general class sex differences are identified along with class mean scores for all descriptors.

Work Habits and Effort

The average attitude group scores for "motivation to do school work" (Class Ranking #6), "careful/deliberate worker (Item #2) and "persistence" (Item #9) are presented together in Table 3. They give an indication of the teacher's assessment of the effort expended and the general work habits of the students in the four attitude-to-student groups.

	motivation	careful	persistence
first 3 choices	2.0	1.3	1.0
attachment	2.4	1.7	1.4
indifference	4.2	4.0	4.2 .
rejection	5.5	5.0	5.5
concern	6.7	6.7	7.0
whole class	3.5	2.9	2.8
girls	3.7	3.2	2:9
boys	3.3	2.6	2.8

ATTITUDE GROUP SCORES FOR WORK HABITS AND EFFORT

Table 3

F tests using Scheffé procedures which were used to find the

differences among the four attitude groups found significant differences between the attachment group scores and all other attitude group scores

for all three measures.

The effort exerted by a student is seen to be personally rewarding to a teacher. This teacher was asked if the effort a child made was important to him. "Sure that's part of the interaction, maybe part of the way in which personalities mesh. You cannot avoid the thought that if someone is putting forth some kind of effort to respond to your work then the likelihood is that you are going to feel better about that and spend more time with him...There may be mitigating circumstances. I think of someone like Ian (Case 14) ... It's also important for the brighter kids to try. I think, for example, if there's a bright child who puts in little effort the kind of interaction they would have with the teacher would be different from the child who is working hard...The effort is a reward for the teacher, no doubt about that. That encourages you to extend further effort."

Weiner (1979:17) stated, "Surely a teacher will not particularly like a student who does not try, and failure perceived as due to a lack of effort does not elicit sympathy."

Part of the liking for the attachment group students is due to the consistently good effort they display. In the group of 16 attachment students there were some slight sex differences that favoured the girls. The teacher credited them with better work habits and with better effort expended over all. These findings are consistent with Brophy and Good(1974) and Brophy et al's (1976) findings for attachment group girls and the attachment group in general.

	motivation	careful	persistence
attachment girls	1.9	1.3	1.1
attachment boys	. 3.0	2.1	1.6

Table 4

ATTACHMENT GROUP SEX DIFFERENCES FOR WORK HABITS AND EFFORT

Willis and Brophy (1974:524) noted that attachment boys had "high ability ...and rewarded the teachers by being dependent on them and by doing well in school work" but they said of attachment girls specifically that they were 'achieving" and "good independent workers" suggesting somewhat more initiative on the part of attachment girls. The present findings tend to support some differences favoring attachment girls on the work habits and effort dimensions.

Indifference girls and boys showed some differences as well. This information is presented in Table 5.

	motivation	careful	persistence
indifference girls	5.0	4.0	5.0
indifference boys	3.7	4.0	3.7
(A score of one is high)			

Table 5

INDIFFERENCE GROUP SEX DIFFERENCES FOR WORK HABITS AND EFFORT

Indifference boys were seen to be more motivated and persistent than the two indifference girls. Negligible sex differences existed for the concern group and the rejection group consisted of two girls only.

If all three sets of scores are collapsed and averaged, a summary view of effort and work habits as assessed by the teacher would result in the findings presented in Table 6.

			Table	≥ 6	•			
	ATTITUDE	GROUP	SCORES	ON THE	COMBINED	WORK	HABITS	
. •				T MEASI			an an Th	

attachment group	1.83
indifference group	4.13
rejection group	5.33
concern group	6.80

Scores for all groups would appear to fit patterns established in other research studies. Although positive attitudes to school were not measured as such in the present study an overt indication of same would be effort expended. As Brophy and Good noted (1974:153) noted, concern students were "in need of encouragement" and were "notably lacking in self confidence" which would interfere with the amount of effort put forth by these students.

For the class in general, the teacher assigned higher scores on all three traits to the 13 girls and the 13 boys were seen as somewhat less motivated, less careful and less persistent.

Age and Maturity

In March the average of of students in the class was 8 years 7 months. Ages of the students ranged from 8 years 1 month to 9 years 5 months.

The youngest attitude group was the indifference group with an average age of 8 years 5 months. The eight attachment girls were actually younger with an average age of 8 years 4 months but in all the attachment group age average was 8 years 6 months. The two rejection girls averaged 9 years 0 months in age. The concern group averaged 8 years 8 months but because Ian had repeated, his age was 9 years 5 months compared to the two other concern group members; Bonnie at 8 years 3 months and Tom at 8 years 4 months.

All girls in the class had an average age of 8 years 6 months and the boys averaged 8 years $\frac{7}{2}$ months in age.

Teacher-assigned ratings for "maturity" resulted in the attitude group averages shown in Table 7. The first maturity rating obtained was provided in the form of a class ranking (#5) and the second

	1			_		
2	ь		\sim			
а	U	1	с.			

	Age in Months	Maturity I	Maturity II
first 3 choices	104.3	2.3	1.3
attachment	102.0	2.6	1.8
indifference	101.3	5.2	4.8
rejection	108.0	5.0	5.0
concern	104.0	6.0	5.7
whole class	103.61	3.69	3.07
13 girls	102.61	3.46	2.92
13 boys	103.61	3.92	3.23

ATTITUDE GROUP AVERAGES FOR AGE AND MATURITY

set of maturity scores were derived from Item #5 on the 13 Attribute Scales which the teacher completed for each individual student about two weeks later. Both sets of scores which purport to measure the same student trait favor the students assigned to the attachment group with the attachment girls receiving slightly higher ratings than attachment boys.

The three indifference boys scored an average of 5.0 on both maturity measures but the two indifference girls received a lower (than boys) score on the maturity I scores and a higher (than boys) score of 4.5 on the maturity II scores.

Rejection girls received 5.0 on both measures. The girl in the concern group received a 6 and then a 5 on the two scores for maturity.

The Pearson correlations used to detect the relationship between the Maturity I and maturity II scores were:

.885 p<.001 for the whole class

.888 p<.006 for all girls

.906 p <. 001 for all boys

The teacher was asked in May to clarify several discrepancies in

these rankings. For example, Brenda (Case 7) had received a 2 and a 5 on the maturity ratings. " I would use the upper one...Sometimes in our one on one interactions after school she's fairly immature but when it comes down to her approach to her work, the academic side of things, then she's very interested, persistent, works very effectively. These are all signs of maturity. So one one day I must have been thinking of some interaction that we had for she is the kind of kid who will stand two inches from your nose to make sure you heard her and go on and on. I probably was thinking more of an after-school thing."

Cooperation

Scores on measures of cooperation were obtained from the teacher. Cooperation I was the 6th class, ranking and cooperation II appeared as Item #6 on the 13 Attribute Scales. The correlations for the two cooperation ratings were:

.817 p<.001 for the whole class
.806 p<.001 for all girls
.904 p<.001 for all boys

The teach apparently assessed the attachment group students as more cooperate and compliant than the other three groups. Overall rejection groups scores are the lowest and this result concurs with

findings from previous research. Indifference students are not seen to be particularly cooperative. Concern students are almost as low in assessed cooperation as the rejection group students. Other studies imply that concern children are more cooperative and "conforming to class rules" (as reported in Brophy and Good, 1974:188)

. than the findings from this study indicate.

	 Cooperation I 	Cooperation II
first 3 choices	2.7	1.3
attachment	3.0	1.6
indifference	4.2	3.2
rejection	6.0	5.5 .
concern	6.3	4.0
whole class	3.84	2.50
13 girls	3.46	2.61
13 boys	4.23	2.38

Table 8ATTITUDE GROUP SCORES FOR COOPERATION

In the first set of scores for cooperation the 13 girls are seen to be more cooperative but the second set of scores which are closer in numerical value indicate that the teacher viewed the 13 boys as more cooperative overall.

Intelligence and I.Q.

The teacher ranked the whole class for his assessment of their "general intellectual ability" (Class Ranking #1). Scores for the attitude groups are presented in Table 9. In addition the three I.Q. scores (Verbal, Quantitative, and Non-Verbal) have been combined and averaged to produce a "total I.Q.score" and are presented as well as an "objective anchor" (Brophy et al, 1976) for the teacher's assessment of ability level.

The teacher assessed the first three choice attachment students as very high in general intellectual ability and indeed their total I.Q. score was well above the class average of 112.2. In general the 16

	Teacher-Assigned General Intellectual Ability	Total I.Q. Scores	
first 3 choices	1.7	121.2	
attachment	2.6	118.5	
indifference	5.2	106.13	
rejection	6.0	99.3	
concern	6.0	97.7	
whole class	3.76	112.28	
13 girls	4.07	111.94	
13 boys	3.46	112.6	

Table 9ATTITUDE GROUP SCORES ON INTELLECTUAL ABILITY AND I.Q. MEASURES

attachment students were rated above the class average by the teacher for general intellectual ability and the Total I.Q. score for the group seemed to justify his high assessment of them.

His assessment of the indifference group's intellectual ability may have been somewhat low since a score of 4 would have been "medium" or average and their I.Q. average of 106 indicates average ability overall but certainly less than average for <u>this</u> particular class which was the teacher's reference basis.

The rejection scores on general intellectual ability are essentially the same as the concern group scores. In previous research findings the rejection students, although rated low by the teacher in intellectual ability, are sometimes found to be of average or better intelligence. In the present study their Total I.Q. scores are still within an "average" range but are much below this class' average of 112. Concern students, as in most other studies of attitude groups had the lowest ability levels and were perceived to have low ability by their teachers. Within the groups there are some interesting sex differences and these are presented in Table 10.

|--|

SEX DIFFERENCES IN ATTITUDE GROUPS FOR GENERAL INTELLECTUAL

ABILITY AND TOTAL I.Q. SCORES

c.	General intellectual Ability	Total I.Q. Scores
8 attachment girls	3.0	118 .
8 attachment boys	2.3	119
2 indifference girls	6.0	102.2
3 indifference boys	4.7	108.7
l concern girl	5.0	108.0
2 concern boys	6.5	92.6
(A	score of 1 is high)	

The teacher assessed the attachment boys and the indifference boys higher than the girls in those two groups on the dimension of general intellectual ability. The total I.Q. scores show that the eight attachment boys have made marginally higher scores then the eight attachment girls but the three indifference boys' average total I.Q. is about six points higher than the average total I.Q. of the two indifference group girls. The concern group differences might be explained by the fact that Bonnie's (Case 18) placement in this group was largely because of medical) social, and emotional problems in addition to certain academic problems.

I.Q. scores in the class at large ranged from 84.6 to 135.0 and the class average for total I.Q. scores was 112.28.

The teacher assigned a higher general intellectual ability score for the 13 boys in the class (3.46) than for the '13 girls in the class (4.07). However, the average boys' total I.Q. score is only slightly higher at 112.6 than the girls' I.Q. average of 111.9. Contrary to many other research studies (as reviewed in Brophy and Good, 1974: 204) in which the girls received higher intellectual assessments than boys, the boys in this particular class were rated somewhat higher by the teacher on this dimension. Most other studies used female teachers as subjects. In the Brophy et al's (1976) study, for example, the 27 teacher subjects all were female elementary teachers.

Creativity

The creativity descriptor appeared as Item #7 in the 13 Attribute Scales. It was precisely worded as "non-creative or unimaginative versus creative and imaginative". It is somewhat related in nature to the general intellectual ability dimension and so is discussed next. The non-verbal I.Q. scores have been presented in Table 11 as well as the creativity scores for they represent the best objective and corroborative information available for the teacher-assigned creativity scores.

Č.						
¢,	Creativits				rbal I.Q	· · · · · · · · · · · · · · · · · · ·
first 3 choices	1.3	girls	boys	122.6	girls.	boys
attachment	2.3	2.87	1.62	115.93	114.37	117.5
indifference	4.2	5.50	3.33	107.6	100.5	112.33
rejection	5.0	5.0	-	104.0	104	- · ·
concern	4.0	2.0	5.0	99.0	102.0	97.5
whole class	3.03	3.54	2.53	111.46	109.6	113.23

Table 11

ATTITUDE GROUP SCORES FOR CREATIVITY AND NON-VERBAL I.Q. SCORES

The relative accuracy of the higher creativity scores assigned to boys in the attachment and indifference groups (compared to the scores

given to girls) are reflected in the objective scores on the non-verbal I.Q. test for scores earned by the boys on this objective test are higher than those for girls in the attachment and indifference groups. Bonnie (Case 18) the concern girl, was assigned a fairly high score for creativity but her non-verbal I.Q. score of 102 was much lower than the class average non-verbal I.Q.score of 111.46. However when faced with any pen and paper task, especially one with an imposed time limit she was often unable to complete it. "She's a bright child... except that the physical aspect of writing is so difficult for her that it can be frustrating ... Verbally she's fine, creative, imaginative, perceptive, all of those things but if you asked me how she is going to achieve on a year-end test which requires other skills as well... Verbally she is reasonably creative. She can relate some interesting stories and experiences in an effective way. Her word usage is pretty solid so she expresses herself relatively well when I can persuade her that this is something she ought to do... If she could keep up with the written aspect I could probably have her in the other (good reading) group."

Generally speaking the teacher assessed the 13 boys in the class as more creative (2.53) than the 13 girls (3.53) and the non-verbal I.Q. scores of 113.23 for the 13 boys compared to the score of 109.69 for the girls would indicate that his perception of their relative creativity was fairly accurate.

Calm, Good Self Control

Item #1 on the 13 Attribute Scales dealt with the calm/good self control trait. "Restless, highly active versus calm, good self control" was the exact wording for this measure. Scores for the four attitude

	Calm		
first 3 choices	2.0	Girls	Воув
attachment	2.1	1.50	2.75
indifference	3.2	3.5	3.0
rejection	5.5	5.5	
concern	6.0	6.0	6.0
whole class	3.04	2.76	3.30

Table 12ATTITUDE GROUP SCORES ON THE CALM DIMENSION

\$2

Attachment students, particularly the eight attachment girls, received the highest scores on the calm attribute. Previous studies described the concern students as "restless" and the rejection students as "notably restless". These two groups of students in the present study received the lowest scores on the calm dimension but the "conditional" rejection students' scores were not as low as previous findings would suggest for true rejection students. Pamela (Case 23), one of the rejection girls <u>did</u> receive a 7 on the calm attribute but Sonia (Case 26) on the basis of having her in the class for less than a month, was assessed as 4 by the teacher. The two indifference girls were rated as less calm by the teacher than the three indifference group boys. The concern group students received an assessment of 6 uniformly.

The teacher rated the 13 girls in the class (2.76) as calmer overall than the 13 boys in the class (3.30) resulting in a class mean score of 3.04 for this descriptor.

Noticeable

The noticeable descriptor (Item #12 on the 13 Attribute Scales) was worded "not noticeable versus stands out, very noticeable". Table 13 presents the teacher-assigned scores for the four attitude-to-student

groups and for the class in general.

	Noticeabl	e	
first 3 choices	1.7	girls	boys
attachment	2.2	2.5	1.87
indifference	4.8	4.5	5.0
rejection	2.5	2.5	-
concern	3.0	5,0	2.0
whole class	2.80	3.0	2.61
A score of 1 is hig	h.		

 Table 13

 ATTITUDE GROUP SCORES FOR THE NOTICEABLE ATTRIBUTE

The lowest overall score on this dimension was received by the indifference group students as verified in previous research. (Brophy and Good, 1974:133). This attribute is one of the major descriptors of indifference children and was originally thought (by Silberman, 1969) to be the main reason why teachers interacted so infrequently with this group of students. Within-group differences show that the three indifference boys (5.0) are even less noticeable to this teacher than the two indifference girls (4.5).

Attachment boys (1.87) were fated as more noticeable than attachment girls (2.5) and the two concern boys were seen to be much more noticeable than Bonnie, the only concern girl.

Generally all boys in the class (2.61) were assessed by the teacher as more noticeable than all girls (3.0) in the class in keeping with previous research findings (Brophy and Good, 1974:231). ٩.

Eye Contact

Item # 13 on the 13 Attribute Scales was "averts eyes versus looks you straight in the eye." The lowest scores went to the indifference students on this trait. The indifference boys (4.0) averted their eyes even more than the indifference girls (3.5). Brophy and Good (1974: 153) had described how the indifference students "notably avoided eye contact." Scores for the attitude groups for eye contact are presented in Table 14.

Table 14

Eye Contact			
first 3 choices	1.3	girls	boys
attachment	1.6	1.62	1.62
indifference	3.8	3.50	4.0
rejection	3.5	3.5	-
concern	3.3	3.0	3.5
whole class	2.38	2.30	2.46

ATTITUDE GROUP SCORES FOR THE EYE CONTACT ATTRIBUTE

Attachment boys and girls received equal scores on this dimension while concern and rejection group scores were almost as low as scores received by indifference groups. Overall the boys and girls in the class were rated fairly evenly on this measure.

Attractiveness

•The "unattractive versus attractive" measure was Item #8 on the 13 Attribute Scales. Scores received by the different attitude groups and the class as a whole are presented in Table 15.

The class average for all 26 students was 1.65 with the boys somewhat higher than the girls in perceived attractiveness. The first three attachment students were given the highest possible scores as were
	· · · · · · · · · · · · · · · · · · ·		
first 3 choices	1.0	girls	boys
attachment	1.2	1.37	1.0
indifference	2.8	2.0	3.33
rejection	1.5 ,	1.5	_
concern	2.3	4.0	1.5
whole class	1.65	1.69	1.61

Table 15ATTITUDE GROUP SCORES ON THE ATTRACTIVENESS ATTRIBUTE

the eight attachment boys (who included two of the three first choice attachment students). Attachment girls were seen as a bit less attractive.

Rejection group girls were rated higher than the class average. The concern group average is lower than class average because the score assigned to Bonnie was 4 compared to the two concern boys' scores averaging 1.5.

Indifference childrenwere seen by the teacher as the most unattractive group and received an average score of 2.8 on this measure. The three indifference boys (3.33) were perceived by the teacher to be far less attractive than the two indifferent girls (2.0).

Нарру

The teacher assigned ratings for how "happy" he perceived the children to be. This measure appeared as Item $#3^\circ$ on the 13 Attribute Scales and was worded simply as "unhappy versus happy." The resulting group averages are presented in Table 16.

The teacher rated the attachment group high on this attribute with no differentiation between the attachment girls and the attachment boys. Indifference group average for this attribute was below the class average with indifference girls rated as less happy than the three

first 3 choices	1.3	girls	boys
attachment	1.5	1.5	1.5
indifference	3.6	4.0	3.3
rejection	5.0	5.0	-
concern	5.7	7.0	5.0
whole class	2.65	2.84	2.46

ATTITUDE GROUP SCORES ON THE HAPPY ATTRIBUTE

Table 16

indifference boys. The individual scores of 3 and 7 in the rejection group averaged out to 5 for the two members of the group. The concern group averaged 5.7 for although both Bonnie and Ian received a teacher rating of 7 for being happy, Tom (Case 15) was assigned a higher score of 3 for this dimension. Poor achievers tend to be unhappy and this is partly due to a lack of success in their school work. Two of the three ; concern children were affected by additional problems which accounted for the teacher-assigned scores of 7 for these two students.

All girls in the class were rated as a bit less happy (2.84) than the 13 boys (2.46) in this study.

SUMMARY

The 16 attachment students in the present study were of average age, were assigned high expectancy scores for achievement in both subject areas, were rated as highly intelligent, mature, cooperative, and attractive. The teacher assigned high scores for their workrelated behavior and effort expended. In short they were model students who behaved well, tried hard, and were seen as having academic potential. The first three choices of attachment students received even higher scores on all measures and a general positive halo effect appeared to be associated with these children.

The five indifference students were younger than most children in the class, were perceived to have lower than average ability, and received lower than average achievement expectancy scores. They were perceived to be less persistent, careful, and motivated. The teacher felt that members of this attitude group were not making an adequate effort. They were viewed as less cooperative and mature than most children in the class. They received very low scores for being noticeable, maintaining eye contact with the teacher, and for being liked by the teacher.

The two conditional rejection girls were viewed as attractive, fairly noticeable, uncooperative, immature and restless. They received low scores on the work-related attributes as little effort on their part was perceived by the teacher. He was, however, very concerned about their academic progress. He held low achievement expectations for them. They were not well-liked by the teacher and were seen to be basically unhappy students.

The three concern students were perceived to exhibit very poor initiative and poor work habits and were assigned low achievement expectancy scores by the teacher. Generally speaking they were immature and uncooperative and fairly unhappy. The teacher's concern for their mental and academic well-being was considerable. Two of the three students had emotional problems that required patience and understanding which the teacher seemed willing to provide. They received higher scores on "wanting to keep" attribute than either the indifference or rejection students and were perceived to be in need of constant encouragement which appeared to result in sporadic bursts of effort on their part. In general the descriptions of the four groups tended to confirm findings from many other studies and it would appear that characteristics of students such as their effort and cooperation appeared to elicit specific affective reactions from the teacher in this study.

CHAPTER V EXPECTANCY EFFECTS

This chapter presents the results of the search for any product expectancy effects and/or any process expectancy effects in operation in this study. By examining the relationships among different kinds of data collected, certain trends were noted and speculations are offered for their possible cause.

PRODUCT EXPECTANCY EFFECTS

- 2.1 What evidence is there, if any, of product expectancy effects? What is the relationship of teacher-held expectancy to end of the year achievement results?
 - 2.11 for students assigned to high, middle, and low "probable highest achiever" levels (for general achievement in all subjects)?
 - 2.12 for students assigned to high, middle, and low "probable achievement in language arts" levels?
 - 2.13 for "good" and "poor" reading groups?
 - 2.14 for students assigned to high, middle, and low "probable achievement in math" levels?
 - 2.15 for boys and for girls?
 - 2.16 for the four attitude-to-student groups?

The Product Measures

In expectancy research, "product" or "outcome" measures are frequently the scores received on a standardized test. The particular product scores used in this study were the scores obtained on school system-developed tests in language arts and math which were administered to all students in Grade 3 in Edmonton in the spring of 1979. The student subjects in this study, who could be described as middle-high S.E.S. children with an average I.Q. score of 112, would be expected to score well on these tests compared to the larger student population in the city. In language arts the class mean of 63.3 (out of a possible 77; S.D. 8.84) for the "comprehension" section was well above the test mean of 56.3 (S.D. 12 for the "decoding" section of the language arts test the class in the section of a possible 63; S.D.4.62) compared to the test mean section S.D. 7.38). The actual percentage scores derived functhes are cores ranged from 66 to 98 and the average class percentage methods 5.1% in language arts.

In make the class mean was 56.2 (out of a possible 60; S.D. 3.52) compared to consist mean of 48.3 (S.D. 9.13). The percentage scores in math ranged from low of 80 to 100 and the class average percentage mark was 93.6%. This suggests that the "product" measures available to the researcher did of provide much discriminatory information about the students in this class since the student subjects scored so well on these tests. This was especially so in math. However, these were the only product measures collected and are consequently the ones used in the analyses in this study.

Expectancy Measures

The teacher provided general overall performance expectancy ("probable highest achiever") scores for all students and also the more subject-specific performance expectancies for math and language arts achievement. These three lists differed as the students were regrouped to reflect the teacher's different subject area expectations for them.

Originally the expectation data were gathered by having the teacher assign the students to one of seven distinct clusters on a class ranking form for math and language arts. In the case of the probable highest achiever attribute measure the teacher assigned each student to a point along a line continuum (which was later converted to a numerical score from one to seven). These 1-7 expectancy scores were collapsed into "high", "middle", and "low" ratings by combining scores of 1 and 2 for "high", scores of 3, 4, and 5 for "middle", and scores of 6 and 7 for "low". The membership in the three groupings for each expectancy measure changed slightly as the teacher regrouped the students to reflect his specific expectancies for "probable highest achiever", "probable achievement in math", and "probable achievement in language arts".

Overall General Expectancy and Product Expectancy Effects

As recorded in March, the "probable highest achiever" (#4 Attribute) provided an indication of how the teacher anticipated the students would perform on overall achievement in all subject areas. It was a general prediction of probable achievement and was not subject-specific. The possible expectancy scores ranged from 1 (highest) to 7 (lowest). Table 17 shows the achievement (product) scores received in language arts and math on the end of the year exams by these "high", "middle", and "low" probable highest achievers.

Probable Highest Achiever Groups	Language Arts Scores	Mathematics Scores
High N=12	92.9%	97.5%
Middle N=10	80.5%	89.8%
Low N=4	73.02	91.0%

Table 17 PROBABLE HIGHEST ACHIEVER GROUPS AND RODUCT SCORES

The range in language arts achievement scores is about 20 points across the three levels of expectancy and about 7.5 points in math achievement. The teacher's predictions seem more accurate in the language arts area than for math.

Language Arts Expectancy

A more precise expectancy prediction for language arts achievement was obtained when the teacher ranked the students for "probable achievement in language arts". The membership in these "high", "middle", and "low" language arts expectancy groups was roughly the same as the three groupings in the "probable highest achiever" rankings. The correlation between the two sets of rankings was strong (r=.9110, p<.001). Table 18 presents the results of the language arts exam for the three language arts expectancy groups.

Table 18LANGUAGE ARTS EXPECTANCY GROUPS AND PRODUCT SCORES

Language Arts	Language	Arts Achievem	ent 🔹
Expectancy Groups	Scores	Girls	Boys
High N=8	1693.6%	94.0% N=3	93.4% N=5
Middle N=12	* \$86.2%	87.8% N=6	84.5% N=6
Low N=6	71.5%	72.7% N=4	70.0% N=2

When examining the actual language arts scores achieved by each opectancy group a range of 22 points can be noted between the highest group average and the lowest group average. When grouped this way the girls' scores ranged over 11 points and the point spread for the boys' scores is 23.

Reading Groups Expectancy

The 26 students in the three above language arts expectancy groupings were placed into two reading groups for parts of the language arts instructional program. These groups have been labelled the "good" and "poor" reading groups for discussion purposes. The average language arts expectancy scores and the actual language arts achievement scores received by members of these two reading groups are of interest and are presented in Table 19.

Table 19

READING GROUPS' EXPECTANCY SCORES AND ACHIEVEMENT SCORES

Reading Groups	Language A	rts Achieyement	
	Score	Girls -	Boys
Good Group	90.3%	91.0%	89.8%
	N=17	N=7	N=10
3	E =2.5	E=2.6	E=2.5
Poor Group	75.2%	76.8%	72.0%
•	N=9	N=6 •	"N=3
٠	E=5.9	E=5.6	E=6.3
All 26 Students	85.1%	84.5%	85.7%
1	N=26	N=13	N=13
•	E=3.7	E=4.0	E=3.4

Overall the teacher expected the 13 boys (E=3.4) to achieve better language arts scores than the 13 girls (E=4.0), which they <u>did</u>. Upon close examination of the sub-groupings it was noted that the teacher expected the 10 "good" reading group boys to achieve slightly higher scores than the 7 "good" reading group girls (which they did <u>not</u>) and the teacher expected that the three "poor" reading group boys would receive <u>lower</u> language arts scores than the six "poor" reading group girls (which they did). The resultant difference between the averages of girls' marks (comparing the two reading groups) was 14 points and slightly more for boys, comparing the average boys' marks in the two reading groups, at 18 points.

The result of comparing these language arts expectancy groupings, including both reading groups, with the end of the year achievement scores could be summarized as follows. In March the teacher reported his expectation that the boys' end of the year achievement in language arts would be more varied in range then the girls' which turned out to be the case. His predictions, when compared with the "product" measures, which determined the "accuracy of prediction", produced the following negative and significant correlations (1 was the highest score in expectancy ratings): r=-.893 p<.001 for the whole class; r= -.9294, p<.001 for boys only; and r= -.8661, p<.001 for girls only. In other words, the teacher was fairly accurate in his expectancy predictions for language arts achievement, especially for boys.

Math Expectancy

Initially the teacher ranked all 26 students for "probable achievement in math" by assigning them to one of seven clusters on the ranking form. The resultant numerical scores were collapsed into "high", "middle", and "low" groups (as described above) for math expectancy. The membership in these groups was roughly the same as for the language arts expectancy groups. The membership correlation between the two groups was .8100, p<.001 and between the math expectancy groups and the "probable highest achiever" groups was .8500, p<.001.

The scores attained by the students belonging to these three math

expectancy groups are presented in Table 20 as well as the average

expectancy (E) rating for each group.

* One boy in the "low" math expectancy group moved away in the spring before the math tests were taken. This is the only data missing from his file and although he is excluded from the math analyses, his data is included in all other analyses in the study.

Table 20

	· · · · · · · · · · · · · · · · · · ·			
Math Expectancy Math Achievement Scores				
Groups		Girls	Boys	
High	* 197.5%	98.5%	96.3%	
	N=8	N=4	N=4	
Middle	94.0%	93.2%	94.7%	
	N=13 +	N=6	N=7	
Low	85.0%√	84.0%	88.0%	
	N=4	N≠3	N=1*	
All 25, students	93.6%	92.7%	94.7%	
	N=25	N=13	N=12	
	E=3.6	E=3.7	E=3.5	

MATH ACHIEVEMENT SCORES OF MATH EXPECTANCY GROUPS

The specific math expectancy scores were more accurate for predicting math achievement than were the more general "probable highest achiever" ratings (See Table 17). The difference in average math achievement scores across the three math expectancy groupings was 12.5 points. For girls in particular, the average group scores differed by 14.5 points and for boys across all three groups, about 8 points. The teacher held a slightly higher average math expectancy for the 13 boys (E= 3.5) compared to the 13 girls (E= 3.7). The boys did perform slightly better overall, but less well than the girls in the high math expectancy group in particular. The correlations for these math expectancy predictions (gathered in March) compared to the end of the year math achievement test scores were -.7622, p<.001 for all 26 students; -.6643, p<.009 for boys; and -.8027, p<.001 for the girls. Thus the teacher was somewhat less successful in predicting math achievement scores than he was in predicting language arts achievement scores. Looking only at math predictions, the teacher was more accurate in predicting the girls' math achievement than in predicting the math achievement scores which the boys would receive.

Attitude-To-Student Groups and Expectancy

The 26 student subjects had been placed in one of the four attitudeto-student groups by the teacher. In earlier studies which investigated teachers' attitudes towards students, as few as one student was nominated to each group and the resultant findings tended to be fairly clear-cut. Very briefly, students assigned to the four attitude-to-student groups often possessed the following characteristics (Willis and Brophy, 1974: 520):

Attachment - model students, high achievers, who respond warmly.

Concern- low achievers, dependent on the teacher, making extensive but appropriate demands, receiving most total interaction.

Indifference- recipients of low rates of interaction and teacher apathy, passive, often display poor attitudes to school (no universal characteristics really).

Rejection- often low achievers, behavior problems, defiant.

By asking the teacher to place all 26 students into the four groups the more prevalent attitude group expectancy effects may have been somewhat obscured but general trends may indicate an expectancy effect due to placement in certain attitude-to-student groups. By regrouping the students this way and again looking at their <u>expectancy</u> scores in language arts and math and their <u>achievement</u> scores in the same subjects a more complex set of relationships develops because the interaction of expressed teacher attitude-to-student with sex of student, expectancy scores for students, and achievement scores received by students has been taken into account.

A separate analysis was carried out for the "first three attachment choices" because of the larger number of students (16) assigned to the attachment group. The information pertaining to expectancy and achievement in language arts for the attitude-to-student groups is presented in Table 21.

Table 21

ATTITUDE GROUPS' LANGUAGE ARTS EXPECTANCY AND ACHIEVEMENT SCORES

Attitude to Student Groups	L'anguage An	ts Expectancy	and Achievement
Student Gibups	Scores	Girls	Boys
All Attachment Students	89.4% N=16 E= 2.8	88.0% N=8 E=3.3	90.9% N=8 E=2.4
First Three Choices Attachmen	93.3% N=3 E=1.3	98.0% N=1 E=1	91.0% N=2 E=1.5
Indifference Students	78.7% N=5 E=4.6	73.5% N=2 E=6	82.3% N=3 E=3.7
Rejection Students	79.0% N=2 E=4.5	79.0% N=2 E=4.5	Ð
Concern Students	76.3% N=3 E=6.3	89.0% N=1 E=5	70.0% N=2 E=7
All 26 Students	85.1% N=26 E=3.7	84.5% N=13 E=4.0	85.7% N=13 E=3.4

On average the teacher held a higher expectancy (E=3,4) for boys than for girls (4.0). Within the attitude-to-student groups he assigned much higher language arts expectancy scores (E=2.4) for the eight boys to whom he was "attached" than to the eight "attachment". girls (3.3). The attachment boys scored 90.9% on the language arts exam compared to the 88.0% language arts score received by the eight attachment girls, therebly confirming his prediction.

Within the attachment group itself, the language arts expectancies held for the "first three choices" were considerably higher (E= 1.3) as were their achievement scores averaging 93.3%, than the whole attachment group's expectancy scores and language arts achievement scores; and these, in turn, were higher than scores for all other attitude groups.

It is of interest to note that one "concern" girl, Bonnie (Case #18), who scored 89%, was in the concern group primarily because the teacher was concerned about her emotional and social development. These factors plus medical problems interfered with her academic development. The teacher recognized her strengths in the language arts area and assigned her an expectancy score of 5.

Generally speaking "indifference" students were average ability students, not obviously different from fellow students who were assigned instead to the attachment group. One might speculate that by being placed into the "indifference" group that a student is likely to be assigned a lower expectancy score and subsequently the student is likely to receive a lower "product" achievement score(about 11 points lower in the case of the indifference students in language arts achievement in the present study). Thus assignment into certain attitude-to-student groups may function as an expectancy of its own and may serve to influence the academic expectancy that the teacher develops for a particular student. Willis and Brophy (1974:527) suggested that the sheer personal reactions and feelings towards students held by the teacher can influence expectancy ratings for these students.

The attitude-to-student groups were re-examined similarly in terms of the <u>math</u> achievement information. Sex differences within attitude groups were noted again as well. The resulting average scores for both math expectancy and math achievement by the attitude-to-student groups are presented in Table 22.

Table 22

ATTITUDE GROUPS' MATH EXPECTANCY AND ACHIEVEMENT SCORES.

Attitude to		tancy and Achievement		
Student Groups	Scores	Girls	Boys	
All Attachment Students	96.5% N=16 E=2.5	96.9% N=8 E=2.5	96.1% N=8 E=2.5	
First Three Attachment Choices	98.3% N=3 E=2	100% N=1 E∓1	97.5% N=2 E=2.5	
Indifference Students	92.4% N=5 E=4.4	91.5% N=2 E=5.0	93.0% N=3 E=4.0	
Rejection c Students	83.5% N=2 E=5.5	83.5% N=2 E=5.5	0	
Concern Students	84.0% N=2 E=7	80.0% N=1 E=7	88.0% N=1 E=7	
All 26 Students	93.6% N=25 E=3.6	92.7% N=13 E=3.7	94.7% N=12 E=3.5	

The overall scores on the math exam ranged from 80% to 100% with an average class mark of 93.64%. In math, between-group differences were not as pronounced as in language arts. Overall the teacher expected the boys (E=3.5) to achieve higher marks in math than the girls (E=3.7), which they did.

It is notable that the only "concern" girl, Bonnie, received the lowest score of the class on the math exam and had been assigned an expectancy score of 7 by the teacher compared to her higher expectancy and achievement scores in language arts (see Table 5-5).

The indication is not as strong for math as for language arts that the product expectancy effect is in operation. The average scores for the different attitude-to-student groups reflect some differences but stronger main effects may remain undetected because of the lack of discriminatory power of the product measures and because of the large number of students assigned to certain attitude-to-student groups.

Summary of Product Expectancy Effects

The teacher did hold differing performance expectations for the 26 students in his class. He held an average higher expectancy for boys for both language arts and math achievement and they did achieve better average scores in both academic areas than the girls in this study. This trend differs from most previous research findings as noted by Brophy and Good (1974:204):

...when a sex difference is operating it appears that teachers are more likely to overrate the intelligence and potential of girls and underrate the intelligence and potential of boys.

Brophy and Good continue (p.231);

These findings are remarkably consistent across many different studies by many different investigators, and they stand as general sex differences. However, there is an additional complication which we have alluded to at times but have not specifically discussed: teachers' expectations and attitudes usually interact with the level of achievement of boys much more than with the level of achievement of girls. Thus, for example, while boys in general are more salient in the classroom, the high achieving boys tend not only to be more salient, but to be seen as active, well-adjusted, successful, and generally positive in the eyes of the teacher. The low achieving boys are also more salient, but they tend to be seen as lazy, immature, maladjusted, and troublesome.

By comparing the teacher's general expectancy for all academic subjects for the "high", "middle", and "low" groups on the "probable highest achiever" measure with the end of the year marks, the results show this was a better projective indicator for language arts achievement than for math achievement. The more specific "probable achievement in language arts" and "probable achievement in math" expectancies provided much better predictive accuracy with the respective end of the year tests.

The teacher was able to make more accurate predictions in the language arts area and although the boys' results were more wide-spread, he was able to predict their particular language arts achievement better than for girls. Conversely in math he was far less accurate in his predictions for boys' achievement in math.

By regrouping the students into attitude-to-student groups and by comparing average attitude group <u>expectancy</u> scores with average attitude group <u>achievement</u> ("product") scores, the results indicate that teacher predictions were accurate. Likewise general accuracy is noted in the math expectancy scores held for the attitude-to-student groups. However, in math two "first choice attachment" boys and one "concern" boy performed surprisingly better than the teacher-held expectancy would indicate. In March the teacher had intimated the possibility that Tom (Case #15) should repeat Grade 3. In a June interview the teacher explained how much more effort Tom had been making of late and how much better he was progressing academically.

Overall the teacher held fairly accurate expectancies in March for the results his students would achieve on the end of the year exams in the academic subjects of language arts and mathematics.

PROCESS EXPECTANCY EFFECTS

If teacher behavior has the potential of communicating differential performance expectations to students (as Lockheed, 1976:15 and Cooper, 1977:419 have suggested) then process expectancy effects can be said to beoccurring. For instance, teacher behavior such as the selection of students into different reading groups could be considered as a process expectancy behavior. In this section only the teacher's verbal behavior is considered. Aside from the actual words the teacher may use to communicate a direct expectancy message to the student, more subtle forces are at work during verbal exchanges with the teacher and his class.

Often students in a classroom do not receive equal amounts of verbal interaction or the same quality of verbal interactions from the teacher (Haigh, 1974:104; Brophy and Good, 1974:14; and Firestone and Broudy(1975:544). Several reasons may account for this fact; The student may display certain characteristics ("attributes") such as shyness or aggressive behavior which might provoke two different verbal reactions from the teacher. The teacher may hold different affective feelings (or "attitudes") toward students but may not realize that he is actually avoiding some, criticizing others, and spending a large proportion of class time enjoying exchanges with others. Research has shown that the sex role behavior of girls in elementary school is usually more compliant and conforming than that of boys and that the resultant higher percentage of verbal interaction with boys may be in reaction to their more unpredictable behavior (Brophy and Good, 1974: 229).

Most teachers hold private differing performance expectations for the students in their classrooms but certain teachers tend to reveal these expectations through the amount and kind of verbal interaction they exchange with individual students. Researchers (Brophy and Good, 1974:330; Haigh, 1974:98; and Muttart, 1977:128) have found that overt communication of expectancies is exhibited in varying degrees by different teachers. The teacher who has a working knowledge of naïve psychology and who is aware of the "expectancy" phenomenon is often on guard against exemplifying it in his own teaching and is able to correct the natural human tendency to react to others on the basis of performance expectations held for them. Smith and Luginbuhl (1976:267) and Brophy et al (1976:46) emphasized the need for making teachers aware of their natural tendency to treat high and low ability students differentially.

Brophy and Good (1974:114) described three types of teachers in the attempt to explain why some teachers are more susceptible to displaying expectancy effects. The <u>overreactive</u> teacher notes the existing differences

in students, keeps these expectations rigid and acts in a way that further exacerbates these differences in students. The expectancy effect would be most obvious with this type of teacher. The reactive teacher gives and takes in his dealings with the students allowing some to dominate and some to recede from the verbal limelight. His behavior is somewhat controlled and influenced by the actions and characteristics of the students with whom he is interacting. "If the teacher merely passively reacts to the differential pressure that different students give her, she will be spending most of her time with boys" (Brophy and Good, 1974:234). Traces of the expectancy effect can be found with the reactive teacher. Brophy and Good (p. 327, and preface p.viii) claimed that most teachers would fall into this "reactive" category. The proactive teacher is also able to assess differences in students and does hold differing performance expectations for the students in her room but is able to keep these perceptions in check while compensating for student differences and is able to diagnose and prescribe what is best for each student, purposely minimizing personal affect towards the student and controlling and shaping student behavior in order to challenge each student academically in the most beneficial way. Each of these three types of teachers holds performance expectations for her own students but each reveals these privately-held convictions to varying degrees. Due to time pressures, complexity in the classroom, the amount of their teaching experience, and the degree of their awareness of the expectancy effect, most teachers would fall into the reactive

category.

()

It appears from research studies that the degree to which expectancy effects are exhibited is quite unique to each teacher. Haigh (1974:98) speculated that "one reason for negative findings in some studies may be that teacher expectancy operates as an idiosyncratic effect" dependent on a wide variety of factors.

To the extent that they are communicated, performance expectancies self-perpetuate because the student learns to conform to them over time. It may be that the age of the student is one factor that allows the effect to operate. Kester and Letchworth (1972:54) who conducted a study of expectancy at the junior high school level, suggested that "associated with the idea of the younger child being more susceptible to the teacher's expectations is the idea that the teacher may be seen as a more valuable source of rewards in the lower grades, where in the seventh grade, students may be less anxious to conform to teacher expectations."

Also these revealed expectancies communicate information to peers in the classroom. Research has indicated that student peers have fairly accurate perceptions of which class members get the most verbal attention, and what kind of attention (Silberman, 1969:405). This has the side effect of shaping peer behavior toward the individual student who is the recipient of teacher-communicated expectancies. Silberman (p.406) maintained that although the teacher's remarks are aimed at one pupil, they are overheard by all pupils and tend to guide the perceptions of all and the behavior towards those students by peers. Nash (1976:28) asked 30 students (aged 12 years) to rank order each other by ability. He found a considerable degree of shared perception among teacher and pupils.

Dusek (1975:680) described the expectancy effect as the functional

relationship between the teacher's own self-generated expectancies and

the student's academic achievement.

To the extent that the teacher self-generated expectancy, as reflected in class ranking, is based on sound objective data regarding student abilities, then the teacher is not biasing the student's education. Rather the teacher's differential behavior toward the student in the classroom may reflect effective teaching-style differences for students with differing needs.

It is to these differential teaching behaviors, or process expectancy effects, which seem to have been exhibited in the present study that the discussion turns.

Frocess Expectancy Measures

product expectancy effects.

As mentioned above the assignment of students into one of two reading groups is, in itself, a kind of process expectancy effect.

Any differential teacher behavior exhibited toward these two groups was noted.

The teacher's dyadic verbal interaction with each of the 26 students in the class was examined and the data was used to develop 26 individualized profiles of this interaction in the subject areas of language arts and math and in the three grouping modes of "individual", "small group", and "whole class" teaching situations. It is these dyadic verbal interactions between the teacher and the individual student that serve as the <u>process</u> expectancy effect measures in the present study. Thus it was possible to consider whether process expectancy effects were in operation with a particular student (see Chapter VI) and combine and re-combine individual profiles into the several grouping patterns as in the previous section on For 35 of the 60 variables on the DIOS (dyadic interaction observation system) the frequencies for each individual student were calculated. As a result, ratio percentage scores were derived by dividing the actual number of pupil responses by the total possible student opportunities to respond. A tailor-made computer program was used to adjust for any student absences so the student's ratios used were truly the ones operating for the times the particular student was present during instruction when the dyadic verbal interaction data was collected.

If all 26 students had been present for all interaction observed and if equal amounts of verbal interaction had been received by all students, each student would have a ratio of 3.8 on each variable on the DIOS instrument. However, due to the adjustments that were made to account for absences the class average ratios for these variables are often somewhat higher, in the 4.2% range.

In all, 2715 separate interactions were recorded over 25.35 hours of classroom instructional time in language arts and math combined and over the time period extending from March 1 until May 16, 1979. In language arts specifically, 1524 interactions were recorded representing 14.85 hours of language arts instructional time and 1191 interactions were collected during 10.5 hours of math instructional time. The 2715 provided the opportunity that each student might have as many as 100 interactions with the teacher which might be assumed to provide an indication of the typical dyadic verbal interaction received by the child over the school

year.

For each student's verbal file, ratios for specific verbal categories were calculated for "combined subjects' instructional time", for "language arts time" (further subdivided into eight different types of language arts lesson settings) and for "math instructional time" (further subdivided into five different types of math instructional settings). For each student the summary totals of these ratios were also subdivided into "individual", "small group", and "large group" settings. Therefore , if a certain student received a high percentage finteraction during math lessons such as many "process questions", many "behavioral reminders" or "Sinits" (student-initiated) it could be determined whether most of these occurred in the "individual" (one-to-one private) grouping mode or whether most of it took place during "whole class" instruction.

All of the above considerations of how teacher expectations might be communicated to students gave rise to Research Question 2.2, the communication of these expectancies to students.

2.2	What	evidence is there, if any, of process expectancy effects?
	2.21	for students assigned to high, middle, and low "probable highest achiever" levels (for general achievement in all subjects)?
ι.	2.22	for students assigned to high, middle, and low "probable achievement in language arts" levels?
	2.23	for "good" and "poor" reading groups?
:	2.24	for students assigned to high, middle, and low "probable achievement in math" levels?
	2.25	for boys and for girls?
	2.26	for the four attitude-to-student groups?

Sex Differences and the Process Expectancy Effect

By concentrating on sex differences, other more influential causes of differential verbal interaction are temporarily disregarded. (These sex differences are examined first and reappear in subdivisions in the expectancy groups examined later). However the results of looking for sex differences on eleven summary interaction variables with both subject areas' data combined produced a fairly conclusive set of findings which are presented in Table 23.

The boys incurred a greater percentage of verbal interaction for all eleven types of interaction. On a straight comparison by sex, the 13 boys in this classroom did receive more total dyadic verbal interaction than the 13 girls in the class.

Table 23

SEX DIFFERENCES AND THE PROCESS EXPECTANCY EFFECT

Type of Verbal Interaction	Girls, N=13	Boys, N=13
Total Interaction	3.74	4.61
All Tinits	3.62 *	4.75
Process Questions	3.57	4.68
Product Questions	3.78	4.57
Get Attention Qu.	2.80	5.61
Behavioral Reminders	3.34	5.11
Teacher Comments	3.54	4.96
Sustaining	4.01	4.40
Tinit Praise	3.91	4.56
Tinit Criticism	3.67	4.72
All Sinits	3.94	4.34

Attitude-To-Student Groups and the Process Expectancy Effect

The four attitude-to-student groups received differing amounts and kinds of verbal interaction as in other studies (Good and Brophy, 1972:620 and Good and Brophy, 1974:130). The average of the summary types of dyadic verbal interaction are presented in Table 24.

Fable 24		
-----------------	--	--

ATTITUDE GROUPS AND AMOUNT AND KIND OF VERBAL INTERACTION RECEIVED

Verbal	Attitude-To-Student Groups				
Interaction Variables	Attachment	First Three Choices	Indifference	Concern	Rejection
Total Interaction	4.08	4.73	3.40	5.50	4.85
All Tinits	3.87	4.57	3.76	6.10	4.90
All Process Questions	4.77	5.50	2.90	3.80	2.45
All Product Questions	3.96	4.07	4.00	5.70	4.05
Get Attn Questions	3.62	4.87·	2.84	8.07	6.60
Behavioral Reminders	3.46	5.70	3.66	6.80	7.90
All Sustaining	3.93	4.07	4.78	4.07	5.25
All Sinits	4.53	5.03	2.74	4.17	4.50

The <u>concern</u> group got the most total interaction (5.03) (as Silberman, 1969:406, had found) and the <u>indifference</u> group got the least(3.40). Specific kinds of verbal interaction were more prevalent with certain groups. As might be expected, <u>rejection</u> students received the highest number of behavioral reminders and the most criticism. One of the two students in this group, Sonia, was the new student who joined the class in April (see Case Study #26). She received a considerable number of verbal interactions from the teacher who was in the process of trying to assess her academic standing and who purposely sought her out in math to try and determine the extent of her knowledge here. Although he was satisfied that she fitted in well in the language arts area he felt concerned that she was far behind the rest of the class in math. As a result the average math Tinits for the rejection group are 5.80 in math compared to the attachment group average of 3.63. The rejection group averages in certain types of verbal interaction may have been higher because of this reason.

The <u>concern</u> group were the recipients of the most Tinits (6.10) overall. Many of these fell into categories like Get Attention Questions (8.07), Behavioral Reminders (6.80) or Teacher Comments(8.17). Compared to the attachment group (3.96) they received a greater percentage of all Product Questions (5.70). This was particularly true in math (6.33). For the total Process Questions however the concern group received 3.80% compared to the attachment group average of 4.77% and the first choice attachment group average of 5.50. However the amount of Process Questions directed at the concern group depended on the subject area for they received 7.86% Process Questions during math instruction compared to the 2.33% of Process

questions they received during language arts instruction.

Within the attitude-to-student groups sex differences were noted favoring the boys who received greater amounts of the various types of verbal interaction from the teacher. Both attitude group differences and sex differences are to be kept in mind when considering the other factors which seemed to influence this teacher and his dyadic verbal interaction

with the students in his classroom. Additional differences in the amount and kind of verbal interaction received are due to the differential expectations the teacher held toward class members and these are considered next.

General Academic Performance Expectancy and Process Expectancy Effects

The "probable highest achiever" rankings reflected the teacher's overall achievement expectancy for his students. It is a predictive measure for general achievement in all school subjects. The verbal categories presented in Table 25 are the general summary verbal categories that combined instructionaltime from both subject areas. Trends can be noted that seem to favor the highest expectancy group on several verbal dimensions. Haigh (1974:104) found a significant relationship between teacher rankings and the number of teacher-initiated contacts with students. He concluded that the quantity of verbal interaction was influenced by teacher expectations.

In their review of the research on expectancy, Smith and Luginbuhl (1976:265) cited studies concerned with how teacher expectancy affected interaction patterns. Some found quantitative differences (Jeter, 1972; Rothbart et al, 1971) and others found qualitative differences as well (Beez, 1970; Brophy and Good 1970; Rosenthal and Jacobson ,1973. Good, 1970:192; Rubovits and Maehre, 1971). Cooper(1977:470) agreed that teacher expectancy for students contributed to feedback and interaction differences exhibited when dealing with students in the classroom.

Firestone and Broudy (1975:544) felt that a "knowledge of interaction patterns significantly increased one's ability to predict academic

performance independently of I.Q." which suggested the process expectancy effect role of verbal interaction in the classroom.

Table 25 presents the total verbal interaction percentages received by high, middle, and low general performance expectancy groups on the "probable highest achiever" measure.

Table 25

GENERAL ACHIEVEMENT EXPECTANCY AND PROCESS EXPECTANCY EFFECTS

Verbal		Pro	obable	Highest	Achieve	r Grou			
Interaction	High	girls	boys	Middle	girls			girls	boys
Total Interaction	4.00			3.50			5.20		
All Tinits	4.0			3.86			5.40	3.90	6.90
Process Q.	5.3-	*		3.16	Math	Tinits	3.05	3.85	8.55ء
Product Q	4.00			4.00	Math P	rocess	π	2.65	9.70
Get Atn Q	3.67			3.26			4.50 8.23	3.20	6.35
Behavioral Reminders	3.69	2.56	4.50	3.67			7.23	6 1.5	8.00
Teacher Comments	3.33 L		*	4.19			7.18	0.45	0.00
Sustaining	3.88			4.61					
Tinit Praise	3.81			4.50			4.17		
Tinit Crit	4.00			3.44	1.93	5.70	6.68	3.25	<u>~0.45</u>
All Sinits	4.67			3.07			5.20		

The data in Table ²⁵ show that the low group got the most verbal attention but a high percentage of this was correctional for maintaining good behavior or for focusing attention on the lesson. When the data for the most instructional kind of verbal interaction , the process question, are averaged across subject areas and students, the low group received the smallest amount (3.05%). Upon closer examination one would find that process questions in math accruing to the boys in this same low group averaged 9.70% but this information becomes lost in general averaging. The low group received the most praise and criticism and it initiated (Sinits) the greatest amount of interaction with the teacher. By contrast the high group received the greatest amount of the process questions overall and the least amount of praise.

The three sets of expectancy rankings (for general highest achiever, for achievement in language arts, and for achievement in math) were roughly the same in membership but some students occupied quite different place orders on these lists. The correlations for the lists for rank order of group membership were in the .8 range,

The scores received by the high, middle, and low groups on this general expectancy ranking reflect differential interaction patterns. However in this study this general ranking compared to the summary verbal interaction categories, while it does reveal some different, masks some interesting subject area differences which can be picked up by comparing the subject-specific expectancy rankings with the respective subject area verbal interaction data.

Language Arts Expectancy and Process Expectancy Effects

It was decided that looking for any differences among high, middle, and low language arts expectancy groups using only the data collected during language arts instruction would be appropriate. The verbal interaction data is presented in Table ²⁶. For each group an average score was computed. The groups were further sub-divided by sex (see Table 27 so any within-group sex differences could be detected.

Table 26

LANGUAGE ARTS EXPECTANCY AND PROCESS EXPECTANCY EFFECTS

Language Arts Interaction	Language	Arts Expectancy	Groups
Variables	High	Middle	Low
Total L.A. Interaction	5.21	≠ 3.34	4.18
L.A.Tinits	4.76	3.73	4.01
L.A.Process Q	6.42 *	* 3.79	1. 52
L.A.Product Q	5.08 T*	* 3.78	3.50
L.A.Get Atn Q	4.08	2.09	7.30
Behavioral Reminders	3.46	3,41	6.42
L.A.Long Duration	5.20	3.79	3.92
L.A.Explanation	2.89	4.39	6.53
L.A. Comments	3.20	3.19	6.03
L.A.Sustain	4.93	3.99	3.10
L.A.Tinit Praise	4.48	4.01	4.13
L.A.Tinit Criticism	3.76	2.70	7.35
L.A.Sinits	6.18 *	2.51	4.53
L.A.Sinit Prai	5.95 ±		6.27
L.A.Sinit Crit	3.13	2.78	8.33
L.A. Achievement Scores	93.62%	86.167	71.50%
		and the second	the second s

•		- Q-				•	-				
	Language Arts		Lang	Language Arts	s Expectancy	1 .	Groups				•
	unceraction Variables	HIGH	girls	boys	MIDDLE	girls	boys	TOW	girls	boys	•
1	Total L.A. I/A	5.21	5.13	5.26	3.34	2.83	3.88	4.18	3.45	5.65	
	L.A.Tinits	4.76	4.56	4.88	- 3.73	3.06	4.40	4.01	3.15	5.75	
	L.A. Process Q	6.42	6.83	6.18	3.79	2.91	4.66	1.52	1.13	2.30	. •
•	L.A. Product Q	5.08	5.36	4.92	3.78	3.55	4.01	3.50	2.70	5.10	
•	L.A. Get Atn Q	4.07	0.0	6.52	2.09	0.0	4.18	7.30	6.25	9.40	
	Behavioral	3.46	2.63	3.96	3.40	2.23	4.58	° 6.41	5.67	7.90	
		5.20	5.06	5.28	3, 79	3.00	4.58	3.91	2.12	7.50	
	L.A. Explain	2.89	1.96	3.44	4.39	4.96	3.81	6.53	4.07	11.45	
	L.A. Comments	3.29	2.53	3.76	3.18	1.78	4.60	6.03	4.98	8.15	
	L.A. Sustain	4.92	5.40	4.64	3.99	3.53	4.45	3.10	2.15	5.00	
	L.A. Tinit Prai	4.47	4.16	4.66	4.00	3.35	4.66	4.13	3.57	5.25	
	L.A.Tinit Crit	3.76	2.93	4.26	2.70	2.20	3.20	7.34	8.07	5.90	· •
•	L.A. Sinits	6.17	6.23	6.14	2.51	2.33	2.70	4.53	4.07	5.45	
	L.A. Sinit Prai	5.94	4.93	6.56	1.77	1.68	1.85	6.26	4.62	9.55	
	L.A. Sinit Crit	3.12	0.0	5.00	2.78	5.55	0.0	8.33	12.5	0.0	an a

 Table 27

 LANGUAGE ARTS EXPECTANCY AND PROCESS EXPECTANCY EFFECTS

In Table 26 the high language arts expectancy group received a great deal more total language arts interaction than the middle group; in particular, more Tinits, more #2 Tinit-Process questions (significantly more than either of the two groups), more #1 Tinit-Product Questions (significantly more than either of the other two groups), more praise, more sustaining behavior. By contrast the low language arts expectancy group received less of the above variables but did receive more overall Tinit criticism, Sinit criticism, and more explanation, behavioral reminders and more Get Attention Questions.

The middle language arts expectancy group received less total language arts interaction than either the high group who got the majority of the process and product questions or lows who received the most teacher comments, behavioral remeinders and get attention questions. The middle group also received less Tinit and Sinit praise and criticism than either group especially in comparison with the high expectancy group.

Table 27 presents the same information but sex differences within the three groups are noted. In the high group although boys generally receive the higher percentage of verbal attention it is interesting to note that the high girls outscore the high boys in the number of process and product questions they receive in language arts as well as more sustaining behavior from the teacher. Girls in the high group required <u>no</u> get attention questions and fewer behavioral reminders than the boys in the high group.

Compared to the boys in the same group the girls in the middle language arts expectancy group received less of most verbal interaction, <u>none</u> of the get attention questions, but did receive more explanations. The girls in the high and middle language arts expectancy groups appeared, from this data, to be exhibiting cooperative classroom behavior for the most part while data from the boys in the high and middle groups suggest more misbehavior although they received both more praise and criticism from the teacher.

The greatest sex differences within groups occurred in the low language arts expectancy group. Here significant differences were found in the number of product questions, explanations, and in the length of the interactions, with the boys receiving many more of these. Overall the boys in the low group received (5.65) more total interaction in language arts than the girls(3.45), more language arts tinits (5.55) than the girls (4.01), more process questions (2.30) than the girls (1.13), and more behavior related interaction as well. Looking only at the data collected one could speculate that the low girls in language arts were not challenged to the same extent academically as the boys in the low language arts group, for on the instructionally related verbal behavior such as process questions, product questions, sustaining, explaining and praise they received much lower amounts. It is notable that they did receive more Tinit(8.07) and Sinit(12.5) criticism than the boys (Tinit= 5.90 and Sinit=0.0 respectively).

Reading Groups and the Process Expectancy Effect

Members from the middle language arts expectancy group either joined the high group to form the "good " reading group or the low group members to form the "poor" reading group. Trends remain generally the same as for the three language arts expectancy groups. Although the poor group received more total language arts interaction and more language arts Tinits, these contacts were managerial in nature with get attention questions, behavioral

Language Arts	•	READ	ING GROU	PS		O				
Interaction	Good Re	ading Gro								
Variables	N=17	Girls N=7	Boys N=10	N=9	Girls N=6	Boys N=3				
Toțal L.A. Interaction	4.04	3.67	4.31	4.23	3.41 *	5.87د				
L.A. Tinits	4.07	3.54	4.44	4.20	3.32*	5.97				
L.A.Process Q	5.07 *	4.75	5.29	2.20	1.53	3.53				
L.A.Product Q	4.20	3.97	4.36	3.96	3.40~*	5.10 ح				
L.A.Sustain	4.42	4.30	4.51	3.41	2.65	4.93				
L.A. Explain	3.41	4.01	2. <u>98</u> *	6.34	3.93 *	11.06				
L.A. Comments	3.29	2.52	3. <u>83</u> *	5.00	3.43	<u>8.</u> 13				
Behavioral Reminders	3.21	2.17	3.94¥	5.83	4.80	7.90				
Tinit Praise	4.00	3.30	4.50	4.51	,3.97	5.60				
Tinit Crit	3.67	3.14	4.05	4.90	5.38	3.93				
L.A. Sinits	4.00	3.90	4.08	4.30	. 3.61	5.66				

READING GROUPS AND PROCESS EXPECTANCY EFFECTS

reminders and criticism accounting for this greater percentage. The more instructionally oriented interaction such as process questions, product questions, and sustaining was accorded to the good group in large measure.

Within-group sex differences were more apparent in the poor reading group with boys receiving more of all types of interaction except for criticism.

The group membership of 17 in the "good" reading group was predepinantly of attachment students, except for one rejection girl, Sonia (the new girl), and three indifference boys. The group of nine students in the "poor" reading group was comprised of one rejection girl, three concern children, two indifference children, two attachment children, and one conditional attachment boy, Nicholas. Therefore since part of the differential interaction afforded both groups is believed to be influenced by the teacher's affective response (attitude) toward the student (see Table 21) it is not overly surprising to find that the good group, made up mostly of highachieving attachment students, should receive more interaction of the instructional type and the poor reading group who were mostly low achievers and children belonging to attitude groups other than attachment should be the recipients of more overall interaction, often involving non-academic matters. Perhaps as a compensatory measure, the teacher directed more praise to the poor reading group which also received more overall criticism.

Thus for the language arts interaction the amount and kind of interaction received depends very much on the way the children are grouped. A general statement concerning girls would be that the highest language arts expectancy girls were challenged to a greater extent with academic questions than high expectancy boys but when the two reading groups' girls are compared, the girls in the good group do not appear to receive as many process and product questions. The four girls in the lowest language arts expectancy group received a minimal number of process questions and a significantly lower number of product questions than their boy counterparts. Although two girls joined the low expectancy girls to form the poor reading group, the results are much the same.
Girls seemed to receive a lesser amount of various types of interaction from which they might have benefitted academically. In addition, they received more criticism than praise. Conversely, the boys for whom he held a low language arts expectancy seemed to receive compensatory attention. The teacher provided proactive behaviors to assist low expectancy boys achieve better results in language arts but seemed to neglect the low girls in various ways which led the researcher to speculate that a negative process expectancy effect was in operation for the low girls during language arts instruction.

Math Expectancy and Process Expectancy Effects

than praise overall,

The verbal interaction for the high, middle, and low math expectancy groups appears in Table 29. Most of the total math interaction was received by members of the <u>low</u> math expectancy group. ¹ In particular, they received the most math <u>process</u> questions, the most math <u>product</u> questions (significantly more than the high group received), the most sustaining and the most explaining. They received the most praise and the least amount of criticism. In addition, they continued to receive the most non-academic kinds of verbal interaction; namely behavioral reminders and get attention questions. The high math expectancy group received the fewest number of product questions, sustaining, explaining and praise interaction, fewer process questions than the low group and they received a lot more criticism

The middle group varied in the amount of certain kinds of math interaction it received. The group received the most criticism of all

Table 29

MATH EXPECTANCY GROUPS AND PROCESS EXPECTANCY EFFECTS

	1		
Math Verbal	MATH	EXPECTANCY GROUPS	· · · · · · · · · · · · · · · · · · ·
Interaction Variables	High, N=8	Middle, N=13	Low, N=4
Total Math I/A	4.13	3.82	5.66
All Math Tinits	3.84	3.88-*	6.12
Math Process Q	4.53	3.68	6.16
Math Product Q	3.53	4.12	5.88
Math Get Atn Q	5.03	2.83	6.88
Behavioral	4.90	3.23 *	
Math Long Duration	1.49	5.14	6.01
Math Explain	1.14	4.34	9.96
Math Comments	3.85 *	3.60 *	<u>1</u> 4
Math Sustain	3.66	4.65	5.18
Math Tinit Prai	2.86	4.42	6.04
Math Tinit Crit	4.45	4.92	2.24
Math All Sinits	4.89	3.67	4.38
Math Sinit Prai	5.30	2.55	5.88
Math Sinit Crit	7.09	1.28	6.68
Math Achievement	97.37x *	9400% *	85.00%

three groups, the lowest amounts of behavioral reminders and get attention questions and the least amount of math interaction overall. Within-group sex differences for math expectancy groups are presented in Table 30. General sex differences were overshadowed by group differences although boys in all three math expectancy groups received more total interaction and more total math Tinits. However, some interesting within-group sex differences occurred. The low group

Math Interaction			MA	MATH EXPECTANCY CDOUDS	LANCY CE	Serio				F
Variables -	HIGH	21118	hovs	ATINUT W						
Total Math	4.13	3.75	4 50	3 87	STITS	poys	FOW	girls	boys	
Interaction				70°C	3.11		5.66	4.40	7.55	
All Math Tinits	3.84	3.55	4.13	3.88	3.62	4.10	* 6,12	4.50	8.55	
Math Process Q	4.53	4.18	4.88	3.68	4.48	3.30	6.16	3.80	9.70	
								1		
Math Product Q	3.53	3.70	3.35	4.12	3.70	4.71	* 5.88	4.43	8.05	
Math Get Atn Q	5.03	5.95	4.10	2.83	1.82	3.70	6.88	1.67	14.70	-
Behavioral	4.90	3.50	6.30	3.23	2.07	4.24	6.30	4.93	8.35	
Long Duration	1.49	1.03	1.95	5.14	6.23	4.20	6.01	4.50	10.55	
	Ļ						*			
Math Explanation	1.14	0.95	1.33	4.34	5.72	3.16	* 9: 96	6.60	15.00	
Math Comments	3.85	2.73	4.98	3.60	3.23	3.91	7.14	5.30	9.90	
Math Sustaining *	3.66	3.60	3.73	4.65	5.82	3.64	5.18	4.80	5.75	
Math Tinit Prai	2.86	3.88	1.85	4.42	4.02	4.76	6.04	4.50	8.35	· · ·
Math Tinit Crit	4.45	4.8 3	4.07	4.92	1.73	7.64	2.24	1,87	2.80	
Math All Sinits	4.89	4.,30	5.48	3.67	4.15	3.26	4.38	3.97	5.00	0
Math Sinit Prai	5,30	4.15	6.45	2.55	2.77	2.37	5.88	6.10	5.55	
Math Sinit Crit	7.09	4.18	10.00	1.28	278	0.0	6.68	5.57	8.35	
Math Achievement	HIGH=	97.37%		MIDDLE=	94.00%		LOW= 8	85.00%		
	с		Table 30							

ð

٦.

ÿ

÷

MATH EXPECTANCY GROUPS AND PROCESS EXPECTANCY FFFECTS

boys received 14.7% of the Get Attention questions compared to the 1.67% received by the low math expectancy girls. The low group boys also received much more instructional interaction with 9.70% for Process questions and 8.05% for Product questions than the girls who received the respective scores of 3.80% and 4.43% for the same •variables. The boys received twice as many explanations and interactions of long duration than the girls in the low group received.

A reverse trend of that encountered in language arts interaction seemed to be operating for math interaction. More instructional verbal interaction (Process and Product questions) was accorded the low math expectancy group. The comparison of averages for the subject areas of language arts and math are presented in Table 31.

Subject Area Interaction Differences

For six of the seven interaction variables represented in Table 31, the high group in language arts received the greatest percentage and the low group in math received the greatest percentage. Only in the Sinit category did the highs continue to maintain the greatest amount for both subject areas. Brophy and Good (1974:103) had noted that high expectancy students manage to create more response opportunities for themselves by intiating them. Brophy and Good (1974:336) stated that "teachers usually do not compensate for differences in student initiation rates so that highs receive noticeably more teacher contact." Hence differences are allowed to widen. "Teachers enjoy contact with high group students. These contacts are rewarding and they subtly condition and reinforce teacher behavior."

Table 31

EXPECTANCY GROUP SCORES FOR LANGUAGE ARTS AND MATHEMATICS

LANGUAGE	ARTS		MATHEMATIC	S	•
Language Arts Total Interaction	(* ^{5.21} (* _{3.34} 4.18	Middle	Mathematics Total Interaction	3.82	High Middle Low
Language Arts Tinits	4.76 3.73 4.01	Middle	Mathematiçs Tinits		High Middle Low
Language Arts Process Questions	(*6.42 (*3.79 *1.52	Middle	Mathematics Process Questions		High Middle Low
Language Arts Product Questions	//5.08	High Middle	Mathematics Product Questions	$\begin{pmatrix} 3.53 \\ *4.12 \\ 5.88 \end{pmatrix}$	Middle
Language Arts Sustaining	4.93 3.99 3.10	Middle	Mathematics Sustaining	4.64	High Middle Low
Language Arts Tinit Praise	4.48 4.01 4.13	Middle	Mathematics Tinit Praise	4.42	High Middle Low
Language Arts Sinits	(*6.18 2.51 4.53	High Middle Low	Mathematics Tinits		High Middle Low
Language Arts Achievement	(*93.62 86.16 (*71.50		Mathematics Achievement	¥97:37 (*94.00 (*85.00	Middle

Perhaps because more "quality" instructional interaction (Process questions, Product questions, and Sustaining behavior) was directed at the low group in math, the range in average achievement scores was less for math (12 points) than for language arts (22 points). Perhaps because of the grouping patterns utilized for language arts instruction, the teacher's mental set towards the two different reading groups may have contributed to the resultant differential treatment exceived. Perhaps remedial teaching in math may be easier generally and the skills required for remedial work in language arts may be

Ľ

more complex. At any rate, and for whatever reason, there was a difference in both the quality and quantity of interaction directed towards high and low groups in the two subject areas.

165

It was found that the middle groups in both subject areas either received an amount between the average high group and low group scores or received the lowest amount of interaction for

all variables.

Brophy and Good (1974:341) sum up the differing teacher treatment of highs and lows as follows:

Highs receive better treatment in part because of their own behavior and their general attractiveness to teachers. Teachers may have some minor aversions to selected lows, but a major related problem is that they do not possess specific and effective skills for dealing with failure. Their own resultant uneasiness and their students' discomfort in public situations subtly encourages teachers to avoid students who fail frequently and/or to react emotionally to their failure with sharp criticism. Furthermore, since classrooms are so busy and teachers possess few conceptual labels for monitoring their own behavior, teachers are largely unaware of their behavior. This lack of awareness allows the process to continue so that, ultimately, teachers overreact to student behavior (anticipating failure and so on).

A Contingency Model for Expectancy Effects

When differences for sex were examined for this class, it was found that there was a difference in the amount and kind of verbal interaction recorded, and that difference favored the boys.

When interaction was examined from the perspective of attitude-to-student groups, differences in the amount and kind of interaction were again found. For example, attachment students received more Process questions, concern students received more Product questions as well as more overall interaction, and indifference students often received the least amount of interaction for the variable under consideration.

Differences in the amount and kind of interaction were found for the three levels of expectancy for both subject areas and in addition these differences appeared to be unique to the subject area.

Many different variables appeared to be operating to produce process expectancy effects in this classroom. It was as if dyadic interaction was based on a contingency model. For example, if a student were a low expectancy girl in language arts, she would likely receive few Process questions (for, in addition, she would automatically be a member of the "poor" reading group). If a student were a low expectancy math girl, she would receive a much higher percentage of Process questions as the teacher concentrated on looking for students who needed help in this subject area. If the student were a boy, but also a low expectancy math student, he would receive more Process questions because the teacher tended to interact more frequently with boys in both subject areas.

Ð

No simple explanation of expectancy effects could describe the interaction pattern in this classroom satisfactorily, if indeed a pattern exists. In this study support exists for sex differences in interaction favoring boys (Brophy and Good, 1974:297), for attitudeto-student differences favoring concern children and avoiding indifference children (Brophy and Good, 1974: 125 and 140), for high, middle, and low expectancy differences in general achievement and in language arts (Haigh, 1974:109 and Brophy and Good, 1974:14) and for these reverse results with high, middle, and low math expectancy





effects (Good and Grouws, 1977:52). Singly each explanation is much too simplified, for it appears that the interaction of all these factors acted to individualize the treatment given to each student, which, in turn, resulted in differing averages for the various groups. Brophy and Good (1974:327) hinted at the need for adopting an eclectic approach when examining this research area:

It has been argued that teachers' expectations other than their expectations for individual students need to be studied, and that a combination of general affective and cognitive expectations is more likely to predict teacher behavior than a single set of expectations.

While the following proposition may still be an overly simplified way of looking at this phenomenon of process expectancy effects in this classroom and the ways in which this teacher treated each student, it is offered as an attempt to visualize the interaction among the variables which all contribute to the process expectancy effect (see Figure 1).

The model presented in Figure 1 presents a Contingency Model for Process Expectancy Effects. By viewing its three dimensions it may be possible to speculate whether this combination of factors serves to explain process expectancy effects in operation generally and for this classroom under study in particular. Sex differences are reflected vertically, subject area differences by the third dimension and levels of expectancy are depicted in horizontal layers. Attitude-to-students is represented by a plus to minus continuum which represents "attachment" to "concern" to "indifference" to "rejection" and this is placed vertically within each layer of

expectancy. The perceived effort of the student as reported by the

teacher and calculated by taking an average of scores assigned for

"motivation to do school work", "persistence" and "careful, deliberate worker" is reflected on a plus to minus continuum running horizontally across each layer of expectancy stopping in the middle to reflect sex differences. In Figure 2 the three subject area settings (combined, language arts, and math) have been separated in order to plot each student's position using information provided by the teacher on all of these dimensions. This diagramatic plotting indicates that although the three configurations are basically the same, there are certain differences seemingly contingent on instructional setting.

SUMMARY

The first section of this chapter presented a series of findings suggesting the existence of product expectancy effects in the classroom. Teacher-provided expectancy scores for students individually and for groupings were compared to their end of the year achievement marks and positive correlations were found for both language arts and math.

In the second section of the chapter, process expectancy effects were noted by examining the quantity and quality of verbal interaction exchanged between the teacher and members of high, middle, and low expectancy groups, reading groups, attitude to student groups, and the individual student.

It appeared as if expectancy effects were in operation to a certain degree in this classroom but were contingent on various factors. A model showing these factors in interaction helps to describe the contingency basis on which expectancy effects seemed to operate in

this particular classroom.

CHAPTER VI THE STUDENT PROFILES

In this chapter all 26 case studies are presented. These individual student descriptive profiles were developed by combining all teacherprovided data about each student and the additional data about each istudent that was obtained during the three month study.

In the first general interview, conducted in March, the teacher was asked to describe each individual student. In the first three weeks of the study the teacher was requested to rank order all students in the class on six ranking scales which included both expectancy and attribute measures. He was asked to rate each student individually on 13 additional student attribute measures. During the course of the study as comments were made about students either during stimulated recall

interviews or during general interviews, they were recorded, transcribed, and eventually combined with numerical data concerning the particular student. At the end of the study, data on achievement, I.Q. and age were added to the file information. As the study progressed certain students were selected for more intense observation in order to determine their rate of engagement (or time on task) during ongoing classroom instruction. Where available, this information in percentage form has been included in the case study profiles.

In the first part of each case study both the numerical scores and the comments provided by the teacher are interwoven in the attempt to describe each student as the teacher perceived him or her. Such information, presented in narrative form, includes the teacher's expectancy for the student's achievement, ratings'on the various personality attributes, teacher-expressed affect toward the student, and teacher comments that helped to explain and corroborate the ratings given.

Additional information about age, I.Q. and achievement is then presented. Selected aspects of the dyadic verbal interaction exchanged by this particular student and the teacher are presented in the form of percentage ratios and are summarized in table form. A discussion of the information in the case study follows and briefly integrates the characteristics pertinent to the student himself and to his relationship with the teacher.

Table 32 presents each of the "rankings" which appear in numerical form (a score of 1 to 7) in each of the student profiles. A brief description of what each of the 'six measures represents is included as well as the class average score for each of the six rankings.

Descriptor Label	Class Average	More Complete description
probable achievement in language arts	3.69	How well the teacher expected the student to perform on end of the year language arts tests.
probable achievement in math	3.62	How well the teacher expected the student to perform on end of the year math tests.
general intellectual ability —	3.77	The teacher's assessment of the 'relative intelligence of the _ student compared to his classmates.
cooperation (I)	3.85	How cooperative and compliant and and conforming the student was to classroom procedures and rules.
maturity (1)	3.69	how mature the teacher felt the student was in relation to his classmates
motivation to do school work	3.50	The teacher's assessment of the the student's apparent motivation.

Table 32 THE SIX RANKING SCALES

Table 33 presents the brief descriptor used for each of the 13

attribute scales, the class average for each of the 13 attributes, and

the descriptive sentence presented to the teacher about each of the 13 attributes.

Descriptor Label	Class Average	Definition of each attribute to be rated from 7 (low) to 1 (high).
calm	3.04	Restless, highly active versus calm, good self control.
careful	2.96	Careless, hasty worker versus careful, deliberate worker.
happy	2.64	Unhappy versus happy.
probable highest achiever	3.31	Probable lowest achiever versus probable highest achiever.
mature (II)	, 3.08	Immature versus mature.
cooperative, (II)	2.50	Uncooperative, defiant versus cooperative, compliant.
creative	3.04	Non-creative or unimaginative versus creative and imaginative.
attractive	1.65	Unattractive versus attractive.
persistent	2.88	Gives up easily, needs to be prodded versus tries hard, persistent worker.
like to keep	2.69	Would like to have removed from class versus would like to keep for another year for the sheer joy of it.
of concern	4.65	Doesn't require special attention
		versus concerns me a great deal. I would like to be able to devote much more attention.
noticeable	2.81	Not noticeable versu very noticeable.
Eye contact	2.38	Averts eyes versus looks you in the eye.

Table 33 THE THIRTEEN ATTRIBUTE SCALES

The twelve dyadic verbal interaction ratios that have been included

in each student's profile in table form are explained briefly in Table 34.

Table34DYADIC VERBAL INTERACTION VARIABLES

<u>ن</u>،

١

All Tinits all teacher-initiated (Tinit) interactions which included the six question types and teacher comments making seven types of Tinits overall. Product Q A Product question was one of the seven types of teacher-initiated interactions which asked a factual question. Process Q A Process question was one of the seven types of teacher-initiated interactions and which required that the student integrate facts in order to explain the question asked. It was considered to be a higher level question. Get Atn Q A Get Attention question was one of the seven types of teacher initiated interactions which was optimarily designed to recapture the student's lagging attention. It was often anticipated in advance by the teacher that the student would not be able to answer the question because of inattention. Comments A comment was one of the seven types of teacher-initiated interactions bate was not a question and which often included a behavioral reminder, praise, criticism, or some personal observation. Sustaining Sustaining behavior referred to the fact that further probing question, had been asked following an initial question, either because the initial question had successfully responded to/one of the student who had successfully responded to/one of the student of the cacher addres wand/or effort. init Crit Tinit criticism was received by the student in the context of a teacher initiated question of either this work or effort expended. shavioral Behavioral reminders, or order received a comment from the teacher addres for order of the student of a student in the context of a teacher-inititated question of either this work or effort ex	Descriptor	Definition
Included the six question types and teacher comments making seven types of Tinits overall. Product Q A Product question was one of the seven types of teacher-initiated interactions which asked a factual question. Process Q A Process question was one of the seven types of teacher-initiated interactions and which required that the student integrate facts in order to explain the question asked. It was considered to be a higher level question. Get Atn Q A Get Attention question was one of the seven types of teacher initiated interactions which was oprimarily designed to recapture the student's lagging attention. It was often anticipated in advance by the teacher that the student mould not be able to answer the question because of inattention. Comments A comment was one of the seven types of teacher-initiated interactions which was oprimarily designed to recapture the student would not be able to answer the question because of inattention. Comments A comment was one of the seven types of teacher-initiated interactions that was not a question and which often included a behavioral reminder, praise, criticism, or some personal observation. Sustaining Sustaining behavior referred to the fact that further probing questions had been asked following an initial question had been answered incorrectly or because the teacher wanted to extend the answer given. init Prai Tinit Praise refers to praise received by the student in the context of a teacher reliated question or comment and was a negative evaluation of either his work or effort expended. shavioral Behavioral reminders occurred when the teacher found it n		
Product Q A Product question was one of the seven types of teacher-initiated interactions which asked a factual question. Process Q A Process question was one of the seven types of teacher-initiated interactions and which required that the student integrate facts in order to explain the question asked. It was considered to be a higher level question. Get Atn Q A Get Attention question was one of the seven types of teacher initiated interactions which was eprimarily designed to recapture the student's lagging attention. It was often anticipated in advance by the teacher that the student would not be able to answer the question because of inattention. Comments A comment was one of the seven types of teacher- initiated interactions that was not a question and which often included a behavioral reminder, praise, criticism, or some personal observation. Subtaining behavior referred to the fact that further probing questions had been asked following an initial question, either because the initial question had been answered incorrectly or because the teacher wanted to extend the answer given. init Prai Tinit Praise refers to praise received by the student who had successfully responded to one of the six question types or who had received a comment from the teacher about the positive evaluation of his work and/or effort. init Crit Tinit criticism was received by the student in the context of a teacher-initiated question or comment and was a negative evaluation of either his work or effort expended. behavioral Behavioral reminders occurred when the teacher found it necessary to comment on a student's classroom behavior.	All Tinits ,	Included the SIX question types and teacher
of teacher-initiated interactions and which required that the student integrate facts in order to explain the question asked. It was considered to be a higher level question. Get Atn Q A Get Attention question was one of the seven types of teacher initiated interactions which was oprimarily designed to recapture the student's lagging attention. It was often anticipated in advance by the teacher that the student would not be able to answer the question because of inattention. Comments A comment was one of the seven types of teacher- initiated interactions that was not a question and which often included a behavioral reminder, praise, criticism, or some personal observation. Sustaining Sustaining behavior referred to the fact that further probing questions had been asked following an initial question, either because the initial question had been answered incorrectly or because the teacher wanted to extend the answer given. init Frai Tinit Praise refers to praise received by the student who had successfully responded to one of the six question types or who had received a comment from the teacher about the positive evaluation of his work and/or effort. init Crit Tinit criticism was received by the student in the context of a teacher-initiated guestion or comment and was a negative evaluation of either his work or effort expended. shavioral Behavioral reminders occurred when the teacher found it necessary to comment on a student's classroom behavior. 1 Sinits All Sinits refers to all student-initiated verbal interactions which included both questions and comments.		A Product question was one of the seven types of teacher-initiated interactions which asked
Instruction question was one of the seven typesof teacher initiated interactions which wasoprimarily designed to recapture the student'slagging attention. It was often anticipated inadvance by the teacher that the student would notbe able to answer the question because ofinattention.CommentsA comment was one of the seven types of teacher-initiated interactions that was not a questionand which often included a behavioral reminder,praise, criticism, or some personal observation.SustainingSustaining behavior referred to the fact thatfurther probing questions had been asked followingan initial question, either because the initialquestion had been answered incorrectly or becausethe teacher wanted to extend the answer given.init PraiTinit Praise refers to praise received by thestudent who had successfully responded to one ofthe six question types or who had received acomment from the teacher about the positiveevaluation of his work and/or effort.init Critthe context of a teacher-initiated question orcomment and was a negative evaluation of eitherhis work or effort expended.behavioralBehavioral reminders occurred when the teacherfound it necessary to comment on a student'sclassroom behavior.1 SinitsAll Sinits refers to all student-initiatedverbal interactions which include both questionsand comments.	Process Q	required that the student integrate facts in order to explain the question asked. It was considered
In Comment was one of the seven types of teacher- initiated interactions that was not a question and which often included a behavioral reminder, praise, criticism, or some personal observation.SustainingSustaining behavior referred to the fact that further probing questions had been asked following an initial question, either because the initial question had been answered incorrectly or because the teacher wanted to extend the answer given.init PraiTinit Praise refers to praise received by the student who had successfully responded to one of the six question types or who had received a comment from the teacher about the positive evaluation of his work and/or effort.init CritTinit criticism was received by the student in the context of a teacher-initiated guestion or comment and was a negative evaluation of either his work or effort expended.ehavioralBehavioral reminders occurred when the teacher found it necessary to comment on a student's classroom behavior.1 SinitsAll Sinits refers to all student-initiated verbal interactions which included both questions and comments.	Get Atn Q	agging attention. It was often anticipated in advance by the teacher that the student would not be able to answer the question because of
further probing questions had been asked following an initial question, either because the initial question had been answered incorrectly or because the teacher wanted to extend the answer given.init PraiTinit Praise refers to praise received by the student who had successfully responded to one of the six question types or who had received a comment from the teacher about the positive evaluation of his work and/or effort.init CritTinit criticism was received by the student in the context of a teacher-initiated question or comment and was a negative evaluation of either his work or effort expended.ehavioralBehavioral reminders occurred when the teacher found it necessary to comment on a student's classroom behavior.1 SinitsAll Sinits refers to all student-initiated verbal interactions which included both questions and comments.	Comments	and which often included a behavioral reminder, praise, criticism, or some personal observation.
Init PraiTinit Praise refers to praise received by the student who had successfully responded to one of the six question types or who had received a comment from the teacher about the positive evaluation of his work and/or effort.init CritTinit criticism was received by the student in the context of a teacher-initiated question or comment and was a negative evaluation of either his work or effort expended.ehavioralBehavioral reminders occurred when the teacher found it necessary to comment on a student's classroom behavior.1 SinitsAll Sinits refers to all student-initiated verbal interactions which included both questions and comments.		an initial question, either because the initial question had been answered incorrectly or because
 the criticism was received by the student in the context of a teacher-initiated question or comment and was a negative evaluation of either his work or effort expended. ehavioral Behavioral reminders occurred when the teacher found it necessary to comment on a student's classroom behavior. 1 Sinits All Sinits refers to all student-initiated verbal interactions which included both questions and comments. 	R	Tinit Praise refers to praise received by the student who had successfully responded to one of the six question types or who had received a comment from the teacher about the positive evaluation of his work and/or effort.
 Behavioral reminders occurred when the teacher found it necessary to comment on a student's classroom behavior. 1 Sinits All Sinits refers to all student-initiated verbal interactions which included both questions and comments. 	init Crit	comment and was a negative evaluation of either
verbal interactions which included both questions and comments.	ehavioral ·	Behavioral reminders occurred when the teacher found it necessary to comment on a student's
Accepted Sinits accepted refers to the ratio of student-	11 Sinits	verbal interactions which included both questions
	s Accepted	Sinits accepted refers to the ratio of student-

174

•

•

	initiated questions or comments accepted by the teacher in comparison to all of the Sinits posed by the student to the teacher , which were ignored, delayed, or not accepted.
Total Verbal	Total verbal refers to the total, across both subject areas, including all Tinits and Sinits, amount of dyadic verbal interaction in which the student was involved.
	WADTE VERBAL INTERACTION VARIABLES

The average class score for most of these dyadic verbal interaction variables across the three subject area settings (math, language arts, and both subjects combined) was between 4.1% and 4.2% in the majority of cases. Individual student's scores can be set in perspective by keeping that class ratio in mind.

It can not be stated strongly enough how deeply grateful the researcher was for the candid honesty and full cooperation of this teacher without whom the study would not exist. Such information was elicited on the assurance that it would be treated confidentially and used for research purposes so his contribution can not be acknowledged in a more direct way. The student subjects were assigned pseudonyms and identity numbers in the interests of confidentiality. These identity numbers were used for the purpose of brevity when recording the rapid-paced live verbal interactions and for later analysis purposes. It is assumed that the teacher provided his honest reactions and assessments of the students in his classroom. The researcher had every reason to believe that he did so once a comfortable and trusting rapport had been established and anonymity was assured.

One reason for obtaining the teacher's perceptions of his students using varying methods was to be able to determine how consistently the

numerical scores were substantiated and corroborated by his expressed thought processes about students during actual instruction, and also by his general comments about them. This information was also compared to the more objective information about the student and to the dyadic verbal interaction experienced by the student. The major reason for obtaining his perceptions of students is the premise that a teacher acts on the basis of his perceptions of students and a knowledge of these may help to explain the teacher's treatment of individual students.

The length of each case study is a source of information in itself. In addition to the basic information provided by the teacher for each student, differences occurred in the amount of additional information the teacher was willing and able to provide about the student. This may be indicative of the relative salience of certain students in the classroom.

In order to help to counteract the effect of 'first impressions' the reader might form, it should be noted that coincidentally Case studies #1 and #2 were about students who belonged to the indifference group and some negative impressions about them are expressed. Although these teacher impressions seemed to be well justified they were not characteristic of the ways in which he viewed most members of the class.

The final case (Case #26) in the chapter concerns a girl who joined the class about one month after the study had commenced. The researcher deemed this an opportune occasion to collect the thoughts about this student as the teacher formed his opinions about her. The case study is therefore longer as more time was spent talking about the formation of his impressions and the information and cues he used

in doing so. If particular interest are both the gradual change in his affective reaction to her from the time she entered the study to the time the study ended and also the reasons provided for the eventual shift in his feelings toward her.

These case studies of the individual students in the class were the 'building blocks' for the class grouping patterns considered elsewhere . The knowledge that the individual characteristics of students become lost when combined with those of other students into group scores led the researcher to develop these individualized profiles which are of interest in themselves and which contain the information which was combined in various ways with other students' data to attain group scores which were also of importance in the study.

Case #1 178 Sharon Page 1

The teacher described her as a moderately "noticeable (4) and "attractive" (3) little girl. Her parents (one of whom was a teacher) were interested in her progress. She was socially unpredictable in that she occasionally became "negative towards friends and school work" and was a bit "sneaky" at times. The teacher found her a little "enigmatic" overall. He rated her as fairly "unhappy" (6).

Partly because she was "quiet" and he did not "find her personality as attractive as some of the others" he considered placing her in either the rejection or the indifference group. He assigned a 4 in "wanting to keep for another year". However he decided that the "indifference" category, minus the provided definition, for he felt he <u>could</u> judge her academic ability and achievement level) seemed a more appropriate placement. He felt she was a student who actively rejected teacher overtures by "groaning when asked to do something" and by "seldom approaching tasks with any enthusiasm". She was habitually late coming into a group which he interpreted as "attention-seeking behavior." The type of behavior was exhibited "frequently enough that it was offputting." She was not a frequent volunteer. Not surprisingly he ranked her low on two-separate ratings of "cooperativeness/compliance" (6 and 6). Overall he felt "sort of blah about her" and claimed she did not "stand out in class either positively or negatively."

His a sessment of her "general intellectual ability"(6) was low as was the rating for "creativity" (6). His expectation for her as a "probable highest achiever" was low as well (6) as was his expectation for her achievement in language arts (6) but was higher for her "probable achievement in math" (4). She was in the lower reading group. He felt the work related attributes (calm (5), careful/deliberate worker (6) and persistence (6)) were consistently low and he noted that "she needed quite a bit of direction... reinforcement... and attention." His perception of her time on task depended on the task in question. For example he claimed "she is not a good attender in language arts." The teacher assigned 6 and 6 for the two ratings for "maturity". The teacher was very "concerned" (1) about her academic progress.

Additional Data

age: 8 years 10 months (mean 8 years 7 months)

I.Q.: Verbal 97 (mean 112.7) Quantitative 101 (mean 112.6) Non-Verbal 104 (mean 111.5)

Achievement: Language Arts 81 (mean 85.07) Math 98 (mean 93.64) In June..."She's always done quite well on math."

Time on Task

	Teacher Directed	Self-Paced	Total
Lang Arts	94%	64%,	81%
	3.63 hours	2.76 hours	6.4 hours
Math	93%	72%	80%
	1.6 hours	2.36 hours	3.9 hours
Combined	93%	68% •	81%
Subjects	5.23 hours	5.13 hours	10.36 hours

Engaged time data show that she kept on task better during independent math activities than for independent language arts activities.

Case #1 Sharon Page 3

180

Verbal Interaction Data:

Sharon	" Language Arts	Math	Both Subjects
All Tinits	3.3	- 3.5	3.4
Product Q	2.6	3.9	3.2
Process Q	2.5	13.9	3.0
Get Atn Q	0.0	0.0	0.0
Comments	4.0	<u>,</u> 2.1	3.1
Sustaining	3.8	5.2	4.3
Tinit Praise	1.6	4.0	2.6
T. Criticism	5.9	, 0.0	3.6
Behavioral	3.9	0.6	2.5
All Sinits	6.1	3.6	5.1
Ss Accepted	25/29 +	10/11	35/48
Total Verbal	4.2	3.5	3.9

The teacher's lower assessment of her general intellectual ability (6) seemed to be substantiated by her I.Q.scores. He recognized her better performance potential in math by assigning a higher expectancy rating in this subject area. He felt she was "not a good attender in language arts." She did achieve a very high math mark on the math achievement test but a lower than average mark on the language arts test She belonged to the poor reading group who as a group made lower language arts marks. The behavioral reminders (3.9) and the Tinit criticism (5.9) she received in language arts did not occur during Math instruction. No Get Attention questions were received by her at all so these interactions were because of behavioral infractions. More praise (4.0) was received during math instruction than during language arts (1.6) instruction. More Product questions (3.9) and Process questions (3.9) were asked of her during math compared with the

Case #1 Sharon Page 4 181

language arts Product questions (2.6) and the language arts Process questions (2.5) she received. Both averages here were well below the class mean of 4.11 and 4.07 for language arts Product and language arts. Process questions respectively.

Overall she received (3.4) a less than class average amount (4.18) of Tinits and more of the Tinits she received occurred in private settings than in public settings. She initiated (5.1) more Sinits than the class average of 4.1. Almost twice as many of her Sinits were in language arts. The 9.4 Sinit rate that occured in language arts private settings compared to her L.A.public Sinits (2.2) was frequently a result of her inattention during the giving of directions for independent L.A. work activities.

Overall her verbal interaction average was slightly less than class average. She was involved in more language arts interactions because of a high percentage of Sinits in private L.A.settings.

	یو روید به م <u>دید</u> میشود بو ر <u>امید</u> ایم از مراجع میکود. او			· · · · · · · · · · · · · · · · · · ·	
		******			Case #1
		# CC + -C + * () + + + + + , , ,	•		Sharon
	•	· · · · · · · · · · · · · · · · · · ·			Page - 5
			. ສ ສຸສາດດທະຫ		
				0400 <u>0</u> 10400	N N N N N N N N N N N N N N N N N N N
			U .u -t0, ·····		
	$\langle \Lambda \rangle$	C C C C C C C C C C C C C C C C C C C	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	2-400,000	0-0-0-+0mm
					1 1000000000
5		1210101010	-6-6-6v	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	100 g 0 8 0 0 8 0 0 8 0 0 8 0 0 8 0 0 8 0 0 8 0 0 8 0 0 8 0 0 8 0 0 8 0 0 8 0
Ē			×00+0-0	NØ0-0-0-	000000nnnn
No.				00 P 0 - 0 - 0 - 0 - 0	noggoorgnnn
9	$ \begin{array}{c} \circ & \circ & \circ & \circ \\ \hline \\ & \\ \end{array} \\ \\ \end{array} \\ \\ \\ \\ \end{array} \\ \\ \\ \\ \\ \end{array} \\ \\ \\ \\ \\ \\ \\ \end{array} \\$	000000000000000000000000000000000000000	0 X X C		- 0 0 0
4		- n 0 n - 0 0 0 0 0 0 0 n	L O		-00-00-0-0
5	100-00000000°	-0000v00h00m	ACR 400 400 400 400 400 400 400 400 400 40	58 359 72 72 72 72	24.000000.00.00
	o o	ភ្ជាល្ច	. ► 8 ←	й, то	2 - 0 4 <u>0</u>
2	၀ ၁၀၀၀၀၀၀၀၀ ၀၀၀၀၀ ၀	000000000000000000000000000000000000000			
1. 3	20000000000000000	000000000000000000000000000000000000000	4 00004+00	000 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 4 0 0 0 0 4 0 0 0 0 7 0 0 7 0 0 0 0 7 0 0 0 0 7 0
		e de la que entre da Siel De	40004004 00004004	4007 NU0-	8 - 000 - moning
Ø	2.5000.485556000.005	NO-00018-91	אר פר		а ,
è	109 125 125 125 125 125 125 125 125 125 125	40 00000000000000000000000000000000000	2000 100 100 100 100 100 100 100 100 100	105 351 201 201 201 201 201 201 201 201 201 20	20.024.2.0.023
	000000000000000000000000000000000000000		i pinin itali itali		₹
. <u>,</u>	-0000000000-000-	400000000400-	LOUR COLL		
S.	2 m m m 4 0 0 0 0 - 4 0 0 -		NI INDODOD	<pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre><pre></pre></pre>	V57 V60 V60 V60 V60 V60 V60 V60 V60 V60 V60
à.		0,000-00-00,00	1	v	>>>>
20.5	00000000000000000000000000000000000000	0000000000000000000000000000000000000		<u>0</u> - 0	20 0
3		00000000000000			0 0000000000000000000000000000000000000
: 0		n0000000000000	n r o n o o n g	l	0 00000000000
Sha	73.80.081.139.239. 79.00.139.09.139.0 74.80.081.139.09.139.0			1	
•	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~				- 00000 m
8	ဝဝဝရှိဝန္နီဝဝဝဝရှိနိုဝဝဝ	C C C C C C C C C C C C C C C C C C C	00111	00000000000000000000000000000000000000	
	00000000000000000	00000000000000000			\mathbf{O} \mathbf{G}
110	-0000000000000000	00000000000000	noo-o		0 0 - 0 0 0 0 0 7 0 1
			0070.0070		
5				4 0	ก วทุงอุนอุนอุนอุน
27		000000000000000000000000000000000000000	4000000000000000	0011 0021	
L LE	- ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	00000000000000			
RESE		<i>p</i>	APT + 0000000		
` ۵. ` ار •	848448484444 848948 848948 848948 8489 849 84			8 0 - 17 0 - 19 19	
Idn		0,00,00,00,00,00,00 0,00,00,00,00,00,00 0,00,0	and the second se	a sa	
-NS P	00000000000000000000000000000000000000	00000000000000000000000000000000000000		Coosooo Coosoo - coosoo - coosooo - coosoo - coosoo - coosoo - coosoo - coosoo - coo	
	*****	+00000000-0+0	TUOGOOGBOO		,
INTE	* 0 * * * * * * * * * * * * * * * * * *			-000000 00	00000000000
1	3.0.0.00000000000000000000000000000000	8-00000000000	V <u>=</u> 0-0000	ဥဝဝဝဝဝစ်က တင	
1. 7. 1	00 ² 000000000000000		5 6 6 5 5	22	
50		Caeeca00000000	000000000		
. <u></u>	~~~~~~~~~~~~~~~~~	00000000000000			
ACT					00000000000
INTER	4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	မ္ကိုင္ငံတဲ့အဝန္႕ရာဆိုဝိုင္ခ်က္ကို	20000000000000000000000000000000000000	3:5555 LA	000000000000
			У) ĻĨ	۲	
1111	V 433 V 434 V 434	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	N-1-2	N85888E	86882 <u>55779</u> 00
TOTAL	وتعتدده فبردية الأبتار المرجبة بجاور والمتقومين	**************************************	Z - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 -	No.	Production of the second secon
5		i	+		A Contraction of the second se
			V Vid	L 9.05802	2 180252

Case 1 Sharon Page 6

 A state of the sta		· · · ·							
•	*	LAN	GARTS	TNC	AUCIVIT	L GROUPT	NG MOD	E	
	S:	EC AL			EC AL	URMAILCS		COI	HOINED
ALL TINITS		1. 219			218				
COMMENTS	E		0.053		108				
PRUDUCJ Q	G			1					0.050
PRUCESS Q	0		0.0		·. ···				0.016
CHOICE	0			e					
SELF PO	0	. 0							0.0
NON ACA	2	. 73.					0.		0.0
GET ATTH	. 0			ŏ					0.010
BEISAVIORAL	- 4		0.062	Ĩ.			0.		0.0
TINT PRAISE	0	. 14		0			. ب		0.046
THUI CRITIC	0	9.	0.0	· 0.				40.	0.0
LONG	0	46.	0.0	Ĵ.			0.	16.	0.0
TINIT SUST 'N	0	3.	0.0	Ő.			3.		0.033
ALL SINITS	21	224.	0.094	11.	193.	0.0	0.	4 .	8.0
SINIT PPAISE	0		0.0	0	1931.	0.057			0.077
SINIT CRITIC	1.	2.		. D.	0.	0.0		8.	0.0
				Ψ.	υ.	9 999	. 	2.	0.500
				SMALL	690.00				•
		LANS	ARTS	4	PATH	GROUPING	MODE		and the second
	SPE	C ALL	RATIO	SPE	G ALL	EMATICS		COMB	INED
ALL TINITS	9.	178.	0.051	0	0.	RATIO			RATIO
COMMENTS	0.	33.	0.0	ŏ.	<u> </u>	9.999		- 178, -	0.051
PRODUCT O	3.	56.	0.054	ŏ.	Ø.	9.099	0.	33.	0.0
PROCESS Q	2.	43.	0 047	ŏ.	0.	9.919	3.	56	Q C54
CHOICE O	1.	۹.	1.000	o ŏ.			2.	43.	0.047
SELF R O	o .		0.0	ŏ.	Q.,	9.999		1.	1.000
NON ACA	3	28.	0.107	ŏ.		9.993	0.	17	0.0
GET ATTN	0.	. 0.	992.9	ŏ.		999	3	28.	0.107 -
BEHAVIORAL	0.	28.	0.0		0.	9.999 0		0.	9.999
TINIT PRAISE	0	18.	0.0	0. C.	0.	9 999	0.	28.	0.0
TINIT CRITIC	0.	7	0.0	ŏ.		8.893	Ο.	418 ····	0.0
LONG	2.	25.	0.080	ŏ.	0.	9.999	0.	7.	0.0
TINIT SUST IN	3.	. 74	0.125		· · ·	9.999	2	25.*	0.080
ALL SINITS	4	67.	0.040	Č.		9 599		24.	0.125
SINIT PRAISE	0.	5.	0.0	ŏ.	0.	9 999	1 4 14	67.	0.060
SINIT CRITIC	0.		0.0	ŏ.	ŏ.	9.999	0.	5.	0.0
	.			· ·	υ.	9.999	0.	1.	0.0
1 1			M	HCLE	-	ROUPING	Men -	. • • • •	
	a filma	LANG A			MATHEN	ATICE	MODE		
ALL WYSLIGHT	SPEC	ALL	RATIO	SPEC	ALL	RATIO	-	COMBIN	
ALL TINITS CONMENTS	18.	652.	0.028	21.		0.032	SPEC	ALL	RATIO
PROTUCT	5.	125	0.040		134	0.0	39. 1		S C.30
PROCESS 0	7.	318.	0.022	13		0.041		259.	12019
	3.	130.	0.020	3.	92.	0.033		632.	0.035
CHOICE Q	0.	1.	0.0		13.	0.033		242.	0.025
SELF R O	2.	29.	0.065	4	67,	0.060		14.	0.071
GET ATTN	. t.	16. 1.1	0.063	ο.	24	0.0	6.	96.	0.063
BEHAVIORAL	9.	13.	0.0	0.	19.	0.0			0.025
TINIT PRAISE	5.	135,	0.037		127.	0.0	0.		0.0
TINIT CRITIC	2.	9G.	0.021	4	73.	0.055			0.019
LONG	2.	18,	0.111	0.		0.0			0.036
TINIT SUST'N			0.033		50.	0.120			0.063
ALL SINITS	1.	78,	0.013		87.	0.053			0.073
SINT DOLLAR		184	0.022			0.0			0.030
SINIT PRAISE Sinit Critic			0.0	0,		0.0			0.013
*****	Ó,	1. 1	0.0	0.		* • *	0.	26. \(0.0

Case #2 Keith Page 1

Keith was described as a "fairly large kid... a bit clumsy... In Phys. Ed., For example, he's not a good athlete." Keith was "not interested in rough-and-tumble sporting activities that most Grade 3 boys are interested in so he'll be skipping with the little girls."

The teacher described him as "not interacting terribly well with other kids in the class" and he suspected "he would be a long way down on a sociogram... He's not an outgoing individual who makes friends easily ... He's quiet and at the same time he'll be a bit of a tettletale at times ... I have to remind him not to tell on others, and once isn't enough... You know he'll do things that cause him to be a little alienated, nothing terrible but kids pick up on that right away." His parents were concerned about his interaction with peers and the fact that other kids said unkind things to him and on one occasion his father discussed Keith's social development at length with the teacher. When the researcher asked whether these impressions could act as a "label", or something that influenced him, the teacher stated that "it's something that /I use in dealing with him." The teacher rated Keith as fairly unhappy (6) and feasoned that it was because of the social isolation he experienced. "You often see him alone at recess bouncing a ball by himself." On two separate occasions the teacher assigned Keith a 5 for maturity.

Keith was considered as a possible candidate for the <u>rejection</u> group for, like Sharon, he was a "quiet person" whose "personality was not as attractive as some of the others." He rated him a 5 in physical attractiveness and a very low (7) score for being noticeable.

Case \$2 \$ Keith 185 Page 2

He also received a low score (6) for the amount of concern the teacher felt about him and 6 for "wanting to keep for another year." The teacher was dissatisfied with the provided definition for indifference for Keith because he "knew a lot about him" but he did feel "blah" and "indifferent" about him. Eventually he placed him in the indifference category. Questioned again in May, the teacher still hesitated to classify him as a rejection student. In an interview on June 7th the teacher stated, "He and I... it's kind of funny, and it's probably a genuine change in my attitude toward him too ... we've talked to each other a bit more and we've...He's been working a little harder. He's done very well. He seems, I don't know, a little happier with the way things are going, something along these lines. That can change my approach to things ... " and later ... My changed attitude toward Keith its not a violent shift it's more an adjustment in my feelings about the kid." Nonetheless at this time he reassigned Keith to the indifference category "without the definition" but with no hint of rejection.

The teacher rated Keith as below average (5) in general intellectual ability, 5 in creativity, and his expectations for Keith's achievement in both language arts and math were in the middle cluster (4) of the class. He was assigned a 5 for probable highest achiever. He was deemed "good solid average. Math is quite strong, language arts a little less so." He was in the good reading group. "He has a funny kind of voice in a way and that threw me off initially at the first of the year but his reading (oral) is fine."

During the oral reading of the novel, <u>Watership Down</u>, the teacher believed that Keith was "turned off completely." The researcher observed

Case #2 186 Keith Page 3

that he usually sat behind the teacher, out of the line of yision. The teacher was convinced that Keith "positioned himself purposely" and "did not choose to be in the forefront answering." He could not ascertain why Keith was disinclined to participate in these read-aloud sessions but suggested "maybe he's the kind of kid who may not like to be put on the spot, to be asked a question." If asked a question in this context he "quite often will not answer but will just sit there." The result was that the teacher basically avoided asking him questions in this setting. He received a fairly low score (5) for eye contact with the teacher.

The teacher assessed his work habits in the 3 to 4 range (motivation-4, persistence-4 and careful/deliberate worker-3). Two separate ratings for cooperativeness/compliance were both 3. The teacher described him as "reasonably conscientious" and "able to get down to task without a lot of direction." He received a high score of 1 on the calm dimension. Additional Information

Age:8 years 5 months (mean = 8 years 7 months)

I.Q.: Verbal-111 (mean= 112.73) Quantitative- 97 (mean = 112.65) Non-verbal -100 (mean = 111.46)

Achievement:

Language arts 86% (mean = 85.07) Math 97 (mean = 93.64)

Discussion.

Keith appeared to have about average or less than average ability as evidenced by scores on I.Q. tests. The teacher assigned a less than average (5) rating for intellectual ability and (5) for creativity. <u>His expectations for Keith's math and language arts achievement were</u> just average at 4. His math mark was a bit higher supporting the Verbal Interaction Data

Case #2 187 Keith Page 4

Keith	Language Arts	. Math	Both Subjects
All Tinits	2.3	2.0	2.2
Product Q	2.6	3.1	2.8
Process Q	2.5	2.9	2.6
Get Atn Q	0.0	0.0	0.0
Comments	1.8	0.8	1.4
Sustaining	3.8	.1.7	3.1
Tinit Prai	3.1	3.0 *	3.1 🔹
Tinit Crit	0.0	4.8	1.8
Behavioral	1.8	0.0	1.0 .
All Sinits	1.7	2.3	1.9
Ss Accepted	8/8	6/7	14/15 -
Total Verbal	2.1	2.1	2.1
1 m /	1		• · · · · · · · · · · · · · · · · · · ·

Engaged Time

	Teacher directed	self-paced	Combined		
	settings	independent	settings		
language	93%	85%	89%		
	3.68 hours	2.81 hours	6.5 hours		
Math	88%	71%	77%		
	1.88 hours	3.13 hours	5.01 hours		
Combined	91%	77%	84%		
	5.56 hours	5.95 hours	11.51 hours		

teacher's belief that he was a bit stronger in math but language arts was average. The teacher felt he put forth a fair effort and the timeon-task data would support this. His percentage of time on task falls within the mid-80% range. He spent more time on task during independent language arts activities than independent math activities. The researcher noted that when he was "off task" he often stared ahead with a blank expression for long periods of time or interacted quietly with his neighbor. Sharon.

His indifference status appears to have resulted because he was not noticeable to the teacher, was quiet, and was seen as a social

Case #2 Keith Page 5

isolate. He had conditioned the teacher to avoid him during the novel reading sessions. This particular setting was one of the teacher's favorite times with his class and he enjoyed sharing the story with them and hearing their reactions to it as it progressed. The fact that Keith so obviously was disinterested must have been disappointing to the teacher. Keith's somewhat effeminate and lethargic behavior did not fit the more stereotyped Grade 3 boy's behavior and this appeared to be what the teacher implied.

Overall Keith received a lower than average amount of verbal interaction from the teacher. He was involved in only 2.7% of the more instructional Product and Process questions. Keith initiated a low percentage of Sinits with the teacher, a behavior characteristic of indifference children. Keith received more Tinits in private (2.7) settings than in whole class settings(1.8). This was particularly true in language arts where the private setting Tinit percentage was 3.2%.

It was evident on the basis of observations during the study that the teacher was developing more positive feelings toward Keith as the year ended and was beginning (by his own account) to interact with him more on a personal level which, in turn, appeared to affect Keith's sense of well being and accomplishment.

Case #3 Geoffrey Page 1

The teacher rated Geoffrey as a very attractive (1) boy who was one of the initial selections for the <u>attachment</u> group and as someone "he would like to keep for another year" (1). He described him as a "bit of an individual, not really going with the herd all the time...I like good old Geoff because I admire idiosyncratic behavior and Geoffrey calls his own too...Well he's not one of the sheep.#

He rated him as fairly "happy" (2). "He takes an interest in other things that may not be the sort of normal things for kids of his age. He's perhaps more interested in books than he is in hockey." The teacher did not judge him to be socially isolated but noted "he tends" not to get into the mainstream with the better athletes in sporting activities." It was as if he preferred to be doing something else.

On two separate ratings for "cooperativeness/compliance" he received a 2 and a 4. For although he worked well, as soon as he was finished he was often disruptive. For example he would "tap pencils, stick them in his ears, or wander around the room." The teacher tended to excuse a considerable amount of Geoffrey's disruptive behavior. "Geoffrey sometimes makes noises just for the hell of it but he is a good friend of mine and I put up with his idiosyncracies because of that." On a similar occasion when he had completed his work quickly, the teacher reasoned on his behalf that "he didn't get anything wrong on his sheet. It was boring for him and so he was fiddling around, surely distracting a couple of others in the vicinity so I got a little annoyed with him but maybe that's <u>unfair</u>." And although the teacher found the distraction "quite annoying, quite frankly", he also added that

Page 2 "in defence of the kid he knew what we were doing, you know" and "that kind of behavior <u>really</u> isn't their (his) fault, that's really <u>my</u> fault and I know that."

After a particularly "wiggly" and "squirmy" week, the teacher was reflecting on the possible cause. "It could be with Geoffrey that I've just started noticing it and that he's been wiggly all year" or "maybe he's getting attention that he's needed but hasn't got all year. Sometimes we take kids for granted when they are doing well." and the teacher noted that he would "have to have things to occupy him in other ways."

For one math activity Geoffrey could not find his scissors and he became most insistent that the teacher interrupt the class to make a general announcement about them. "His damn scissors. I don't know. Geoffrey hadn't returned his report card to me yet, over a month, the little buggar, and then his scissors go missing for 30 seconds and he's upset about it." The teacher "tried to ignore it but it didn't work. He was very insistent." But he had to be content with the loan of a pair. On two separate ratings of maturity Geoff received a 2 and a 3.

On balance for the year the teacher rated him as a 4 "or a bit above " on the "calm/good self control" dimension. He rated him 2 for being "noticeable" and 2 for maintaining "eye contact". Geoffrey was assigned a 6 indicating a small amount of "concern" on the part of the teacher but "this has more to do with the behavioral aspect." There was no concern about Geoff's academic work.

The teacher described Geoffrey as "bright" and rated him 1 on "creative/imaginative", 2 on "general intellectual ability", and 1 for 190

Case #3 Geoffrey

Case ∦3 Geoffrey Page 3

"probable highest achiever". The teacher's expectation for Geoff's achievement in language arts was 1 and 2 for achievement in math. "He's a good solid average student; quite well above average." He was in the better reading group. "He likes to read. He's a good reader. He likes math and does well at that." In discussing the year-end achievement results in June the teacher pointed out that Geoffrey had "done really well. He had 57 out of 60 on the math and was over the 90th percentile in the reading area. Maybe he's real smart and I haven't given him enough to do there..."

The teacher perceived that Geoffrey did his work "quickly, effectively, and accurately. Until his work is done, Geoffrey attends well." The teacher assigned the following scores :: "careful/deliberate worker"- 2, "persistent" - 1, and for "motivation to do school work"- 3. "He's a funny kid in a lot of ways, you know. Geoffrey, in math for example, he gets right down to work and does it but then when he's finished, if there's a two minute hiatus he's ... He wouldn't be 'off task', it's when he's <u>finished</u>. You see that's a problem and you're observing him not doing anything. Is he off task? or is he finished? (The teacher's usual expectation was that a free reading book kept in the desk might fill in any extra time quietly.)

Additional Data

Age: 8 years 7 months (mean = 8 years 7 months)

I.Q.: Verbal 122 (mean = 112.73) Quantitative 133 (mean =112.65) Non-Verbal 122 (mean = 111.46)

Achievement: Language Arts 94% (mean = 85.07) Math 95% (mean = 93.64)

Engaged Time Data

Case #3 Geoffrey Page 4

192

	Teacher Directed	Self-Paced	Total Time 89% 5.58 hours		
Lang Arts	90% 2.9 hours	88% 2.68 hours			
Math	ath 84%		79% 2.35 hours		
Combined Subjects	88%, 3.81 hours	84% 4.11 hours	86% 7.93 hours		

Verbal Interaction Data

	• •		
Geoffrey	Language Arts	Math	Both Subjects
All Tinits	4.9	7.1	5.9
Product Q	2.4	5.1	3.8
Process Q	4.6	9.8	6.3
Get Atn Q	20.0	5.3	11.8
Comments	7.5	11.1	9.0
Sustaining	3.1	8.0	4.8
Tinit Praise	5.8	1.2	3.9
Tinit Crit	12.5	5.9	10.2
Behavioral	8.7	13.8	10.8
All Sinits	3 ₁ 8	5.1	4.3
Ss Accepted	14/16	10/14	24/30
Total Verbal	4.5	6.6	5.4
		the second se	the second s

Geoffrey's I.Q. scores indicated that he was indeed a "bright" boy, well above this class' average. He worked quickly and effectively but then lacked the self-control to use his extra time in a way that conformed to classroom rules. It was difficult to assess and record his time on task for this reason and the scores assigned may be too low. But if the implicit understanding was that free reading or quiet drawing,

writing, or interacting with a friend was acceptable then most of Geoff's antics after his work was completed would have to be considered

as "off task" behavior.

His achievement, though good, might have been even better. He had the fifth highest mark in language arts and the 13th highest mark in math. With a quantitative I.Q. score of 133 it might be expected that he could have achieved a perfect score on the math achievement test as four of his classmates did.

On several occasions the teacher implied that Geoff was bored and that he knew the content well already and he therefore claimed responsibility for Geoff's misbehavior during instruction. He also reflected in June that he may not have challenged Geoff enough in the reading area.

Geoff received high percentages of most verbal contacts with the teacher. Aside from the fact that the teacher appeared to enjoy interacting with him, many of the contacts were for disciplinary reasons; hence the 11.8 for all Get Attention questions, 10.8 for Behavioral contacts, and 10.2 for total Criticism. The teacher directed a high percentage of high level, or Process questions, to Geoff in language arts (4.6) and especially in math (9.8) resulting in the high overall average of 6.3 for all Process questions. Geoff was more inclined to initiate Sinits during math instruction.

Case 3 Geoffrey Page 7

		1997 - 1997 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -							
CASE: 3								1224	
				INDI		GFOUPING			· · ·
and the second second	SPE		ARTS			EMATICS			BINED
ALL TINITS	10.		RATIO 0.053					C ALL	
COMMENTS	5.	106	0.057	7.	161.		20.		
PRODUCT Q	Ő.	9.	0.0	ΰ.		0.099	0.		
PROCESS Q	ŏ.		0.0		. 11.	0.0		19.	
CHOICE Q	Ő.	1	0.0	0.	0.	9,999	0.	1.	
SELF R Q	ō.		8,999	ŏ.	3.	00		э.	0.0
NON ACA.	3		0.048				. U.	97	0.052
GET ATTN		3.	0.333		1.	0.0		4	0.250
BEHAVIORAL	4.	56.	0.071		32.		10.		0.114
TINIT PRAISE	8.	13.	0.077		11.	0.0	1,	24.	0.042
TINIT CRITIC	1.	9.	0.111		3		1.	12.	0.083
LONG	З.	39.	0.077	0.	33	0.0	Э.		0.042
TINIT SUST'N	0.	1.	0.0	Ö.		0.0	· 0.		
ALL SINITS	· A	187	A 014	5.	474	0.029	- 9 .		
ALL SINITS SINIT PRAISE	t	3	0.333	Ō.		0.0	· 1.	7.	0.143
SINIT CRITIC	0.	1.	0.0	Ó.	Ο.		ο.	1 1 1	0.0
		LANG	ARTS	MALL		GROUPING		COM	BINED
		ALL	PATIO	SPEC		RATIO	SPFC	CUM2 C (ALL	DATI
ALL TINITS	6.	178		0.	0	9.999	6.	178.	0.034
COMMENTS	2.	33.	0.061	ο.	ŏ.	9.999 9.999 9.999 9.999 9.999 9.999 9.999 9.999 9.999 9.999 9.899 9.899 9.899 9.999 9.999	2.	33	0.061
PRODUCT Q	0.	56.	0.0	Ō.	Ő.	9,999	ō.	56	0.0
PROCESS Q	1.	43.	0.023	0.	ō	9,999	1	43	0.023
CHOICE Q	0.	1	0.0	0.	Ö.	9.999	ο.	1.	0.0
SELF R Q	9 .	17	0.059	0.	0.	9.999	t.	17.	0.059
NON ACA.	2.1	28.	0.071	Ο.	Ο.	8.999	2.	28.	0.071
GET ATTN		0.	9.995	0.0	0	9.999	0.	0.	9.999
BEHAVIORAL		4 0.	0.071	0.	0.	9.999	2.	28	0.071
TINIT PRAISE	- 1 -	18	0.056 0.143	Q.	0.	9.999	1.	18.	C.056
TINIT CRITIC	- F -	7.	0.143	Ο.	0.	9.999	1.	7.	0.143
LONG	0.	25. 24.	0.0	0.1	0.	9.999	0.	25.	0.0
TINIT SUST N	0.	24.	0.0	ю.	0.	9.999	0.	24.	0.0
ALL SINITS	1.	67.	0.015	0.	0.	9 999	· 1.	67.	0.015
SINIT PRAISE	• • •		0.0	0.					
SINIT CRITIC	0.	1	0.0	0.	0.	9.999	0.	1	0.0
			. v			GROUFING	mor	0	
		LANG	ARTS		MATHE	MATICS		COMB	INED
		ALL	RATIO	SPEC	ALL	RATIO	SPEC	ALL	RATIC
ALL TINITS		576.	0.052	45	609.	0.074	75.	1185.	0.063
COMMENTS		114.	0.096	14:	119.	0.118	25.	233.	0.107
PRODUCT Q	8.	264.	0.030	17	293. (0.058		557.	0.045
PROCESS Q		·T43.	0.056	. 8	81.	0.099			0.071
CHOICE O	0.	1.1	0.0	- f. ·		0.091	1. 4	12.	0.083
SELF P Q		4 ,1 +	0.037	3.	65	0.046	4.	92.	0.043
NON ACA.	0.	10.	0.0	1.	22.	0:045	1.	37.	0.027
GET ATTN BEHAVIORAL	2.	494	0.167	- 5 .	. 18.	0.056			0.100
TINIT PRAISE	14.	124	0.097	14.	113.	0.124	26.	237.	0.110
TINIT CRITIC	5. 2.	- 1C	0.056	1.	73.		6.1	163.	0.037
LONG	1.	10.	0.125	1.	14.	0.071	3.	30.	0.100
TINIT SUST'N	3.	71	0.018	1.	43.	0.071 0.023 0.082	2.	99	0.020
ALL SINITS	-11.	168	0.042		49.	0.082	7.		0.058
SINIT PRAISE	1.	19.	0.003		103.	0.086	20.	273.	0.073
SINIT CRITIC			0.0		1	0.143	2.	26. €.	0.077
						U	- 4		
Case #4 Edward Page 1

The teacher rated Edward as very "attractive" (1), "happy" (1), "quite small", "short, freckles, and some new front teeth". He was rated 1 for "wanting to keep for another year" and was one of the three first choices for the attachment group."He's the sort of a kid that <u>I</u> was. He's into sports. He's a bit of leader in class. He's the one...He can run things when I'm not there...He's quite "cooperative" (1 and 3) and actually he's <u>younger</u> than a lot of the kids too, by <u>months</u>, not by a long time, but he's younger than most of them... He's just a real nice kid to be around." He was assigned a 1 for being "noticeable" and a 1 for maintaining "eye contact".

He sat beside Geoffrey and was frequently involved in mischief although he often managed to escape notice. On one occasion while viewing a videotape, the teacher admitted, "It's funny. I focussed on Geoffrey but Edward is no angel either...but he's so cute, what do you do?" and of both of them... "Well they're <u>finished</u> their work sobut they're not exactly being helpful." Edward received a 3 and a 1 on two separate ratings of "maturity".

The teacher assigned a 2 for "general intellectual ability", ("He's <u>really</u> bright.") and 2's for "creativity and imaginativeness", and for "probable highest achiever". For expected achievement in language arts and math he was assigned a 2 and a 3 respectively. "You know he's good at everything he does, no problems." The teacher was not at all "concerned" (7) about this academic progress. He rated Edward as 2 in "motivation to do school work", 1 in "persistence", 1 for "careful/ deliberate worker", and 1 for being "calm",

Case #4 Edward Page 2

Additional Information

Age: 8 years 11 months (mean= 8 years 7 months)

- I.Q.: Verbal 106(mean= 112.73) Quantitative 109 (mean = 112.65) Non-Verbal 116 (mean = 111.46)
- Achievement : Language Arts 88 (mean = 85.07) Mathematics 100 (mean = 93.64)

Engaged Time Data

	Teacher-Directed	Self-Paced	Total Time on Task
Language Arts	99%	94%	98%
	1.51 hours	.51 hours	2.03 hours

Verbal Interaction Data

Edward	Language Arts	Math	Both Subjects
All Tinits	4.4	4.1	4.2
Product Q	5.1	4.7	4.9
Process Q	5.0	3.9	4.6
Get Atn Q	0.0	0.0	0.0
Comments	2.9	2.5	2.7
Sustain	2.9	6.9	4.3.
Tinit Praise	3.1	9.1	5.7
Tinit Crit	5.9	4.8	5.5
Behavioral	3.5	2.4	3.0
All Sinits	5.7	2.3	4.3
Ss Accepted	22/27	7/7	29/34
Total Verbal	4.8	3.6	4.3

Discussion

Edward was one of the three first choice attachment students who was well liked by the teacher and seen to be fairly cooperative. He received a less than average amount of behavioral reminders and although the researcher noted he was frequently quite involved in some antics with Geoffrey he maintained the impression of being on task. The 3 out of a possible 55 critical contacts he received resulted in a higher than normal ratio of criticism but this teacher behavior was exhibited infrequently overall.

Edward had a greater percentage of total verbal contacts in language arts with a higher than average percentage (5.1 and 5.0) for language arts Product and language arts Process questions respectively.

He initiated a high percentage of Sinits (5.7) in language arts but a less than average amount (2.3) in math. From the teacher he received a high percentage of Sustaining (6.9) and Praise (9.1) during math instruction. The teacher had assigned a high expectancy for Edward's language arts achievement and an expectancy score in the middle range in math. However, Edward achieved a perfect score of 100% on the math exam.

Curiously the teacher felt he was younger than most students but in fact he was the third oldest child in the classroom. He was also perceived to be very bright but his I.Q. scores indicate that he was <u>below</u> average intelligence for this class. It appeared that by perceiving him as being younger and brighter than he really was, the teacher may have held a high expectancy for him. This may have been due to Edward's work habits for on the basis of two hours of observation in language arts Edward appeared to be an on-task student. The effort rating he was assigned was very high.⁶⁰

In all Edward was one of the top students who tried hard, did well, and got along well with the teacher.

Case #5 Lena

199

Page 1 (1), "noticeable", (1), "happy" (2) and "calm (2) little girl. Although not one of his initial choices, Lena was placed in the attachment group. "I don't find her personality as attractive as some of the other kids'... She's attention-seeking in a funny sort of way. She quite frequently will ask questions that she knows the answer to, and knows it very well. (because of the answers she gives me all the way along the line)...You know it's the little things, and it seems so damn petty... Well, you don't chew gum in class. I haven't come to school a day this year when Lena wasn't chewing gum. That's not an important item and I don't overstress it...but collectively, all those little irritants ..." She was assigned a 1 for maintaining "eye contact".

The teacher described Lena as a very "attractive"

She received a 4 in "wanting to keep for another year". "After a year it's time to move on sometimes, but if I got her again it wouldn't break my heart either. Perhaps she is a little 'defiant' of authority and yet very interested in doing the things that are necessary to achieve good results." His comment in May was " I must say, to put it bluntly, I like her a lot better now than at the start of the year... That sort of attitude and approach on her part has disappeared, practically."

He rated her 3 and 4 on two separate ratings of "cooperativeness" but found her cooperative while waiting. "Lena is finished and she's good that way. She's not going to bug me about it. She'll get something out and start coloring or whatever, you know, for a moment or two while I help some of the others." He gave her credit for working well with her neighbor, Ellen, and often helping her quietly and discreetly. "If you get little groupings like that, there's nothing wrong with that at all." He assigned her a 2 and a 1 in "maturity".

He described her as a "very bright student" and rated her 3 on "general intellectual ability" and 2 for "probable highest achiever". He assessed her as 4 for being "creative". She was in the better reading group and his expectancy for her achievement in language arts was a 2. Although "good in language arts" he saw her as "much stronger in math", and the 1 assigned for expected math achievement reflected this. On a couple of occasions. as evidenced by his interactive thoughts, he appeared to use her as a gauge while teaching mathematics. "I was a little surprised to find people like Lena making mistakes...She's usually pretty good and I'm starting to think- this is when I'm starting to pull my hair and wonder if I've got to reteach all of this or..."

As for her actual work habits, he described her as highly "motivated to do school work" (1), "persistent" (1), and a "careful deliberate worker" (1) and he had little "concern" (7) about her progress.

Additional Information

Age: 8 years 5 months (mean = 8 years 7 months)

I.Q.: Verbal 120 (mean = 112.73) Quantitative 134 (mean = 112.65) Non-Verbal 118 (mean = 111.46)

Achievement: Language Arts 93% (mean = 85.07) Mathematics 100% (mean = 93.64)

Engaged Time Data: Little data were available.

201

Lase	1/ ງ
Tomo	

				Lena
Engaged Time Data	• `			Page 3
	cher-Directed	Self-Paced	Total	Time on Task
Language'arts	90% 20 minutes	0% 5 minut	es .	72 25 minutes
Mathematics	91% 17 minutes	100% 31 minu	ites	97% 48 minutes
Combined Subjects	91% 37 mi nutes	86% 36 minu	ites	88% 1.2 hours

Verbal Interaction Data

an Se

η. Lena Language Arts Mathematics Both Subjects All Tinits 5.1 4.8 5.0 Product Q 5.4 5.3 5.3 Process Q 9.0 4.0 7.3 Get Atn Q -0.0 18.8 9.4 Comments 2.9 4.3 3.6 7.6 Sustain 7.5 7.6 Tinit Praise 3.1 412 3.3 3.2 Tinit Crit 0.0 5.0 1.9 Behavioral 2.6 6.9 4.4 All Sinits 4.8 6.2 5.4 Ss Accepted 21/23 14/19 35/42 Total Verbal 5.1 5.2 5.1

Discussion

Lena appeared to be a self-assured student who had academic potential and who achieved good marks. By the teacher-assigned scores she was seen to be somewhat uncooperative. She was involved in a high percentage of verbal contacts with the teacher for three reasons: The teacher initiated many quality instructional contacts with her. For example, she received 9% of all the Language Arts Process questions the teacher asked. She was the recipient of a moderate amount of behavioral

Case #5 Lena Page 4

202

reminders and Get Attention questions particularly in math settings. Also she created many verbal contact opportunities for herself by initiating a high ratio of Sinits (5.4 overall), particularly in math (6.2) settings.

The teacher rated her as 3 in general intellectual ability and 4 in creativity although he commented that she was a "very bright student" and her I.Q. scores, particularly the Quantitative I.Q. score, are well above average for this class. One could speculate to what degree his somewhat negative affect (4 in "wanting to keep for another " year, mean = 2.69) might have contributed to his general impression of her potential (as expressed in March). It was interesting to note the positive change in teacher affect toward her over the course of the year.

Little time on task data is available for Lena but the teacher's assessment of her work habits reveals she was perceived to have excellent work habits and achievement motivation. Perhaps the indiscretions that did happen during math instruction may have resulted because'she experienced little difficulty, worked quickly, and then had time to spare.

She was involved in a high percentage of verbal contacts and was considered to be one of the top achieving students.

Case ∦6 Ellen Page 1

Ellen was portrayed as a very "attractive" (1), "happy" (2), "cooperative" (2 and 2) and "mature" (2 and 2) little girl. "She's quite quiet but gets along quite well with other students in the class...in general a nice kid to have in class." She was assigned a 1 on the "calm" attribute. She was in the "attachment" group and received a 3 for "wanting to keep for another year". The teacher found her "persistent" (1) ("She works hard; she tries hard at things."), a "careful/deliberate worker" (1), and "keen" or "motivated to do school work" (2). However she seemed hesitant and unsure and often showed her work in progress to the teacher for approval. "She's not one of those who jumps in with both feet by any means but...you know she really has come a long way this year too, I believe. She really is developing quite a bit of self confidence now...She had quite a bit of success this year largely attributable to the way that she's worked. I think she's beginning to recognize that she can do things just as well as the others." She received a 3 for maintaining "eye contact."

The teacher felt that sitting beside Lena and the interaction that occurred between them was beneficial to her. The assumption the teacher made was "she might be yacking with Lena but they're working."

He assessed her as 6 in being "noticeable". "You know the kind of little girl she is. She doesn't demand attention in any way really...In a group situation it's unlikely that she'll volunteer very frequently. She's not a wallflower or anything like that but she's certainly not a very assertive person." This perception of Ellen was utilized in an interactive decision concerning the selection of students. "Ellen is one that I asked because she's a quiet little girl and I often overlook her so

Case #6 Ellen Page 2

I try to say to myself '<u>There's</u> an individual that I want Page to ask a question just to make sure that I know where she's at because she's not the kind, if she doesn't understand, to ask me(in a whole class setting)."

She was rated 4 in "probable highest achiever". She was in the better reading group and the teacher's expected language arts achievement rating for her was 4. Although "she's a stronger math student that she is in language arts" her expected mathematics achievement rating was 4 as well. "In math I'd say she is about average... Math is the kind of thing where you can perfect a skill whether you're in the top range or not and that's what she tends to do. By doing a little extra work she can perfect a skill and be able to work away at it but in terms of initial insight, for example, she's not by any means the first one to pick things up." He assigned a 5 in "creativity" and a 3 in "general intellectual ability"..., not one of the top students ..." The teacher was not all that "concerned" (6) about her academic progress.

Additional Data

Age: 8 years 7 months (mean 8 years 7 months)

I.Q. Verbal 114 (mean = 112.73) Quantitative 117 (mean = 112.65). Non-Verbal 109 (mean = 111.46)

Achievement: Language Arts 86% (mean = 85.07) Math 100% (mean = 93.64)

Case #6 Ellen Page 3

Engaged Time Data

	Teacher Directed	Self-Paced	Total Time on Task
Language Arts	92%	83%	88%
	3.33 hours	2 . 9 hours	6.2 hours
Mathematics	81%	90%	86%
	1.16 hours	1.66 hours	2.8 hours
Combined	89%	86%	88%
Subjects	4.5 hours	4:56 hours	9.06 hours

Verbal Interaction Data

· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·		
Ellen	Language Arts	Mathematics	Both Subjects
All Tinits	2.8	4.1	3.4
Product Q	3.1	5.3	4.1
Process Q	5.0	3.9	4.6
Get Atn Q	0.0	5.0	2.8
Comments	1.1	2.5	1.8
Sustaining	6.7 .	6.9	6.7
Tinit Prai	2.3	3.0	2.6
Tinit Crit	0.0	0.0	0.0
All Sinits	3.6	4.2	3.8
Ss Accepted	16/17	13/13	29/30
Total Verbal	3.0	4.1	3.5

Ellen was of average age and had made slightly higher

than class average scores on the Verbal and Quantitative I.Q. tests. She was a cooperative, quiet little girl who made a consistently good effort. The teacher recognized that she was dependent on him for feedback and encouragement and he attempted to give her Sustaining support and praise. Although her overall Praise percentage was 2.6, in individual grouping modes she received 7.1% praise in language arts, 3.8 of the praise allotted in individual group settings in math and 5.0% in individual group settings overall.

206 Case #6 Ellen Page 4

She initiated a high percentage of language arts Page 4 Sinits (5.8), math Sinits (6.2) and overall Sinits (6.0) in the <u>individual</u> grouping mode. Many of these were approval-seeking contacts. In both small group and whole class settings she initiated very low percentages of Sinits. Apparently she felt more comfortable approaching the teacher on a one-to-one basis.

305

She received a high percentage of Product questions in math (5.3) (more in individual settings) and a high percentage of Process questions (5.0) in language arts (most of which occurred in the good reading group setting). Corrective disciplinary contacts were few and the Get Attention question percentage of 5.0 she received in math was a result of the one contact out of the total of 20 such contacts the teacher delivered to the class. Her overall verbal interaction percentage is lower than the class average for, as the teacher had to keep reminding himself, she was quiet and was easily overlooked.

Case #7 Brenda Page 1

The teacher rated Brenda as 4 in "attractive" (mean 1.65), Page 1 2 in "happy", and 2 in "noticeable/stands out". She received a 1 and a 2 in two ratings of "cooperativeness". She was placed in the "attachment" group and was rated 2 on "wanting to keep for another year". She was assigned 2 for "eye contact". Her father was anxious that she do well in school. He was not satisfied with general comments about her progress. "He really pressed me for an exact ranking and so I said, 'about 12th of 13th on certain things'...He was going to tell her this as a means of persuading her to do better and move up." By comparison Brenda was not so "fiercely competitive. At the same time she likes to know where she's at and likes to achieve as well as she can... She's also competitive in that 'she likes to be the first done."

He assigned a 1 in "calm", a 1 in "persistence", 2 in "motivation to do school work", and 2 for being a "careful/deliberate worker." However, he rated her 4 in "probable highest achiever", 5 in "general intellectual ability", and 4 in "creativity". "On balance she's a good solid average student." She was in the lower reading group. "Brenda is an on-task individual who works pretty hard...She just doesn't have the skills to be in the other group... Primarily and initially it was some little testing things we did at the start of the year to group according to ability and someone like Brenda, her comprehension skills aren't at the level that I really need for someone in that back group where I can sort of say 'read the directions' and then leave them." He rated her as 5 in "probable language arts achievement" and a 3 in "probable math achievement". "She's much stronger in math. She's one of the top students in math and the reading thing bothers her a little so she needs a lot of approval in the reading area."

Depending on the subject area she appeared to exhibit different degrees of self confidence. In math "she volunteers quite frequently" and her ability in math seemed to influence his selection of students to answer math questions. An interactive decision while teaching math was the choice of Brenda "just to get it underway here; somebody who probably knew the answer." However, in language arts settings... "Brenda isn't a frequent volunteer but she contributes when I ask her; not a fully developed answer but at least she knows where we're at and that sort of thing." On another occasion, during the reading of the novel by the teacher, Brenda gave a lengthy answer prodded along by sustaining feedback from the teacher. His interactive thoughts during this interchange were: "She was succeeding and so I thought 'Well here's someone who doesn't often get a chance to respond in this situation because there are the lions who are always wanting to answer'. So I thought 'I'll keep her going as long as I can' ... Very often I know she has the idea but putting it into words is difficult for her."

She received marks of 2 and 5 in two separate ratings of "maturity". When questioned about this discrepancy in May the teacher explained, "I would use the <u>upper</u> one. Sometimes in our one on one interactions after school she's fairly immature but when it comes down to her approach to her work, the academic side of things, she's very interested, persistent, works very effectively. Those are all signs of maturity. So on one day I must have been thinking of some interaction we had after school. She's the kind of kid who will stand two inches from your nose to make sure that you heard her and go on and on ..."

208

Case #7 Brenda

Page 2

Additional Information

Age: 8 years 2 months (class mean = 8 years 7 months)

I.Q.: Verbal 113 (class mean = 112.73) Quantitative 110 (class mean = 112.65) Non -Verbal 110 (class mean = 111.46)

No Engaged Time Data available

Achievement: Language Arts 83% (mean = 85.07) Math 93% (mean = 93.64)

Verbal Interaction Data

Brenda	Language' Arts	Math	Both Subjects
All Tinits	2.7	2.5	2.6
Product Q	4.7	2.7	3.7
Process Q	2.6	3.1	2.8
Get Atn Q	0.0	0.0	0.0
Comments	0.4	1.7	1.1
Sustaining	4.6	2.0	3.7
Tinit Praise	7.1	4.5	5.9
Tinit Crit	0.0	5.6	2.1
Behavioral	0.5	0.6	0.6
All Sinits	3.4	4.4	3.8
Ss Accepted	12/13	9/12	21/25
Total Verbal	2.9	3.0	3.0

Discussion

Brenda was younger than most students in the class. Her somewhat unruly hair and protruding teeth accounted for her low score on "physical attractiveness". Her scores on I.Q. tests indicated she was of average ability for this class. Her score on the comprehension section of the language arts achievement test was 58/65. The class mean on this section as 63.3 and the city test mean was 56.3. His

assessment that she lacked some necessary skills needed for the good

Brenda Page 3

Case #7

Case #7 Brenda Page 4

210

reading group may have been well founded. Her overall achievement in Innguage arts confirmed his lower expectation for her and likewise her good score on the math achievement test reflected his higher assessment of her ability in this subject area.

His effort attribution for her was excellent and her good behavior and cooperation in class made him eager to reward her efforts with supportive praise and sustaining behavior; particularly in language arts. There is no instance of criticism in her 36 language arts contacts with the teacher.

Her overall verbal interaction record indicates a less than average amount of verbal contact with the teacher. Her sought her out (4.7) for language arts Product question Tinits but overall Tinit percentages for her were low. Perhaps due to her confidence in math she initiated a slightly higher than average percentage of math Sinits with the teacher.

Case ∦8 Anna Page 1

The teacher described Anna as a very "attractive" (1), "happy" (1), "noticeable" (1), and "creative" (1) little girl. Her parents were interested in her academic progress. Her cultural background was Hungarian and "she was quite interested in talking about her heritage. For example, when they were working on some Christmas things, she talked about Christmas in Hungary and brought a bunch of items and articles and so on to describe the differences."

She was in the attachment group and he assigned a 1 for "wanting to keep for another year". "She has a personality that I like... She's kind of an amusing little girl. I like her...Good sense of humor." She received a 1 for maintaining "eye contact".

She was assigned a land a 4 on two separate ratings of "cooperativeness". On both ratings for "maturity" she received a 2. He noted there were"a few interesting little problems in language arts... more maturational than anything...She does reversals from time to time, that sort of thing, letter reversals in her written work."

He described her as very "calm" (1), "persistent" (1), a "careful deliberate worker" (1), and "motivated to do school work" (2). He assigned a 2 in "probable highest achiever" and 2 for "general intellectual ability". "She's a good solid average student, not in the top two or three in the class, but a good solid student." She received a rating of 2 in "probable achievement in language arts" and a 3 for "probable achievement in math." She was in the better reading group. "She's a better than average reader...better than average in language arts...She's quite well spoken I find, and she <u>is</u> careful about the words she does use. I don't find her to be someone who can't form the

Case #8 thoughts or who has the thoughts and can't speak them... Page 2 verbally she's a very effective student...She's careful about what she says when she's in a formal answering situation and I find that her answers are usually pretty high quality answers." However he noted that "some of her word attack skills are not as well developed as you might expect for someone who is as verbally productive as she is... She's a little behind in those things." He had low "concern" (6) about her academic progress.

Additional Information

Age: 8 years 4 months (mean 8 years 7 months)

• I.Q.: Verbal 133 (mean= 112.73) Quantitative 123 (mean = 112.65) Non-Verbal 118 (mean = 111.46)

Achievement: Language Arts 91% (mean = 85.07) Mathematics 95 (mean = 93.64)

Engaged Time: Little data was available

		cher-Directed	Self-Paced	Total Time on Task
Jang	uz.	100% 16 minutes	89% 45 minutes	92% 61 minutes
rba.	<u>1 Ii n</u>	Data		
	Anna	Language Arts	Math	Both Subjects
	All	5.1	2.9	4.1
	Prod	6.4	2.5	4.5
	Proce Q	7.0	4.9	6.3
	Get Q	0.0	0.0	0.0
	Comme	2.9	2.9	2.9
	Sustain	3.8	0.0	2.5
	Tinit Prai	3.5	3.0	3.5
	Tinit Crit	8.8	4.8	7.3
	Behavioral	3.5	1.8	2.8
	All Sinits	7.6	- 4.5	6.4
	Ss Accepted	36/36	11/14	47/50
	Total Verbal	5.9	3.4	4.8

Discussion

Case #8 213 Anna Page 3

Anna was a bit younger than most of the other students and had very high I.Q. scores. The teacher assigned her high effort attribution scores. The one hour of observation of her work habits showed she was on task a high percentage of the time. She had good achievement scores and achieved a mark of 59/63 on the decoding sub-test in language arts. Although she was assigned a middle expectancy in math, she achieved a better than average mark on the test.

She was considered to be amusing, personable, and verbally skilled. It seemed to be rewarding to the teacher to interact with her. She feceived a higher than average percentage (4.8) of all interaction with the greater amount occuring in language arts (5.9) compared to math (3.4) settings. In public language arts settings she received 6.3 of the Product questions and 8.0 of the Process questions. In math, although she received a lesser amount of math interaction overall, a ratio of 4.9 Process questions were asked of her and of these 9.1 occurred in private math settings.

The Tinit interactions in which she was involved were mostly instructional interactions of either Process or Product questions. In addition she initiated a high percentage of Sinits- 7.6 in language arts and 4.5 in math - resulting in a higher than average Sinit (6.4) ratio overall. This implies that she had sufficient self-confidence to initiate contacts with the teacher and the greater percentage of these book place in public settings.

She received a low percentage of behavioral reminders suggesting that her classroom behavior and cooperation was satisfactory despite

214

Case #8

Anna the one score of 4 she received on one of the "cooperation" Page 4 measures. Higher ratios of Tinit Criticism resulted from only 4 such contacts for there were only 55 instances of Tinit Criticism delivered to the whole class.

In all she appeared to be a high expectancy student, who worked hard, was liked by the teacher and who received a high percentage of quality verbal interaction during instruction.

Case #9 Emily Page 1

The teacher described Emily as a very "attractive"(1), "noticeable" (1) and "happy" (1) little girl. She was in the attachment group and received a 2 for "wanting to keep for another year." "Well she's a <u>real</u> nice little girl, a big smile all the time... She's a 'with-it' kid, and I think sort of knows it. She's got a personality that's very attractive to other people...She's one of my real favourites in the class, just a super kid. She's happy, that's the way I see her...She also enjoys a bit of banter with me, that sort of thing." She received a 1 for maintaining "eye contact".

She received a 1 and a 2 on two ratings of "maturity" and a 1 and a 2 on two ratings of "cooperativeness/compliance". The teacher assessed her as 2 in "general intellectual ability", a 2 in "creativity", and 2 in "probable highest achiever". "She's bright. She's actually a pretty good student." He assessed her as 2 for "probable achievement in math" and 3 for "probable achievement in language arts. "Emily has a little trouble with some word recognition things, spellings, and a few items like that... It almost seems like a perceptual distraction/disorder. She sees parts of words but will give a wrong ending, not too frequently. You could hardly notice it. It makes her perhaps a little apprehensive about her reading and so she's a little nervous about it too ... She's one who is really showing improvement in these areas... It's not a major problem ... I could say she's not as good in language arts, for example, her reading, because her word recognition skills aren't as strong and yet on the creative side she's quite a creative individual-, not as 'up there' as some of them but she's a fairly creative individual who writes stories

C. S. C. S.

215

Case #9 Emily Page 2

that are interesting and imaginative."

He rated her as "motivated to do school work" (2), very "persistent"

(1), and a "careful, deliberate worker" (1). He was not "concerned" (7) about her academic progress.

Additional Information

Age: 8 years 10 months (mean = 8 years 7 months)

I.Q.: Verbal 119 (mean = 112.73) Quantitative 119 (mean = 112.65) Non-Verbal 110 (mean = 111.46)

Achievement: Language Arts 88 (mean = 85.07) Mathematics 97 (mean = 93.64)

No engaged time data available.

Verbal Interaction Data

Emily	Language arts	Math	Both Subjects
All Tinits	3.5	3.5	3.5
Product Q	3.3	3.1	3.2
Process Q	4.8	3.9	4.5
Get Atn Q	0.0	0.0	0.0
Comments	2.3	2.5	2.4
Sustaining	2.0	5.2	3.2
Tinit Praise	2.5	5.1	3.6
Tinit Crit	6.9	9.5	8.0
Behavioral	1.8	1.8	1.8
All Sinits	2.6	2.9	2.7
Ss Accepted	12/12	8/9	20/21 。
Total Verbal	3.2	3.4	3.3

Discussion

Emily was a cooperative and mature little girl who received few behavioral corrections and whose I.Q. scores and achievement scores were higher than average for this class. The teacher held a lesser expectancy for her in language arts due to some word recognition problems he felt she had but rated her high on her creative work in language arts.

She was seen to be pleasant and he enjoyed interacting with her. She initiated a low percentage of Sinits with the teacher but more were initiated during private or individual settings in language arts (4.0) and in math (3.6).

Her high percentage of Tinit Criticism resulted from only four such comments out of the 50 the teacher delivered during times Emily was in attendance. She received a good percentage of Process questions in both language arts (4.8) and math (3.9). The researcher asked in June if the teacher would care to make any more statements about Emily as little information had been volunteered about her overall."Gee, its odd...Very clearly she's one of my favorite people in the class you know and she's just a tremendous little individual as far as I'm concerned...What can I say? She's bright. She's happy."

Case #10 Brent Page 1

218

The teacher described Brent as very "attractive" (1), "noticeable" (1) and "happy" (1). He was in the attachment group and received a 1 in "wanting to keep for another year". The teacher assigned a 1 and a 3 on two separate ratings of "maturity" and a 1 for "general intellectual ability". He was assigned a 1 for "probable highest achiever" and 2 for "creativity". "Well in terms of background experience and so on he's probably the brightest, well no, certainly close, certainly one of the brighter kids in the class... lots on the ball, Verbally he is an extremely bright child whose background of information and experience is well... The travelling he's done this year is really something. His Dad is a fairly successful lawyer and he takes Brent with him on a lot of business trips for a couple of days at a time. One trip he stopped a couple of days in Toronto, New York, Atlanta... I haven't travelled like that in my life...gives him a real sense of size of place and geography or even politics to a degree. He's the one who knows what the election is all about, this sort of thing. And at this grade unless you teach it, most kids are unaware and he's very aware of Canada and the political and provincial boundaries. Other children are not in this class... He has firsthand information. He has a good memory and seems to have a pretty stimulating kind of background."

The teacher was not at all "concerned" about Brent's academic progress. He was assigned a 1 in "probable math achievement" and a 2 in "probable achievement in language arts". "The interesting thing about him is that, especially in something like a written language arts assignment, he doesn't produce the kind of work that you would think, based on his verbal skills,

Case #10 Brent Page 2

he would be able to do...It's not because I've been able to detect any deficiency in his skills, it's just this inclination to stall things as much as anything. Not a serious problem." The teacher suggested that perhaps "it doesn't turn him on...I'm not sure whether it's a reluctancy to go that extra mile and do that or whether it's a case of not being able to generate the thoughts without questioning or whatever, I don't know. It <u>is</u> kind of interesting."

The teacher rated Brent as very "persistent" (1), a very "careful/ worker" (1) and "motivated to do school work" (2). He received a 2 on the "calm" attribute and a 1 and a 4 on two separate ratings of "cooperativeness". "He's an enthusiastic kid... Brent loves to volunteer. One of the problems I have with Brent is getting him not to shout out and very frequently, well , he is getting better, but often I have to remind him that 'Brent the procedure we use is hand-raising so I can get to a variety of people'." This behavioral tendency formed an antecedent for an interactive decision to select Brent to give an answer in one instance. "Also I like to satisfy Brent's need to speak out. If I frustrate that then the speaking out behavior is more frequent." His "persistence" seemed somewhat dependent on the subject area in which Brent was involved. "Brent needs a little more direction. He needs a little more telling him to please stick to task than ... "(The children often worked together and were permitted to converse about their work). "If it were creative writing and Brent's chatting, Brent is not going to be on task. If we're working on math Brent may very well be on task. If we're working on another area of language arts Brent may very well be on task."

Additional Information

Age: 8 years 10 months (mean = 8 years 7 months)

I.Q.: Verbal 127 (mean = 112.73) Quantitative 128 (mean = 112.65) Non-Verbal 110 (mean = 111.46)

Achievement : Language Arts 96% (mean = 85.07%) Mathematics 97 (mean= 93.64%)

Engaged Time Data

Only a small amount of engaged time data is available for Brent

	Teacher-Directed	Self-Paced	Total Time on Task
Language Arts	100%	87%	93%
	30 minutes	38 minutes	68 minutes

Verbal Interaction Data

Brent	Language Arts	Math	Both Subjects
All Tinits	6.5	4.4	5.4
Product Q	6.4	4.5	5.4
Process Q	10.6	5.8	8.7
Get Atn Q	0.0	0.0	0.0
Comments	3.9	3.3	3.6
Sustaining	2.3	5.2	3.4
Tinit Praise	7.2	2.0	4.6
Tinit Criticism	2.9	4.8	3.6
Behavioral	2.8	5.3	4.0
All Sinits	15.1	11,3	13.4
Ss Accepted	49/57	31/35	80/92
Total Verbal	9.2	6.2	7.7

Discussion

Brent was perceived to be a bright and enthusiastic student whose self confident personality and wealth of background experience prompted

221

. him to make substantial verbal contributions during ongoing instruction.

He was an attachment student for whom the teacher held high expectancy. Brent achieved excellent results on the end of the year exams.

The relatively large proportion of interaction in which Brent was involved is partly explained by the teacher's perception that his general knowledge enabled him to be a source of information for the others, by the fact that he was enthusiastic and wanting to participate, and partly because the teacher recognized that Brent would channel his enthusiasm into disruptive behavior if thwarted in his attempts at interaction.

Brent initiated an unusually high percentage of Sinits in language arts (15.1), in math (11.3), and averaging 13.4 overall. Brent was the recipient of 10.6% of the Process questions in language arts and 5.8 in math. He received a high percentage of Tinit Praise (7.2) in language arts but not in math (2.0).

Brent seemed to monopolize more than his share of dyadic verbal interaction. He was a high achieving student with whom the teacher enjoyed interacting and who almost agressively created more verbal opportunities for himself. He seemed to benefit from a positive halo effect but appeared to deserve the high assessment he received from the teacher.

Case #11 Simon Page 1

Simon was described as "attractive" (1), "noticeable" (2) and "happy" (2). He was in the attachment group and was assigned a 1 for "wanting to keep for another year." The teacher described him as "likeable" and "nice to have in the class." His parents were here for a one year sabbatical leave from England. "He started the year in grade 4 but his age (8 years 3 months) is correct for grade 3. He should never have been placed in grade 4 and he was having difficulty there."

Simon received a 1 and a 2 on two separate ratings of "cooperativeness". He was assigned 2's on both ratings of "maturity" and 2 on the "calm" attribute. He received a score of 2 for maintaining "eye contact". For "probable achievement in math" and for "probable achievement in language arts" the teacher assigned a 3 in each case. Simon was rated 2 in "general intellectual ability", 2 for "probable highest achiever", and 1 for "creativity". "He's a very creative individual. He writes real good stories for me... He is verbally quite strong and even his written work is quite imaginative but some of the technical skill is absent and it's largely because of curriculum differences... §imon has a lot to contribute..You've probably picked up that he's fairly silent. I feel his comments are generally of pretty high quality."

ς,

In assessing Simon's work habits the teacher assigned a 1 for "persistence", 2 for "careful/deliberate worker", and 3 for "motivation to do school work." "He has some difficulty with eye/hand coordination, although not terribly significant. There is a background of that in the family. The mother has that disability...Though he's persistent enough

Case #11 Simon Page 2

that he does it, often the result is not very attractive. The

teacher was not "concerned" (7) about Simon's academic progress. Additional Information

Age: 8 years 3 months (mean = 8 years 7 months) I.Q.: Verbal 126 (mean = 112.73) Quantitative 125 (mean = 112.65) Non-Verbal 118 (mean = 111.46)

Achievement : Language Arts 91% (mean = 85.07) Mathematics 98% (mean = 93.64)

Engaged Time Data

Only 2.1 hours of observed time on task data are available.

	Teacher Directed	Self-Paced	Total Time on Task
Language Arts	95%	92%	93%
	.95 hours	1.11 hours	2.1 hours

Verbal Interaction Data

Simon	Language Arts	Math	Both Subjects
All Tinits	4.7	3.5	4.1
Product Q	3.8	2.5	3.2
Process Q	5.5	5.8	5.6
Get Atn Q	6.3	5.0	5.6
Comments	5.5	2.9	4.3
Sustain	2.9	0.0	1.8
Tinit Prai	5.5	4.0	4.8
Tinit Crit	5.9	4.8	5.5
Behavioral	5.7	4.7	5.3
All Sinits	2.3	3.2	2.7
Ss Accepted	10/11	9/10	19/21
Total Verbal	3.9	3.4	3.7

Discussion

Although he was rated (2) high for "probable general achiever"

the actual subject-specific achievement expectancies assigned were

Case #11 Simon Page 3

both 3's which put him in the "middle" expectancy group. In both subjects he made higher marks than the class average so the teacherrecorded expectancies for language arts and math predicted somewhat lower results than he actually achieved. He was rated fairly high on effort expended and that plus a higher than average I.Q. may have accounted for his achievement.

Although (2) quite "noticeable" he was also described as "fairly silent" which may explain his lower than average Sinit ratios. In public settings he was more inclined to initiate math Sinits (5.2) than language arts Sinits (1.1) but overall he averaged in the 2.7% range for Sinits compared to the class average of 4.1%.

He received a lower than class average percentage of Product questions in language arts (3.8) and math (2.5) but a higher than class average percentage of Process questions(5.5 and 5.8 respectively). He received a high percentage of the language arts Process questions (14.0) which were posed in the "good" reading group setting. These ratios of Process questions he received from the teacher indicate a higher teacher-held expectancy for Simon than the "middle" expectancy ratings would suggest.

More of the 4.8 Tinit Praise he received was given during whole class instruction.

The high rates of behavioral reminders result from periodic inattention. Saying his name or asking a Get Attention question were usually sufficient tactics for correcting the situation.

Basically he was a good student who achieved well and who was liked by the teacher.

The teacher described Marilyn as a "nice little girl"; "attractive" (1), and "happy" (2). "She gets along well with the other kids in the class." He assigned a fairly low score of 5 for the "noticeable" attribute and a 2 for the "calm" attribute. "She's one of the ones you can miss. It's partly because of her location in the class. She's at the back. It's partly because she's not ... you know, someone like Nicholas demands attention. She doesn't. She's not that kind of individual." She received a 4 (mean 2.6) in "wanting to keep for another year " and he placed her in the indifference group (using the provided definition which implied that the teacher was a bit unsure of a student's academic progress). " I have a general idea of how things are going, where her strengths and weaknesses are generally, but if somebody wanted to pin me down about something specific I'd have to say 'Hey, let me check my records, this sort of thing'...(talking of Marilyn, Sharon, and John, who were all indifference students)... They don't stand out in class. I guess in a sense their demands on me are not as great in both respects; they don't stand out positively and they don't stand out negatively either." He felt generally "blah" about these indifference children although he did not dislike them enough to consider placing them in the rejection category. He perceived Marilyn as having "somewhat negative approaches to some of the tasks, or appearing to have."

She received 2's on both ratings of "cooperativeness" and a 3 and a 5 on two ratings of "maturity". "She's a chatty little thing. She seems like such a sweet little girl, and <u>is</u>, she <u>is</u> a nice little kid, but if I'm not keeping my eye on her she really will chat a lot...She's

226

perceptive enough to know when to be doing her chatting ... Yeah if en I've got two groups together, very often she'll you ever wat be a 'ring 1 bunch of chatters up in that group but I have to be at bad knows my ability to control her directly is the time she will be chatting away... Sometimes reduce d th they ig e th eye contact. (She received a 2 for maintaining "eye contact" ly). It's a little game, you know. It's quite interesting. Marilyn 1 ttle leader there. When I'm not there she'll be chatting nd with Trisha but back and forth, back and forth and every with Pamel I see Marilyn sort of...She's certainly interested in time I loof what's going around her. There's not doubt about that!"

The teacher assigned a 6 for "general intellectual ability", a 5 for "probable highest achiever", and a 5 for "creativity". He described her as a "lower han average student in this class." He assigned a 6 for both "probable to we ment in language arts" and for "probable achievement in math". She was in the lower reading group.

The teacher rated her as having below average "persistence" (4, mean =2.8), as 3 for being a "careful/deliberate worker" and as 5 for being "motivated to do school work". So his effort attribution for her was fairly low overall. "With a lot of positive reinforcement she will try pretty hard." He was somewhat "concerned" (3) about her academic progress. A sample of the teacher's interactive thoughts concerning Marilyn reflect his assessment of her ... For one "selection" he had picked her partly because she was in the back row and because " I like to check up and make sure she's got the skill" and on another occasion..." I asked her because she wasn't there at all."

In June the researcher asked the teacher to verify Marilyn's placement in the indifference group. "Marilyn would certainly be put in that category, with the actual definition, although now she <u>doesn't</u> really any more because now, since it was brought to my attention, you know, you start doing some work that way."

Reflecting in June the teacher stated..There were probably others in the class that I should have felt more concern for and didn't. I think of someone like Marilyn, for example, who, you know, was not a very noticeable kind of person in the class. She's not a particularly disruptive individual but she probably deserved more of my time than I ever gave...or if she didn't deserve it, maybe as a professional teacher I should have spotted that and said 'Hey, if I can motivate them to a greater degree, then I can get them going', but we didn't interact...Do you see what I'm saying?"

Additional Information

Age: 8 years 1 month (mean = 8 years 7 months)

I.Q.: Verbal 104 (mean = 112.73) Quantitative 110 (mean = 112.65) Non-Verbal 97 (mean = 111.46)

Achievement: Language Arts 66% (mean = 85.07) Mathematics 85 (mean = 93.64)

Teacher-Directed Self-Paced Total Time on Task Language Arts 97% 75% 86% 2.41 hours 2.2 hours 4.6 hours Mathematics 99% 98% 982 1.36 hours 2.16 hours 3.53 hours Combined 97% 86% 912 Subjects 3.77 hours 4.36 hours 8.15 hours

Engaged Time Data

Verbal Interaction Data

Marilyn	Language Arts	Math	Both Subjects
All Tinits	3.7	5.0	4.3
Product Q	3.3	5.6	4.4
Process Q	1.5	5.8	3.0
Get Atn Q	12.5	5.0	8.3
Comments	5.9	3.7	4.9
Sustaining	1.9	12.1	5.5
Tinit Praise	3.9	5.1	4.4
Tipit Crit	8.8	0.0	5.5
Behavioral	7.0	4.1	5.8
All Sinits	4.4	• 3.9	4.2
Ss Accepted	17/21	11/12	28/33
Total Verbal	3.9	4.7	4.3

Discussion

Marilyn was younger than the average age of the class by six months. She received a 3 and a 5 on the two "maturity" ratings. She seemed to have some difficulty attending to task. Scores received on the effortrelated variables are in the 3-5 range. Of the 2.2 hours (during five different language arts lessons) of observation during "self-paced" language arts activities, Marilyn was on task only 75% of the time. Over the 2.1 hours of observation of math "self-paced" activities she was observed to be on task for 98% of the time. His expectation scores for both subject areas were 6's and her below class achievement scores verified his predictions for her low academic achievement ther I.Q. scores were well below average for this particular class.

A number of contributing factors- her younger age, her maturity, her effort, her intelligence- resulted in her lower than average achievement.

Although she was seen as "cooperative" (2 and 2), Page 5 she caused others to be distracted while she was off task, particularly in small reading group independent activities when the teacher worked with the other group. Although not rated as overly "noticeable"(5), she attracted a dot of Behavioral reminders (5.8) overall with 7.0% occurring in language arts settings. She received 8.3% of Get Attention questions overall with 12.5 occurring in language arts settings generally and 15.4% occuring during "whole class" language arts instruction. This meant the teacher was attempting to keep her attention focused on the lesson in progress.

Her Sinit rate was average; above average in language arts (4.4), which was not usually characteristic of an indifference student.

Overall she received an average amount of dyadic verbal interaction but a closer examination reveals more of the attention she received was of the monitoring sort. The instructional Tinits (Product questions) she received were lower in language arts (3.3) although they averaged 4.4 overall because of the higher number of Product questions received during math instruction(5.6). However, the Process questions she received totalled 3.0% overall with only 1.5% occuring in language arts settings.

Although he recognized her problems and lack of achievement, perhaps the teacher was discouraged by the lack of apparent effort she appeared to exert overall. At any rate he felt somewhat unhappy at the end of the year with the relationship that had existed between Marilyn and himself and with the fact that she had not achieved a higher level

of academic success.

229

ंडे

Case #12 Marilyn

Case #13 Grant Page 1

The teacher rated Grant as a very "attractive" (1), "happy" (1), and "noticeable" (1) boy. "I guess if there's a 'typical boy' in quotation marks, he would be it...very interested in sports...plays baseball, soccer, and games like that...a little mischevious at times but lots upstairs and so he can afford the time off." He was placed in the attachment group and had received a 1 in "wanting to keep for another year." "He has a good sense of humor." He was assigned a 1 for maintaining "eye contact" and a 2 for the "calm" attribute.

He received a 2 and a 4 on the two measures of "cooperation/ compliance". He would desplay impatience at times during lessons. "That bugs me. When someone asks, 'Do we get out early' in the middle of a lesson. That tells me...you know...when you have the feeling that a lesson is going badly and then someone pipes up 'Can we get out early?' you feel like strangling someone at that point you know. It's kind of annoying." and on another occasion... "Pace is important. Grant gets a little frustrated with the pace after a while. He's the one who asked 'Can we get going?' and I gave him hell for it. I told him I thought it was inappropriate. What do I do? I don't want to be asked that question. The impatience of youth...maybe I should be more tolerant, I don't know." He was reading a book during the math lesson on another day ... "To my chagrin I noticed that Grant was off task but then his reason was that he knew this stuff. Maybe its not fair for me to tell him that he can't do other things when he does know but I sort of insist anyhow...Generally I feel that children should be discreet enough not to be doing those kinds of things. It's more of a general kind of rule and he's not a frequent offender."

Case #13 Grant Page 2

Grant kept testing the teacher regarding another Page 2 general rule that had been made about having to sit on the floor in a group while the teacher read the novel to them. " I feel like an idiot saying 'Hey you can't sit on a chair'." Grant was rated 1 on both measures of "maturity".

The teacher assigned a 2 in "general intellectual ability", 1 for "probable highest achiever", and a 1 in "creativity". "He's a very bright little boy." He was assessed as a 2 in "probable achievement in language arts" and also for "probable achievement in math". The teacher felt he was "excellent in both subject areas."

Grant was rated as 2 in "persistence", 1 in "careful/deliberate worker", and 3 in "motivation to do school work". When asked to clarify his assessment of Grant's work habits the teacher said, "I don't know if that's entirely accurate. It's better than fair. What may have influenced what I put down or said is that he sometimes won't do his work in class time...It's not a case of refusal but he would prefer to take it home... Sometimes staying on task at school is difficult for him, or appears to be, because he doesn't do it. But anytime I say 'Okay, Grant, that's fine, but you best have that done and here tomorrow' he always does it and does it well. So is that lacking in 'persistence'? I don't know."

The teacher was not "concerned" (7) about Grant's academic progress. Additional Information

Age: 8 years 5 months (mean = 8 years 7 months)

I.Q.: Verbal 128 (mean = 112.73) Quantitative 140 (mean = 112.65) Non-Verbal 137 (mean = 111.46)

Achievement: Language Arts 93% (mean = 85.07) Mathematics 98% (mean = 93.64)
Case #13 Grant Page 3

Engaged Time Data

ļ	Téacher-Directed	Self-Paced	Total Time on Task
Mathematics	96% 24 minutes	89% 1.25 hours	91% 1.65 hours
······		nours	1.03 nours

Verbal Interaction Data

Grant	Language Arts	Math	Both Subjects
All Tinits	3.3	2.3	2.8
Product Q	4.1	2.1	3.1
Process Q	3.2	1.0	2.5
Get Attn Q	6.3	11.1	8.8
Comments	3.0	2.2	2.6
Sustaining	8.2	0.0	5.5
Tinit Prai	1.7	2.2	1.9
Tinit Crit	0.0	5.6	2.4
Behavioral	3.0	3.2	3.1
All Sinits	2.7	2.6	2.6
Ss Accepted	10/11	7/7	17/18
Total Verbal	3.1	2.4	2.8

Discussion

5.00 100 100 Grant was a couple of months younger than most class members but he received top scores on the "maturity" ratings. He had extremely high I.Q. scores and was rated high (2's) for expectancy measures. The teacher considered his academic work to be excellent.

His scores of 2 and 4 on the "cooperation" attribute may have been assigned because of his obvious impatience with delays in instruction. He was well liked and was a member of the attachment group.

The teacher felt that his effort was adequate but not perhaps his best, and he assigned marks ranging from 1 to 3 on the effort-related

attributes.

Case #13 Grant Page 4

It would be reasonable to expect that a high-achieving boy in the attachment group would be involved in a high percentage of dyadic verbal interactions and in addition would create many Sinit opportunities for himself. However, overall his percentage (2.8) of involvement in verbal exchanges was lower than class average. Only language arts Product questions (4.1) were within normal range, while math (2.1) and all Product questions in both subjects (3.1) were less. He received a low ratio (3.2) for language arts Process questions but the ratio of language arts Process questions rose to 5.3 in the reading group setting. In math, the teacher was more inclined to address Product and Process questions to low math achievers and indeed Grant received only 1.0 of Process questions in math settings.

He initiated a low percentage of Sinits (2.6) overall but during reading group settings the ratio of Sinits rose to 5.9%. The Get Attention questions and Tinit Criticism ratios are based on few teacher-delivered instances thus the three Get Attention questions and the one instance of Criticism he received resulted in inflated percentages. In all there are surprimingly few verbal contacts with Grant. It may simply reflect the teacher's perception that he was an excellent student who needed little assistance.

If I.Q. scores are a good indication of potential and if effort attribution was accurate then Grant seemed to be able to achieve excellent results without having to exert any more effort than was necessary for the task.

Case #14 Ian Page 1

Ian was described as a very "attractive" (1), "noticeable" (2) but very "unhappy" (7) little boy. The teacher assigned a 3 in "wanting to keep for another year" and was very "concerned" (1) about him. "Ian has had a lot of school problems. Ever since the outset he's really had a lot of difficulty... He repeated Grade One. In a sense he's 'turned off' school... He's sort of laid-back, easy-going, and what the hell it'll all come out in the wash sort of thing... Ian was assigned a 3 for maintaining "eye contact"."

"Ian has eye/hand coordination problems making any written assignment difficult for him...Part of the problem is this year his Mom died after a lengthy illness...He's handled it very well but sometimes he just 'spaces out' and I let him go. I don't want to put too much pressure on a kid who has a lot to live with right now...It just isn't the time to put intensive catch-up kinds of activities on him..."

Ian was placed in the concern group. "I could spend a lot of time with someone like Ian. Mostly cause I like the kid, I guess, and because he has some problems...Ian isn't the brightest lad across the board, he just <u>isn't</u>. Not withstanding that he's a nice lad to work with. He's a kid who, I think with a lot of attention, a lot of drill, a lot of workwell he's not going to be a 'scholar' but obviously I think a one on one situation would improve whatever skill level...whether it would be retained, I don't know ... I think I understand his situation. It's been a pretty rugged year for him.

Ian received a low rating (7) in "general intellectual ability" and received a 4 for "creativity". " I wouldn't say that he has an awful lot

of ability" and in addition he received the lowest expectancy Ian Page 2 rating of 7 for both math and language arts achievement and a 7 for "probable highest achiever". "With a lot of individual help he comes along, but it's painstaking in math. Math is the area I've probably spent the most time with him on. He's come along fairly well dealing with things that I didn't think at the outset he would be able to do."

Ian was in the lower reading group and was considered to be at the bottom of that group." Both Ian and Tom go to the resource room for individual help in language arts three times a week" (two of which were during math lessons)." They miss more than I'd like but they're receiving the extra help in the language arts area and you sort of have to weigh the benefits you know .. " The teacher found that their returning during math lessons often created a distraction. "It's hard enough to keep them on task when they're here for the whole lesson but when they're sort of wandering in then it becomes more difficult because I have to very carefully go over directions with these two usually and often its right in the middle of the lesson and it really interrupts the flow and it bugs me a bit..." On one such occasion Ian requested to go back out to the washroom and was refused permission and then he informed the teacher he had left his pencil at the resource room. The teacher's interactive thought at this moment was ... "Well I don't often think 'kill' but, you know, I sort of say to myself 'Gee what's going through this kid's head?' You know it's pretty scattered thinking ... It's pretty distracting. I think I'm a little annoyed, but with Ian again, I don't like to show my annoyance.

In the language arts setting of oral reading of Page 3 the novel by the teacher Ian was "turned off". "He doesn't know what's going on. I basically avoid asking him questions" in that setting.

Ian received a low score (7) for being a "careful/deliberate worker" and also for the "calm/good self control" dimension. The teacher assigned a 7 for both "persistence" and "motivation to do school work". His behavior was inconsistent in that "he tunes in and out for days and then works hard... I can't figure him out. At <u>times</u> I can't anyway. He <u>seems</u> to respond well to positive reinforcement at times..It's really hard to stay on task given the kind of situation that he has...From class experience Ian is quite often off task, specially if he's finding a task difficult." A reported interactive decision occurred when the teacher perceived that "he wasn't paying attention, or I wasn't satisfied that he was, and I thought 'proximity' will help. So I sat on his desk. He'll work sporadically while I'm there. You see he can explain this to me. He can tell me....then I would look over and I'd see he was doing nothing. The effort wasn't too great."

Concerning teacher-directed small group reading situations the teacher commented, "One thing I think it would be safe to say is that Ian doesn't concentrate on the task at hand, even in a situation like that. He's yawning, stretching....Ian spends a lot of time off task and I <u>know</u> that and one has to sort of balance off how much time to spend keeping him on task specifically and that's a judgment that you have to make and its difficult...or maybe I could ask you..'How much time <u>should</u> I spend keeping that child on task?' Even if nothing is going on essentially he'll daydream. I would hate to guess what percentage of his time he spends off task. God knows where he is ... "

The teacher felt that Ian was "approachable" Page 4 and although he had learning problems, Ian appeared to be "open" about them. "The thing is he doesn't see himself as a very bright individual. That can often be a cause of unhappiness. In his interaction with peers I think he does fine but I don't think there's any 'future' to his vision of himself... I'm pretty anxious that he will at least leave grade 3 feeling that he's a decent sort of fellow anyway with some of the basic skills that are required... School is not going to be easy for him. It's probably going to get harder for him ..."

The teacher assigned a 5 and a 6 in the two ratings of "maturity" and a 4 and a 7 on the two measures of "cooperation". An observation of a videotaped lesson was..."I was a little annoyed with him there because he was being quite disruptive. Not only that but he and Tom were off task completely and indeed as it turned out later, when I went back he hadn't finished his work. I forgave him though." Of Ian and Bonnie (the concern girl) the teacher said, "They really can't keep their act together as far as having their equipment and so on and it can be a little frustrating at times when you have to start each lesson assembling materials for them. I'm not inclined to provide them with a new pencil each day. I'm just not going to do it."

When requested in June to identify which was the more accurate rating for "cooperation" the teacher stated, "I'd say it was the lower, toward the lower side than the medium. You know how it is with Ian but he has fluctuations too. There are rare times when Ian will really, if you encourage him and so on, things are going well, then he is very "cooperative". There are other times when he's just not interested so its difficult. But on balance I would put him on the lower side."

Another reflection in June was "Ian, who had some very obvious problems, probably wasn't the most 'responsive' kid I've ever taught in terms of responding to the kinds of approaches that I took. At the same time, I felt a great deal of concern for him."

Additional Information

Age: 9 years 5 months (mean = 8 years 7 months)
I.Q.: Verbal 80 (mean = 112.73)
Quantitative 82 (mean = 112.65)
Non-Verbal 92 (mean = 111.46)
Achievement : Language arts 69% (mean 85.07)
Mathunfortunately he moved away before taking
this test.

Engaged Time Data

	Teacher-Directed	Self-Paced	Total Time on Task
Language Arts	75%	47%	62%
	3.56 hours	2.85 hours	6.4 hours
Mathematics	53%	72%	67%
	.50 hours	1.25 hours	1.75 hours
Combined	72%	54%	63%
Subjects	4.06 hours	4.1 hours	8.16 hours

Verbal Interaction Data

Ian	Language Arts	Mathematics	Both Subjects
All Tinits	6.7	9.3	7.8
Product Q	5.6	9.0	7.1
Process Q	1.5	13.7	4.7
Get Atn Q	18.8	29.1	24.2
Comments	10.7	8.1	9.6
Sustaining	6.7	8.7	7.3
Tinit Prai	7.0	5.1	6.3
Tinit Crit	11.8	5.6	9.6
Behavioral	10.5	• 7.7	9.4
All Sinits	4.4	5.6	4.8
Ss Accepted	17/21	14/15	31/36
Total Verbal	6.0	8.3	6.9

238

Case #14 Ian Page 5

Discussion

Case ≇14 Ian Page 6

The teacher displayed a considerable amount of empathy and understanding toward Ian but at the same time wanted to prod and encourage him along. It often seemed a hopeless and thankless task but occasionally a response of sustained effort on Ian's part encouraged the teacher to keep trying to accomplish as much as possible with him.

Ian was involved in a high percentage of verbal interaction with the teacher. In all, he received 7.8% of the teacher-initiated interaction for the class during the times he was in the room. More of this occurred in math (8.3) than in language arts (6.0) and much of it took the form of Explanations (15.1) and Sustaining behavior (7.3).Part of the attention in math was due to the fact that he missed part of math time twice a week while at resource room and the teacher had to make sure that he could easily rejoin class activities.

He was asked more Product questions (9.0) in math and considerably more Process questions (13.7) in math than in language arts (only 1.5%). This could be due to Good and Grouws' (1977) finding that process questions were effectively used during math explanations. The general tendency of this teacher was not to ask as many Process questions in math as he posed in language arts.

In keeping with the teacher's perception of Ian's short attention span, he received a high percentage of Get Attention questions (24.2%) More of these Get Attention questions were posed in math (29.4%) than in language arts (18.8). He often received substantial criticism and behavioral reminders for the same reasons although the teacher tried not

240 Case ≇14 Ian Page 7

to show his annoyance. Ian received a high percentage of

e F (praise for his efforts in language arts (7.0) and math (5.1).

In all Ian monopolized a lot of teacher verbal attention because he was a likeable, low-achieving student whom the teacher wanted to help and encourage. Ian was a concern student, and because he was frequently perceived to be off task and not paying attention he received large amounts of corrective and sustaining verbal behavior as well. The teacher described Tom as a fairly Page 1 "attractive" (?) and "noticeable" (2) boy. He rated him as 3 on the "happy" attribute (mean = 2.65) and reasoned that his difficulties with schoolwork were disturbing to Tom. "It's the self image. He always has a big smile on his face, but how much does that mean?" Tom was assigned a 4 for maintaining "eye contact". The teacher was greatly "concerned" (1) about Tom's academic progress and had placed him in the concern attitude group. Tom received a 4 in "wanting to keep for another year."

241

Case #15

The teacher assigned a 6 for "general intellectual ability". "Across the board he just isn't terribly bright...Progress with him is pretty painstaking." Tom would be moving at the end of the year to another school district otherwise the teacher would have attempted to have him placed in a special class situation. "I would have had him tested to see if he was a reasonable applicant for adaptation." The teacher was considering failing him. His parents were very concerned about him. His younger brother was one year behind in school and they did not want both boys to be in the same grade. "I have mixed feelings about it but I'm going to go with their feelings on it...Actually, given the succees rate of repeaters, I think they're right too...Well it doesn't matter to me...You know, the kid <u>likes</u> himself and if I do that, there will probably be enough rivalry develop that would make things difficult..."

The teacher assigned a 7 for both "probable achievement in math" and for "probable achievement in language arts". He assigned a 6 for both "probable highest achiever" and for "creativity". Tom was in the lower reading group and was considered to be " one of the lowest in the group"...He doesn't read in a flowing fashion...He does have trouble with word recognition sometimes. Tom attended Case #15 Tom resource room for individual help in language Page 2

arts three times a week and as a result missed some instructional time in math.

He had "quite severe eye/hand coordination problems which make written assignments painful for him". He was assigned a 6 for both "motivation to do school work" and for being a "careful/deliberate worker." He received a 7 on the "persistence" attribute. He was assigned a score of 5 for the "calm/good self control" attribute. On the two measures of "maturity", Tom received a 6 and a 7 and for both measures of "cooperation" he received a 4 and a 6 respectively. "Tom needs a lot of work and tries pretty hard. He's willing to, intermittently at least, he's willing to force himself to concentrate...Some things he will really strive to complete and finish and other times he's sort of not willing to do that." The teacher described him as "open" and "approachable" regarding his learning difficulties,

In the June interview the teacher was greatly encouraged by Tom's test results and his recent academic effort. "What a super job this kid's been doing. He's been working very hard. His parents have been working hard with him at home. He's been getting some positive results and feels a lot better about what he's doing, of course. He can't spell to save his life and he still doesn't transfer but he'll memorize those words and he gets one hundred percent on them ... and with math tests, if you know your basics and drill, drill, drill... that has made all the difference in the world..."

"His reading - he was in the third percentile in decoding and in the

Tom Page 3 he moved to about the 35th. It's a pretty marked improvement...His math score is 53 out of 60... I wouldn't fail him now... His scores are high enough..He's got lots of problems and he's going to need understanding teachers but I wouldn't fail him now...

Additional Information

Age: 8 years 4 months (mean= 8 years 7 months)

I.Q.: Verbal 96 (mean = 112.73) Quantitative 103 (mean 112.65) Non-Verbal 103 (mean = 111.46)

Achievement: Language Arts 71% (mean 85.07%) Mathematics 88% (mean = 93.64)

	Teacher-Directed	Self-Paced	Total Time on Task
Language Arts	84%	58%	70%
	2.53 hours	3.Q8 hours	5.61 hours
Mathematics	64%	52%	56%
	.83 hours	1.75 hours	2.58 hours
Both Subjects	79%	56%	66%
	3.36 hours	4.83 hours	8.2 hours

Engaged Time Data

Discussion

Tom was a few months younger than the average age for this class and he was considered to be quite immature by the teacher, His I.Q. scores were well below average for this particular class and the teacher assessments of his general intellectual ability, his creativity, and his expected achievement were very low.

Tom was pleasant and the sporadic effort he exerted was mutually encouraging to both himself and the teacher. 243

Case #15 Tom Page 4

p			
Tom /	Language Arts	Mathematics	Both Subjects
All Tanits	4.8	7.8	6.0
Product Q	4.6	7.1	A . 5.6
Process Q	3.1	5.7	3.8
Get Atn Q	0.0	0.0	0.0
Comments	5.6	11.7	8.2
Sustain	3.3	2.8	
Tinit Prai	3.5	11.6	. 6
Tinit Crit	0.0	0.0	0.0
Behavioral	5.3	9.0	6.6
All Sinits	6.5	4.4	5.8
Ss Accepted	28/29	8/11	36/40
Total Verbal	5.3	6.8	5.9

Verbal Interaction Data

The engaged time data represented 8.2 hours and showed that he was only on task approximately two-thirds of the time overall, and was off task even more during independent self-paced activities. He often found the work discouraging, especially written work, and tended to become distracted quite easily.

His parents, who were most concerned offered help and encouragement at home and the concerted effort resulted in reasonable achievement by the end of the year. His mark in language arts was 14 points lower than the class average mark but his math mark was only five points lower than the class average. On balance, the teacher felt that with continued effort Tom could continue to experience a certain amount of success in school. In June, because of all the hard work Tom had been doing the teacher remarked," Tom is kind of a rewarding experience right now."

Tom received higher than average ratios of verbal interaction overall.

His ratio of Product questions was 4.6 in language Case #15 Tom Page 5 received more of the Product questions in individual settings (12.5 in individual language arts settings, and 13.5 in individual math settings). He received a high ratio of Praise overall with 3.5 occurring in language arts (7.7 in the poor reading group instructional setting) and 11.6% of Praise occurring in math (17.4% in self-paced activities and 8.7 in public math settings). He received no instances of Get Attention questions and no Criticism at all. Higher than average amounts of Behavioral reminders were necessary, particularly in math settings.

Tom initiated a great many Sinits with 5.8% occurring overall and 6.5 in language arts and 4.4 in math. He was not hesitant to ask the teacher for help and assistance when he needed it.

Tom was a low achieving student who tried hard at times. He concerned the teacher who was willing to help and encourage him. He was able to command considerable verbal attention from the teacher and, as it turned out, his gains in achievement over last year's results were a source of great satisfaction to everybody concerned.

Case #16 Matthew Page 1

Matt was described as an "attractive" (1), "happy" rage (2), but not all that "noticeable" (4; mean =2.8) boy. "He's not a demanding individual in the classroom but he's sure a nice boy." He was assigned a 2 for maintaining "eye contact" and a 2 in "wanting to keep for another year." He was placed in the attachment group. "He's in with the sporting thing so he's got lots of after-school friends and so on."

Matt was assigned 2's for "probable highest achiever", for "probable achievement in language arts", and for "creativity". He received 3's for "probable achievement in math", and for "general intellectual ability". "Matt is a better than average student. He's by no means the <u>top</u> student in the class. He's pretty,good at most of the things that he does."

Matt was rated as 2 on the "calm" attribute and was considered a fairly "careful/deliberate worker" (2), and "persistent" (2). He was assigned a 3 for "motivation to do school work." "Sometimes he's a little slower than I would like him to be...He's not the fastest worker but he's an accurate and methodical worker who is a pretty bright fellow... His pace is a little slow, that's all." The teacher was not "concerned" (7) about Matt's academic progress. He received a 2 and a 3 on the two measures of "cooperation" and a 2 and a 3 for both measures of "maturity". Overall the teacher provided little information about Matt during the interviews.

Additional Information

Age: 8 years 10 months (mean 8 years 7 months)

I.Q.: Verbal 114 (mean= 112.73) Quantitative 122 (mean= 112.65) Non-Verbal 130 (mean= 111.46)

Case #16 Matthew Page 2

Achievement: Language Arts 93% (mean= 85.07%) Mathematics 95 (mean= 93.64%)

No Engaged Time Data available.

V	ert	bal	Int	e	cac	ti	on	Data

[
Matthew	Language Arts	Mathematics	Both Subjects
All Tinits	5.1	2.7	4.0
Product Q	4.9	1.7	3.3
Process Q	6.5	2.9	5.3
Get Atn Q	6.3	0.0	2.8
Comments	4.0	3.3	3.7
Sustain	4.8	1.7	3.7
Tinit Prai	4.7	2.0	3.5
Tinit Crit	0.0	0.0	0.0
Behavioral	3.9	2.9	3.5
All Sinits	4.0	2.9	3.6
Ss Accepted	19/19	7/9	26/28
Total Verbal	4.7	2.8	3.9

Discussion

Matt had more ability than was readily apparent if I.Q. scores are any indication. He was quiet and not as noticeable as other students. His overall Sinit percentage is less than average which implies that he was not overly assertive. Although the teacher held a high effort attribution for him, his slow working pace may have influenced the teacher's perception of his ability. He was assigned high expectancy scores for "probable highest achiever" and for language arts achievement but only middle expectancy scores for math achievement and for general intellectual ability. It looked as though he should have

Case #16 Matthew Page 3

been a top math student as the potential seems to have Page 3 been there. The verbal interaction during math was curiously lower than during language arts instruction. He received less than half of the amount of Product and Process questions he received in language arts and he approached the teacher with fewer Sinits during math instruction.

Disciplinary contacts were lower than class average which provides some indication of his classroom behavior and cooperation.

Matt's total Tinit percentage was an average 4.0. Most of these (4.6) occurred in large group settings. Of the 3.3 Product questions ratio all were posed during large group instruction with none involving Matt in either individual group or small group settings. He did receive a higher than class average amount of Process questions indicating that the teacher felt he was capable of answering high level questions. The greater percentage of these Process questions occurred in large group

settings during language arts instruction.

The teacher seemed to lack any additional background information about Matt but considered him to be well-liked and an able student.

248

John was portrayed as a fairly "happy" (2) boy Page 1 but a bit less "attractive" (2; mean - 1.65) and less "noticeable" (3; mean = 2.8) than other classmates.

John was assigned a 4 for both "general intellectual ability" and for "probable highest achiever" and 2 for the "creativity" attribute. He was rated as 3 for "probable achievement in language arts" and 4 for "probable achievement in math". He was in the good reading group. "He's somebody I would typify as being a reasonable, good, average studentnothing outstanding but nothing I need to worry about a lot...In terms of academics if I wanted to put my fingers on where he's at I might have a little more trouble doing that." The teacher was slightly "concerned" (3) about his academic progress.

John received ratings of 3 and 5 on the two measures of "cooperation/compliance" and a 5 on both ratings of "maturity". "John is one of the youngest kids in the class. He's fairly bright and I think any problems that John has are related to age level and maturation... I would typify him as a little immature in some of his dealings with me and with some of the other kids as well... Probably given the catch-up in maturity he would probably be a very solid student." He was assigned a 3 for maintaining "eye contact".

You know, some kids are leaders, and I'm not saying he $\underline{isn't}$ or won't be, but he's younger. He's not the most visible person in the class."

John was assigned a 3 for "motivation to do school work", 3 for "persistence" and 4 for both "careful/deliberate worker" and the "calm" attribute. John exhibited behaviors which the teacher found off-putting.

The teacher had reminded him during class to buy a Page 2 scribbler instead of having to use sheets of paper. "He hasn't had one all year. There's no problem with money. It's just he hasn't bothered to."

"Occasionally, almost in a purposeful way, he's negative and thinks it's cute to tease a little bit... There are days when I think he doesn't pay attention as closely as he ought to and so he doesn't know quite where we are... This morning when I was talking to him about spelling, that was earlier on, he basically was being a bit lazy. He wanted me to give him some information and he had the information." And on another occasion... "The only one who didn't get the directions there was John. Oh man, that annoys me cause he'll do that sometimes. He must just click off. Cause when everybody else is getting it and one kid isn't, then that's listening... It's too bad the camera isn't on him because he must be fiddling around or phased out or dazed out ... I had asked him a question as we went around. He was very insistent. You see he makes a hell of a lot of noise until you ask him a question, quite often. He'll do things like that... Here's when we're doing the estimating and the measuring with the card of the other shape and he made a heck of a racket and he did have it done and that was fine but insisted that I ask him and I was basically finished with that part of it but he still had his hand up and he was wiggling and jiggling and so I thought 'Well I may as well ask him'."

While confirming the rating of "calm" (4)..."Well you know the kind of demands that he makes when he wants to volunteer. It's not sufficient to put up a hand. It has to be 'Oh, Oh, Oh, ' and you know 250

this kind of thing...Sort of lacking in self control." John The teacher felt these shortcomings in behavior "sort of coincide with the maturity thing."

John was rated as 4 in "wanting to keep for another year" (mean= 2.69) and he was placed in the indifference group. As for others placed in this attitude group the teacher felt that John stood out "neither positively or negatively in class" and generally felt "blah" or "indifferent" about him.

Additional Information

Age: 8 years 3 months (mean = 8 years 7 months) I.Q.: Verbal 113 (mean = 112.73) Quantitative 104 (mean = 112.65) Non-Verbal 118 (mean = 111.46)

Achievement : Language Arts 82% (mean 85.07) Mathematics 92% (mean 93.64)

Engaged Time Data

	Teacher-Directed	Self-Paced	Total Time on Task
Language Arts		77%	83%
L	41 minutes	22 minutes	63 minutes

Discussion

John was seen to be a fairly immature little boy who, at times, exhibited irritating behavior and whose attention often wandered from the lesson. As this necessitated extra monitoring behavior from the teacher he was seen to be lacking in the "calm/self control"

(4) area as well as being somewhat uncooperative overall.

The assigned effort-related scores were in the 3 to 4 range which implied that he did not try as hard as he might and did not exhibit much enthusiasm at times. The teacher liked him less well than others in the class and assigned him to the indifference group. Verbal Interaction Daya

Case #17 John Page 4

John	Language Arts	Mathematics	Both Subjects
All Tinits	5.4	5.7	5.5
Product Q	4.9	5.5	5.2
Process Q	6.0	3.5	5.2
Get Atn Q	0.0	5.9	3.1
Comments	4.9	6.7	5.7
Tinit Prai	6.1	4.1	5.3
Tinit Crit	6.1	4.1	5.3
Behavioral	5.6	6.7	6.0
All Sinits	1.7	2.4	2.0
Ss Accepted	7/7	5/6	12/13
Total Verbal	4.3	4.8	4.5

His expectancy scores were in the 3 to 4 range which projected "average" achievement of end of the year results. He made below average marks in both language arts and math.

John received higher than average ratios for verbal attention which is unusual for an indifference student. Most of this attention consisted of Tinit Process questions, Tinit Product questions and Tinit Comments. Although somewhat inflated, the Tinit Criticism ratios indicate that it was sometimes considered necessary for the teacher to address such comments to John but most of these occurred on a one to one basis in private settings.

A high percentage (5.9). of Get Attention questions were posed in math to keep him on task. (The researcher noted that he would sometimes sit and just stare ahead, looking somewhat "blank" for periods of time. This particular behavior is one sociated with indifference students from other studies.)

Conversely, in keeping with typical indifference student behavior, John

John Page 5 did occur during whole class settings but the overall percentage of Sinits was lower than class average. ,

Although the teacher kept in constant contact with him and afforded high percentages of instructional interaction, John chose not to reciprocate to any great extent by initiating contacts with the teacher.

253 Case #17 John

Case #18 Bonnie Page 1

The teacher perceived Bonnie to be relatively "unattractive" (4, mean=1.65), not all that "noticeable"

(5, mean=2.81), and a very "unhappy" (7) little girl who had a lot of problems. "She's hyperkinetic and on medication which slows her down. Some days she's 'not here'. Her mother manipulates her medication and I think if Bonnie seems to be a little rambunctious, Bonnie's mother ups the dosage of it." In May he added, "She goes through phases too. To what degree the drugs have an effect on her mood I really can't say. I know, for example, last week she was down in the doldrums. Well it wasn't the drug thing it was that Mom and Dad were out of town, gone on a holiday somewhere. That happens to lots of kids but they don't respond in the same way. It really seems to have an effect on her."

"She's had severe eye/hand coordination difficulties. She's very slow at any written assignment." Concerning her drawing in the hands on the clock faces in math the teacher explained, "Her handwriting capability is such that sometimes you can't tell where something is pointing. So anyway I had the impression that she had the idea despite the drawing itself... I should show you her handwriting scribbler. It's 'something else'. She's coming along but it's slow progress and she knows it...If you asked her to sit down and write the letters of the alphabet cursive style, or even print them, it would take her... well, if you didn't supervise her it wouldn't be done...It's very discouraging for her to have to do written work. I try not to put too much pressure on her for written work, you know. She'll sometimes play that off against me saying, 'Well should I only do this much?' and if it's a reasonable amount I'll say, 'Yeah, that will be fine.' ... In a timed math test, for example, if I gave 30 practice questions, she'll get six done and

and I'll say to her 'Well that's enough. I can see Case #18 Bonnie that you know how to do that,' this sort of thing. But Page 2 she knows exactly what's going on and she finds that quite frustrating ... When she first came to me in the beginning of the year she was almost a disaster. The 'avoidance' techniques she had developed were many and varied and inventive and long lasting and so on... We had to have some frank discussions about some of the problems she has and she's very aware of the difficulties she has... My approach now is that if I see sort of a backslide into some of the old habits, then I try and jump on her pretty quickly ... I'm not big on keeping kids after school. Yesterday, during art, I had Bonnie working because essentially, almost in an act of defiance, she had not done some work hoping I'd say 'Well', or just ignore. I decided I'm not ignoring that !... So she spent the better part of a period doing what should normally take even her only twenty minutes... As I'd mentioned she was very poor at the first of the year, just getting organized and finding things. If I didn't say anything to her she wouldn't work all day. She wouldn't disrupt at all but she wouldn't do any work. She's not anything like that now but of late she isn't as good as she has been and I decided we're reaching the end of the year. 'Let's try and straighten ourselves out'... Then she tells me quite a few little fibs about where her work is and who did this or who did that. She's quite adept at avoidance techniques although they don't work too" well with me anymore."

During one recess the teacher spoke to her privately. "She initially tried to blame everybody else. 'Nick's bugging me.' Well I had a chat with her telling her that 'Hey, we've got to get it together. The work has to be done.' And it's not because she's having difficulty with the

work, it's the other thing- getting herself organized, getting going. Today she's a bit 'spaced out' but she's not angry with me for having said these sorts of things."

The teacher assigned a 5 (mean=3.7) for "general intellectual ability", a 4 (mean=3.3) for "probable highest achiever", and a 2 (mean=3.03) for "creativity". "She's bright enough, good solid average student. Of course, this is typical of a kid with her disabilities. She's a bright child." He assigned a 7 for "probable achievement in math but had difficulty assigning a ranking for "probable achievement in language arts". "She's better in language arts except that the physical aspect of writing is so difficult for her that it can be frustrating... Here's a dilemma ... Well it is achievement ... Verbally she's finecreative, imaginative, perceptive- all of those things But if you asked me how she is going to achieve on a year end test which requires other skills as well, then my answer would be that she's somewhere here." He assigned a 5 (mean=3.69) for "probable achievement in language arts". She was in the poor reading group but primarily because of eye/hand coordination problems. "She reads quite nicely orally, very expressive and she has good word recognition skills and good comprehension skills and can exprese them very adequately in the oral mode ... Verbally she is reasonably creative. She can relate some interesting stories and experiences in an effective way. Her word usage is pretty solid so she expresses herself relatively well when I can persuade her that this is something she ought to do... If she could keep up with the written aspect I could probably have her in the other (good) group. Indeed, I think she reads well orally and with better comprehension than some of the people that I have back there but there's the other work load. They

have to do more independent work at the back... Bonnie Page 4 The reason she goes to resource room is more for the oneon-one attention than anything else because she's a good reader. She responds pretty well to a lot of positive reinforcement."

Bonnie received low scores on the work-related attributes: "careful/deliberate worker" (7), "persistent" worker (7), "motivated to do school work" (7), and for possessing "calm/good self control" (6). "She's a frustrated girl in a lot of ways because she knows that she's bright but at the same time keeping up with the brighter members of the class is something that she just has to say 'Well mentally I can do that but I'm unable to...' She knows how to do all of these things but actually the physical aspect of writing it down...She's discouraged and she doesn't work at it all the time."

An interactive reflection while watching a videotaped short math quiz in progress... "I stand there quite a while at Bonnie because I know she knows a heck of a lot more than she ever does and even though it's a written thing and she has difficulty with that- nevertheless it's just a case of writing numbers and concentrating long enough to figure answers and while I stood there she did seven or eight in a row and really produced- because I was there! She wouldn't have done it normally ..." He returned to stand beside her minutes later after he had moved on.. "She stopped. As soon as I'd left she'd stopped."

She was assigned a 4 and a 6 on the two ratings of "cooperation" and a 5 and a 6 on the two ratings of "maturity".

At times she's been upset with the peer group things. Kids are not as kind to her as they might be. At the same time it's a two-way street. It always is with kids." The teacher noted that Bonnie was 257

sometimes 'isolated' in playground situations. "It Bonnie Page 5 depends on the week and there are times when she completely isolates herself and...Of course, there's always a conflict that precedes the isolation and so she has some problems in that area at various times.. ..inconsistent in that way."

When questioned as to whether her medical problem affected his perception of her he replied, "Yes, it does. The tendency might be to say that this girl has a problem but it's being looked after medically but I have to deal with her within a certain framework. It's a little different from the class."

She was assigned a 3 for maintaining "eye contact" and was assigned a 4 in "wanting to keep for another year." She "concerned" him very much (1) and he placed her in the concern attitude group. "In terms of personality my first choice for the group would be Bonnie." In June he reaffirmed his concern for her. "I'm really concerned about her.. I'm concerned that she's emotionally disturbed...She's just so up and down and has such wide swings in mood and so on..." Although the family was very obviously well-to-do and appeared as if they were supportive of her, "the girl's very unhappy, quite frankly. I'm really quite concerned about her but I don't know if all my attention would do any good... I'm really concerned about her state of mind..."

Additional Information

Age: 8 years 3 months (mean = 8 years 7 months)

I.Q.: Verbal 118 (mean = 112.73) Quantitative 104 (mean 112.65) Non-Verbal 102 (mean = 111.46).

Achievement: Language Arts 89% (mean 85.07%) Mathematics 80 (mean = 93.64%)

No Engaged Time data available.

Case #18 Bonnie Page 6

Verbal	Inter	ractiòn	Data
the second s		and the second second second	-

hand the second s	· · · ·	•	
Bonnie	Language Arts	Mathematics	Both Subjects
All Tinits	4.6	• 4.3	
Product Q	4.9	3.8	4.4
Process Q	2.1	4.2	2.9
Get Atn Q	0.0	0.0	0.0
Sustaining	2.7	0.0	1.8
Tinit Prai	2.4	7.1	4.8
Tinit Crit	0.0	0.0	0.0
Behavioral	5.6	3.0	4.4
Comments	6.3	7.0	6.7
All Sinits	2.0	1.8	1.9
Ss Accepted	7/7	4/4	11/11
Total Verbal	3.8	3.7	3.7

Discussion

Bonnie was younger than most of her classmates and was deemed to be quite immature by the teacher.

The typical concern student is a low achiever of low intelligence but whose effort encourages the teacher to keep working with such a child. His problems are mainly academic in nature. Bonnie's case was somewhat different in that she was reasonably intelligent, <u>could</u> do the work but exerted little effort. She was almost continually discouraged. Her medical and emotional problems greatly interfered with academic enthusiasm and progress and the physical problems of coordination only exacerbated her difficulties.

The teacher's perception of his role and responsibility can be seen clearly in his interaction with Bonnie. He privately revealed a lot of empathy and concern for her but when working with her, he firmly insisted that she work up to the level he felt was manageable and reasonably possible for her. He talked to her directly about her academic problems in an understanding way and helped and encouraged her to stay on task and to keep trying.

Therefore she required a great deal of supervision and monitoring from the teacher who experienced few rewards from working with her. Perhaps he felt that because many of the problems she experienced had external causes he seemed willing to maintain a consistently supportive effort in working with her. He recognized her academic strengths and weaknesses and felt she was capable of more than she was physically able or motivated to do. Her language arts mark was above class average which shows she did have ability there which could be demonstrated at times. His expectancy for math achievement had been low and her mark received was lower than class average by about 13 points.

Her verbal interaction percentages are lower than average and those, that occurred were mostly Tinit Product questions or Comments containing either Behavioral reminders or Praise. The greatest percentage of these verbal interactions occurred in the small group setting or while she worked independently in individual settings. She received a lesser amount of Process questions overall. On no occasion did she receive any Get Attention questions or Tinit Criticism for although such interactions might have been appropriate the teacher hesitated to provide negative feedback to her.

She initiated a low ratio of Sinits overall and did not seem to approach the teacher when she experienced difficulty. However, he

was quick to spot her dilemma and to provide Case #18 Bonnie the required assistance. Page 8 261

She was a very unhappy and discouraged child who had many problems which the teacher could not solve. The resource room appointments were primarily set up to help to provide some extra attention and support for her but it looked as though overall the teacher felt that Bonnie's problems were only getting worse and he was still most concerned about her at the end of the year.

Nicholas was rated as a very "attractive" (1), Nicholas Page 1 very "noticeable" (1) and fairly "happy" (2) boy. He was actively involved in baseball, soccer, and other games with the boys in his class.

From the previous year's reports his parents had been led to believe that Nick was a much better student. They were most concerned about this teacher's assessment of Nick's level but agreed that it was more realistic. "They're nice folks. They're a little older. They've basically raised their family and they are involved in a lot of things. Nick is the youngest child by quite a number of years. He has everything that you could imagine that could be bought an I'm wondering if there's more money spent than time and that might partially explain his need for attention - as much attention as he demands in a class setting. His Dad just says 'Well he's spoiled'. 'Wel!,' I said, 'I'm not going to argue with you on that one'...

"Nick's an imp is as good a way of putting it as any. I like him though." One day he wore an obscene T-shirt to class. On another occasion he had blood smeared all over his arm. After a particularly squirmy day . the teacher remarked, "He walked in here this morning and I could see right now Nick's in one of those humours and I know damn well we're in for a good day... Of course, he's really 'noticeable'. He demands attention and if you don't give it to him... You always question someone who is willing to set themselves up as the 'class clown', which Nick does from time to time, and the people in the class know that it's Nil turn for a little pat on the back sort of thing. How 'happy' is an individual like that? Nicholas seems like a happy enough individual but underlying that there must be some need for attention that goes

Case #19 Nicholas Page 2

there , he's everywhere'. It's well known throughout the school. That's the kind of label that he has earned... Comparing him to the start of the year he's improved so dramatically in terms of staying in his seat; staying on task. That may seem strange when you look at him and see him squirming, but it's <u>true</u>. And he really <u>has</u> come a long way in that respect. I was just about tearing my hair at the beginning of the year. He was much more difficult to control and more persistent in his non-malicious but scatterbrained behavior... I don't know if it was an attention-seeking thing or whether it's just the chemistry that doesn't allow him to sit still but anyway he needed a lot more correction early on than he does now, a lot more reminders to stay on task... I don't know whether he's better or I'm more tolerant. Maybe I've learned to ignore it... You know it bothered me a lot more earlier in the year... You know you could spend two-thirds of your day saying 'Sit still'. I really very rarely bug him about that."

unsatisfied...He has the label of 'he's here, he's

Not unexpectedly, the teacher assigned the lowest possible score (7) on the "calm/good self control" attribute. He was rated a 5 and 6 on the two measures of "maturity" and 4 and 7 on the two measures of "cooperation/compliance". "Nick , despite the fact that he'll occasionally run himself into hot water with me, he's not a malicious person. He's not trying to do someone in. He's also very honest. He's not going to fib me about whether he's been here or there or doing this or that. When confronted I get the straight goods." On the other hand he was often uncooperative. For example, after an interruption caused by a spider... "This annoyed me in here. Very persistent in here. I began

Case #19 to get a little ticked off. 'Not another word, Nick'. Nicholas 'Okay', he says." Another distraction during a lesson was caused by him. "Nicholas had run to the back of the room to throw something in the garbage. Of course, he knows that I'm not very fond of that." He received a 2 for maintaining "eye contact".

The teacher assigned a 5 in "general intellectual ability" (mean = 3.76), a 5 for "probable highest achiever" (mean= 3.30) and a 3 for "creativity". The teacher rated Nick as 5 for both "probable achievement in language arts" and for "probable achievement in math" although by his comments the teacher implied that Nick was better in math. "There's no question about that he's a stronger math student" and "pretty solid in math. Not terribly good in reading but not the worst either." When questioned about this discrepancy the teacher reconsidered and suggesmed, "Maybe I should have him up there (middle group =4)" He was in the slow reading group. "Nick, I think, gets a little uptight as he begins to read and he starts..gets the first four words wrong and then once he settles into it, reads the rest of the passage very well. But I've noticed this on many occasions. He gets flustered and then bombs a few words. He tries to read too quickly for one thing... "He started to huff and puff and I tried to slow him down but maybe I just put him more on the spot, I don't know ... He's in such a rush to whistle through. I think that's true of his written work too. He really takes very little care in it. I guess he looks upon it as work and work is undesirable and so get through it as quickly. as possible- that sort of approach ... He's often in a hurry to get through his work."

"In the reading situation Nick is the kind of a kid who reads

something by looking at a few words, doesn't really understand, doesn't really comprehend what's going on. Case #19 Nicholas Page 4

He's always in such a hurry to get through things. My big problem is getting him to slow down and answer the questions that are asked, to read the directions carefully, and do what's asked of him in the situation."

He was rated low on the work-related attributes, receiving a 7 (mean= 2.9) on "careful/deliberate worker", a 6 for "motivation to do school work" (mean=3.5) and a 4 for "persistence" (mean = 2.8). The teacher was quite "concerned" (2) about his academic progress.

The teacher assigned a 3 (mean = 2.69) in "wanting to keep for another year" and had difficulty placing Nick into one of the attitude groups. "Someone like Nicholas...where will I put him? I'm not. indifferent about him. I'm not sure I'd want him for another year...I don't reject him...He almost doesn't fit! He's not like these kids who are sort of ones I see without marvellous personalities and that sort of thing...I guess I'd be closer to attached to him more than anything else but not as close as some of these others, Maybe I'll do this to him ' attached *'.

Additional Information

Age: 8 years 8 months (mean = 8 years 7 months)

I.Q.: Verbal 105 (mean = 112.73) Quantitative 92 (112.65) Non-Verbal 90 (111.46)

Achievement: Language Arts 76% (mean = 85.07%) Mathematics 88% (mean = 93.64%)

Engaged Time Data

Case #19 Nicholas Page 5

	Teacher-Directed	Self-Paced	Total Time on Task
Language Arts	95%	61%	77%
	2.4 hours	2.8 hours	5.28 hours
Mathematics	72%	76%	74%
	1.8 hours	2.25 hours	4.1 hours
Combined	85%	68%	76%
Subjects	4.3 hours	5.1 hours	9.4 hours

Verbal Interaction Data

Nicholas	Language Arts	Mathematics	Both Subjects
All Tinits	6.4	5.9	6.2
Product Q	5.1	4.7	4.9
Process Q	6.0	3.9	5.3
Get Atn Q	12.5	10.0	11.1
Comments,	8.1	8.7	8.4
Sustain	4.8	3.4	4.3_
Tinit Prai	6.3	7.1	6.6
Tinit Crit	0.0	9.5	3.6
Behavioral	7.9	10.6	9.0
All Sinits	6.1	9.1	7.3.
Ss Accepted	22/29	the second s	42/57
Total Verbal	6.3		6.5

Discussion

The teacher seemed to understand Nicholas and his classroom behavior and he was patient and tolerant in his dealings with him. Nick certainly was a noticeable boy who required almost constant attention and supervision. He was very physically active as well. The teacher was amused by him at times and quite annoyed with him at others.

Nicholas scored less than average marks on the I.Q.tests which put him well below average intelligence for this particular class. He found certain tasks difficult and his coping strategy often seemed to be to

Case #19 Nicholas Page 6 267

The teacher felt he exerted little effort overall and did not take much pride in doing a good job. Most attachment students are high achievers, who try hard and who are very cooperative and compliant. Nicholas seemed to be none of these. His redeeming features seemed to be his honesty, his sense of humour, and his obvious liking for the teacher.

get the job out of the way as quickly as possible.

Nick was involved in an unusually (6.5) high percentage of verbal contacts, both Tinits and Sinits and both instructional and more managerial contacts. He received more overall attention during math periods but got more instructional contacts of the Process question and Product question type during language arts lessons.

Get Attention questions and Behavioral reminders were understandably high. Most of the kinds of attention he received occurred in one-to-one situations while he was working independently. The only Tinit Criticism he received, which resulted in inflated percentages due to the fact that the teacher gave infrequent criticism, amounted to 28.0% and this was incurred during individual settings and these were the only instances of Criticism received by him. The teacher was fairly supportive with Tinit Praise in all three settings and for both subject areas.

Nicholas was not hesitant to create verbal opportunities for himself. His Sinit ratios are high, particularly in math. However, a fair number of his overtures were inappropriate of badly timed and were therefore refused.

The kinds of work performance and classroom behavior exhibited by Nicholas made extra demands on the teacher for he required constant monitoring by the teacher who as a result became involved in a greater
Case #20 Mason Page 1 268

Mason was described as a reasonably "happy"(2),

not all that "noticeable" (5, mean = 2.8) and not all that "attractive" (3, mean = 1.65) boy.

The teacher assigned a 5 (mean = 3.76) for "general intellectual ability" and 4 for "probable highest achiever" and 'a 3 for "creativity". He assigned 4's for both "probable achievement in language arts" and for "probable achievement in math". He was in the good reading group.

Earlier in the year his parents were very distraught about the teacher's overall assessment of Mason's performance. For instance with respect to his creative writing..."He hadn't handed me in <u>anything</u>. The quantity and the quality were really not good: His mother was quite upset about it so I showed her some of the stuff. Well she went home and must have read the riot act because thereafter the quality of his work in creative writing situations has been really quite good and that is why I say 'Here's a kid who is not really a self-motivated individual but who has enough ability to be able to do maybe a little better than he shows I guess'...After last year she thought he was a top-notch student I think her expectations were that her son was a much stronger student than he really is and I think that she'll find that he's <u>not</u>. I mean he's a good , solid average student, but he's not someone whom I'd consider to be a strong student. There are lots of other kids in the class who perform at a much higher academic level than Mason."

"Mason sits up near the front. I guess I would typify him as being a fairly sort of average student. He's not a real insightful kind of student, not like some of the top students. At the same time he'll fool you from time to time and really do a top-notch job but he's not a highly motivated individual. He's not someone who's going Page 2 to get this done right now and he's excited about it...He's the kind of kid that I find- probably that's the reason that he's close to me- I have to stay on top of the situation with him. I have to make sure that he's doing what I'm asking him tordo. At the first of the year I had some difficulty keeping him on task." An interactive thought the teacher expressed about selecting Mason to give an answer was..."And, of course, I often find that I get two-thirds of the way through and I'll find that Mason hasn't had an opportunity to say much and I make a point of asking him."

The teacher assigned a 4 for "motivation to do school work", a 5 for "careful/deliberate worker", and a 5 for being a "persistent" worker. " I do notice that unless I'm directing his activity, he'll direct his own but not in a way that I necessarily like... If we talk about lacking in 'persistence' in language arts I would say that he's not as persistent an individual. He will quietly sit and hope that the issue will be avoided. Not as true in math because he is a better student in math." The teacher was fairly "concerned" (4) about Mason's academic performance.

A later reflection in May about the creative writing was... "He's doing a better job. In fact he turned in a 'book' to me the other day. Yeah, its about that thick!"

"He's not a disruptive person but for example you may have noticed that sometimes when I ask them to free read and I'm sitting at the back here filling out attendance sheets or something like that, I'll go up to the front and Mason is <u>not</u> free reading. He's not making any noise or bugging anybody but he's staring around, twiddling his thumbs, drawing a picture:" 269

Case #20

He was assigned a 4 in "calm/good self control". When asked to elaborate on this seemingly low score the Case #20; Mason Page 3

teacher said," He may have the disadvantage of proximity to me and you see everything a kid's doing...'Self control' and 'calm' may not always go together. I think, for example, when I offer time for free reading, Mason quite often won't, that's 'self-control' isn't it? In terms of being 'calm', he's quite a placid young fellow so I may have misinterpreted in his case." He was assigned a 4 for maintaining "eye contact".

He was rated 4 and 5 on the two ratings of "maturity" and he received a 3 and a 5 on the two measures of "cooperation/compliance".

The teacher assigned a 5 (mean = 2.69) on "wanting to keep for another year" and placed Mason in the indifference group.

Additional Information

Age: 8 years 8 months (mean = 8 years 7 months).

I.Q.: Verbal 109 (mean = 112.73) Quantitative 108 (mean = 112.65) Non-Verbal 119 (mean = 111.46)

Achievement : Language Arts 79 (mean = 85.07) Mathematics 90 (mean = 93.64)

Engaged Time Data

	Teacher-Directed -	Ē	Self-Paced	Total Time on Task	c
Language Arts	1007 16 minutes	Γ	962 45 minute	977 ol minutes	

Discussion

Mason exemplified the typical indifference student. He was not very noticeable, not well-liked by the teacher, and did not exert very much effort. It was necessary for the teacher to monitor his activities

Verbal Interaction Data

Case #20 Mason Page 4

		in the second second	· ·
Mason	Language Arts	Mathematics	Both Subjects
All Tinits	12.5	4.4	3.4
Product Q	2.8	6.1	4.4
Process Q	.1.5	.0.0	1.0
Get Atn Q	0.0	5.0 /	2.8
Comments .	.3.3	3.7	3.5
Sustaining	3.8	6.9	.4.9
Tinit Praise	2.3	3.0	2.6
Tinit Criticism.	5.9	4.8	5.5
Behavioral	2.6	3.5	3.0
All Sinits	0.4	0.6	0.5
Ss Accepted	2/2 * •	2/2	4/4
Total Verbal	1.8	3.4	2.5

for he was not responsible enough to remain on task unless supervised.

His I.Q. scores are lower than average for this class except for the Non-Verbal one and the teacher assigned one of his higher scores for "creativity" which is seen to be associated with non-verbal intelligence. A negative halo effect seemed to be associated with his projected achievement and the assessment of his work habits. His achievement was below average in both subjects, moreso in language arts. He had been assigned 4's on both subject area achievement expectancy ratings although the teacher felt that he was slightly more able in math.

The small amount of engaged time data spow he was well on task for the hour observed during language arts but this may not have been indicative of his usual engaged time rates.

He was involved in less than average amounts of verbal interaction. In the math lessons coded, Mason experienced a great deal of attention

Case #20 Mason Page 5

because he was having trouble telling time and the rage of teacher spent a lot of class time(both days when "time" was taught and recorded)trying to help Mason. This accounts for the 6.1% of Product questions in math which may not represent the usual amount of interaction experienced.

The teacher initiated very few Process questions with him; none were asked during math even in those sessions which involved him so intensively. His overall Process question ratio was only 1.0%. Mason initiated a negligible amount of Sinits with the teacher which is characteristic of indifference children. The higher Tinit Criticism ratios he received are inflated because so few instances of Tinit Criticism Occurred overall.

In all Mason received a less than average amount of verbal interaction and seemed, generally, to be somewhat lethargic in his approach to schoolwork. 272

Case #21 Trisha Page 1 273

Trisha was described as a very "attractive" (1), "happy" (2) but not all that "noticeable" (6) little girl. "She's a very quiet little girl but she's a little sweetheart... She's not a real outgoing, assertive individual. She gets along well with the other kids. She's. by no means isolated." She was assigned a 3 for maintaining "eye contact." The teacher assigned a 5 for "general intellectual ability", and a 5 in "creativity". She received expectation ratings of 4 for "probable highest achiever", a 5 for "probable achievement in math, and a 6 for "probable achievement in language arts". "She's average, good average in math but she needs the resource room activity in language arts. She's pretty low in language arts... She's behind in some of her language arts skills, word recognition skills particularly." She was in the lower reading group.

The teacher rated her as a 2'and a 3 on the two ratings of "cooperation". Examples of her cooperation occurred when she rejoined the class after attending the resource room. "Trisha's easy to deal with though, because I briefly give her instructions and she'll be alright." She was assigned a 3 and a 5 on the two ratings for "maturity". "I forget her exact age, but she's a little younger than most of the others..But she's a little small, not that that has necessarily a lot to do with maturity."

Ratings of 2 were assigned for "persistent" worker, for the "calm" attribute, and for being a "careful/deliberate worker". She received a 4 in "motivation to do school work". "She's a good worker. She tries

hard."

He was fairly ("concerned" (2) about her academic progress. She was rated as 2 in ("wanting to keep for another year" and was placed in the attachment group.

Overall few teacher comments about Trisha were made.

Additional Information

Age: 8 years 4 months (mean = 8 years 7 months) I.Q.: Verbal 103 (mean = 112.73) Quantitative 105 (mean = 112.65) Non-Verbal 103 (mean = 111.46) Achievement: Language Arts 71% (mean 85.07)

Mathematics 93 (mean 93.64)

Engaged/Time Data: Only about one hour s worth in language arts is

available.

	Teacher-Directed	Self-Paced	Total Time on Task
Language	100%	82	94
Arts	41 minutes	. 22 minutes	63 minutes

Verbal Interaction Data

	· · · · · · · · · · · · · · · · · · ·		
Trisha	Language Arts	Math	Both Subjects
All Tinits	1.1	1.3	1.1
Product Q	2.1	1.5	1.9
Process Q	0.0	0.0	0.0
Get Atn Q	0.0	0.0	0.0
Comments	0.4	0.6	0.5
Sustaining	1.0	2.6	1.5
Tinit Prai	3.3	0.0	2.2
Tinit Crit	0.0	0.0	0.0
Behavioral	0.0	0.0	•0.0
All Sinits	1.4	1.3	1.4
Ss Accepted	6/6	2/3	8/9
Total Verbal	1.2	1.3	1.2

Case #21 Trisha Page 3

275

Discussion

Trisha was viewed as a low achiever. Although he was quite "concerned" about her academic progress and had rated her general intellectual ability as low Trisha was placed in the attachment group, which is usually composed of high achievers and intelligent students. Her effort, persistence, and general cooperativeness may have influenced this decision. In addition, he described her as a very "attractive" little girl.

Her language arts achievement ,(71%, mean=85.07), was low as predicted, but she seemed to exceed his expectation for her achievement in math by scoring an average mark for the class. She received more explanations and sustaining behavior in math. The small amount of engaged time data collected in language arts indicated a high percentage of time on task. Her verbal percentages are consistently low, perhaps because of her "unnoticeable" personality and because girls tended to receive less interaction anyway. Her placement in the lower reading group would account for less attention received in language arts as well. For instance, for the 2226 interactions for which she was present(she missed periods of time by attending resource room) none of the 256 Process questions posed by the teacher were asked of her and only 1.9% of all Product questions were asked of her overall. She received no Get Attention questions, Behavioral reminders, or Criticism in any setting which helps to confirm that she was cooperative and compliant. She initiated few Sinits with the teacher (1.4) overall except during small group reading instruction when her Sinit percentage rose to 4.5 %. In individual math settings she

Case #21 Trisha Page 4 276

initiated a few Sinits (2.2%) but no Sinits were initiated in whole class math instruction. It seemed as though she preferred to approach the teacher in settings other than whole class instruction.

In all, although liked by the teacher, Trisha was not involved in much dyadic interaction during the 20 hours in which her verbal interaction data was collected.

277 Case #22 Merrill Page 1

Merrili was described as a very "attractive" (1), very "happy" (1), not all that "noticeable" (3) little boy of Oriental descent. "Mother spends time helping in the school library. She's a nice lady- good

parents."

The teacher assigned a 1 for "general intellectual ability", 1 for "creativity", and 1 for "probable highest achiever". Merrill received a 1 for "probable achievement in language arts" and a 3 for "probable achievement in math". "He's a good enough math student. In fact I'd say he's a good math student but...At the start of the year I had to get him to do a little extra work on his times tables because he didn't pick them up quite as fast but he knows them all now and works with them effectively. He's not the kind of individual who's an imaginative math student. He has a good memory and can learn things, that's the kind of math student...Math isn't quite as strong. I told Mom and so they drill him all the time and he's coming along quite well. He's a bright kid and it's just that his strength is more in the language arts area."

During a math lesson when several students including Merrill were experiencing difficulty, the teacher used his knowledge of Merrill's ability as a gauge. "He's very bright in fact. Just as I was talking to Merrill. I decided, or it snapped into my mind that 'Hey, there's quite a few people who aren't getting this first part. I'd better look at another way of teaching this'."

"In language arts he's easily the best person that I've got...superb! He's creative. Imagination and skill level in written work is far beyond grade three level. Very literate. His writing ability far exceeds any

Case #22 Merrill * Page 2

sort of expectations that I would normally have of a grade three individual..and then I've taught grade five as well and far exceeds most grade five students that I've ever had. I should show you some of the stuff he writes. It really is imaginative and, of course, the word usage and the proper spellings; all of those things are intact and so his written skill level is feally pretty advanced. He's a little shy and he's not quite as outgoing and verbal as some of the other kids." The teacher was not at all "concerned" (7) about Merrill's açademic progress.

Merrill was assigned a 1 for being a "careful/deliberate worker". 1 for being a "persistent" worker, and 2 for being "motivated to do school work". "He's usually a good worker." He received a 1 and a 3 on the two measures of "cooperation" and also a 1 and a 3 on both measures of "maturity". He was assigned a 2 for maintaining "eye contact".

He was rated 2 on the "calm/good self control" attribute. Concerning the "calm" rating, the teacher elaborated, "He's better than fair. Based on what I know about him, based on his academic ability, that kind of thing, I might say that his self control isn't as high as I might expect... but his is not a <u>negative</u> thing... just that he can't control his <u>enthusiasm</u> at times."

When questioned again about Merrill's shyness the teacher said, "Yes I would say that he's shy, and it's kind of odd because on the one hand I'm saying that it's hard for him to contain his enthusiasm and on the other hand there are times...for example, in a creative writing situation, something he knows he does exceedingly well, he won't want to share his

Case #22 Merrill Page 3 279

story with the rest of the group where other kids are really in there clamoring to do so...He's not painfully shy but he's an individual who is sometfmes relectant to talk."

```
Additional Information
```

Age: § years 8 months (mean = 8 years 7 months) I.Q.: Verbal 130 (mean = 112.73) Quantitative 110 (mean = 112.65) Non-Verbal 117 (mean = 111.46)

Achievement : Language Arts 96% (mean = 85.07%) Mathematics 98% (mean = 93.64)

No engaged time data is available for Merrill.

Verbal Interaction Data

Merrill	Language Arts	🧿 Math	Both Subjects
All Tinits	5.3	3.1	4.3
Product Q	6.6	4.7	5.7 (
Process Q	7.5	1.0	5.3
Get Atn Q	6.3	• 0.0	2.8
Comments	1.5	2.1	1.8
Sustain	6.7	1.7	4.9
Tinit Prai	5.5	3.0	4.4
Tinit Crit	0.0	,4.8	1.8
Behavioral	1.8	1.8	1.8
All Sinits	3.4	2.9	3.2
Ss Accepted	15/16	8/9	23/25
Total Verbal	4.7	3.0	3.9

Discussion

Merrill was a good student who worked hard and achieved good marks. He was cooperative and conforming, well-liked by the teacher and

somewhat quiet and shy.

Case #22 Merrill Page 4 280

His total verbal ratio of 3.9 is somewhat less than class average but closer examination of his verbal file reveals very high rates of involvement) in Tinit (5.3) contacts. He received 6.6 of all the language arts Product questions and 7.5 of all the language arts Process questions. He received a high rate of Praise (5.5) and Sustaining (6.7) behavior in conjunction with his language arts verbal interaction.

Totally in math he was involved in a lesser amount of interaction receiving 4.7 of the math Product questions but only 1.0% of math Process questions. He initiated a lower than class average of Sinits especially in math (2.9) settings. Perhaps because of his apparent shyness the teacher compensated by attempting to draw him out by involving him in Tinit contacts.

The teacher rated him as highly intelligent, literate, and creative. His Quantitative I.Q. score of 110, which was about average for this class, may have contributed to the lower assigned expectancy score of 3 received by Merrill for "probable achievement in math." However, he achieved a score of 98% on the end of the year math test which may have been largely due to good working habits and support from his parents.

The majority of verbal contacts Merrill received were of high instructional quality and were initiated by the teacher.

Pamela The teacher described Pamela as a very "attractive" Page 1 (1), "noticeable" (1) but very "whhappy" (7) little girl, "She's young..the mome background." (In March; before Sonia, the new girl, arrived, Pamela was the).. "only kid in the class with the single parent* situation" (excepting Ian, whose mother had died). "Mom has separated from a common-law relationship but has a lot of difficulty with the situation. It's in a state of flux all the time ... new fathers, uncles, one of those." In April, Mother came for an interview. She was concerned about the poor report card and confirmed that the "family situation was in an uproar." Since the father had no legal right to see the children she had demanded that he leave them alone ... "but the two girls had been sneaking around and seeing their father. Mom doesn't get home from work until about 5:30 and the Dad picks them up from school from time to time so it creates a lot of anxiety and conflict for them. Mom has said 'No' and yet they're still doing it... It's hard for a child that age to understand these things." In June he reflected that her home situation and family problems had functioned as a kind of label which may have influenced his perceptions and treatment of her.

She received 7's for both ratings of "cooperation/compliance" and 7's for both ratings of "maturity". He assigned a 5 (mean = 2.69) on "wanting to keep for another year" and assigned her, with certain reservations, to the rejection group. "I suppose the one who would be the most marginal in terms of personality traits would be Pamela... It's essentially a personality thing... Yesterday, for example, she was chewing a great wad of gum, so much so that she couldn't talk, so I asked her to spit it out and she did. Five minutes later another wad. So I asked her to spit it out and I asked her 'Do you have any more? 281

Case #23

'No.' So fine. Nothing more with me but at recess, 'No.' So fine. Nothing more with me but at recess, Just after science, the science teacher mentioned to me that she had Pamela spit out two great wads of gum and 'What the heck's going on?' This teacher sees Pamela in much the same fashion as I do, you see, a little <u>sneaky</u>; I guess is the word. So I came in and roasted her a bitnot for chewing the gum, but for the <u>lying</u>, and that's what it is- and its out and out. Its attention-seeking behavior, and I <u>know</u> that, and I haven't responded probably in the way I should. I should try more effective means to try and change the behavior. I just haven't <u>done</u> it for whatever reason." She was assigned a 4 for maintaining "eye contact'.

Another example he provided was... "If I walk into the main building (from the portable), she'll be down at the office wondering if she can use the phone. She'll be sort of wandering around, sneaking around, trying to avoid me because I've told the kids 'Look, if you need to use the phone, don't bother the office people. Come and see me and I'll judge whether you ought to be using it.' The answer's been 'No' often enough that she's got to usurp the authority- the hell with 'it- and she's got to get by me one way or another."

At another point the teacher was mentioning that scissors get lost mysteriously through the run of the year,..."Pamela had none at the start of the year and now she has two. So you know what's happened but I'm not going to make an issue of it as long as they share."

He described her as "perpetually late coming into a group...In a class situation she does things that are attention-seeking, you know, the out-of-seat activities...I'll be in the middle of a 'stimulating'

282

lesson and it's 'bathroom time'. ..Well-planned Case #23
pamela
distractions, I guess more than anything.. "I should Page 3
be nicer to Pamela than I am... I leave..feeling that often.
You know, she's one of those kids that you sometimes just don't interact
terribly well with." Nevertheless, his assignment of her to the
rejection group was "conditional" as he "did not feel terribly
negative about any of them."

283

He assigned her a 7 for "general intellectual ability", a 7 for "probable highest achiever" and a 7 for "creativity". She received a 6 for "probable achievement in math" and a 6 for "probable achievement in language arts" as well. "She's a low average student...a little better in math than in language arts. I would say that she's not a terribly bright individual. She doesn't work to capacity." She was in the poor reading group and she attended resource room for individual help in language arts three times a week. Overall he was very "concerned" (1) about her academic progress.

She was rated 7 on the "calm/good self control" attribute, 7 for being a "careful/deliberate worker", 7 for "persistence" and 7 for being "motivated to do school work." On one occasion she'd returned from the resource room and had managed to complete quite a bit of the worksheet although she had missed most of the math period. "It's funny. Sometimes she'll move along more quickly than I think." But more often she was perceived to be "off task"...Reflecting while watching # videotaped lesson the teacher said, "I think I have a little private chat with Pamela here saying 'Stay on task' but then, of course, she doesn't." And on another occasion... "I went up to the front (to the poor reading group) because I was watching Pamela and she was way off task. There were others too but she was the worst...She had put Case 23 Pamela her stuff away and I wanted to see what she had Page 4 done cause I hadn't seen her do any work. She had done very little ..." She was one of several who habitually tuned out' during the reading of the novel. As a result the teacher avoided directing questions at her in this setting.

Additional Information

Age: 9 years 1 month (mean 8 years 7 months) I.Q.: Verbal 90 (mean = 112.73) Quantitative 91 (mean = 112.65) Non-Verbal 111 (mean = 111.46)

Achievement: Language Arts 71% (mean 85.07%) Mathematics 87% (mean = 93.64).

Engaged Time Data

	Teacher -Directed	Self-Paced	Total Time on Task
Language Arts	91%	57%	76%
	1.4 hours	1.1 hours	2.5 hours

Discussion

Pamela was one of the oldest students in the class but was perceived to be one of the most immature. Perhaps because of this the teacher actually perceived her to be "young". Knowledge of her home problems helped to explain why she exhibited certain behaviors at school and enabled him to be fairly tolerant of her almost constant misbehavior. A negative halo effect appeared to influence his perception of her various attributes. She was considered to be lacking in general intelligence and her I.Q. scores indicate that she was considerably below average ability for this particular class. He felt that she did. not use what ability she did have to any great advantage and his effort

Case #23 Pamela Page 5

Pamela Language Arts Mathematics Both Subjects All Tinits 4.5 4.2 4.4 Product Q 2.8 3.9 3.2 Process Q 0.5 1.4 0.7 Get Atn Q 12.5 0.0 8.7 Comments 9.6 5.2 7.8 Sustain 1.9 2.3 2.0 Tinit Praise 5.5 1.3 3.9 Tinit Crit 17.6 5.6 13.5 Behavioral 11.8 7.7 10.4 All Sinits 4.4 6.2 5.1 13/16 Ss Accepted 19/21 32/37 Total Verbal 4.5 4.8 4.6

Verbal Interaction Data

attribution for her was the lowest possible. His expectations for her achievement were accordingly fairly low as well. Her actual achievement scores were below average but she scored only six points lower than class average in math compared to the fourteen points lower than average in language arts. His assessment that she was a better math student seems to have been accurate.

Although placement in the rejection group never did reach the definitive stage, it is interesting to note the various attributes she possessed which are common to the "typical rejection students" from other studies. Brophy and Good (1974:164) noted that teachers tended to reject children who are overly active and assertive, especially if they are low achievers as well. Findings from many studies reviewed by Brophy and Good suggest that teachers perceive the demands made by rejection students is overwhelming and illegitimate. These students are frequent behavior problems. They receive a high number of behavioral and critical comments. The teacher tended to avoid them in public settings. On the basis of examining date on 37 rejection boys and 25 rejection girls assigned to rejection groups by 28 teachers (less than three per room) an interesting sex difference was noted that applies in the present case perhaps. Rejection boys usually came from intact families in which both parents were living while teachers made frequent negative comments about the families of rejection girls (broken home, or poor parental cooperation) along with the more usual comments about failure to pay attention, poor general ability and poor work behavior which would apply to rejection students from both sexes.

The small amount of engaged time data available suggests that she wasted a fair amount of academic learning time by being off task which substantiates the teacher's perception of her poor work habits and effort.

Pamela received average amounts of total verbal interaction but less than average amounts of the more instructional contacts. Overall in language arts, of the 201 Process questions posed only one was addressed to her resulting in a ratio of 0.5% for Process questions in language arts. Product questions in language arts were somewhat higher at 2.8. (Most of these (5.9%) were addressed to her in individual settings and only 2.2% in public settings). More instructional contacts were afforded in math. She received a high percentage of Get Attention questions, Behavioral reminders and Tinit Criticism. She received a fair amount of Tinit Praise in language arts (5.5 overall) with 11.1% accruing to her in the poor reading group setting but not as much praise

286

was given to her during math. The teacher appeared to be taking advantage of opportunities to praise her when he could find them.

She initiated a higher than average number of Sinits with more occuring in math than in language arts.

In all Pamel'a was a demanding girl who made her presence felt in' the classroom. Her lack of effort and cooperation and defiance on occasion resulted in the teacher conditionally assigning her to the rejection group. Although he acknowledged and regretted holding such feelings toward her and he attempted to deal with her patiently, many of the verbal contacts that resulted were of a disciplinary and managerial nature and were not the sort of verbal contacts that are of greatest educational benefit to a student. 287

Case #23

Pamela Page 7

Maureen was described by the teacher as a very "attractive" (1), very "happy" (1), and "noticeable" (2) little girl of Oriental descent. "She's one of the kids who takes the bus right after school. Quite frankly I don't know very much about her extra curricular activities...The interacting often takes place in the informal part of the day on an informal casual basis after school... She's not somebody who's assertive in that sense at all." She was seen as a "quiet", "cooperative" (1 and 1), "mature" (1 and 2) little girl and the teacher assigned a 1 for "wanting to keep for another year." She was placed in the attachment group. She was assigned a 1 for maintaining "eye contact".

He rated her a 3 in "general intellectual ability" and 2 in "creativity". He assigned a 3 for "probable achievement in language arts" and a 1 for "probable achievement in math" and a 1 for "probable highest achiever". On balance the teacher's comments implied a higher language arts expectation than he had actually assigned. "She's pretty strong in language arts, not nearly as strong as in math though. Math is her stronger subject, as per the stereotype." In June the researcher asked for further clarification concerning the lower language arts expectation and creativity assessment he held for her. "Well maybe it's my expectation. She is so high in other areas. But, for example, if we look at a piece of creative writing, the stories are simplistic and not particularly imaginative... Her's are a good average <u>length</u> but they're pretty dull stuff. But in other areas she's a superb student. For example in reading, she reads well." (She was in the good reading group).

"She's a pretty strong language arts student. She's not the most creative individual when it comes to expressing her own thoughts. In math, of course, she's a very capable student."

3

Maureen was assigned a top rating of 1 on the following work-related attributes: "calm", "careful/deliberate worker", "persistent worker", and for being "motivated to do school work." The teacher was not at all "concerned" (7) about her academic progress. It was noted by him that she did interact frequently with Joanne during lessons but not in a disruptive way. "Very frequently I find that when they are exchanging words it is work-related. Maureen and Joanne have been seated beside each other all year long. They're good friends. There's a lot of interaction between them in a class situation but they help each other. That's fine! "

Additional Information

Age: 8 years 7 months (mean *= 8 years 7 months)

I.Q.: Verbal 122 (mean = 112.73) Quantitative 133 (mean = 112.65) Non-Verbal 117 (mean = 111.46)

Achievement: Language Arts 94% (mean 85.07) Mathematics 97 (mean 93.64)

No Engaged Time Data available

Discussion

Maureen was of average age, possessed a very high I.Q. and had excellent work habits and good achievement results.

For a student of such high calibre, she was involved in very

few verbal interactions with her teacher. She was perceived as non-

1.6

Maureen	Language Arts	Math	Both Subject	
Abl Tinits	1.5	2.2		
Product Q	2.0		1.8	
Process Q		3.9	2.9	
	2.0	1.0	1.6	
Gèt Atn Q	0.0	0.0	0.0	
Comments	0.7	0.4	0.6	
Sustain	1.9	1.7		
Tinit Praise	0.8		1.8	
Tinit Crit	I.	2.0	1.3	
	**0.0	0.0	0.0	
Behavioral)	0.9	0.0	0.5	
All Sinits	0.8	1.3	1.0	
Ss Accepted	4/4	3/4		
Total Verbal	1.3		7/8	
1	1.5	1.9	1.6	

Verbal Interaction Data

assertive and very quiet. She initiated very few Sinits with the teacher averaging 1.0% (mean =4.1) overall although this percentage rose in the small group (reading) mode to 4.5. Perhaps, as a result of her non-salience, the teacher tended to underrate her intelligence by assigning a 3 for general intellectual ability, although, if I.Q. is any indication, she appeared to have possessed very high academic potential. No criticism, no get attention questions, and only 2 out of 398 behavioral reminders were given to her which indicates she was indeed a student who conformed to classroom rules.

She was a high achiever who put forth an excellent effort, was cooperative and mature. However her amount of involvement in Product questions and Process questions was 2.0% for each in language arts and the ratios received for math Product questions (3.9) and for math

291

Process questions (1.0) were below class average as well.

Because she was involved in so few verbal interactions (43 out of 2715) the accompanying verbal behaviors of Praise and Sustaining are low for her as well.

Although the teacher may have realized that Maureen needed little instructional assistance, he usually aimed to actively involve students verbally in ongoing instruction. Maureen seems to exemplify the good and quiet student who can escape verbal attention in a busy classroom.

3

6.5

Ph-

Joanne was rated as a very "attractive" (1), Very "happy" (1) and "noticeable" (2) little girl. Case #25 Joanne Page 1

The teacher assigned a 1 for each rating of "cooperation/compliance" and a 1 for each rating of "maturity". "She's a real nice kid; bright, inquisitive, all kinds of nice things to say about her. She does a good job. She'd certainly be an obvious leader in class. She gets along well with others. She's in a sense a little more mature than many of the other children in the class." She was assigned a 1 in "wanting to keep for another year" and was one of the "first three choices" for the attachment group. "I really like Joanne, you know, and the thing is we have a lot of interaction after school and so on: She's enthusiastic about everything. She's into this and into that and a kid who really has her head together, knows where she's going and yet she's still a kid. She's not a pseudo-adult."

She received the top rating of 1 for <u>all</u> of the expectancy measures: "general intellectual ability", "probable highest achiever", "probable achievement in math", and "probable achievement in language arts", and for the "creativity" attribute. His comment in May was, "Here's a kid who does awfully well. Gee, I marked the first half of her reading test. She has 100% so far, so you know she does pretty well. It's not a mirage, her capability, and I know her other teachers who have had her in the past have felt the same way...so." He was not at all "concerned" (7) about her academic progress.

When questioned about his selection of students for answering a question (while viewing a video tape) he said, "These kids <u>deserve</u> to be asked cause they know this stuff quite well. I've been surprised so rarely this year by Joanne <u>not</u> knowing something that I sometimes say

292

to/myself 'Hey I've hardly asked this kid a question all day. When you teach a large group like this I try and get to everybody... As I've mentioned before that sometimes it's easy to overlook people who do things <u>right</u> all the time and she doesn't <u>demand</u> attention the way some children do."

On another occasion concerning the selection for answering... " I ended up with Joanne - not because I wanted to hear her read again but because it occurred to me that when she had read I hadn't said anything about her reading and that I ought to say 'Well, Okay very ... good' or at least that, you know ... She enjoys our interaction but it's on a somewhat different level...It's not the 'see-what-I've-done-praise-me' kind of attention...Yeah, she's really quite mature I think. There's a bit of banter which is quite unusual between adults and children of that age."

While watching a video tape and reflecting on the quality of answers received ... "Joanne gave me a very good answer here- the kind of answer I've learned to expect from her. She had whispered to me before 'Hey, I've got an answer' ... and conversely, "Joanne's in a funny mood today, sort of a little baby talk and that sort of thing... So I thought it was not a well thought-out answer and certainly hot her usual standard answer...She's being a little silly today for whatever reason. I allow her the occasional slip. Why not? You know I moved away from her pretty quickly. If I'm not getting a good answer, if it's a silly answer, and that's what it was on that occasion, I guess she would get that 'message'. I don't know if it would prevent her from being silly again...She might just have one of those days, you know. But we understand each other pretty well." Her work-related behavior received top ratings of 1 as well. She was deemed to be highly "motivated to do school work" (1), a very "careful/deliberate worker" (1), a "persistent" worker" (1) and very "calm" (1).

While viewing a video tape again ... "I don't know who's wandering around there..I didn't see...Joanne. Oh well it's alright if it's Joanne. She has a good reason, no doubt." And on another occasion when questioned whether Joanne might be 'off task'..."If she's off task briefly, so what? You know because she's a bright little girl who probably has the concept whether she's completely task-oriented or not." Basically the teacher felt ..."There is no problem with her paying attention." She usually chose something non-disruptive when she finished her work early, which frequently happened...like reading a book. "Joanne and Maureen may have been checking each other's work too, that sort of thing. They do quite a bit of that. They learn from each other. That's why I would never move them. They're discreet about it."

Additional Information

Age: 8 years 7 months (mean = 8 years 7 months)

I.Q.: Verbal 127 (mean = 112.73) Quantitative 126 (mean = 112.65) Non-Verbal 130 (mean= 111.46)

Achievement: Language Arts 98% (mean = 85.07%) Mathematics 100% (mean = 93.64%)

No engaged time data available.

Discussion

Joanne was obviously one of the teacher's favorite students and she seemed to deserve the positive halo effect reflected in her teacher294

Case #25

Case	#2	5
Joann	ie i	
Page	4	

Joanne	Language Arts	Mathematics	Both Subjects
All Tinits	3.5	3.7	3,6
Product Q	4.3	2.5	3.5
Process Q	4.5	7.8	5.6
Get Atn Q	0.0	5.0	2.8
Comments	1.8	3.7	2,7
Sustaining	4.8	0.0	3.1
Tinit Praise	. 5.5	5.1	5.3
Tinit Criticism	0.0	4.8	1.8
Behavioral	1.8	5.3	3.3
All Sinits	6.3	6.8	6.5
Ss Accepted	28/30	15/21	40/51
Total Verbal	4.4	4.5	4.5

Verbal Interaction Data

at school.

assigned scores. She was the kind of model student who was a classic attachment candidate. She was cooperative, mature and had a personality the teacher liked. The teacher was careful not to show the favoritism felt in his dealings with her although, as he said, "We understand each other pretty well."

She was intelligent and put forth her best effort consistently and thus, not unexpectedly, achieved excellent results. She was cooperative during lessons although she liked to tease the teacher subtly. For example, she was tapping his shoe one day as he was reading to the class seated on the floor at his feet. Some of the substantial number of Sinits were refused perhaps for reasons along these lines.

Although her overall verbal interaction percentage is within average range (4.5) most of the interaction occurred as Process questions which were the highest instructional type and as Sinit interactions which she created for herself so frequently. In all, Joanne was well liked by the teacher and her friends and was having an enjoyable and successful year

295

Sonia was described as an "attractive" (2), not all that "noticeable" (3), not all that "happy" little girl. The teacher assigned 3's on both ratings of "maturity" and a 4 and a

5 on the two measures of "cooperation/compliance". She received a 3 for maintaining "eye contact" with the teacher.

Sonia was rated as 5 for "general intellectual ability", 4 for "probable highest achiever", and 3 for "creativity". The teacher assigned a 5 for "probable achievement in math" and a 3 for "probable achievement in language arts". He was fairly "concerned" (2) about her academic progress.

She was deemed to be not overly "motivated to do school work" (4), not overly "persistent" (4), a less than average "careful/deliberate worker" (3), not all that "calm" (4). The teacher assigned a 4 for "wanting to keep for another year" and tentatively placed her in the rejection group.

Sonia's case results in a particularly interesting profile because she joined the class on April 8, 1979, as a new student. The researcher capitalized on this opportunity to gather formative impressions of her during her first three weeks in the class. The above ratings were provided after about a month when the teacher felt better able to assess her on the various measures.

The following perceptions gleaned both from regular interviews and during stimulated recall interviews date back to the first teacher comments provided to the researcher about Sonia and are presented in chronological order. It was expected that during the process of forming his impressions about her the teacher's thoughts might be contradictory at times and would provide the range of background Page 2 information that contributed to the formation of his perceptions. It is particularly interesting to note her classroom behaviors on various occasions which appear to contribute to his eventual placement of her into the rejection group, albeit conditionally. April 9: His first expressed impressions were..."a nice little kid, fairly outspoken. I think in judging from her Cum.card that the situation at home isn't as stable, Mom and Dad have separated, as one would normally find in this community... She's average in language arts, just on first impressions. She's a little behind, compared to this group, in some of the math skills...

April 10: "I'm actually reasonably pleased with the way that she's settling in. Based on the first day I was asking myself the question 'Is this going to be a kid whom I'm going to have to be correcting the 'speaking out' all the time?' I got that impression initially because she was quite outgoing. She was very willing to chat, chat, chat all the way. Most kids when they come to a new school situation are quite meek and a little frightened and this sort of thing and she was <u>none</u> of that and I thought 'Oh, Okay. Here we go'. I'm quite satisfied now that she is well mannered and that's good as far as I'm concerned."

From other conversations the teacher stated that at the first of the school year, when getting to know his students, he rarely looked at the cumulative record card but preferred to form his own impressions. For the sake of expediency and because it was late in the year he did consult the information contained therein. " I had taken a quick look at the Cum. card for at this late date in the year, you want to decide

Case #26

what groupings to put a kid in and so on. She'll fit in Page 3
and be average in the top group" (reading). She had initiated
some questions during the class. "Yeah, I don't think she's shy."
The teacher had to make use of Get Attention questions when.
it appeared that her attention was wandering. "She wasn't quite on
task at that time which was the reason I asked her the question. She
had a 'glazed eye' look. It's interesting, you know, she says that she
has taken this particular lesson at her other school and the
assumption is then, that in her own mind, that she knew the information
(math) quite well. That was not the case, as it turned out... "

50

She was caught whistling during the lesson... "I don't know if it was a nervous habit or what..." Still later during this same math lesson she pulled out a reading book and became engrossed in it instead of listening. "Maybe it's unfair, but I like kids to at least look semi-interested and if you're reading a book in math class, that has nothing to do with math, then you've demonstrated ... I guess I'm always a little insulted by that sort of thing ... She looked quite remorseful at this point. She's not frightened but at that particular point she knew very well that she was wrong and that it was quite clear that I could show her that 'Hey, you don't understand this as well as you ought to'. Her questions were (a) not finished and (b) some of the ones she had finished were wrong, were incorrect, and she didn't really have the idea of certain parts of it anyway...Maybe that's all the correction that will be needed. She doesn't strike me as an individual who goes out of her way to be testing, you know. She's not one of those who will push, push, push to see how far ... I haven't known her long enough to

Case #26

know whether in a situation like that she would Case #26 Sonia Page 4 ask for help or whether normally she pulls out a book and does her own thing when things are difficult. I don't really know yet."

April 12: "These two, Marilyn and Sonia, I'm going to have to move them, separate them, because they've developed a quick friendship and are chatty. They're not disruptive but they quietly chat very often and so you may notice from time to time I sort of focus attention on that whole corner." Again he directed a Get Attention question at her. "I think she was off task. I can't remember exactly but you know the old 'glazed look'. I don't know if it was there or sometime later... " Concerning her potential ability and performance in math... "These are the first samples of multiplication work that I've had from her and she's far behind the rest of our group in that respect so I'm going to have to do some things to help her catch up with the rest of the group. She seems to be a little stronger in language arts than she is in math...just initial assessment."

"She and I will get along very well. I just want to impress on her early that - well to get her accustomed to the kinds of things that all the other people in the class are accustomed to, regarding me. Yeah, I have no fear that she and I will get along. I'm sure that we will."

Long Easter weekend intervened....

April 17: "I'm reasonably satisfied that she's fitting well into the class. ... I've noticed that little Sonia, the new girl, doesn't participate in this kind of situation, so it would seem this far, quite as willingly." The situation concerned was volunteering during small group reading sessions.

299

The teacher asked her teacher-initiated questions Sonia Page 5 in whole class situations..."Yeah I tend to get to her in the full class. But she doesn't volunteer as often as some of the other children in this group setting anyway." It was noted that Sonia was answering most questions in the lesson correctly... "Well she got <u>those</u> certainly and I'm still in the process of assessing her. It's interesting right there. She didn't even have her work <u>out</u>; you know. I don't know what she was doing but I was

surprised when I got there."

"As far as Sonia's concerned, I'm reasonably satisfied that she's fitting in fairly well. She seems to be getting on well enough with the other kids. I don't see any signs of conflict with them or anything like that as yet. So my initial reaction is that 'Here's a nice little girl who's going to fit well into the situation'. I think in math, in some of the basic skill areas that we've worked pretty hard on this year, that she's a little bit behind the class and I'm going to have to get some flash cards to her and to spend a bit of time with her to catch up. I want to know where she's at before we have these big year-end tests , in May sometime ... As I go around on one of those quick quiz things I often will throw in questions that will give me a pretty good indication of whether a kid has a skill...and I know from those that she's having trouble with multiplication..."

"'Could you please wait for a few minutes?' That sort of thing bothers me. When you're in the middle of something like that and someone asks you if they can go to the bathroom"..You know, it's not a hard and fast rule although I do tell kids to be a little discreet when they..." There were only two minutes to go when she had asked. "There are times 300

Case #26

to ask and there are times not to ask."

Case ∦26 Sonia Page 6

Part of the math activity was to emboss over coins with a pencil and fill in the appropriate amount of money using that method on a worksheet. He took it for granted that it would be fun and easy for them but the students experienced some difficulty in holding the coins steady. "Most of them handled it pretty well but there were some of them who said 'I can't do this!' Oh, and Sonia, at the back, I had to sort of direct her 'Now this is what I wanted you to do' not because it was so terribly important but I don't want her to get the idea that she can sort of do everything her own way...that sometimes...well let's just take the time to do it this way." It was noted that she had seemed to be getting exasperated in the attempt to do the exercise..."Yeah she was and at one stage I said 'You're not going to do it your way, you're going to do it my way' more or less and that was the message. And so I spend some time with her to show her that she indeed <u>could</u> do it if she wanted to."

April 19: "Sonia is an interesting little girl...Something happened yesterday and I don't know much about it yet but she was late getting back after recess, quite late, late enough that I went looking for her. This was yesterday afternoon, and she arrived and I asked her 'Where have you been?', you know, as I was wandering the halls and saw her. And I guess she had been involved in some kind of conflict. She'd been threatened apparently by three or four grade six girls, or so the story goes. My question is somebody new getting involved in that kind of a situation so quickly? 'What's going on?'. I haven't the whole story yet. I haven't decided yet whether she was sort of being a little upstart or whether this is a genuine Case #26 Sonia Page 7

problem that she has run into and I'm wondering what the reasons are. So I'm waiting and watching with interest to see what develops because it's sort of unusual to be in that situation ... I haven't had a chance to talk to the principal about it. She was down there chatting with him and then making these accusations that these people were harassing her..."

She had bitten the inside of her mouth and wanted to go inside to the main building to the washroom to look in the mirror. "Well yeah, I'm not buying that sort of stuff. She was disappointed that I wouldn't let her...The 'big hole in her mouth' won't heal just by going and looking at it in the mirror ... Anyway I'll be watching to see what develops on the playground with her..."

During the math class that day, Sonia took out a comic book and began to read it. Along with being told to pay attention she was informed that you don't read comics in math class for any reason unless I give you special permission.' It was pointed out that this was a similar incident to the one last week in which he had 'had it out with her'. "Yeah, I <u>did</u>. Well. Okay, here's this little interaction I have with Sonia. I guess I sort of think...I guess it maybe irks me a little bit to have someone tell me 'I hate doing that', that sort of thing..." "The kids in this class are largely from stable home situations where respect is an important thing and there's lots of money and they're kind of status quo people by and large and she may be somewhat outside that norm in the class... but she <u>seems</u> like a nice enough kid. I'm not concerned I'm going to have any real problem with her. I just think her approach and attitude are somewhat different than many of the other kids in the class."

The teacher was asked if he would draw any similarities between Sonia and Pamela in that way.

"Well to the extent...I mean "they're much different children, but to the extent that there are things I can notice about their behavior that are in some ways different from most of the other kids and that both of them are from broken home situations. I suppose there's <u>that</u> similarity."

The teacher was asked if his authority ever got challenged by any of the students, and if so, on what occasions?

" I think thus far, maybe it's early to say. I would say that I haven't run into that at <u>all</u> in this class until now. You know the business of ... either that or it's been in a friendlier manner or something. I don't know. It hasn't been as noticeable or maybe as overt or something ... You know she's been around a couple of weeks. I'm quite certain that - well it's highly unlikely that she and I will have difficulties that reach crisis proportions or anything like that. I'm not worried about that at all."

The teacher was asked what "maturity" ranking he would give her. "Well 'maturity' encompasses many things. You know, worldliness and responsibility are far different. I think I'd put her slightly above the middle just now, I might change my mind on that." April 26: The teacher stopped to spend some time at Sonia's desk during the lesson in progress. " She and I had a little battle earlier this morning with regard to work left at home and I suspect work not done. And I wanted and <u>did</u> make very clear that that's 303

Case #26
Case #26 unacceptable and that's why I had her in at Sonia Page 9 recess briefly to tell her those things and so then I thought I better say a word or two her to show her that ... " When asked if he had any further thoughts about Sonia the teacher said, "Well I'm still up in the air about her. I think she's maybe one who will try and get away... I think this morning when she didn't bring her work, for example, my suspicion is, and it's without evidence, is that the work just simply wasn't done and, for example, they had some other work to do too, that was the spelling that I had assigned, and she hadn't that finished either. So I'm going to have to keep a fairly close eye on her to be sure that she, in fact, completes tasks. She gets along with kids in the class and I'm satisfied that she is fitting in fairly well but I think she's sort of an 'independent thinker', shall we say, and wants things to go her way and she's not... I wouldn't put her down as being the most conscientious student by any stretch of the imagination but I think those are things that can be worked out fairly easily. When someone comes in as late as she has it's hard to know what routines she's accustomed to." When asked whether her work was neat he said, "It's fine. She doesn't appear to have any handwriting difficulties or anything of that nature."

The situation with the grade six girls had resolved itself without further incident ."There hasn't been a recurrence. I don't know how it was instigated or anything about it... " "She'd be an average student generally. a little lower than average perhaps compared to some of the people in some areas, most particularly in math. She's a little above average perhaps ... I shouldn't say above

304

average of the class, but above the lower group Case #26 Sonia that I have anyway on language arts but not too, Page 10 well founded in some of the math areas. It's going to take some work and also naturally there's the process of- she's learning the routines, my expectations and so on and so forth. I wouldn't say that she's testing exactly, well maybe she is...It's not an overt kind of testing. It's not like that. It's more a situation where she's feeling her way around and finding out what she can get away with a little bit and that

sort of thing. So I'm being fairly firm with her as a result, in situations like that."

May 10: "I think we're at the point now where I know her strengths and weaknesses. She seems to be a much more effective language arts student than she is a math student. I think her language arts ability is pretty strong in many ways. She's sometimes a little careless in her work. She can be. "

" I think she likes intrigue a bit. She can be a little sneaky at times. I find her down in the janitor's room. She has developed a friendship with the janitor who lets her into all the places that are generally restricted. As a new student she seems to know the ropes better than most do after five or six years at this school which is interesting. But I don't see any particular harm in it."

He was asked how the social adjustments were..."She seems to get along pretty well with the kids. She's kind of an independent soul. She'll look after herself if need be."

June 7: He was giving serious' consideration to placing Sonia in the rejection group and decided on a "not definite-conditional" kind of

placement. "I'd put Sonia in that sort of category. She's sort of, oh- I don't know ... She's a funny kid..just the chemistry's wrong, something like that." 306

Case #26

Sonia Page 11

Additional Information

Age: 9 years 0 months (mean = 8 years 7 months)

I.Q.: Verbal 104 (mean 1¹2.73) Quantitative 103 (mean = 112.65) Non-Verbal 97 (mean = 111.46)

Achievement: Language Arts 71% (mean 85 07%) Mathematics 80% (mean = 93.64)

No Engaged Time Data available.

Sonia	Language Arts	Mathematics	Both Subjects
All Tinits	3.3	7.4	5.4
Product Q	3.3	6.3	4.9
Process Q	1.0	11.1	4.2
Get Atn Q	0.0	5.9	4.5
Comments	5.9	9.6	7.9
Tinit Prai	5.0	9.6	7.5
Tinit Crit	6.3	0.0	4.0
Behavioral	4.2	6.5	5.4
All Sinits	1.6	6.9	3.9
Ss Accepted	3/3	9/10	12/13
Total Verbal	2.8	7.3	5.05

Verbal Interaction Data

Discussion

Sonia was quite a bit older than most of her classmates. Her behavior and attitude were on the whole rather self-assertive when compared to the more compliant members of the class. Her home background may help to explain why she had developed certain behavior patterns. It looked as though she had a little less respect for the teacher and might have had better luck getting her own Sonia Page 12 way with her former teacher.

His expectancy of 3 in language arts as well as the comments he made about her ability and performance in that subject area suggested that he thought she would achieve better results in language arts. Her score was 14 points lower than class average. In math the expectancy (5) assigned was lower and it was obvious from his comments that he felt she would perform less successfully in this subject area. Her score of 80% was 13 points lower than class average. So she achieved well below average results in the end of the year tests.

It is difficult for a child to adjust to a new school part way through the school year. As it turned out, this appeared to be a regular occurrence for her. The researcher needed to double check on one of the⁴ I.Q. scores and so phoned the school in January, 1980 to confirm the information. She had left the school, in which the study had been conducted, in September. The researcher phoned the new school only to find that she had moved again in December and now attended yet another school in the city. Eventually the information needed was retrieved but the additional information gained was of interest for this child had been forced to adapt to four different schools in less than 10 months. Such adjustments require certain coping styles and she appeared to have developed them but not all of them fitted the conforming child model as might be expected.

The teacher's effort attribution for her was in the medium range so she was not seen to be a high-achieving, enthusiastic student and in addition she occasionally made subtle challenges to his authority which, although they were not serious in nature, became mildly irritating to the teacher. 307

Case #26

Her verbal interaction ratios are based onCase #26the 1464 interactions for which she wasSoniaPage 13

present. More of these interactions occurred in math partly because he was trying to assess her capability in that subject area and partly because he spent more time in math instruction helping low achieving students generally.

She initiated a great many Sinits (6.9%) in math but only 1.6% in language arts. He seemed to feel she did not wish to volunteer and participate in language arts settings although he felt her ability level was quite high in this subject area. Math settings evoked more Get Attention questions and Behavioral reminders which were directed to her. Her overall higher than average verbal interaction ratios may be mostly due to the fact that she was a new student settling in and the teacher was concentrating special attention on her as he tried to form his impressions of her behavior, ability, and performance.

- 308

SUMMARY

Individualized profiles were developed for each of the 26 student subjects in this study. Although each profile contains the same basic and requested information which had been provided by the teacher in the form of numerical scores and comments, the case studies differed in various ways; The numerical scores (from 1=high to 7=low) revealed a range of teacher assigned ratings on the 19 measures and most of these scores were substantiated by teacher comments. This indicated that the teacher could make distinctions among the 26 students on these 19 student attribute measures. The length of each case study varied and depended on how much information the teacher had provided about the student during interviews. This in itself indicated that the teacher actually knew more about some students than others. Not all students were chosen by the researcher for further examination of their use of

acad eigh hours student engaged ig time. Eventually the choice was narrowed to a core of idents who were observed for an average of about nine uctional time in language arts and mathematics. For other amount of time observed was less or non-existent, but data were included in the profile if available.

The formation contained in these student profiles was obtained over time and represented a considered reflection about each student on the part of the researcher who chose to omit and include teacherprovided information; piecing this information together with other sources of data about each child. The foremost aim was to represent the child as the teacher portrayed him to the researcher as faithfully as possible.

The actual collection of the data from the teacher extended over the three month period. Class rankings were completed on separate days and the teacher was asked to concentrate on each child carefully while he assigned scores on the 13 Attribute Scales which were completed on an individual basis during a two week period. Data such as I.Q. scores, age, and achievement information were collected at the end of the study. After analyses, the average mean scores for verbal interaction data were computed for each individual student and data on the use of academic learning time, where applicable, were averaged and totalled for each student. Thus, information about students was gathered continuously over the three month period using different methodologies. The portions included in the individualized case study profiles were at the discretion of the researcher.

The first section of each profile purposely alternated between the presentation of teacher-assigned numerical scores and teacher comments taken from the interview transcripts which served to corroborate the assigned ratings. This format was adopted to minimize the reliance on numerical tables and to keep the presentation of the somewhat repetitious, but basic, information for each student as readable as possible. Obviously some sections of the data were more efficiently and comprehensively presented in table form such as the individualized mean scores received on the verbal interaction contacts with the teacher.

In the judgment of the researcher a reasonable portrayal of each student resulted. Although brief, the accounts enable the interested reader to delve into the case studies in order to make fine comparisons between any two students, for example. No two are alike, nor were they viewed identically by the teacher who assigned the scores and made comments

310

about each student. With this information at hand it is possible to speculate about possible causes for differential verbal interaction received (or for the amount of time spent on task by the student).

The profiles might have been enhanced by more home background information about the children, indications of pupil attitude toward school, and/or by peer opinions/information. However, the case studies succeeded in presenting basic teacher-provided information in order to describe each student subject in this classroom under study.

CHAPTER VII TEACHER THOUGHT PROCESSES

This chapter presents the results of the micro-analysis of the stimulated recall data obtained from the teacher subject in this study. The transcripts of the stimulated recall interviews were analyzed using the content analysis system developed and utilized by Conners (1978). Twelve categories comprised the system and most of these were further subdivided to "ascertain the categories' substantive components and explore the richness of the data (Conners, 1978:117)."

3

Micro-Analysis of Interactive Thoughts

In the present study quantification of the stimulated recall data was to be used to answer Research Questions 4.0 and 4.1, namely:

- 4.0 What information is obtainable from the interviews held with the teacher?
 - 4.1 Which kinds of thoughts about students in particular were reported during the stimulated recall interviews?

⁴ Previous studies using stimulated recall methodology (Marland, 1977; Conners, 1978; and Tuckwell, 1980) have been interested in the full spectrum of teachers' interactive thoughts. In the present study it was re-emphasized at the beginning of each stimulated recall interview that the researcher was primarily interested in those thoughts., perceptions, and feelings about students which the teacher experienced during, instruction and the teacher was encouraged to focus on and verbalize those particular interactive thoughts. As a result of this emphasis, it is presumed that interactive thoughts concerning other aspects of teaching may have been withheld and that percentages of tategories reported in the findings from the present study may differ from the other studies which were interested a fuller range of teacher interactive thoughts.

During stimulated recall interviews, the usual aim of the researcher

312

is to maintain the teacher's focus on his interactive thoughts and to discourage non-interactive comments, using careful question probes which restrict attention to the lesson being' viewed or which subtly guide the interview back to the stimulus lesson. While this aim was followed in principle, often the teacher would divulge non-interactive thoughts about students or background information about them that impinged on his interactive thoughts. These expressed thoughts were not discouraged because they were valuable sources of information but they could not be coded as interactive thoughts. The probing questions used in this study were not kept as pointedly to interactive thoughts. Any additional background information was noted and was explored more fully in later general interview settings.

Not unexpectedly, some differences were found between the percentages of thoughts in categories in Conners' work and those in the present study. More student information was obtained from this teacher overall. For instance about 10% of reported thoughts in Conners' study were coded as Information-Pupil compared to the 16.3% in the present study. Conners' teachers averaged about 16% for thoughts coded as Perceptions which included sub-categories of perceived student Academic Performance, perceived Student Verbal Behavior, and perceived Student Movement compared to about 21.7% of expressed thoughts in the present study.

Tuckwell (1980) implied that a teacher might be hesitant to restate a recurring thought and he suspected that formerly expressed thoughts in his study were sometimes withheld although they continued to exert an influence. Also he felt his teachers treated the stimulated recall interviews as ongoing conversations and so he questioned whether the 313

percentage of interactive thoughts expressed included all of the influences operating in a given instructional setting.

At any rate, the system proved to be a useful and adequate mechanism for quantifying the thoughts that were expressed by the teacher in the present study. For both Conners' study and the present one, although the rank ordering differed, the five most frequently mentioned thought categories were: Perceptions, Instructional Moves, Information-Pupil, Self Awareness, and Interpretations.

The twelve categories in the Conners' Micro-Analysis system are now presented with a very brief description of the kinds of interactive thoughts they include as well as some illustrative examples taken from stimulated recall data in the present study. The relative frequency of use expressed in percentage form is given as the categories and subcategories are discussed. See Figure 3, Page 315.

Teacher Perceptions

Interactive thoughts coded in this category deal with sensory awareness of such sub-categories as Student Verbal Behavior, Student Noise /Unacceptable Behavior and Student Expression. The highest percentage of this teacher's interactive thoughts were classified as Perceptions. This trend may help to substantiate the high scores the teacher received on Items #19 (Awareness) and #20 (Withitness) on the high inference instrument used in Project Quest. Within the Perceptions category there were eight sub-divisions. The bulk of this teacher's Perception interactive thoughts (44.7%) centered on the Academic Performance of the students. These thoughts may have resulted because he did walk around the room monitoring seatwork,

Figure 3

Summary of Categories in Content Analysis System

1. <u>INSTRUCTIONAL MOVES</u> Feedback--Teacher; Structuring; Organizing; Control--Discipline; Presenting; Reviewing; Repetition; Reinforcing; Motivating; Feedback--Pupil; Involving; Transfer; Learning Processes; Miscellaneous.

2. PERCEPTIONS

Academic Performance; Student Verbal Behavior; Student Movement; Student Expression; Student Presence--Absence; Noise --Unacceptable Behavior; Equipment---Materials; Miscellaneous.

3. INTERPRETATIONS

Academic Performance; Student Attention--Motivation; Student Movement; Verbal Behavior--Noise; Student Cognitive Processes; Materials--Content; Feelings; Miscellaneous.

4. EXPECTATIONS

Task Performance; Cognitive Performance; Affective Performance.

5. MEDITATION--PUPIL

6. <u>SELF-AWARENESS</u>

Instructional Awareness; Instructional Evaluation; Instructional Reflection; Content Reflection; Personal Reflection; Affective Dissonance.

7. BELIEFS

General Pedagogical; Developmental; General Learning; Specific Learning Principles; Memory; General Psychological Principles.

8. OBJECTIVES

General; Lesson Specific; Lesson Facilitating.

9. LESSON CONTENT

10. INFORMATION--PUPIL

Personality; Academic Ability; Academic Performance; Classroom Behavior; Social Behavior; Home Background; Physical Characteristics; General Information.

11. INFORMATION--OTHER

12. FEELINGS

. from Conners' dissertation, 1978:280.

particularly during math instruction and five of the eight stimulated recall lessons were math instruction. He had received high scores on the high inference behavior (#4) "teacher moved around the room a lot monitoring seatwork" from Project Quest.

<u>Academic Performance</u>. Expressed thoughts belonging to this subcategory resulted from seeing a pupil's work in progress, hearing an answer or noting in some way the degree of success the student was experiencing. These thoughts classed as Academic Performance accounted for about 44.7% of all interactive thoughts belonging in the Perception category.

While I stood there, she did seven or eight right in a row... She'd stopped, as soon as I'd left, she'd stopped... People are responding fairly accurately and effectively... The procedure is exactly the same and she did figure it out eventually- although I don't think she put the right answer but she had the right routine anyway.

Student Verbal Behavior. About 16% of Perceptions expressed had to do with student-initiated verbal behaviors (not pupil responses to teacher-initiated questions which were coded under Academic Performance) These thoughts might be precipitated by a pupil talking, commenting, or asking the teacher a question.

Merrill was sitting there muttering because he's not allowed to answer at this stage of the game. He's the one who asked 'Can we get going? Yeah, let's get going!' Somebody had given me a response 'one-twenty' or something like that. But there were some of them who said 'I can't do this'.

Student Movement. These interactive thoughts involved the expressed awareness of student gross movements such as borrowing an eraser or tidying books. They accounted for about 12% of the Perception category thoughts expressed.

As soon as I said 'Sit down', she stood up and walked to the pencil sharpener...

..but he still had his hand up and he was still wiggling and jiggling...

She was tapping my shoe. I don't know whether it was purposeful or just doodling. It stopped anyway.

Yeah he's got to check out the machinery.. I don't blame him...

He was leaning on the desk.

Noise/Unacceptable Behavior. In this sub-category which accounted for about 13% of the Perception thoughts expressed, the teacher reported being aware of the misbehavior of pupils or of noise that was indicative

of off-task behavior.

Geoff was filling his cheeks with air and making strange noises... That was John whistling...

I found in that lesson that every time we had to change- from the quick quiz to other things- there was a lot of disruption...

... because I was watching Pamela and she was way off task. There were others too but she was the worst. She had put her stuff away.

Student Expression. This sub-category of Perceptions includes the subtleties in student expression such as a worried appearance or a smile that the teacher noticed. Thoughts.coded as Student Expression accounted for only about 2.8% of total Perception category thoughts.

T: ...he doesn't appear to know and then he starts to get flustered and doesn't know either way very shortly.

I: What was he doing at the time? How do you know he was getting 'flustered'?

T: Well flushed a little bit, that sort of thing.

He gets a little frustrated and he starts to huff and puff.

Equipment and Materials. Interactive thoughts about equipment and materials accounted for about 4% of total interactive thoughts in the Perceptions category and were expressed when the teacher noticed some item, equipment, or instructional materials.

His sheet was a little messy ...

It's a little tight in there with all the equipment (video) so people get behind me...

I'm aware of being stuck up at the front with that machine (overhead) too...you lose people at the periphery...

Malfunction. I think that's the cord. The kids are sitting on it.

<u>Student Presence or Absence</u>. Five students in the room had regular resource room appointments three times a week. The four expressed interactive thoughts in this category, which accounted for about 2% of all expressed Perceptions, concerned the comings and goings of these students.

Now Ian and Tom have come in from resource room ...

Yes, she's just arrived back with what- ten minutes remaining ...

<u>Miscellaneous</u>. The interactive thoughts coded as Miscellaneous accounted for over 4% of all Perceptions.

Brent's not feeling too well... quite pale...Mom wasn't home so he couldn't go home...

It was warm...

Oh here's the P.A. announcement. I can't hear it anyway. It must have been something about lunch.

That's when I noticed we're at the end... I noticed that it's after the time...

Instructional Moves

This large category contained about 17% of all the interactive thoughts expressed by this teacher. The category was sub-divided into 14 subcategories. Instructional moves are reported teacher-initiated actions or deliberated actions which could be physical, verbal, or non-verbal and which embodied skills or principles of teaching and learning that the teacher consciously was processing while teaching.

<u>Control/Discipline</u>. The greatest number of interactive thoughts expressed about Instructional Moves fell into the Control/Discipline sub-category and amounted to 19.2% of all Instructional Moves thoughts. They included actions taken and actions considered which would deal with various forms of off-task and deviant student behavior.

I had to resort to this hand-raising... I haven't had to so much lately but today I felt that...

I tried to ignore it but it didn't work.

Here I'm looking up at the front again. That's eye contact I'm trying to make.

When I went up to look at what Pamela was doing ...

Indeed I initially asked him because I thought he was off-task and I wanted to bring him back.

<u>Feedback Teacher</u>. About 12% of the expressed interactive Instructional Moves thoughts were sub-categorized as Feedback-Teacher moves. In these instances the teacher reported that he was actively seeking information about a student's performance, progress, affective state, or general classroom behavior. Information so gained was influential in determining future teacher action.

... just sort of spot checking

۱. ۲

You know I'll know better the next time we talk about it, which will be tomorrow.

Well you can briefly check. If someone can come up and do it then they have the idea.

I went up front. I wanted to see what she had done 'cause I hadn't seen her do any work when I looked up.

Well that's why I asked them to keep those sheets. I'll have a quick look at them. I can tell pretty quick.

<u>Structuring</u>. About 13% of the interactive thoughts about Instructional Moves centered on structuring. The teacher presents' material in sequence so it is clearer. Students are directed to attend and focus on the steps or directions presented. Pupils' thoughts may be clarified and extended through use of structuring actions taken by the teacher.

But it turned out I still had to give him a number of hints and I think he did get it eventually with a lot of help. I decided to give him time, thinking he might be thinking about it. I backtracked when I started finding the same things that I was finding yesterday and tried to look at it in a different fashion. I'm trying to focus their attention on what they might... but I'm searching for someone to say 'Hey, there are some reasons why this guy has done what he's done'...

<u>Involving</u>. Involving describes an instructional move for effecting student participation in an activity. The teacher attempted to involve his students to maintain their attention and interest and to capitalize on student contributions to enhance the lesson. He felt they gained more from the lesson by being actively involved in it. Thoughts about involving accounted for about 10% of all Instructional Moves.

... and I have to help Pamela with that stuff. She wasn't there for the directions.

And I'm keeping my eyes open at the same time to see if somebody's far off-task or having difficulty and I can charge over there first. I want to make sure that I have, that's why they read short passages, I wanted to make sure that I gave everybody a chance to read. Also I suppose part of it is I ask someone from the back to come up and that focuses the attention on the act of moving up so it's not centered so much just around the group near the machine.

Organizing. Interactive thoughts in this sub-category accounted for about 13% of all interactive thoughts about Instructional Moves. Specifically these thoughts dealt with the allocation of time and

materials, the timing and sequence of the lesson, and with moves to keep

transitions smooth and the pace and flow of the lesson under control.

... but at that point I'm not going to stop and tell her how to do it... It's too late...

It was a decision to get things in their hands and let's get going cause I knew that it was going to take a minute or two to get that out.

The reason I was giving some of the others was a constraint of time and indeed on the next section, which is considerably down the road, I did all of them.

Here I want to end it off cause I look at the time and find that I only have four minutes in which to read. This kind of a story I need half an hour.

Feedback- Pupil. About 7% of all interactive thoughts on Instructional

Moves concerned the provision of feedback to the student about work

performance. This usually involved making evaluative statements to

students.

Yeah, well, as I told him though, I wasn't satisfied with the way he had worked.

I'm telling him not to worry if the estimate isn't right on.

I started with Mason thinking 'Well, we'll get this kid off to

a good start right off the bat...

Reinforcing. This sub-category dealt with thoughts about rewarding

students in some positive way for their effort or achievement. About

5% of Instructional Moves were coded as specific Reinforcing moves.

I'm still with Mason...I wanted to...if he was right, if he had a good point, I wanted to be able to say 'Hey, yeah, that's right'.

I ended up with Joanne, not because I wanted to hear her read again, but because it occurred to me that when she had read I hadn't said anything about her reading and that I ought to say 'Well, Okay, very good' or at least that, you know.

50

...and so I'm trying to take the opportunity to praise them as we go along.

<u>Reviewing</u>. Interactive thoughts concerning reviewing comprised about 5% of the Instructional Moves category. Reviewing thoughts centered on the act of integrating and interrelating knowledge the student had

learned previously.

It was just a question of refreshing it for some of them and for others as well...

The sheets that I gave them today were a review of the story which they've read twice.

Usually I start just by getting the details, the factual details of where we are in the story, so it will be a little review, that sort of thing.

<u>Presenting</u>. Presenting refers to those Instructional Moves which impart knowledge and content by using equipment, written materials, or oral presentations which may involve explanations and demonstrations. These thoughts Accounted for about 4% of all interactive thoughts categorized as Instructional Moves.

Okay, this is the last sheet that I handed out to them. I simply illustrate by doing these now pretty much.

So I'm writing down a few of them now...

Now here's where it's forming in my mind that I can approach this step by step, partly by taking one example right through the hour.

<u>Repetition</u>. Repetition was one specific category of Instructional Moves that focused on the Reviewing aspect but included only repetition and drill activities in which some content or skill was emphasized. Thoughts in this sub-category accounted for only about 2% of all

That's why I wanted to give him the example...so he could imitate. We work with them (spelling words) a few minutes through the week. Motivating. Motivating is an Instructional Move primarily used to

instill internet and enthusiasm both intrinsically and extrinsically.

Objects, se ce questions or the teacher's own verbal behavior might be us te. The expressed interactive thoughts in this sub-c gory in number and accounted for some 2% of all

in the Instructional Moves category.

expre

W

th

to Ιs

quit

tho

just a bringing to bear their own experiences, given it situation, wondering if any of them had ever- would like tor be experiences they've had with wild animals.

here quite a while at Bonnie(because I know she knows ot more than she ever does and even though it's written work she has difficulty with that, while I stood there she n or eight.) did s

of Learning. Transfer of Learning was a sub-category Transf of Instructional Moves that is used to help students generalize or transfer knowledge from one learning situation to another. Only one such (rather 1 g) example was included in the teacher's expressed and it had to do with how deceived the students interactive th were by a small handful of coins which really represented a large amount of money. By having the students handle and work with the actual coins the teacher felt they would begin to recognize the difference.

Learning Process. Instructional Moves in this sub-category were often closely related with other sub-categories within Instructional Moves such as Motivating, Reinforcing, and Repetition, as well as to Beliefs. Conners (p.131) instructed, "These instructional moves focus on the learning process and included thoughts about learning being meaningful, pupils learning through association and discrimination,

through problem solving and through using a variety of presentation

modes."

Once you're slowed down with a word or two, the rest of it is more difficult.

So you know, there isn't the change in stimulus that you need to make good use of the time, I suppose.

I think there's some value in spending a little bit of time in working with the actual coins and in becoming accustomed to them.

It is entirely possible that Learning Process moves have been coded within other sub-categories of Instructional Moves or under Beliefs. The focus of the study was not on his perceptions of <u>how</u> children learn and he may have withheld interactive thoughts about learning processes knowing they were not of major interest to the researcher. Perhaps the foregoing helps to explain why only .6% of thoughts under Instructional Moves seemed to belong in the Learning Processes sub-category.

<u>Miscellaneous</u>. Instructional Moves coded in this sub-category dealt with items related to teacher's maintenance of a student's self-concept, or preservation of a pleasant class atmosphere. Thoughts in this subcategory also dealt with providing individual assistance to students or involving them in certain types of decision-making which would directly affect them.

I tried to shut her down in a nice way ..

If I frustrate <u>that</u> then the speaking out behavior is more frequent. It's been a pretty rugged year, I try not to put too much pressure on him.

Interpretations

Expressed interactive thoughts coded in this category were properly those which contained teacher-assigned meanings or explanations of

perceptions. (It was sometimes difficult to distinguish at which point a fact-like Perception became an Interpretation; eg. "He was paying attention.") The category was sub-divided by Conners into eight divisions. Most of the interactive Interpretation thoughts had to do with the attention and motivation of the students, their perceived cognitive processing, and how they were feeling. Short examples, taken out of context, are presented as representative of the sub-categories.

Student Attention and Motivation. The substance of thoughts which fell within this sub-category was of major interest to the researcher in the present study. About 26% of all Interpretations were subsumed under this sub-category. The teacher's assessment of whether students were on task, were paying attention, or were interested in the ongoing instruction were classed as Student Attention and Motivation.

I was satisfied that most of them were on task for most of that time period. Bonnie would dearly love to read the whole thing ... Most of the people in the class are really enjoying the story .. Tom...Nicholas...those people are enjoying it immensely. When I was talking to him about spelling, that was earlier on, he was being a bit lazy.

Student Cognitive Processing. In this sub-category of interactive thoughts called Interpretations, the teacher inferred what mental processes were occurring as students were engaged in learning activities. He expressed his interpretations of their cognitive behavior and thoughts. These assessments accounted for the greatest number of thoughts expressed under Interpretations (about 43%).

I think she knows a heck of a lot more than she ever does ...

Mason is looking for a straight fact answer, there isn't one... ...but even then I was satisfied that she knew what to do and would normally get it correct.

I think maybe she <u>did</u> get a hint because then she elaborated and talked about the orders he was always giving.

Feelings. The teacher interpreted how students were feeling by attending to facial expressions and gestures. These interactive Interpretations of pupil emotions as expressed by this teacher accounted for about 21% of all Interpretations. He had been assigned high scores on the high inference behavior #26 (Teacher responded accurately to both obvious and less obvious meanings, feelings, and experiences of the children.") from Project Quest.

He's either quite tired or quite bored. She looked quite remorseful at that point... It makes her perhaps a little apprehensive about her reading and and so she's a little nervous about it too.

<u>Academic Performance</u>. Interactive thoughts dealing with Interpretations of Academic Performance accounted for about 5% of all Interpretation • thoughts. The student's written or verbal performance is assessed by the teacher. (Many of these thoughts seemed to the present researcher to fit equally well under Student Attention and Motivation or under Cognitive Processes).

Part of the problem for Ian and Tom, who didn't really know how to do this very well was...

Jim's having trouble.

<u>Student Movement.</u> The teacher interpreted a student's gross movements when the intent of the student was not apparent. Only one such thought was coded in this sub-category. Yeah it was unintentional, but he was bumping the machine.

<u>Verbal Behavior and Noise</u>. This sub-category included the teacher's interpretations of either classroom noise or a student's verbal behavior which did not seem task-oriented. (No examples were coded in this category. The present researcher suspected that if any potential examples existed they were coded just as appropriately under interpretation of Student Attention and Motivation or perhaps under Perceptions of Noise/Unacceptable Behavior.)

<u>Materials/Content</u>. This sub-category was meant to include the teacher's interpretation of the suitability of instructional materials for students. (No examples from the present study were coded here but is was suspected that potential examples may have been coded in either the Instructional Evaluation or Instructional Reflection sub-categories of the Self-Awareness general category or in the Expectations general category.)

Miscellaneous. The miscellaneous sub-category included teacher interpretations concerning certain unusual classroom behavior, home background, or organizational matters. Only four thoughts were coded in this sub-category which totalled 3.3% of all Interpretations.

I thought she was going to swallow the damn thing... And I thought it might have been because he couldn't see...

Information-Pupil

Conners stated that "this thought unit is concerned with information a teacher carries around in her head that centers on pupil characteristics

327

such as personalizy, academic potential and performance, general classroom behavior, social behavior, and home background information (p.164)." About 16% of all expressed interactive thoughts in the present study concerned information about pupils. (It was sometimes difficult to decide when the background information about pupils that influenced the teacher became non-interactive in nature and therefore classed as "Case Study" type of information). The teacher referred to many thoughts about pupil characteristics, behavior, ability and home background which immediately contributed to the interactive thought expressed and were therefore felt to be impinging on teacher thoughts and shaping teacher treatment of students during instruction. The general category of Information Pupil was further sub-divided into nine categories. Examples from transcripts in the present study have been presented out of context to illustrate thoughts coded within the subcategories.

<u>Academic Performance</u>. The academic performance sub-categories appear under the three general categories of Perceptions, Interpretations and Information-Pupil. Under Information-Pupil, this sub-category accounted for 29.4% of all pupil information thoughts and were considered to include those thoughts concerning the teacher's assessment of usual work habits and performance of either individual students or of the class as a whole.

He forgets his rules...

In a task like this (test) almost all the kids get down to it and do it the best they can. They work whatever speed they are able. 328

Pamela, who's easy to get off task...

I haven't known her (Sonia, Case 26) long enough to know whether she would normally, in a situation like that, ask for help and whether normally she pulls out a book and does her own thing when things are difficult...I don't know yet.

... or with Trisha, who generally gets her work done ...

<u>Classroom Behavior</u>. This sub-category of Information-Pupil thoughts was used about 28% of the time by the teacher when expressing thoughts about pupils. Conners described this sub-category (p.170) as "containing both individual and group references and dealing with activities and verbalizations of the pupil or group that the teacher considers typical behavior." Conners reported that his nine teachers referred to this sub-category most frequently when considering Information-Pupil thoughts. In the present study this was the second most frequently used sub-category; Academic Performance being the more frequently used.

She's a chatty little thing. If I'm not keeping my eye on her she really will chat a lot.

She'll sometimes play that off against me saying, Well should I only do this much?'

Interesting to watch the ones who are perpetually late...Pamela, Sharon..are often late coming into a group, a couple of others... Geoff sometimes makes noises just for the hell of it...

Personality. About 21.8% of this teacher's expressed interactive thoughts coded as Information-Pupil referred to personal characteristics of the students.

He's not the kind of kid who's really turned on by school work... He's sort of laid-back, easy-going, it'll all come out in the wash sort of thing...

Grant has a good sense of humor ...

They aren't as imaginative generally ...

You've probably picked up that sometimes he's fairly silent... She doesn't strike me as an individual who goes out of her way to be testing, you know, she's not one of those who will push, push, push to see how far...(Sonia)

He's an enthusiastic kid.

<u>Academic Ability</u>. The thoughts in this sub-category deal with the teacher's assessment of the general academic potential of students. In this study thoughts coded in this sub-category accounted for about 10% of all Information-Pupil thoughts.

You know he's very bright.

She seems to be a little stronger in language arts than she is in math, just an initial impression.. (Sonia)

He's got lots on the ball...

Well maybe it's not too surprising that Tom would ask a question like that...

Social Behavior. This sub-category relates to peer relationships.

Many non-interactive comments described inter-student relationships but only one such interactive comment was coded in this category.

Maureen and Joanne, seated beside each other, they're good friends: <u>Home Background</u>. This sub-category dealt with thoughts about home background of students processed interactively while instructing. Only about 4% of Information-Pupil thoughts were coded in this section.

She might be somewhat outside the norm in this class...

Brent, whose background of information is just...the travelling he's done this year is really something...gives him a real sense of size of place, geography, and even politics.

There's no problem with money ...

Physical Characteristics. Thoughts in the sub-category centered on physical coordination or physical appearance. Infrequent mention was made of such characteristics in the teacher's interactive thoughts which accounted for only 2.5% of all Information-Pupil thoughts.

He has a funny kind of voice in a way and that threw me initially at the first of the year but his reading is fine...

...with Bonnie sometimes there's difficulty. Her handwriting capability is such that sometimes you can't tell where something is pointing...I should show you her handwriting scribbler. It's something else. She's coming along. She's much better than she was. He's got a rugged voice, yeah...

<u>General Information</u>. The interactive thoughts contained in this sub-category included background information about pupils which was other than information about the home situation. Few examples were coded in this category and they amounted to 2.5% of the total Information-Pupil

thoughts.

He hasn't had one all year ...

It's interesting, you know, she says she has taken this particular lesson at her other school...

Of course his time was shortened because of his absence in the resource room.

Self Awareness

About 12% of all interactive thoughts from the stimulated recall data interviews were coded as Self Awareness thoughts. Conners (p.146) described thoughts in this category as those which indicated that teachers were "conscious of various facets of their performance in the classroom" and were aware of their behavior as they interacted with the pupils. This awareness involves, evaluation and emotions about one's performance, deliberations on instructional strategies, and reflections on lesson content as well as on one's personality, appearance, and general teaching style. Conners divided this category into six sub-categories which are briefly described and illustrated with exerpts from the present study's stimulated recall interviews.

Instructional Evaluation. The teacher expressed both positive and negative evaluations about his teaching. Such self-assessment thoughts could be about whole class instruction or the handling of individual teaching situations. Totally such judgmental thoughts accounted for the bulk (64.1%) of all interactive thoughts classified as Self Awareness

thoughts.

This is a situation where you hand out a highly motivational kind of thing and then try to settle people. Maybe I tried to do that too quickly.

It was not one of those situations where it required intense concentration so they would tire out. There was enough activity.

That remark I just made to Brenda was almost off the cuff, and it's not very effective, I know.

I thought I was fairly careful in giving the directions... but we've done that before, that sort of thing, and that's why I wasn't quite as plain as I ought to have been.

That's unfair to Tim. I didn't ask him to explain it.

The only difference that I think.. I should have presented it one way one day and the other way another day, instead of trying to cover the whole thing at once.

You see, I'm still pawning both approaches, which especially with someone who is confused, is not the best thing.

Instructional Reflection. Instructional reflections refer to those interactive moments of decision where a course of action is deliberated. Reflections in the category also involve interpretative assessments of whether students are attending and understanding. About 22.5% of the thoughts coded as Self Awareness were coded in this sub-category.

I started this lesson wondering where am I going to find us today? Where will they be at? I don't know whether that's listening skills...Maybe I was giving too many instructions at once.

... cause I have to make a judgment... so because one or two turn off what do I do about it?

Well I guess I kind of thought that Okay, that maybe I shouldn't have jumped on him right away. Maybe I should wait a minute or two and let him refresh his memory.

...and I'm starting to think...this is when I'm starting to pull my hair and wonder if I've got to reteach all of this...

Instructional Awareness. Thoughts in this sub-category express an awareness of the effects of strategies used and the impact that such things as the organization and sequence of presentation might have on students. Only about 5.8% of all Self Awareness thoughts fell within

this sub-category.

This is a terrific opportunity to ask the sequence of questions that we're supposed to ask, you know, the higher order questions. You can sprinkle them. You know I can ask a lot of just straight lower level questions of facts and so on and build up to the other.

I wondered if I had the hour hand a little bit off place and that was confusing him...That wasn't the case.

When you're trying to function with one of these machines- with the overhead- and operate a class as well...using the Quest items, I'm probably not as 'with-it' as I ought to be.

Personal Reflection. About 5.8% of all Self Awareness thoughts were

judged to be Personal Reflections. They included expressed impressions of how the teacher might be perceived by students. Matters of concern

could be general style of teaching exhibited, or personality

characteristics displayed.

And I thought that I was a little abrupt. I was not feeling terribly well right there and I became a little abrupt with the kids.

Mind you I must say that I was a little crabby today.

Geez...do I sound hostile this morning? (I felt hostile all morning)...

<u>Content Reflection</u>. In this sub-category thoughts indicating an awareness of factual mistakes or lack of specific knowledge were to be coded as Content Reflections. No instances of such expressed thoughts were noted in the stimulated recall data from the present study.

Affective Dissonance. Affective Dissonance is exemplified when a teacher expresses an emotional reaction to teaching behavior exhibited. Examples of such dissatisfaction with personal performance only appeared twice in the transcripts of the present study and account for only 1.6% of thoughts in the Self Awareness category.

I wanted to get a little further than I did this morning...

I guess I would say I was pretty unhappy with this lesson.

Feelings

About eight percent of all expressed interactive thoughts belonged to the Feelings category. Conners described such thoughts as "emotions experienced by the teacher during the lesson " (p.175). No sub-categories were created but the emotions in the present study expressed such feelings as frustration, annoyance, enjoyment, concern, amazement, and satisfaction.

This annoyed me in here. Nicholas was very persistent here. I began to get a little ticked off.

I love this. This is the part of teaching that I like the best. This fantasy is my thing. I think it's very imaginative, colorful, interesting. It's stuff that I enjoy myself so I'm kind of on a trip myself and I love listening to the answers I get. I love the insightfulness of some of the kids.

It's amazing. I was amazed at how many people tried to withhold ...

I'm a little concerned about Ian in the last couple of days. He's really been acting out.

I felt like I was scampering from one thing to the next. This is the kind of time that I actually enjoy; being able to get around without a great hubbub around you, on a one on one.

You don't want to hear what I was thinking on that one!

Beliefs

Beliefs referred to the accepted values, teaching principles and practices which influenced the ways in which the teacher behaved. These beliefs might be about learning processes, both general and specific, or about the appropriateness of using certain teaching techniques with children. The beliefs expressed accounted for about 4.3% of all interactive thoughts and although the teacher was asked to focus on his interactive thoughts about student performance and characteristics, it is interesting to note that some beliefs were expressed as well. Conners sub-divided the Beliefs categories into five sub-categories which are now presented and illustrated with examples from the stimulated recall transcripts.

<u>General Pedagogical Beliefs</u>. General beliefs about students and about the role of the teacher formed the substance of General Pedagogical Beliefs. Overall about half (57%) of the Beliefs were coded within this sub-category.

Proximity is important, and also the ability to move.

If people are working on task- even if they're chatting with someone else, I think that you can learn a lot.

When you teach a large group like this, I like to get to everybody.

I often ignore hands up. I just don't even- the problem with taking hands up, volunteers, is that the same people volunteer all the time.

I guess quite often, especially in math, I question people who might be having difficulty because it's a fast way to find out if they have the concept. £0

When you're working around a machine like that, the people who are right there are going to be focused in usually, you know but at the periphery sometimes you run into... so it's just a matter of movement, get people focusing in on the business of moving up there.

Developmental Beliefs. The developmental stage of pupils

influences the way they learn. Beliefs in this category considered which

behavior was most appropriate for certain age groups. About 19% of

all Beliefs were coded in this sub-category of Beliefs.

You know they want to contribute and participate. Kids have the idea that teachers know everything.

When you hand out something like that (money) it's unrealistic to expect children this age to keep their paws off it.

... because most kids at this age level are reasonably successful and even if they aren't you can keep them going...

...and children are perceptive; more perceptive than often we give them credit for. They know if they can do these things or not and if I'm telling them 'Great', and they aren't, then they know.

General Beliefs Concerning Learning Principles. These thoughts

expressed the beliefs that active involvement, problem solving, or repetition, to name three, would be meaningful to children. Only five thoughts of this nature were coded from the transcripts which accounted for about 11% of all Beliefs coded.

... because there is some truth, too, at least I think there is,

if you require people to write something, even if they know, that the information will stay with them longer.

If you can get little groupings like that, there's nothing wrong with that at all.

You know there isn't the change in stimulus that you need to make good use of the time I suppose.

<u>Beliefs Concerning Specific Learning Principles</u>. Beliefs in this category dealt with principles to be used in more specific learning situations. The motivation, reinforcement, and maintenance of selfconcept for students were coded in this category as well. Only four thoughts were designated as belonging to the Specific Learning

 $\mathcal{S}^{\mathcal{S}}$

337

Principles sub-category and accounted for only about 9 % of all

Beliefs thoughts.

q

You know when you have the chance to work with things that you can sort of manipulate with your hands, then it's not a difficultarea's not a difficult concept really, I don't think.

Well the other thing is that this is a good opportunity to encourage people to read... They've demonstrated publicly that they can read you see and if you encourage in that sense it may encourage them to read privately too.

Well money makes a hell of a lot more noise than most teachers want but the thing is I think it's good for them to have it in their hands, a concrete object, and to work with it in that way.

Beliefs Concerning Psychological Principles. Beliefs in this

category concerned the social and personality development of children and their individual differences. Only two such beliefs were identified in the stimulated recall transcripts which accounted for about 5% of

total Beliefs expressed.

I think that all the other kids are aware that there was some difficulty and to brush it under the carpet and then deal with it elsewhere I didn't feel, maybe I'm wrong, was terribly important.

Generally I feel that children should be discreet enough not to be doing those kinds of things. It's more of a general rule.

<u>Beliefs Concerning Memory</u>. Beliefs Concerning Memory included beliefs that cues or hints could aid the student to remember factual knowledge. This category seemed to fit more appropriately under one of the broader belief categories. At any rate, no examples from the transcripts were coded in this sub-category.

Meditation Pupil

ssed in this category were those dealing with the

out the mental processes of a group or of a

specific student. <u>Why</u> the pupil experiences cognitive difficulty or is inattentive and <u>how</u> the pupil understands or remembers are the types of thoughts to be included in this category. Conners did not subdivide this Meditation Pupil category. Nine thoughts from the stimulated recall transcripts from the present study were coded in this category. This Meditation Pupil thought category accounted for only about .9% of all interactive thoughts expressed.

I wondered if she read the story once before and thought she had read it and therefore didn't reread it when I assigned it or something.

I don't think even yet that he has confidence in his ability to tell time. Like on those exercises that I was handing out, I don't think he really had the confidence that he was going to answer the questions correctly.

The assumption is then, in her own mind, that she knew the information quite well.

Lesson Content

Thoughts in this category(which was not subdivided) were those concerning the teacher's concentration on subject matter or the evaluation of specific content or those referring back to content from previous lessons. Only 11 such references were made in the stimulated recall transcripts from the present study accounting for only 1.1%

of all interactive thoughts expressed.

The first one is an alphabetical order thing...

And they all missed the 6:30 one; only that they had the hour hand pointing to the 6 directly.

... just a review of multiplication, addition, subtraction and a couple of little problems just to keep their hand in...

Objectives

Thoughts in this category dealt with long range and short range pupil outcomes to be achieved. Only seven Objectives thoughts were coded in all and amounted to .7% of all interactive thoughts.

The category was subdivided by Conners into three sections.

<u>General Objectives</u>. Thes *e* were broad objectives that the teacher hoped would be achieved in most lessons. Four thoughts were coded as General Objectives.

... (responding well). That's more or less what I was after in that little exercise.

Lesson Specific Objectives. In the lesson the teacher hoped that pupils would achieve certain objectives that had been decided upon preactively. Three thoughts were coded in this sub-category of Objectives.

Part of the idea of the lesson was to demonstrate that we can use a regular tool to measure area and that squares...well basically what I wanted them to find was that squares are a good way to measure area, or square-type shapes...

Expectations

Thoughts in this category referred to pupil behavior that the teacher expected or anticipated would occur during the lesson. In this study about 3.4% of all interactive thoughts were coded as Expectations. Conners subdivided the category into three kinds of expectations.

Task Performance Expectations. Thoughts coded as Task Performance Expectations referred to the physical performance of an instructional task by the pupil. Twenty-six percent of the Expectations were coded in this sub-category.

You know...the business of coloring over these coins...I think I took it for granted that that would be an easy task and all of a sudden its <u>not</u> that easy to hold it still and those sorts of things.

Actually I had four and I rather thought that we would get through them but it turned out that there were a few who needed a little more help I guess than I had anticipated.

... because I very briefly give her instructions and she'll be alright.
Cognitive Performance Expectations. This sub-category of Expectations is concerned with whether or not pupil performance on a task was correct. The bulk (66%) of the Expectation thoughts in this study concerned the Cognitive Performance Expectations the teacher held for his students.

I thought to myself that she would normally know the answer... The initial ones were ones that I suspected they would all get without difficulty.

... (asked Brenda)..someone who probably knew the answer.

... cause I know, given another 15 minutes at some point, I think almost everyone then will have these ideas down and will be able to tell time effectively.

Affective Expectations. This sub-category included the teacherexpressed expectations for the affective behavior of pupils or the expected <u>misbehavior</u> of pupils. Only two thoughts were coded and they accounted for about 6% of all Expectation thoughts. Both of these thoughts were about Sonia, the new girl (Case 26), as he was forming impressions about her.

She and I will get along very well...

I'm not concerned that I'm going to have any real problem with her.

Information-Other

Conners described the thoughts in this category as two basic types of thoughts which dealt with classroom rules about the organizational aspects of classroom life, or with thoughts irrelevant to the lesson. The category was not subdivided. Twenty-two such thoughts were expressed in the present study and they accounted for 2.3% of all interactive thoughts expressed.

It's a nice day.

We work to bells too, and it's not a satisfactory way to close off a lesson, but there it was.

The old hand signal... They know that if I'm working with someone or if I'm working with a group that I don't like them wandering over to see me...so usually a hand signal will have them return to...

("Could you say that one more time?") That shows you how plugged up my ears were! Can't hear a damn thing today.

I need a more comfortable chair. Normally I have a great big easy chair to read in which is very comfortable for me.

I find it interesting...The kids are so accustomed to you now that they don't even notice (the camera).

One of the most irrelevant thoughts and certainly the most amusing was the following interactive thought coded in the Information Other

category:

I was standing there right then thinking I'm going to stick my tongue out at that camera ... and then I thought- you can see that some of my thoughts are frivolous- and then I thought if I do that I'm going to have a class full of people sticking their tongues out at you!

Summary of the Micro Analysis of Interactive Thoughts

By subjecting the transcripts of the stimulated recall interviews to a content analysis, it was possible to identify the type and quantity of the teacher's interactive thoughts. Conners' (1978) Micro Analysis system was used as outlined by Conners. It was a comprehensive system which more than adequately accommodated the stimulated recall data from the present study. The resultant findings in terms of type and quantity of teacher-expressed interactive thoughts were presented in the first part of this chapter along with illustrative examples taken from the text of the transcripts.

Nine categories and their further subdivision resulted in a total of 60 possible locations in which to code each interactive thought unit. The teacher was asked to focus on the range of thoughts that were of major interest to the present researcher; namely those thoughts and perceptions and assessments about students (and their characteristics, personality, ability, and classroom behavior to name just a few possible examples) which the teacher consciously processed during instruction.

The most frequently used category (22%) was labelled Perceptions within which the teacher focussed most heavily on his perceptions of the students' Academic Performance during instruction.

Instructional Moves occupied about 17% of his interactive thoughts with particular emphasis here on control/discipline moves, organizing moves, and moves to involve students in the lesson in progress.

Almost as frequently processed were thoughts about Information-Pupil (16%) which were considered to be more background information kinds of thoughts but which did impinge on his interactive thoughts during instruction. Knowledge of their usual academic performance, their usual classroom behavior, and their personality attributes occupied most of his attention in this category.

About 12% of the teacher-expressed interactive thoughts dealt with Self Awareness. This self-monitoring tendency was most frequently revealed in his thoughts about evaluation of the instruction he was providing.

Interpretations at 11.7% were nearly as frequently processed as Self Awareness thoughts. Within this category, most thoughts dealt " with his interpretation of his students' cognitive processing and how they were understanding the instruction in progress with a lesser, but substantial amount concerned with his interpretation of their states of attention and motivation.

His expression of his own personal Feelings experienced during instruction accounted for about 8% of total expressed interactive thoughts. The other categories of Beliefs (4%), Expectations (3.5%), Information Other (2.3%), Lesson Content (1%), Meditation Pupil (.9%), and Objectives (.7%) were used less frequently as the percentages indicate.

Of all 60 <u>sub</u>categories the three most frequently used were Perceptions of academic performance (94 thought units), Self Awareness of Instructional Evaluation (77 thought units), and his Interpretations of Student Cognitive Processing (49 thoughts) during ongoing instruction.

The lower percentages expressed in certain of the nine categories may be due to the fact that these accommodated <u>interactive</u> thoughts only and the categories like Lesson Content and Objectives probably would have received more <u>preactive</u> attention from the teacher. Also the fact that he was trying to adhere to the researcher's request that he restrict his reporting to the narrowed focus of interactive thoughts about students during stimulated recall interviews may help to explain this trend.

As a final comment on the use of Conners' Micro Analysis system for coding the interactive data from this study a couple of observations are made. In three of the <u>subcategories</u> no thoughts were coded from the present study as there seemed to be no suitable examples. Also due to the fairly large number of thoughts classified as Feelings (80 thought units) in the present study, this category could have been subdivided in some way to deal more precisely with this information. It is speculated that Conners' nine teachers did not provide as great a proportion of Feelings for the following reasons: Perhaps a personality difference in teachers prevents some from experiencing and articulating such thoughts processed during instruction. A more plausible explanation would be that Conners' teachers were involved in two stimulated recall interviews compared to the eight stimulated recall interviews in which this teacher participated. In addition, many general interviews had helped to establish the quality of rapport which would have to exist before more personal feelings would likely be expressed. It must not be forgotten too that the teacher and the researcher worked together over a three month period.

Macro-Analysis of Interview Data

3

The result of using Conners' macro-analysis system was less than satisfactory for plainly it was not suited to the data from the present study although the system had accommodated the data obtained from the nine teachers in his own study. A perusal of the categories used in Tuckwell's (1980) macro analysis system suggested that they more closely fit the data from the present study.

The teacher subject had not been asked directly to express his "beliefs" about teaching or to identify the general "principles" of teaching which he believed to be influential in his own work. Thus any beliefs and principles which appear to emerge from a macro analysis of the data run the risk of being rather incidental. However, those expressed are deemed to represent the attitudes and priorities exerting an influence at that particular time.

The transcripts of both the general interviews and the stimulated recall interviews were comb-searched using the "sieve code" procedures as outlined by Guetzkow (1950) in order to find expressed priorities and personal convictions about the instructional processes in a classroom, working with children, or about his perceived role as a teacher. The following examples represent these attitudes and priorities which appeared to underlie his routine teaching behaviors,

The teacher tended to depend on his own judgment about students. In speaking particularly about forming initial impressions during the first week of school the teacher commented:

"With respect to the children I try to have them talk a lot and I talk a lot to them and they interact."

What things are you noticing?

"Well lots and lots of things. I can notice how someone speaks and, of course, I may make quick and erroneous judgments about their ability based on what they can do verbally; if not "erroneous", incomplete... ..So certainly I make all sorts of judgments about children very early on in the school year despite the fact that I tend, just as a matter of practice to avoid the officially written records."

You don't go through the Cum. cards?

" No, and I don't ask for last year's report cards either...

... You can't meet somebody without making some kind of judgment about them unless I say to myself 'This is a token introduction and I'll never see this person again' but that's not the way it is in school. You're seeing these people and you're living with them for a year, pretty nearly a year. So I concentrate very hard on what they're doing and and then naturally I make judgments." Do you change them?

"Sure I change lots of judgments about individuals...through the year. I like to remain quite flexible in my judgments about people. Not all of them are changed."

How accurate do you think you were at the first of the year? "Oh pretty accurate .. I think so. When we talk about Keith, for example, my changed attitude toward him it's not a violent shift it's a...an adjustment in my feelings about the kid."

The teacher had a strong set of feelings about the giving of marks, report cards, and the ranking of children in general. He related that during a teacher-parent interview, Brenda's (Case #7) father wanted her <u>exact</u> ranking.

"He wasn't satisfied with the general kinds of comments that one makes...He really pressed me for an exact ranking so I said about 12th or 13th on certain things'."

That didn't go down well?

"Well he told me he thought my assessment of the child was about accurate but he wanted, as a means of persuading her to do better, he was going to tell her this and hope that she can move up. I'm sure that at the next interview he'll be wanting to know 'What's her exact ranking now?' I have visions of that."

Do you indicate to them where you think they are? compared to others? "I don't compare children with others as a general rule and that's what I tried to tell him but he wasn't going to accept that. He was almost on top of my desk. Quite an interesting interview." The comments about Tom's (Case #15) academic performance at the end of the year were not in relation to the performance of classmates but rather to Tom's prior performance.

"What a super job this kid's been doing! I don't know whether you noticed but his math score was 53 out of 60. His reading... I think he was in the third percentile in decoding and the eleventh in comprehension last year and this year he moved to about the 35th. That's a pretty marked improvement!"

The teacher tried to be realistic in the giving of grades on report cards and in his dealings with the parents and the students themselves when assessing the ability and progress of a student.

"Well last report card interview Mason's (Case 20) mother came in pretty hostile and I sort of had to calm her down. Last year on his final report card, when we had a point marking system, you know four is the top grade, he had all fours except for three of his marks that were threes. There's no way that he... I mean I haven't given <u>anybody</u> those kinds of grades!"

Do you save them til the end of the year?

No, I'm not saving them at all. I just don't see people as being that excellent I guess, I don't know. And she's distraught because her son was... well think of it <u>this</u> way. You've been around the class long enough to know that Mason's not the top pupil in the class. What grades do I give the top people if I'm handing those out to <u>him</u>?"

Is there anybody who is receiving fours?

"Sure, lots of them. I give out lots of them but I don't give them in <u>everything</u>, not in every subject... They become meaningless. Well

I talked her out of it. I was very careful and went through the areas that I thought he... I was trying to give her some idea of where her kid was at and she said 'Well Gee, you know, after last year I thought he was a top-notch student.' No, she was pretty accepting. I think when you finally ask people 'Do you want me to say your kid's top of the class if he <u>isn't</u> they sort of say 'No I'd rather have an honest assessment.'...Well it's interesting. I always find parents pretty accepting of the sorts of observations that you make but the problem I've run into this year with two or three people is that they had an entirely different picture last year. And I'm not being criticial of the people who taught them before."

Nicholas (Case #19) had a similar kind of report card. "There is <u>no</u> way...That's just not a realistic report card and the assessment is not as I see it."

... Well the way I look at it is I'm reporting to parents really and it's unfortunate in a sense our style of reporting is that. That's the element of competition - 'I have 14 fours and you haven't got any'-I think we should get away from. We write a lot of comments on them but there's still a sort of rating. Well next year what I'm going to do is I'm not going to hand out report cards to kids, I'm only going to hand them to parents."

What if they don't come in for an interview?

"Well I can't be quite that stringent because we are obliged to get reports home at one point or another but basically I'd like to see parents. I'd like to give the reports to the parents."

But the kids will still see them.

"Except there isn't the immediate hysteria that surrounds the giving

of the report card...and the dashing of hopes."

In speaking about questioning children the teacher commented, "When I get an incorrect answer it's amessage to me....The thing is I can get a correct answer to every question that I ask. I just have to ask the right people you know."

Are you conscious of their feelings?

"Well there's <u>that</u> aspect, of course. I want to have a good reason that a kid can recognize for praising the child and secondly, of course if you're just dealing with information...If I'm bothering to teach it then I want them to have it and children are perceptive, more perceptive than often we give them credit for. They know if they can do these things or not and if I'm telling them 'Great' and they aren't, then they know. So I like them to produce a little bit and I like to praise."

He also commented,

"You can focus in on a child and really embarrass him." He reflected about an interaction with Mason at the conclusion of which he was able to praise him for eventually giving the right answer.

"It would be a bit of a saw-off because I think that he was a little flustered, a little embarrassed and I think I focussed on him a little too long ... Well I hope that he's not upset cause that's detrimental to learning."

The teacher was asked how he thought Nicholas (Case #19) reacted to an instance of assessment concerning his oral reading followed by support and encouragement.

"No, I didn't consider whether the might... I guess, in a sense

unless something really needs to be private I think that... All the other kids are aware that there was some difficulty and to brush it under the carpet and then deal with it elsewhere I didn't feel...Maybe I'm wrong, was terribly important. But I certainly hadn't considered that it was anyway."

The teacher reflected about working with Ian and Tom, the two concern boys. "I think that both these kids, Ian and Tom, are really approachable young people who are...well I've taught in other schools where kids have learning problems but they also have severe personality problems. These kids have problems but they're fairly open about them."

You talk to them about these things?

"Sure. Well you know even kids who aren't smart aren't stupid. They have a very accurate perception of where they are in a class situation in terms of their capabilities and they know who the brightest are and they know who the dullest are...The interesting thing is that these kids have a pretty good perception of where they're at. And the important thing I think is to say 'Well just cause you don't know math doesn't mean you're dumb, it just means you don't know <u>math</u>, that kind of approach."

The teacher took a similar approach with Bonnie (Case #18). She often experienced frustration with not being able to keep up with the written work.

Does she say things to you or do you just know?

"She has on many occasions. We've had some very frank discussions about some of her problems and she's very aware of the difficulties. she has." One objective that the teacher appeared to have for the students in his room was the development of a good self concept and the ability to see their problems in perspective.

You appear to like children.

"Well I may not be able to do everything right, but I do... Children are very important to me."

You were saying one day that the selfconcept part was important to you. "The affective is... I want children to leave my class saying 'I like myself'. If they can do that... If they don't know that two plus two equals four at least they've succeeded at something. Schools are important, but at this age?"

Speaking of Ian and his many problems...

"It's been a pretty rugged year and I try not to put too much pressure on him and yet I want him to ...Well he's a year behind now. I'm pretty anxious that he, at least, leave grade three feeling he's a decent sort of fellow anyway with some of the basic skills that are required. He's a nice boy and school's not going to be easy for him and it's probably going to set border for him." "Kids who like to come to school learn more effectively than those who don't."

When the class as a whole was engaged in independent seatwork activities the teacher spent most of the time walking around monitoring seatwork and providing individual help to students.

"It's a time that I enjoy and sometimes the pace of the lesson will suffer because I'll be spending time trying to make sure that, people have the idea. I find I sometimes think I can have greater impact, well it makes sense, a one to one situation. Well sure it's a good way to evaluate too as far as I'm concerned. I can tell right now if a kid can do that and I sort of mentally store some of that information."

"This is the kind of time that I actually enjoy- being able to get around without a great hubbub around you, on a one to one. Just to briefly check. I find that I can retain quite a bit you know...get a pretty good impression, put it that way."

On other occasions he spoke about students who are able to answer questions correctly all the time.

"But you see on a day to day basis you have to look at other things because you say to yourself 'Okay these people who answer these questions, that's great, but I have a responsibility to these <u>other</u> kids too...to at least try to make sure that they're going to get it. And I hope that I try to do it in a way that... You know, a kid may not know math. Doesn't mean he's dumb, just means he doesn't know math very well and I want kids to go away feeling that; not feeling that 'Hey, I'm dumb'."

"I get almost zero reward from asking people who I <u>know</u> will provide the answer. I don't get a reward from that. I have to test by asking people for myself. I have to test to find out whether children understand the concept. The children that I'm going to ask are the ones who might have the greatest difficulty." In Conners' study, Equality of Treatment was a concern expressed by his teachers. This teacher expressed several thoughts which indicated that he was concerned about fairness and equality of treatment as well.

"One of the problems I have with Brent is getting him not to shout out and very frequently, well he <u>is</u> getting better, but often I have to remind him that 'Brent, the procedure we use is hand raising so I can get to a variety of people'."

"In a situation like this, just try to get everybody once, or something like that and it's going quite well."

"I often ignore hands up. You know I just don't even... The problem with taking hands-up volunteers is that the same people volunteer all the time."

"I want to make sure that I have... That's why they read short passages. I wanted to make sure that I gave everyone a chance to read."

" I have to say to myself that these kids <u>deserve</u> to be asked cause they know this stuff quite well. I've been surprised so rarely this year by Joanne <u>not</u> knowing something that I sometimes say to myself 'Hey I have hardly asked this kid a question all day'. So you have to include, when you teach a large group like this, I like to try to get everybody."

"Ian spends a lot of time off task, quite a lot of time off task, and I know that. And one has to sort of balance off how much time to spend keeping him on task specifically and that's a judgment that you have to make and its difficult." (in fairness to the time demands made by other individuals in the class.) The teacher gave more evidence of favoring <u>inequality</u> of treatment overall however. He regarded his students as individuals it seemed and different approaches were taken with them and disciplining was contingent on various factors, not universally applied. Little evidence of "disciplining" occurred, therefore, because the usual practice was to deal with an incident in a low-key manner, often in a private conversation at recess time or during class.

"I'm not very fond of generally applied punishment or generally applied anger. That kind of thing is not my style and anytime I fall into that sort of trap myself, if I ever make a sort of class -wide threat, I find that it works very poorly for me because I... You must consider people as individuals. I don't like that form of discipline in a school except in <u>rare</u> circumstances ... then invariably what happens is the very person you don't want to step out of line inadvertently does so. Then what do you do? Then you either strap or lose face. That's garbage! No, I think that's a fast way to lose the respect of children if you frivolously apply..." "The thing is whenever I chew out a kid, which is rare, I have a hell of a time keeping a straight face."

Some comments taken out of context may serve to demonstrate that differential treatment was provided to class members. In some cases when obvious misbehavior was occurring through what he considered to be a fault of his own, he chose to ignore it unless it became too great. "The problem is sometimes I'll look up and I'll see someone like Edward fiddling around or Geoffrey or some of those people. And I sometimes have to lay off those people because the kids have the

concept and I'm overdwelling on it and it's not really fair to be ..."

"I don't know who's wandering around there I didn't see ... Joanne.

Oh well, it's alright if it's Joanne. She has a good reason, no doubt." "Also I like to satisfy Brent's need to speak out too. If I frustrate that then the speaking out behavior is more frequent." "Sometimes staying on task in school in class time is difficult for him, or appears to be, because he doesn't do it but anytime I say 'Okay, Grant, that's fine but you best have that done and here tomorrow' he always does it and does it well."

"It's discouraging for her to have to do written work. I try not to put too much pressure on her for written work, you know. She'll sometimes play that off against me saying 'Well, should I only do this much?' and if it's a reasonable amount I'll say 'Yeah, that will be fine'."

"You know, it bothered me a lot more earlier in the year. You could spend two-thirds of your day saying 'Nick, sit still'. I very rarely bug him about that." and also of him..."You always question someone who is willing to set themselves up as the class clown which Nick does from time to time and people in the class know very well that it's Nick's turn for a pat on the back sort of thing. How happy is an individual like that? Nick seems like a happy enough individual but underlying that there must be some need for attention that goes unsatisfied."

"Partly with Marilyn I like to check up and make sure that she's got the skill."

Two thoughts reflect an acknowledgement of the harried pace of classroom life and the demands on the teacher...

"Pace and flow of the lesson. Teacher has to be very careful of those kinds of things because kids will sidetrack you and you know they want to contribute and participate."

The teacher provided praise and encouragement for the students in his room. It was not lavishly or thoughtlessly given and not indiscriminantly applied. The following thoughts reflect his desire to provide meaningful praise and encouragement.

"With handwriting... I don't take any unique or unusual approaches to it or anything but you can sure get a lot of positive feedback to children in a handwriting situation because most kids this age level are reasonably successful and even if they aren't you can keep them going..."

"This ("poor") group I think enjoys the oral reading exercise because I try to build them up to indicate to them that I think that they're doing, in general, a good job and it gives each one a little ... you know, the focus is on them briefly and they can succeed with these stories. They're fairly simple and they can succeed at them. So it's a good opportunity for lots of positive reinforcement." And again of this reading group ...

"The last few weeks I've been quite pleased with the way these people have been oral reading. They've come a long way. Word recognition skills seem to be improving. They're aware of expression. Well most of them understand that when you read orally, you read so that others can hear. And they've really improved since the beginning of the year and so I've been trying to take the opportunity to praise them as we go along...Well the other thing is that this is a good opportunity to encourage people to read I think. They've demonstrated publicly that they can read you see and if you encourage in that sense it may encourage them to read privately too." Reflecting on working through a long interaction with Mason the teacher commented:

'I did a lot of structuring for him and he did answer and I didn't understand his answer. I tried to rephrase it and he didn't agree. So I asked him to rephrase it himself and he said the same thing. If he was <u>right</u>, if he had a good point, I wanted to be able to say 'Hey, yeah. That's right' but I didn't understand and I thought 'Well we've spent enough time'."

"I ended up with Joanne, not because I wanted to hear her read again, but because it occurred to me that when <u>she</u> had read I hadn't said anything about her reading and that I ought to say 'Well Okay. Very Good' or at least that. So she finishes up." "Later on, by design, I went back to him and gave him a short and fairly easy part to read and he did it well and then I could praise him and <u>did</u>. The thing is they're reading publicly so the evidence is there and so my comments ... if I say 'Very good. That's just dandy' everyone knows that that's <u>not</u> the way it was. So I try to make constructive comments ... Well sometimes the remarks are a little pedestrian. It's important I know to think of specific things to say about people's reading and about their work as well and yet you would need a list of 1001 items and sometimes that's difficult ... And that remark I just made to Brenda was almost off the cuff and it's not very effective I know."

The thoughts expressed to this point contain implicit attitudes and priorities about his tole as a teacher and the ways in which he prefers to deal with students. These are the clearest indications available concerning this teacher's system of beliefs and values but one could clearly derive the following priorities:

the development and maintenance of self concept in the student.

the development of a realistic view for the individual student (and his parents) of his strengths and weaknesses as a student.

, the provision of support and encouragement to the student.

Ę.

the regard for each student as an individual requiring differential treatment and consideration while at the same time ensuring fairness and equality of treatment for all students in the class.

The only other thoughts included in this section are those concerning his participation in the research experience itself. He was asked to comment on any effects the presence of the observer and the recording equipment had on him and on his teaching performance. "Well you were around long enough to know that probably I have enough other things going that I really couldn't change my routine significantly... Well I became so accustomed to having you there although there was...you can't avoid the fact that there's someone there and you know they're there but that's fine and for example within the last week there have been a couple of times when I've even sat on my desk which I don't think I would have done if you'd been there but my routine has not been... Well I probably wasn't much different after you were there a couple of days than I was toward the end."

You were used to the Quest coder too.

"So I wasn't too worred about having you or the cameras or the tape recorder or anything else there. Certainly the kids weren't. No one even mentioned to me that you'd gone!"

What about the way that you were thinking about the students?

"<u>Absolutely</u>. As soon as we got into some of these things and you were asking me questions that causes you, it caused me to focus attention on the kinds of questions that you were asking."

Which is sort of too bad from my point of view ...

"Well yeah, but if you ask a question and want someone to focus. The other alternative would be to do it all at the end."

But I needed to know these things in order to develop the 26 portfolios " but through your eyes, how you say you feel about them and some were chosen for intensive observation (use of academic learning time) on basis of your comments.

"Well I think that's an unavoidable difficulty which you'll have to ...

...I think it's quite clear that this year I know more about children in my class than I ever have before- about their personalities. And I've given greater thought to the kinds of individuals that they are and how I interact with them and how I feel about them and things like that...and sure, that focus is a result of the Quest study and your interest...so certainly whether it has had any effect on the children or how I would have felt about them anyway I don't know but I certainly have focussed more than I have before."

The teacher had begun to rethink the questions posed to the poor reading group .

"But that doesn't mean I oughtn't to ask them questions that cause them to think!... One thing that all this has done for me, aside from ruining my health, is it causes one to rethink approaches that you're doing. That doesn't mean that in one fell swoop I'm going to redo those things that I perceive to be weak. At least if I think about them I can make judgments about them and move one way or another and that's useful."

Summary

In the first section of this chapter, the results of the micro-analysis of the stimulated recall interview data were presented and discussed. The micro-analysis system developed by Conners (1978) was employed in order to convert the qualitative interview data into quantitative form which could then be analyzed to identify the

predominant types of interactive thoughts this teacher processed. Although Conners' system had nine major categories, many of which were further subdivided, most of this teacher's interactive thoughts were categorized in the three areas of Perceptions (with a concentration on perceptions of student academic performance during instruction), of Instructional Moves he was making (particularly those dealing with classroom control), and of Background Information about Students which the teacher had brought to the lesson situation. A substantial proportion of teacher-expressed interactive thoughts were Self-Monitoring in nature as the teacher consistently expressed an awareness and concern for the effectiveness of his teaching behavior during instruction. He also expressed a considerable number of personal Feelings concerning different aspects of the ongoing lesson throughout the stimulated recall interviews. These more personal revelations merhaps indicate that a suitable rapport had been established for conducting stimulated recall interviews and for his sharing affective reactions to his teaching performance with the researcher.

A macro-analysis of all interview data, both stimulated recall interviews and the more general interviews which had been conducted with the teacher was performed in order to identify the more global attitudes and priorities which, by virtue of being expressed, appeared to exert a degree of influence on his teaching style. On the basis of such teacher-expressed thoughts, the researcher tentatively concluded that four attitudes or priorities were representative of those guiding

his dealings with students in his class:

-the development and maintenance of self concept in the student.

-the development of a realistic view for the individual student (and his parents) of his strengths and weaknesses as a student.

-the provision of support and encouragement to the student.

-the regard for each student as an individual requiring differential treatment and consideration while at the same time ensuring fairness and equality of treatment for all students in the class.

It is to an examination of the observed teaching behaviors of the teacher subject in the present study that the discussion now turns.

CHAPTER VIII TEACHER BEHAVIOR

The findings in this chapter are in answer to Research Question 5 which, in broad terms, asked "How could the teaching behavior of this teacher subject be described?"

In the first section of this chapter the observed teaching behaviors of this particular teacher as recorded for Project Quest on 31 high inference variables are summarized and used to develop a descriptive profile of this teacher's behavior during instruction.

The second section of this chapter describes the verbal behavior of this teacher using the low inference data collected on the DIOS (dyadic interaction observation system). The many DIOS categories indicate what kinds of verbal behavior the teacher exhibited during instruction. Any differences in the teacher's verbal behavior exhibited during language arts and math instruction are identified as well as any differential verbal treatment of particular students or groups of students.

High Inference Variables

Teaching behaviors as observed on the 31 high inference variables from Project Quest were assigned a numerical value of 1 to 5 or were coded as "not observed." A coder for Project Quest made 10 visits to the classroom (five during language arts instruction and five during math instruction) representing about 10 hours of classroom instructional time over the period ranging from Debruary 19 to May 18, 1979. For the purposes of discussion here the ten sets of coding scores received by the teacher on each of the 31 variables have been totalled and averaged for the three months. These three months coincide fairly well with the March 1 to June 7 timeline observed by the present researcher so that data gathered were pertinent and relevant to this study as well.

On all 31 variables (see Appendix A) the teacher scored above the mean average for the sample of 60 teachers subjects in Project Quest and on eight of these variables he received a top-score of five for every observational visit made. As a result he was one of the highest scoring teacher subjects on these variables which, in the literature from correlational studies, are seen to be highly desirable and effective teaching strategies.

Some of the 31 variables vary in the degree of high inference judgment necessary on the part of the coder. For instance, Items #18 and #27-31 are based on summarized impressions and ratings of the more objective (and low inference) verbal interaction data gathered for Project Quest. On three of these (#28-providing "sustaining" behavior In the form of tlues or rephrasing of questions to encourage and extend pupil answers, sample mean= 3.81; #29- use of "praise", sample mean =3.84; and #31-"acceptance of student-initiated interaction , sample mean=4.02) the teacher received a perfect score of 5 overall. He also scored well on #27-"selecting many different students to answer" and for #30-"using criticism with the more able student" and #18-"using an appropriate mixture of high and low order questions."

The composite profile of this teacher using these scores might result in the following description: The 26 children in his classroom had already internalized a "system of rules and routines" (#1) and, as a result, little misbehavior actually occurred in his room. Any instances of misbehavior that did occur were (#2 and #3) "quickly spotted" and "handled in a low-key manner" (59, usually privately. He was constantly "on the move" (#4) during seat work activity and was thereby able to "spot potential problems" (#2) and to provide individual attention and to (#14) "spot check assignments" as they were being completed. This was particularly true in math classes where the children worked as one large group. "Spot checking (#14) was also accomplished by involving particular students during instruction and presentation of both math and language arts lessons.

He was able to command the students' attention readily and would sometimes use a "hands up everybody"(non-verbal) "standard signal" (#9) to alert them. He received uniformly high scores for this and for "waiting to speak"(#10) or "stopping his instruction in progress until all children were attending" (#11).

Children were able to work on their own on independent math activities and in reading group situations, particularly the better reading group, because he chose "independent activities that were worthwhile" (#6), "task-oriented," and which seemed to "optimize academic learning time" (#8). He utilized a variety of instructional techniques which maintained student interest and involvement.

Using the data from the particular high inference variables, derived from Kounin's (1970) work, which were more global and allencompassing, the teacher could be described as personable, supportive, liked and respected by his students. This teacher consistently displayed a high degree of "warmth" (#25) and "empathy" (#25) toward the students, was "clear" (#23) in presentations to the class and demonstrated "persuasiveness" (#24) or charismatic motivating behavior. He was rated,

high on "withitness" (#19) or awareness of what was going on, and on

"overlappingness" (#20) or the ability to attend to more than one issue at a time, such as evidenced in the reading groups situation. Both were ably demonstrated by this teacher whose instruction was well-paced, with good "momentum" (22) and with "smoothness" (21) in

transitions from one activity or subject area to another. If fine tuning improvements were to be made anywhere, the physical transitions from seats to reading group areas could have been accomplished, perhaps, with more army-like precision but although this was acknowledged by him during interviews as an area for possible improvement, he chose purposely to allow the students a bit of breathing space during these times.

In summary as reflected by Project Quest's data representing about 10 hours of classroom teaching, this teacher displayed a high degree of proficiency in classroom management skills, instructional skills, and in interpersonal skills which are embodied in this high inference data collection instrument.

With both the Quest coder and the present researcher the teacher subject was extremely cooperative, pleasant, and accommodating. Having spent three months at the school it was obvious to the researcher that he was a popular staff member who got along extremely well with colleagues and whose good humor and repartee were appreciated and enjoyed by fellow teachers in staff room conversations.

Teacher Verbal Behavior

Verbal Interaction Data

The verbal interaction data consist of 2715 interactions which were recorded during 14.9 hours of language arts (1524 interactions) and during 10.5 hours of math (1191 interactions) representing 25.35 hours of combined instructional time in all.

While recording each interaction, any number of applicable categories out of a possible 60 categories across the row could be checked. At the beginning of each interaction the identification number of the student involved was recorded in addition to whether the verbal exchange was a Tinit (teacher-initiated) or a Sinit (student-initiated) interaction. In order to identify the type of interaction one selection from the seven types of Tinits was made or one choice from the two Sinit types. The grouping mode in use (individual/independent work activity, small group, whole class) was always checked as well as the duration of the interaction (long or brief). Other information about the particular intefaction was recorded by putting check marks in the appropriate boxes along the row on the Dyadic Interaction Observation System (DLOS)

Instrument.

An examination of the data revealed that about eight categories were very seldom used and these were deleted from the analysis. In addition it was advantageous to combine several categories into one more inclusive category (eg., "Sustaining" is comprised of the three former categories of Repeat Question/Give Clue, Pose New Question, or Repeat Child's Answer). A Sustaining response was used by the three following an incorrect answer or when the teacher wished to example a pupil answer given).

In all, 32 categories were selected for use in order to compute frequencies and percentage ratios for each student. These ratios were calculated for each of eight separate language arts settings (eg., reading

group instruction, working in spelling workbook, listening as a whole class to the teacher reading the novel) and were recombined to compute a total language arts instruction ratio. Ratios were likewise calculated for each of the five math contexts and for all math instructional settings (by combining the five math contexts) and also for <u>all</u> instructional settings in both language arts and math to produce a total verbal interaction ratio for each student. Sixteen of these categories were further separated into three grouping modes (of individual/independent, small group, whole class) in order to see what ratios of the 16 categories selected occurred in which grouping mode. For an example of the individualized ratios computed for each child in the settings for which he was in attendance, see those included in Case #1, Sharon and Case #3 Geoffrey.

Sinit Interactions

From a closer examination of this verbal data it was possible to identify certain characteristics of this teacher's behavior which substantiated data gathered from other sources.

The kelative proportions of Tinits and Sinits reveals a prevalent teacher characteristic; the involvement of students and his acceptance and integration of student ideas. He had received high scores on this high inference behavior #31 ("When pupils initiated interaction, the teacher accepted and integrated the pupil question," comment, or other contribution."). Of the 2715 interactions recorded, 1931, or 71.1% of them, were Tinits and 784, or 28.9%, of them were Sinits. Effectively over one-quarter of all interactions during the 25 hours of instructional time sampled were initiated by the students. These Sinit ratios differ for subject area, for group memberships, and for grouping mode. More Sinits, for example, occurred during the individual or independent work activity mode (49%) than for either small group (29%) settings or whole class (23%) settings.

Specifically for subject area comparisons, Sinits accounted for 31.16% of all language arts interactions but for only 25.9% of all math verbal interactions. This may indicate a difference in the teacher's instructional approach to these two different subject areas. In any event, although percentages differ for grouping mode and subject area, this teacher, on balance, permitted the students to initiate interactions about 29% of the time which is a considerable proportion when compared to the interaction patterns in many classrooms.

The high proportion of Sinits also indicates another feature of this teacher's behavior. By inference, drawn from pre-lessin conversations and observation of this teacher over the three month period, it appeared that he did not engage in extensive preactive planning as a general rule. Rather, the global objectives of the lesson and the necessary instructional materials were assembled and he proceeded by attending to his general lesson plan, to student-provided contributions, to student-provided cues such as inattention or academic performance or to his perceptions of their cognitive processing as the lesson progressed.Zahoric (1970) had noted that teachers who planned extensively were less flexible and more rigid; Marx and Peterson (1975:16) found that teachers who made fewer preactive decisions made more interactive decisions and had students with more positive attitudes; Shavelson (1976) felt that most interactive decisions were modifications of those made in planning; and Clark and

Yinger (1977:292) felt that interactive decisions were more a "process fine tuning and adapting to the aspects of the situation that were unpredictable in principle" (such as specific student responses). It could be argued that the acceptance of a high proportion of studentinitiated interaction is representative of interactive decision making.

Cooper (1977:421) claimed that a teacher was rewarded in interaction with the brighter, high-achieving students and suggested that the teacher's "control over personal rewards", especially while interacting with certain students depended on his ability to control the "timing, duration, and content" of the verbal contacts. Hence, a teacher tended to deal with low-achieving or troublesome students in private settings where all three factors were more likely to be under the teacher's control. Cooper claimed that teachers used criticism in public settings to discourage the overtures of certain students with whom the teacher preferred to deal privately. Little evidence of this behavior was apparent in this classroom. Although no measures exist, one would hardly suppose that this teacher would be considered as "dogmatic" or "authoritarian" by virtue of the Sinit percentages alone.

Tinit Interactions

The amount of involvement with the 26 students did, of course, differ. The average amount of all interaction for all students was 4.17% (due to the adjustments for absences when computing individual ratios) and individual total interaction averages ranged from 1.2% to 7.7%. Generally the 13 boys (average=4.61) received a greater amount of total verbal interaction contacts with the teacher than the 13 girls (average=3.73%).

The total amount of verbal contact with the different

attitude groups regulted in the following percentage totals overall:

3 concern students5.53 rejection students4.85first three attachment4.7316 attachment4.15 indifférence3.4

More detailed information about the verbal interaction received by the four attitude-to-student groups is available in Table 24 in Chapter V, Page 149). The teacher-initiated contacts resulted in the following scores for the attitude groups:

concern6.1rejection4.9attachment3.9indifference3.8

Tinit Types

In all there were seven different types of teacher-initiated interaction chosen for examination in this study. The percentage of occurrence of the seven Tinit types is presented in Table 35.

	Language Arts	Mathematics	Total Combined Subjects	
Product Q	25.0%	30.1%	27.6%	
Process Q	13.1%	8.6%	11.2%	
Choice Q	0.1%	1.0%	0.5%	
Self Reference	3.0%	6.0%	4.3%	
Non-Academic Q	7.8%	6.1%	7.1%	
Get Attention	1.0%	1.6%	1.3%	
Comments ·	17.8%	20.3%	18.9%	
Sinits	31.2%	25.9%	28.9%	

Table 35 PERCENTAGES OF TYPES DE VERBAL INTERACTION It is notable that more Product questions (30.1%) occurred during math instruction than in language arts (25.6%) instruction and that conversely more Process questions (13.1%) were posed during language arts instruction than during math (8.6%) instruction. This could be explained by the nature of the subject matter itself for the math lessons observed frequently involved straight factual recall.

Even more pronounced are the differences in Product and Process questions that occurred among the high, middle, and low expectancy groups for the two subject areas of language arts and math.

Table 36

DIFFERENCES AMONG HIGH, MIDDLE, AND LOW EXPECTANCY TINIT RATIOS

-	I	.anguage A	rts	Mathem	atics	۰.
	Product	Process	Achievement	Product	Process	Achievement
High	15.08	16.42	193.62%	/3.53	4.53	* 97.37%
Middle	(*3.78 *	3.79	* 86.16%	4.12	3.68	94.00%
Low	* 3.50	1.52	71.50%	5.88	6.16	85.00%

The ratios received are presented in Table 36 along with the end of the year achievement results. (For more detailed information see Table 31 Chapter V, Page 164).

Brophy and Good (1974:107) noted that "expectancy group differences will be larger later in the year because (1) polarization has occurred and (2) after initial effort, the teacher has given up upon them." While this may seem somewhat true in language arts, the trend is quite the opposite during math instruction.

It is purely speculative to suggest that math might be an easier subject area in which to accomplish remedial work and this may encourage a teacher to pursue the low-achieving math student and spend more time and attention on him. Table 37 reduces the information in Table 35 to three types of interaction for closer examination.

PERCENTAGES OF TOTAL INTERACTION				
V	Language Arts	Mathematics		
Product Q	25.5%	30.1%		
Process Q	13.1%	8.6%		
Sinits	31.2%	25.9%		

Good and Grouws (1977) conducted a process-product study in fourth grade classrooms during math instruction in order to determine which teaching behaviors resulted in the highest math achievement. They concluded that the "integrative effect of specific math teaching acts in relation to the degree and sequence of other teaching acts accounted for higher levels of success" in some classrooms. If, for example, the teacher was rated high in clarity, and asked lots of product questions, and "used process questions or gave developmental feedback when student responses indicated some error" (p.52), and allowed a high percentage of studentinitiated interactions, then higher levels of math achievement could be effected in a whole class grouping mode. Good and Grouws advocated posing more product (not process) questions in math instruction, They found that classrooms using math ability groupings fell into middle math achievement levels. Less effective math teachers in their study asked more process questions, gave evaluative feedback (not developmental process feedback), had a higher percentage of teacher-initiated verbal contacts, a "heavy" climate in the classroom, and more managerial problems (p_{\neq} 53). The teacher in this study did not group students for math instruction, used more product questions overall, and tended to give

developmental process feedback (in the form of Sustaining and Process Terminal Feedback, both categories on the DIOS) to students experiencing difficulty and encouraged student-initiated participation (about 25.9% of all math interaction).

Differences noted in the expectancy level ratios for Process and Product questions in language arts and math may not be due solely to expectation effects in operation but rather to differential instructional tactics dependent on <u>subject matter</u> being taught, and not upon <u>students</u> being taught.

Effects of Grouping on Product and Process Questions

Another possible explanation for the Product and Process question differences in language arts and math is the fact that groups ("good" and "poor" reading groups) were in operation during portions of total language arts instruction. Perhaps a different "mental set" is an automatic side-effect of segrating a class into different ability groupings and this may have acted to influence the teacher's differential^{*} treatment of these students while they were taught in actual reading group situations as well as for language arts instruction generally. Table 38 (taken from Table 28, Chapter V, page 158) shows the relative proportion of Product and Process questions received by the "good" and "poor" reading group members.

Table 38

LANGUAGE ARTS INTERACTION RATIOS AND ACHIEVEMENT FOR "GOOD" AND "POOR" READING GROUPS

•	Product Q	Process Q	Total Tinits	Achievement
"good" group	4.20	(* 5.07	4.07	90.3%
"poor" group	`3.9 6 ′	2.20	4.20	75.2%

Good and Brophy (1974:118) stated that in the "early grades the important differences are usually in the quality of the interactions since grouping for small group instruction tends to equalize the quantity of interactions that different students in the class have with the teacher."

The following exerpts have been taken from the interview transcripts to show that perhaps a different mental set toward the two reading groups existed. Certainly a recognition of ability differences was expressed. A non-intentional (but not undesirable) side-effect of the study itself resulted in the teacher, who was prone to self-monitoring, starting to re-examine his interactions with the "poor" reading group while the study was in progress. The researcher did not discuss the noted differential treatment with the teacher but questions asked apparently alerted him to assess his questioning patterns used with this group.

He spoke of the questions interspersed during the oral reading of the novel...

"Usually I start just by getting the details, the factual details of where we are in the story so it will be a little review, that sort of thing and then after I will get to the other types of questions - asking them what they think of the various individuals." "This is a terrific opportunity to ask the sequence of questions that we're supposed to ask, you know, the higher order questions. You can sprinkle them...I can ask a lot of just straight lower level questions of facts and so on and build up to the others... "

Speaking of the lesson in progress with the "poor" reading group ...
So there are more factual questions?

"These are just straight fact, just a straight rehash of the story, the key points in the story...and I don't get beyond, them." Speaking of a lesson in progress with the "good" reading group...

"These people do well answering those kinds of questions, the opinion type questions, even though, as I've said before, they're a more difficult question. But these are the kids in the class who in reading sessions will venture a guess as to what's going to happen. They're the ones who respond pretty effectively to the higher level questions as a general rule."

The front group?

"The front group very frequently if you ask the 'what if?' questions, the 'what do you think of?' questions, I'm not nearly as likely to get a response that is...well it depends on the situation, but I'm speaking generally. I won't get as good a response from that front group as I do from the back group. They aren't as imaginative generally and they don't comprehend what they read as well either, you know, that aspect." (April 17th)

During the reading of the novel, Nicholas, one of the "poor" reading group members was asked a higher order question...

"The answer was perfect. I asked the question because I thought he was too wiggly. I thought he was half off task and I wanted to bring him back but he was listening obviously and it wasn't an easy question."

Yes it was one of those 'why do you think' questions...

Yes, and well he had to relate it back to an earlier part of the,

story that was quite a bit earlier and make the connection

... yeah so it was exactly the correct answer."

But he's in the front group and often you say you don't ask those sorts

"That's right and maybe not often enough. That's maybe something that... what happens is that you ask those questions and get <u>nowhere</u>. So you quit asking them and, of course, that's unfair. And usually, in thinking about it now, I can think of occasions when I've asked a number of questions, comprehension questions, that take it a little beyond..."

Well I noticed this morning that in the front group you were asking those higher level questions...

"You see I've been thinking about that and I've thought I have no right to cut them off simply because I'm dissatisfied with the answers on a given day or a given month, if you will. What I have to start doing is restructuring my questions, make them <u>simpler</u> higher level quesions, something like that..."

I remember one day we were talking and you said you don't usually ask them with that group.

"That's true, and in thinking it through more carefully since, however, I say to myself 'Well why would I <u>not</u> ask those sorts of questions?!"

But you must have had some basis for thinking that you don't get anywhere with that sort of question.

"Well I <u>didn't</u> get anywhere with it but in rethinking it maybe my expectations were too great. Maybe I shouldn't always expect to get good answers. Secondly maybe I'd better look at rephrasing questions in such a way that they're more easily understood;" If you don't get good answers still will you abandon them?

"No. Well that's what I'm saying...You know no matter what level of reading a person is at there <u>have</u> to be appropriate level questions at that level and if I'm not asking them then I'm cheating those kids a little bit in getting them to think through those thought processes... "

"Let's put it this way. My primary concern was of a more basic nature; 'Do they have the basic details, facts of the story'and you feed that really before, as a basis for the more complicated higher level questions. But what happened is that I spend too much time making sure that they had the basic factual information but didn't carry it beyond that. That's my responsibility and in thinking it through more recently that these people are certainly capable of answering an appropriately asked question, and constraints of time sometimes come in and you say 'I'll ask questions to see if they have the basics and then I've got to get back to this other group and with this other group I <u>assume</u> they've got the basics and we start with the higher level questions."

So you don't always go down to the level of fact gathering with the back group?

"Sure, I'll frequently assume, based on experience... You have evidence?

"yeah"

Just in the same way that you have evidence with the front group that they don't?

"That's right but bound to be wrong if I do that. Bound to be wrong in both instances... I think that the <u>sequencing</u> of questions is important and it's funny because I <u>do</u> that in <u>Watership Down</u>. It never occurs to me not to ask the basic structuring type questions that give us the basis for the higher level questions. It never

occurs to me not to ask them there but then I get them into these individual groups and I seem to be on the run and this sort of thing . That might be the reason, it's not an adequate ..."

Do you feel differently about the front group? Is that why they're in the front group?

"How I feel about their ability in reading...that's right and what I'm saying now doesn't change the fact that in most areas they are not as capable as... "

There's a difference?

"Certainly there is but that doesn't mean that I oughtn't to ask them questions that cause them to think; not on a different level from the first level of question."

Tinit Comments

Comments were teacher-initiated statements to pupils. Frequently about work-related matters, they could include either positive or negative evaluation or perhaps a behaviotal reminder. Usually at least one additional category was used in conjunction with the comment to classify its import.

Evaluative Tinits. Criticism was occasionally expressed. It occurred a total of 55 times in conjunction with the possible 1931 Tinit interactions. The criticism given was not ability related usually but was directed at the student's behavior. However, the resultant percentages were inflated because of the relatively infrequent occurrence of criticism overall. Marilyn, for example, received three instances of criticism out of the 34 delivered during language arts and her resultant criticism ratio was 8.8% (of all criticism delivered to students in language arts). Instances of criticism occurred in conjunction with only 2.8% or all Tinit interaction overall and could not, therefore, be considered as a predominant verbal teaching behavior.

380

Conversely, there were 227 instances of praise accompanying 11.8% of the 1931 Tinits. Boys (4.56) received higher ratios of praisecontaining-verbal contacts than girls (3.90).

Differences were noted for expectancy levels for in language arts more preise was delivered to high expectancy group members but in math more praise was accorded to low expectancy group members with whom he exchanged more total instructional interactional Tinits in any case. Keeping in mind that criticism was used much less frequently and the ratios are not as reliable as those for praise, Table 39 presents the praise and criticism ratios received by the high, middle, and low language arts and math expectancy groups

					ST GROUPS
•		Lang	uage Arts	Mathen	natics
		Praise	Criticism ,	Praise	.Criticism
ć	HIGH	4.48	3.76	2.86	A.45
	MIDDLE	4.01	2.70	4.42	4.91
	LOW	4.13	7.35 •	6.04	2.24

Table 39 PRAISE AND CRITICISM RECEIVED BY EXPECTANCY GROUPS

From the literature it appears that praise is infrequently used in most classrooms. Silberman (1969:406), found that positive evaluation was the least frequently displayed teacher behavior overall. Haigh (1975:108) claimed that out of his eight teachers only one made any use of praise. Cornbleth, Davis, and Button (1974:57) commented on the infrequent use of both criticism and praise. Bennett (1978:143) claimed there was evidence to suggest that criticism (of academic performance) can be as effective as praise with certain types of pupils. Good and Grouws (1977:52) noted that the effective math teachers in their study praised less than the less effective math teachers. Cooper and Burger (1978:25) concluded that in lab simulations the low ability students received more praise and less use of criticism but in naturalistic classrooms on the other hand, low ability students received the opposite treatment.

Brophy et al (1976:260) ascertained that some praise merely reflects the teacher's attitude to the student and the desire for personalized contact with students who respond favorably. In Table 40 the Praise and Criticism ratios for the attitude groups are presented.

	Tinit Praise	Tinit Criticism
indifference	3.60	5.66
attachment	3.93	3.35
first 3 choices	. 4.96	5.83
rejection °	5.70	8.75
concern	5.90	3.20

TABLE 40 PRAISE AND CRITICISM RECEIVED BY ATTITUDE GROUPS

Silberman (1969) had concluded that teachers tended to compensate for

negative feelings held toward rejection students by praising them at every appropriate opportunity. He also felt that teachers minimized praise directed at attachment students in the effort to avoid displaying favoritism toward these students.

In summary it would be fair to say that praise was a relatively predominant teacher behavior exhibited consistently by this teacher. He had stated that he liked to praise and support the students at every opportunity deserving this teacher reaction. Praise often accompanied Sustaining behavior exhibited by this teacher.

Summary of Verbal Behavior

During the 25.35 hours of language arts and math instruction in which the dyadic verbal interactions were recorded the following teacher verbal behaviors were noted: A high percentage of verbal interaction was characterized by Sinit contacts originating from the students themselves. Over one-quarter of all interaction was student-initiated. The fact that the teacher permitted and encouraged that degree of student participation corroborates the high inference data about teacher "warmth", "empathy", and "acceptance of student degree." The most frequently used Tinit behavior overall was the Product question, followed by Comments to students. Process questions accounted for about 11% of all interactions with a higher percentage occurring in language arts. Other question types were used less frequently.

In conjunction with the teacher initiated contacts, praise was used about 11% of the time which appears to be a higher than average use of praise and this information substantiates his expressed intention to praise

and support the students when it was realistically possible to do so and also supports the high scores assigned to related high inference measures.

The teacher was aware of the importance of questioning skills as expressed during interviews and appeared to pose academically oriented questions to the students for most of the verbal interaction recorded.

Summary

The first section of this chapter presented a brief description of this teacher's exhibited teaching strategies as recorded on high inference measures representing ten hours of instructional time in language arts and math. He performed well on all variables observed and was rated as one of the higher scoring teachers out of the 60 in the total sample of the Project Quest study. This information was used as an objective data source about this teacher's instructional behavior during language arts and math lessons.

The second section of the chapter presented the predominant verbal behaviors exibited by this teacher as recorded on the DIOS instrument during about 25 hours of instructional time.

CHAPTER IX

SUMMARY, CONCLUSIONS, IMPLICATIONS, AND RECOMMENDATIONS° FOR FURTHER RESEARCH

Summary of the Study

Purpose

This study viewed the teacher as an information processor who might act on the basis of causal perceptions of and attitudes held toward students in his classroom. This study investigated the interaction of one teacher and his 26 students using a case study methodology and eclectic data collecting techniques.

Various forms of teacher-provided information about students were compared to the substance of his interactive thoughts about students during actual teaching situations, his thoughts expressed about them in "general interviews, and his expressed affective feelings toward his students.. This information about students was examined in relation to overt teaching behavior exhibited during instruction, in relation to the dyadic verbal interaction exchanged with these students, and to the level of achievement of these students in a bid to seek some understanding of the factors affecting ongoing instructional strategies used by this teacher. The study examined the extent to which traces of the expectancy effect could be found in the behavior of this teacher. A related purpose was to examine the effectiveness of stimulated recall as one of the introspective data collecting devices used in this study.

Methodology

The teacher completed class rankings on six expectancy and personality measures and in addition rated each student individually on 13

્ 384

attributes. Both these sets of information were converted to γ ' quantitative scores and combined for analysis purposes.

Self reports of interactive thoughts about students during eight videotaped lessons were obtained using stimulated recall procedures. These data were coded using a category system designed to facilitate the quantification of these interactive thoughts. Additional introspective information was obtained during five general audiotaped interviews and during many other shorter taped conversations with the teacher. All interview data were subjected to a broad macro-analysis which identified the overarching attitudes and priorities expressed which may have been representative of the teacher's underlying beliefs.

During 10 hours of language arts and math instruction, overt process data on 31 observed teaching strategies exhibited while working with the students were obtained using high inferences measures.

Low inference measures for recording dyadic verbal interaction during 25 hours of language arts and math instruction were used to identify the verbal exchanges occurring between individual students and the teacher. These interactions were analyzed at the observable level to determine the quantity and quality of verbal contacts with the teacher.

Additional data about students were gathered from school records. Individual student profiles were developed for all 26 students by compiling all teacher-provided information and other data sources about students into a case study report on each student.

A sub-sample of students was selected for more intensive investigation of their use of academic learning time. Although not a major focus in the present report, summary percentages of these time on task data are included in the student profiles where available.

Analysis

The teacher-expressed thoughts were analyzed for substance by conducting both a micro-analysis of interactive thoughts from the stimulated recall data and a macro-analysis of all thoughts using all interview data.

Ratio scores were generated for individual pupils for selected categories of verbal behavior exchanged with the teacher. These were used to indicate the relative involvement of the student in dyadic verbal exchanges with the teacher and the amount and type of these verbal contacts and they were combined with other students' data to compute average ratios of verbal interaction for the different grouping patterns under study.

Twenty-six individual profiles were constructed which stand on their own merits as units of analysis but data from these individual mini-case studies were combined into various grouping patterns for comparative purposes.

The attempt was made to triangulate the data sources on teacher exhibited behavior with teacher-expressed thoughts and teacher verbal interaction behavior in the bid to understand and describe how this teacher coped with environmental and student-imposed demands during teaching.

Conclusions

This case study was limited to one teacher's perceptions and behaviors as they occurred in one classroom. Generalization beyond the limits of the study itself is therefore precluded. Certain tentative results can be concluded concerning the extent of the operation of

expectancy effects in this particular classroom, this teacher's predominant teaching behaviors, and the nature of his expressed priorities and attitudes in working with his students.

Although precautions were taken, the obtrusive effects of the data collection may have disturbed the naturalistic environment to a certain degree although over the three month time period, the subjects appeared to eventually ignore the presence of the researcher and the video recording equipment for all practical purposes. Certain adjustments and changes in the teacher's approach during instruction were noted during the three month period, some of which were directly attributable to involvement in questioning and interviews with the researcher.

Conclusions reached in the study are presented as answers to the five research questions that guided the direction of the study and the data collection procedures.

1.0 What are the common characteristics of students assigned by the teacher to each of four attitude-to-student groups (attachment, concern, indifference, and rejection)?

The 16 <u>attachment</u> students in the present study were rated as highly attractive, intelligent, mature, cooperative and enthusiastic students for whom the teacher held high academic achievement expectancy. They were perceived to exert a consistently good effort in their academic work and they achieved exceedingly good results on the end of the year exams. A general positive halo effect, which seemed to be well deserved, was associated with a subgroup of the three initial choices from the attachment group.

The five indifference students were seen to be less mature,

less cooperative, as well as less inteiligent and attractive. They were generally younger than other classmates and received lower teacherassigned expectancy scores and effort attributions for academic work. They were not as well liked by the teacher and they received low scores for being noticeable and for maintaining eye contact with the teacher.

The two (conditional) <u>rejection</u> girls were attractive and noticeable. They were older than most classmates but were viewed by the teacher as immature and uncooperative and restless. They received low teacher-assigned scores for effort expended and were the recipients of low teacher expectancies for academic achievement. He was concerned about their academic progress. They were not well liked by the teacher and were seen to be basically unhappy children.

The three <u>concern</u> students exhibited poor initiative, motivation, and work habits and were assigned very low achievement expectation scores by the teacher who was most concerned about their academic progress. Two of the three students had additional emotional problems requiring considerable understanding, attention, and differential treatment from the teacher.

> 1.1 What were the perceived personality characteristics which might serve to explain the teacher-expressed attitudes and affective feelings toward these groups of students?

The effort attribution the teacher held for a student and his perceived general work habits seemed to color his feelings toward and expectancy held to a certain extent. The amount of cooperation and compliance afforded the teacher during instruction and the general classroom behavior of the student were also appreciated student attributes which seemed to affect the degree of teacher affect expressed toward individual students.

2.1 What evidence is there, if any, of <u>product</u> expectancy effects? What is the relationship of teacher-held expectancy to 'end of the year achievement results?

A sex difference was noted in the correlations of teacher-held expectancy and end of the year achievement results. Boys received higher teacher-assigned expectancy ratings for both subject areas of math and language arts and did achieve slightly higher scores on both subject area exams than did the girls in the study.

Expectations held for the four attitude-to-student groups were positively related to end of the year achievement marks. In language arts the highest expectancies were held for attachment students who performed well, particularly the first three choice attachment students. An almost equal expectancy rating was assigned for language arts achievement to both the indifference and conditional rejection students and indeed their average end of the year language arts achievement marks were nearly identical. The lowest expectancy for language arts achievement was held for members of the concern group who did least well on the end of the year language arts exams.

The same general pattern of correlations held for math expectancy and math achievement results with the attachment students, for whom the teacher held the highest achievement expectancy, performing best on the end of the year math exam and concern students, for whom he held the lowest math expectancy, performing least well of all attitude groups. In this subject area, indifference students' expectancy scores were higher than those assigned to rejection students and their math achievement reflected this teacher-assigned difference in expectancy.

It can be argued that prior student achievement accounts for the formation of teacher expectancy assigned just as easily as the theory that higher teacher expectancy held results in higher achievement. Causal relationships can not be said to exist in this study for the "preponderance of causation" (West and Anderson, 1976;613) was not determined. Certainly, positive correlations existed between the achievement expectancy held for the different attitude-to-student groups and their end of the year exam results and so it could be concluded that product expectancy effects appeared to exist with respect to the attitude to student groups in the study.

High, middle, and low expectancy levels for general achievement, specific language arts achievement, and specific math achievement were examined in relation to the end of the year achievement. It was found that higher positive correlations existed between the specific subject area expectations and achievement in the respective subject areas although the same positive correlative trend was apparent in the more global expectancy measure when compared with general achievement results.

The two reading groups showed definitive evidence of the same positive correlations for teacher-held expectancy and achievement in language arts.

In all, a fairly clear cut positive relationship existed between teacher-assigned expectancy scores (obtained in March) and the end of the year achievement test results. <u>Product</u> expectancy effects appeared to be in operation in the classroom.

2.2 What evidence is there, if any, of process expectancy effects? What is the relationship of teacher-expressed expectations to the amount and kind of verbal interaction exchanged?

The process expectancy effect measures used in this study were the amount and type of different verbal interactions exchanged between a student and the teacher.

Sex differences favored the boys who received both more quantity and better quality of verbal contacts from the teacher than the girls.

The four attitude to student groups' averages of total dyadic verbal interaction show differences favoring the concern students, as in previous research. Rejection students received the next highest amount of total verbal interactions but this could be attributable to the fact that one of these students was the new student settling in and more verbal contact was necessitated. The full group of attachment students received an average amount of verbal interaction but the first three choice attachment students from the larger group received a higher than average amount of verbal contacts. Indifference students received considerably less verbal contact from the teacher.

Generally speaking the total amount of verbal interaction received by the high, middle, and low expectancy groups favored the low group. This trend was especially so in math interactions but the situation was reversed during language arts instruction favoring the high expectancy group. It is notable that for both subject areas the middle expectancy groups received the least overall verbal attention.

The quality of interaction provided followed the same pattern for subject areas. The highs received more quality dyadic verbal interactions in language arts and the lows received more qualitative interaction in math. These ratios were computed for members assigned to specific subject area expectancy levels which differed in membership slightly. The ratios of verbal contacts received by the overall general achievement expectancy levels for all subject areas showed that more Product questions overall accrued to low expectancy students and more overall Process questions were addressed to high expectancy students. It could be tentatively concluded that there was evidence of 392

process expectancy effects in operation in this classroom.

3.0 Using all available information, to what extent is it possible to construct a comprehensive student profile which would describe the student's behavior, his achievement, his membership in various grouping patterns, and the relationship he had with this teacher?

The attempt was made to portray the 26 students in the classroom in quick character sketch style. Teacher-assigned numerical information was interwoven with teacher-provided statements about the student to reveal the most salient aspects of this particular student in the teacher's view. In addition, a somewhat interpretative summary followed each case study presentation. The researcher was satisfied that this was the most faithful and complete picture of each student that the teacher provided.

- 4.0 What information is obtainable from the interviews held with the teacher?
 - 4.1 Which kinds of thoughts about students in particular were reported during stimulated recall interviews?

It was possible to identify and quantify the type and quality of teacher interactive thoughts by subjecting the stimulated recall transcript data to a micro-analysis. Nine major categories, most of which were subdivided, comprised Conners' (1978) micro-analysis category system used in the present study. Most expressed interactive thoughts were classified as teacher Perceptions of student academic performance during instruction, and concern for the Instructional Moves he was making, particularly those concerned with Classroom Control, and as background Information about Students which the teacher brought to the lesson situation. A substantial proportion of interactive thoughts were Self-Monitoring in nature as the teacher expressed awareness of the effectiveness of his teaching behavior during ongoing instruction. He expressed personal Feelings about different aspects of the lesson in progress on many occasions as well.

4.3 To what extent do teacher-reported thoughts clarify/ corroborate numerical scores assigned on the different student ratings?

The researcher found that remarks about students tended to confirm and explain the scores assigned on the various student attribute ratings and teacher expectancy dimensions. Where discrepancies arose, the teacher was asked to clarify or confirm the numerical score assigned.

4.2 What teacher attitudes and priorities are révealed in all interview data?

A macro-analysis of all interview data was conducted in order to identify the most frequently expressed priorities and attitudes which appeared to guide this teacher in his approach to students. It appeared to the researcher that on the basis of the thoughts expressed during

interviews one could conclude that the following summarized set of values

were representative of those guiding his dealings with the

students in his class:

the development and maintenance of self concept in the student.

the development of a realistic view for the individual student (and for hig parents) of his strengths and weaknesses as a student.

394

the provision of support and encouragement to the student.

the importance of regarding each student as an individual who requires differential treatment and consideration while at the same time ensuring fairness and equality of treatment for all students in the class.

- 5.0 How could the teaching behavior of the teacher subject be described?
 - 5.1 What are the predominant teaching behaviors of this teacher?

This teacher was one of 60 teachers from selected grade 3 and and 6 classrooms in the system at large who participated in a process-product study entitled Project Quest. Observable teaching behaviors were recorded and rated. This teacher was well above the sample average on most of the 31 variables measured. On several of the variables, he received top scores consistently during the 10 hours of classroom observations conducted during language arts and math instruction. Based on these measures of this teacher's behavior the teacher was found to be highly proficient in classroom management skills. In particular, he was rated as highly supportive of students and accepting of student-initiated interaction, as using an appropriate mixture of high and low order questions , and as being skilled in handling disruptive situations in a low-key manner. He was given high ratings for monitoring students' work constantly and for possessing high degrees of the more global variables derived from Kounin (1970) such as warmth, empathy, clarity, overlappingness, withitness, persuasiveness, momentum, and smoothness.

5.2. What are the predominant verbal interaction behaviors of this teacher?

From the verbal interaction data collected by means of the dyadic interaction observation system (DIOS) instrument, it was apparent that this teacher accepted a high percentage of student-initiated interactions during actual teaching situations. Most of his teacher-initiated contacts were classified as either Product or Process academic question types. Teacher Comments to students accounted for a high percentage of total teaching interactions. The teacher appeared to use praise more frequently than teachers described in the literature and from the interview data collected it was clear that this was a purposive teaching act to encourage and support student academic progress.

IMPLICATIONS AND RECOMMENDATIONS

Approaches to Research Methodology

Adopting an integrative approach to data collection in educational research is a relatively new occurrence. More exacting studies in the past examined selected variables and the contributions they made to student achievement gains, but in themselves have not fully explained the complexity operant in a classroom. Team approaches have been used with a degree of success but several observers in a classroom run the risk of disturbing the naturalistic environment they wish to examine. Bennett (1978) proposed that studies should not concentrate only on the variables which have been shown to exert influences on instruction but also should look for the interaction among these variables which may reveal promising directions for further research. Mixing qualitative and quantitative data collection methods is not without problems but the results of this study indicate its potential for success in the search for meaning in a classroom. The triangulation of different data sources poses highly complex and time consuming analysis problems. Such studies tend to raise more questions than they immediately answer but the development of new questions to guide future research is not an unwelcome event. The conundrums inherent in adopting novel approaches to classroombased research are thought-provoking, and inventive researchers will rise " to meet the challenge posed. In such a pursuit open-ended studies are required which depart from the more traditional forms of educational research.

Focus on the Student

Increasingly classroom research has begun to focus on the individual student as the unit of analysis. Studies conducted on student effort attribution and motivation show these to be "alterable variables" (Bloom, 1980) which have a major effect on student achievement. Profitable use of academic learning time varies with the student and acts as a mediating variable affecting student achievement. This too is an alterable variable which the teacher can control by working more closely to keep the student attending to task and by breaking down the task itself into more manageable units. More studies of students are needed in order to provide a greater understanding of how self-concept

-5

and self attribution operate with children and to examine ways to alter these in more favorable directions.

One way of investigating student-perceived needs and attributions is by using introspective techniques such as stimulated recall to gain an understanding of which sorts of thoughts are salient for student development and maintenance of self concept. King found stimulated recall techniques suitable at the grade six level. Research is needed to determine whether younger students are able to understand and articulate the impressions gained from working with a teacher in a classroom situation.

Thought Processes

Investigating the mental activity of teachers is still a relatively new area of interest. Research on teacher thinking is predicated on the assumption that what teachers do is governed by what they think. Strictly observational studies of teacher behavior, taken at face value, are subject to misinterpretation without teacher-provided causal explanation for the actions taken. By conducting interviews with the teacher, and stimulated recall interviews seem particularly well suited to this purpose, valuable insights are gained about ways in which a particular teacher thinks, makes decision, solves problems and responds to pupils while teaching. By conducting more studies of this nature it is hoped that the commonalities in teacher thinking, thought processes, and decision making may begin to emerge.

Classroom Interaction

Classroom-based studies investigating classroom interaction have shown that students exert considerable influence on teacher behavior. The extent of mutual influence is dependent on many factors not well understood as yet. Teacher characteristics account for the susceptibility of becoming unduly influenced by students but the particular characteristics of the students themselves are not without considerable influence. The Student Attribute Study, conducted by Brophy et al., 1976, revealed that certain student attributes seemed to affect many teachers in predictable ways. This study offered additional evidence. More studies of student characteristics and the ways in which teachers react to them are needed, however, to extend these findings. By knowing of these trends teachers could guard against exhibiting inappropriate behavior toward students.

Verbal Interaction. Many studies use verbal interaction as the mediating process variable. The examination of the verbal interaction on its own merits reveals the types of questions used by the teacher and can provide the teacher with data on self performance about questioning techniques and feedback procedures. This is one of the alterable variables which can be used to effect better instruction. A related purpose of recording verbal interaction is to identify which class members receive the more qualitative types of verbal interaction and whether the teacher affords all children fairly equal opportunities for verbal contacts with the teacher. However, more subtle communications such as tone of voice, wait time afforded the answering student, and non-verbal gestures (as pointed out by Muttart, 1977:131) are not easily captured on existing low inference measures. Data collection instruments which will capture these subtleties need to be developed to gain the information communicated by these means as well as by the more obvious verbal content.

Expectancy Research

Many studies of teacher expectancy use verbal interaction patterns as evidence of differential treatment provided to students by virtue of the quantity and quality of interaction provided. More research is now needed to find the reasons for the formation and maintenance of teacher expectancy for students. The susceptibility of exhibiting expectancy effects, which does not appear to be universal with all teachers, depends in part on the degree of teacher awareness of the phenomenon itself. Teacher education programs should prepare prospective teachers by acquainting them with the processes of expectancy, thereby sensitizing them to the negative effects that may occur.

<u>Teacher Attitudes to Students</u>. Attitudes to students function as affective expectancy. Conclusive results from previous research have shown that the teacher's affective reaction to students affects treatment of students. In striving to prepare proactive teachers, teacher education programs should present and discuss the findings to date concerning the influence of teacher-held attitudes to students. More research is needed to determine how teacher affect influences teacher behavior with particular students.

Effects of Participating in Introspective Research

Self-confrontation is a threatening and stressful experience unless the teacher is self confident. The technique should be used with a degree of caution. Teachers involved in this form of research find it time consuming, but not completely without benefit for it forces them to consider ways of thinking about what they are doing. If changes are warranted they are made after reflection about alternative ways of doing things. Teacher educators might use the technique particularly during formative # evaluations of student teachers, and administrators concerned with the supervision of teachers in the system should consider using it in the context of a non-evaluative, clinical supervision type of setting where the teacher is provided with information about his behavior. Care in establishing a suitable rapport cannot be overstated before embarking on stimulated recall experiences with teachers in order for it to be a useful experience for either party involved in the research. The extent to which, the teacher should be acquainted with the purpose of the introspective research is open to question. Cooperative subjects attempt to provide what is requested, while at the same time for the researcher a focus is necessary in order to limit the range of thoughts considered to be a manageable number for discussion or analysis.

Reflections on Using an Eclectic Approach

Much more research is required in many of the areas isolated for examination in the present study. The treatment of each area was purposely kept minimal in the attempt to determine how many disparate areas could be considered simultaneously. More areas than have been reported here were examined. One becomes quickly humbled by the complexity of life in the classroom. It is apparent that many more researchers must engage in similar studies and that many more hypothesis-generating studies are required before developing the underlying theories which might explain the interacting variables which account for learning gains and effective teaching. It whets the appetite for more of the same. While a certain amount of preplanning is necessary, keeping the approach and methodology open allows for serendipitous findings. It keeps the 'researcher sensitive to the 'subjects under study and to the varied methodologies needed for research in education.

REFERENCES

	Adame R S	and B.J.Biddle
2	1970	<u>Realities of Teaching: Explorations with Video Tape</u> .
. ,		New York: Holt, Rinehart, and Winston,
	6	
•	Adams, G.R.	and J.C. Lavoie
· -	1974	"The Effect of Student's Sex, Conduct, and Facial ,
		Attractiveness on Teacher Expectancy." <u>Education</u> , 95(1):
		76-83.
	`A	
	1978	R. and R.L.Debus
	1970	"Persistence and the Causal Perception of Failure Modifying Cognitive Attributions." <u>Journal of Educational Psychology</u> ,
		70(2):154-166.
	Bar-Tal, D.	
	1978	"Attributional Analysis of Achievement-Related Behavior."
		Review of Educational Research, 48(2):259-271.
	Beez, V.W.	
	1968	Influences of Biased Psychological Reports on Teacher
		Behavior. Unpublished manuscript, Indiana University.
	Bennett, S.N	
	1978	"Recent Research on Teaching: A Dream, a Belief, and a
	1,0	Model." <u>British Journal of Educational Psychology</u> , 48:
		127-147.
	Berliner, D.	
	1976	"Impediments to the Study of Teacher Effectiveness."
		Journal of Teacher Education, 27:5-13
	Bloom, B.S. 1953	"Thought Desserves de Lostures and Differents II must take 1
	1933	"Thought Processes in Lectures and Discussions." <u>The Journal</u> of General Education, 7(3):160-169.
		<u>or General Education</u> , 7(3):100-109.
	Bloom, B.S.	a da ser en entre de la companya de La companya de la comp
	1974	"Time and Learning." American Psychologist, 29(9):682-688.
	Bloom, B.S.	
	1980	"The New Direction in Educational Research: Alterable
		Variables." Phi Delta Kappan, 61(6):382-386,
25		
	Bowles, F.D.	Upandadan bella da Tananadan Ultrana a comatan Di
	1973	"Decision Making in Instruction." <u>Journal of Teacher Education</u> , 24(1),38-40
		24(1):38-40.
ė.	Braun, Carl	
	1976	"Teacher Expectation: Sociopsychological Dynamics." Review of
		Educational Research, 46(2):185-213.
•		

- Brophy, J.E. and C.M. Evertson 1973 Low-Inference Observation Coding Measures and Teacher Effectiveness, ERIC Microfiche Ed 077 879. Brophy, J.E. and Evertson, Anderson, Baum, Crawford. 1976 The Student Attribute Study, Austin Texas. ERIC documents ED 121 799. Brophy, J.E. and T.L. Good 1969 Teacher-Child Dyadic Interaction: A Manual for Coding Classroom Behavior. Report Series No.27, Texas University, Austin, Research and Development Centre for Teacher Education. Brophy, J.E. and T.L. Good "Teachers' Communication of Differential Expectations for 1970 Children's Classroom Performance." Journal of Educational Psychology, 61(5):365-374. Brophy, J.E. and T.L. Good 1974 Teacher-Student Relationships: Causes and Consequences. New York: Holt, Rinehart, and Winston. Clark, C. 1978 Choice of a Model for Research on Teacher Thinking. Paper presented at A.E.R.A., 1978. Clark, C. and P. Peterson 1976 Teacher Stimulated Recall of Interactive Decisions. Paper presented at A.E.R.A., 1976. Clark, C. and R.J. Yinger "Research on Teacher Thinking." Curriculum Inquiry, 7(4): 1977 279-304. Conners, R. An Analysis of Teacher Thought Processes, Beliefs, and 1978 Principles, during Instruction. Doctoral dissertation, University of Alberta. Cooper, Harris 1977 "Controlling Personal Rewards: Professional Teachers'
 - Differential Use of Feedback and the Effects of Feedback on the Student's Motivation to Perform." Journal of Educational Psychology, 69(4):419-427.

Cooper, N. Colleen

1979 <u>Information Processing by Teachers and Pupils During</u> <u>Mathematics Instruction</u>. Doctoral dissertation, University of Alberta.

Ċ

Cornbleth, 1974	C., O.L.Davis, and C. Button "Expectations for Pupil Achievement and Teacher-Pupil Interaction." <u>Social Education</u> , 38(1):54-58.
Crano, W. a 1978	nd P.M. Mellon "Causal Influences of Teachers' Expectations on Children's Academic Performance." <u>Journal of Educational Psychology</u> , 70(1):39-49.
de Charms, 1 1976	R. <u>Enhancing Motivation</u> . New York: Irvington Press.
Denton, D. 1979	What Elements of Phenomenology Pertain to Educational Research. Paper given at A.E.R.A., 1979.
Dunkin, M.J. 1974	and B.J. Biddle <u>The Study of Teaching</u> . New York: Holt, Rinehart, and Winston.
Dusek, J.B. 1973	and E.J. O'Connell "Teacher Expectancy Effects in the Achievement Test Performance of Elementary School Children." <u>Journal of</u> <u>Educational Psychology</u> , 65:371-377.
Dusek, J.B. 1975	"Do Teachers Bias Children's Learning?" <u>Review of Educational</u> <u>Research</u> , 45(4):661-684.
Eggert, W. 1977 Eisner, E.W.	Teacher Process Variables and Pupil Products. Doctoral dissertation, The University of Alberta.
1976	"Educational Connoisseurship and Educational Criticism: Their Form and Function in Educational Evaluation." <u>Journal of</u> <u>Aesthetic Education</u> , 10 (3-4):135-150.
Eisner, E.W. 1977	"On the Uses of Educational Connoisseurship and Criticism for Evaluating Classroom Life." in <u>The Design and Evaluation of</u> <u>Educational Programs</u> . New York: Macmillan.
1970	. and M.A.Webster <u>Raising Children's Expectations for Their Own Performance</u> . Centre for the Study of Social Organization of School Reports, John Hopkins University, Maryland.
Fasano, J.H. 1977	Pupil Characteristics and Teacher-Pupil Dyadic Interaction. Doctoral dissertation, University of Alberta.

•

'Fiedler, Martha, L. 1975 "Bidirectionality of Influence in Classroom Interaction." Journal of Educational Psychology, 67(6):735-744. Finn, J.D. 1972 "Expectations and the Educational Environment." Review of Educational Research, 42(3):387-410. Firestone, G. and N. Broudy 1975 "Longitudinal Investigation of Teacher-Student Interactions and Their Relationship to Academic Performance." Journal of Educational Psychology, 67(4):544-550. Fuller, F.F. and B.A. Manning "Self Confrontation Reviewed: A Conceptualization for Video 1973 Playback in Teacher Education." Review of Educational <u>Research</u>, 43(4): 469-528. Gage, N.L. 1978 "The Yield of Research on Teaching." Phi Delta Kappan, 60(3): 229-235. Gaier, E.L. 1954 "Memory Under Conditions of Stimulated Recall." Journal of Genetic Psychology, 50:147-153. Good, Thomas 1970 "Which Pupils Do Teachers Call On?" Elementary School Journal, 70(4):190-198. Good, T.L. and D.A. Grouws "Teaching Effectiveness in Fourth-Grade Mathematics 1977 Classrooms." in G.D. Borich (Ed.) The Appraisal of Teaching: Concepts and Process. Reading, Mass.: Addison-Wesley. Guetzkow, Harold "Unitizing and Categorizing: Problems in Coding Qualitative 1950 Data." Journal of Clinical Psychology, 6:47-57. Haigh, N.J. 1974 "Teacher Expectancy: Accuracy, Change, and Communication." New Zealand Journal of Educational Studies, 9(2):97-112. Harré, R. and P.F. Secord The Explanation of Social Behavior. Oxford: Basil Blackwell, 1972 Holsti, O.R. 1969 Content Analysis for the Social Sciences and Humanities. Reading, Mass.: Addison-Wesley.

Jackson, P.

5

,

3

Jackson, P.	
1968	Life in Classrooms. New York: Holt, Rinehart, and Winston.
Jeter, J.T.	
1972	Elementary Social Studies Teachers' Differential Classroom
	Interaction with Children as a Function of Differential
	Expectations of Pupil Achievement. Dissertation Abstracts
	International, 33(4):680.
Jick, Tood.	D.
1979	"Mixing Qualitative and Quantitative Methods: Triangulation in Action." <u>Administrative Science Quarterly</u> , 24:602-611.
. T	이 그 사람은 것 같은 것
	and R.E. Nisbett
1972	"The Actor and the Observer: Divergent Perception of the
	Causes of Behavior." in E. Jones, D. Kanouse, H, Kelley,
•	R. Nisbett, S. Valins, and B. Weiner, (Eds.) <u>Attribution:</u>
	Perceiving the Causes of Behavior. Morristown, N.J.: General Learning Press.
e e e e e e e e e e e e e e e e e e e	General Beating riess.
Kagan, N. ar	nd D.R.Krathwohl, and R. Miller
1963	"Stimulated Recall in Therapy Using Video Tape: A Case Study,"
	Journal of Counseiling Psychology, 10(3):237-243.
Kennedy, Mar	$\sim M$
1978	
	<u>Generalizing from Single Case Studies</u> . Paper delivered at A.E.R.A., Toronto, 1978.
Kester, S.W.	and G. A. Letchworth
1972	"Communication of Teacher Expectations and Their Effects on
	Achievement and Attitudes of Secondary School Students."
	The Journal of Educational Research, 66(2):51-55.
King, Leonar	
1979	An Attributional Analysis of Student Achievement-Related
	Behavior and the Expectancy Effect. Doctoral dissertation.
	University of Alberta.
Kounin, J.S.	
1970	Discipline and Group Management in Classes we want
	Discipline and Group Management in Classrooms. New York: Holt, Rinehart, and Winston.
Kulka, A.	
1972	"Attributional Determinants of Achievement-Related Behavior."
	Journal of Personality and Social Psychology, 21(2):166-174.
Lerner, R.M.	
1969	"The Development of Stereotyped Expectancies of Body Build:
	Behavior Relations." Child Development, 40(1):137-141.
	비행님 그는 사람들은 것 같은 것 같은 것 같아요. 것 같아요. 가지 않는 것 같아요. 것 같아.

÷.



Lockheed, Marliane E.

1976 "Some Determinants and Consequences of Teacher Expectations Concerning Pupil Performance." <u>Beginning Teacher Evaluation</u> <u>Study, Phase II</u>, 1973-1974. National Institute of Education.

Luce, S.R. and R.D. Hoge

1978 Relations Among Teacher Rankings: Pupil-Teacher Interactions, and Academic Achievement: A Test of the Teacher Expectancy Hypothesis." <u>American Educational Research Journal</u>, 15(4): 489-500.

MacDonald, Barry and R. Walker

"The Case Study and the Social Pholosophy of Educational Research." <u>Cambridge Journal of Education</u>, 5(1):2-11.

MacKay, D.A.

1975

1978 Observational Studies of Teaching: A Progress Report. Paper presented at A.E.R.A., Toronto, 1978.

MacKay, D.A. 1979

Project Quest: Teaching Strategies and Pupil Achievement. Occasional Paper Series, Report #79-1-3, Centre for Research in Teaching, The University of Alberta.

MacKay, D.A. and P. Marland 1978 <u>Thought Processes of Teachers</u>. Paper presented at A.E.R.A., Toronto, 1978.

MacKay, D.A. and E.C. Osoba

irections in Teacher Supervision." The Canadian trator, 18(3):1-5.

Magoon,

19

1977

tructivist Approaches in Educational Research." <u>Review</u> ucational Research, 47(4):651-693.

Mahen, R.G. 1977

lationships Between Students' Classroom Behaviors and lected Context, Process, and Product Variables. ctoral dissertation. The University of Alberta.

Marland, P. 1977

Study of Teachers' Interactive Thoughts, Doctoral dissertation, The University of Alberta.

Marliave, R.S. 1976 OI

Observable Classroom Variables, Beginning Teacher Evaluation Study. Far West Laboratory for Educational Research and Development, Technical Report 1-2. San Fransisco.

Marliave, R.S., C.W. Fisher, and M.M. Dishaw

1977 <u>Academic Learning Time and Student Achievement in the</u> <u>Beginning Teacher Evaluation Study</u>. Far West Laboratory. San Fransisco. Marx, R.W. and P.L. Peterson

1975 <u>The Nature of Teacher Decision Making</u>. Paper presented at A.E.R.A., Washington, 1975.

McDonald, Frederick J.

1976 <u>Teachers Do Make a Difference</u>. Educational Testing Service, Princeton, New Jersey.

Miles, Matthew M.

1979 "Qualitative Data as Attractive Nuisance: The Problem of Analysis." <u>Administrative Science Quarterly</u>, 24:602-612.

Moody, Peter

1980 <u>Descriptive Studies of Six Children in Their Classroom</u> <u>Contexts</u>. Doctoral dissertation, The University of Alberta.

Muttart, D.G.

1977 <u>A Study of Teacher-Pupil Interaction as it Relates to</u> <u>Differential Teacher Expectations and Selected Teacher</u> <u>Characteristics</u>. Doctoral dissertation, The University of Alberta.

Nash, Roy

V.

1975

1975

1976 <u>Teacher Expectations and Pupil Learning</u>. London: Routledge and Kegan Paul.

National Institute of Education

Teaching as Clinical Information Processing. National Conference on Studies in Teaching. <u>Panel No. 6 Report</u>. Washington, D.C.

Oberg, Antoinette A.

Information Referents and Patterns in the Curriculum <u>Planning of Classroom Teachers</u>. Doctoral dissertation, The University of Alberta.

Palardy, J.M. 1968

- The Effect of Teachers' Beliefs on the Achievement in Reading of First Grade Boys. Doctoral dissertation, Ohio State.
- Radford, J.

1971

1974 "Reflections on Introspection." <u>American Psychologist</u>, 29(4):245-250.

Rogenshine, B. and N. Furst

"Research on Teacher Performance Criteria." In B.O. Smith (ed.) <u>Research in Teacher Education</u>. New Jersey: Prentice-Hall.

408

Rosenshine, B. and N. Furst

1973 "The Use of Direct Observation to Study Teaching." in R.M. Travers (ed.) <u>Second Handbook, of Research on Teaching</u>. Chicago:Rand McNally.

Rosenthal, R. and L. Jacobson

1968 <u>Pygmalion in the Classroom</u>. New York: Holt, Rinehart, and Winston.

Rosenthal, Robert 1973 "The Pygmalion Effect Lives." <u>Psychology Today</u>, 7(4):56-59.

Rothbart, M., S. Calfen, and R. Barrett 1971 "Effects of teacher's Expectancy on Student-Teacher Interaction." <u>Journal of Educational Psychology</u>, 62(1): 49-54.

Robovits, P. and M.R.Maehre

1971 "Pygmalion Analyzed: Toward an Explanation of the Rosenthal-Jacobson Findings." <u>Journal of Personality and Social</u> <u>Psychology</u>, 19(2):197-203.

Russo, (Atwood) Nancy

1978 <u>Capturing Teachers' Decision Policies: An Investigation of</u> <u>Strategies for Teaching Reading and Mathematics</u>. Paper presented at A.E.R.A., Toronto, 1978.

Ryans, D.G.

1963 "Teacher Behavior Theory and Research: Implications for Teacher Education." Journal of Teacher Education, 14(3): 274-293.

Santrack, J.W. and R. L. Tracy

1978 "Effects of Children's Family Structure Status on the Development of Stereotypes by Teachers." <u>Journal of</u> <u>Educational Psychology</u>, 70(5):754-757.

Seaver, W.B.

1973 "Effects of Naturally Induced Teacher Expectancies." Journal of Personality and Social Psychology, 28(3): 333-342.

Shavelson, Richard 1973 "What is the Basic Teaching Skill?" <u>Journal of Teacher</u> / Education, 14:144-151.

Shavelson, Richard

1976 "Teachers' Decision-Making." In N.L. Gage (ed.) <u>The</u> <u>Psychology of Teaching Methods</u>. Yearbook of the National Society for the Study of Education. Chicage: University of Chicago Press. NSSE Yearbook.

0

đ		n5-4)
	Shavelson,	Richard
	1978	A Model of Teacher Decision Making, Paper presented at
,	2070	A.E.R.A. 1978.
		A.D.A.A. 1970.
	Shavelson	R. and N. Atwood
	1977	
	1977	"Teachers' Estimates of Student States of Mind." British
	· ·	Journal of Teacher Education, 3(2):131-137.
	Shaveleen	
	1977	R., N. Atwood, and H. Borko
	1977	"Experiments on Some Factors Contributing to Teachers'
	s.	Pedagogical Decisions." <u>Cambridge Journal of Education</u> ,
		7(1):51-70.
	Chulman I	
		S. and A.S. Elstein
	1975	"Studies of Problem Solving, Judgment, and Decision Making:
		Implications for Educational Research." In F.N. Kerlinger
		(ed.) <u>Review of Research in Education</u> . Vol. 3. Itasca,
		Illinios: F.E. Peacock Co.
	Cd 1h annual a	
	Silberman, N	
	1969	"Behavioral Expression of Teachers' Attitudes Toward
	•	Elementary School Students." Journal of Educational
		Psychology, 60(5):402-407.
	Cd	
		d E.G. Boyer (eds.)
	1974 -	Mirrors for Behavior III: An Anthology of Observation
1		Instruments. Philadelphia, Pa.: Research for Better Schools.
	Semith E L	
•	1976	and J.E.R. Luginbuhl
	1970	"Inspecting Expectancy:Some Laboratory Results of Relevance
	0	for Teacher Training." <u>Journal of Educational Psychology</u> , 68:265-272.
	2	00:203-272.
	Smith Mutan	a de la construcción de la constru A de la construcción de la construcc
	Smith, Murra 1977	
	1977	Attribution, Achievement Motivation, and Task Difficulty in
	· · · ·	Physical Education. Paper presented at 3rd Annual B.C.
		Conference on Physical Education, Victoria, British Columbia.
	Smyth, W. Jo	hn
	1979	
	19/9	An Ecological Analysis of Pupil Use of Academic Learning
		Time. Doctoral dissertation, The University of Alberta.
	Spaulding,	
	1973	The Contro Analysia Cababata Car Di anti i a site data
	, 12/2	The Coping Analysis Schedule for Educational Settings (CASES).
		Paper presented at A.E.R.A. New Orleans, 1973.
		ERIC documents ED 076 694
	Staffard 7	Polont
	Stafferi, J.	
	1967	"A Study of Social Stereotypes of Body Image in Children."
		Journal of Personality#and Social Psychology, 7(1):101-104.
	1. 4. 114	

a.

6

٠

3

Fi.

<...е

.

 \mathbb{P}^{2}

Stake, Robert E. 1976 "To Evaluate an Arts Program." Journal of Aesthetic <u>Education</u>, 10 (3-4):115-133. Stake, Robert E. 1978 "The Case Study Method in Social Inquiry." Educational Researcher, 5-8 Stallings, J.A. 1978 The Development of the Contextual Observation System. Paper presented at A.E.R.A., Joronto, 1978. Tuckwell, Neil B. 1980 A Study of the Impact of an Intervention Program on Teacher Thought Processes. Doctoral dissertation, The University of Alberta. van Manen, Max 1978 A Phenomenological Experiment in Educational Theory: The Utrecht School. Paper presented at A.E.R.A., Toronto, 1978. Webb, E.J., D.T. Campbell, R. Schwartz, and L. Sechrest 1966 Unobtrusive Measures: Non-Reactive Research in the Social Sciences. Chicago: Rand McNally. Weiner, B. 1972 "Attribution Theory, Achievement Motivation, and the Educational Process." Review of Educational Research. 42(2):203-216. Weiner, B. 1976 "An Attributional Approach for Educational Psychology." In L. Shulman (ed.) <u>Review of Research in Education</u>. (Vol 4). Itasca, Ill.,: F.E. Peacock. Weiner, Bernard "A Theory of Motivation for Some Classroom Experiences." 1979 Journal of Educational Psychology, 71(1):3-25. West, Charles and T.H. Anderson 1976 "The Question of Preponderant Causation in Teacher Expectancy Research." Review of Educational Research. 46(4):613-630. Willis, Sherry and J. Brophy "Origins of Teachers' Attitudes Towards Young Children." 1974 Journal of Educational Psychology, 66(4):520-529.

Wilson, Stephen

1977 "The Use of Ethnographic Techniques in Educational Research." <u>Review of Educational Research</u>, 47(1):245-265.

Winne, Philip H. and R.W. Marx

1977 "Reconceptualizing Research on Teaching." <u>Journal of</u> Educational <u>Psychology</u>, 69:668-678.

Withall, John

1949. "The Development of a Technique for the Measurement of Social-Emotional Climate in Classrooms." <u>Journal of</u> <u>Experimental Education</u>, 17(3):347-361.

Wodlinger, Michael

1980 <u>A Study of Teacher Interactive Decision Making</u>. Doctoral project in progress, The University of Alberta.

Yinger, Robert J.

1978 <u>Fieldwork as a Basis for Theory Building in Research on</u> <u>Teaching</u>. Paper presented at A.E.R.A., Toronto, 1978.

Yoshida, Roland K. and C.E. Meyers

1975 "Effects of Labelling as Educable Mentally Retarded on Teachers' Expectancies for Change in a Student's Performance." Journal of Educational Psychology, 67(4): 521-527.

Zahoric, J.A.

1975 "Teachers' Planning Models." <u>Educational Leadership</u>, 33(2):134-139.



Copies of the various instruments used in the study are contained in this appendix. Since these instruments originated in other studies and were borrowed and used, the reader is referred to the sources cited for particulars concerning the use and application of each instrument. We 3

413

Conners' (1978) Micro Content Analysis System appears only as a summarized list of categories (Page 315) in the present study.

Guidelines for conducting stimulated recall interviews may be found in King's (1979) Appendix B and in Tuckwell's (1980) Appendix E. Adapted from Brophy and Good, 1969; and from Brophy and Evertson, 1973.

ġ

TIME DATE Succt L'ON STATE 36.65 ŝį 1H va-1 11 30 nutil Т IN TERACTION I ALSPERED -W7050 ave-ewut h WZ-Wh an 1er TEVENCE TEM SOF ANNEL -2-WWYELW CUUWQH · S Σ STUDENT-INITIATED П amano San INTE PALION Í -66A1A763 5 3088 & AM 80107452 ð 4503300 11 THE esont-os SERVATION 100 1130 THE REAL 19-Jehu Π 2004-0 Th Í. Įï. T 320 1 ŀ Ŵ STUDENT 201754905 1ŝ H Vess OSF 08 505 204 agu weres Ansider ----4050 112 C710N UP-L-UUNU 11 TO STUDIAT i 428-64 ++ π Π SUBTAIN NEP SU Per 1: 583 82 11 ۲ TEARINAL SUNUL TEA H 402 0 U13 7/01 Т -050EN ----4-78 440 2 U V REOULIN SUDELU TER CHRIFE DYADIC ACO LYSU Ś 8 1 SO SAHONA QUALITY O 14 2024 4505 36.050 1 A ACT.IN 41 INITIATE UNCON П 544 446 Ŀ QUESTIONS LEUD SOS 1 1 not the U20-UW 3 CTOUWNA 01 1 12800301 l 11 wso 4 TEACHER 11 A 505 MARANT 11 4. 44 4 63 4 4 Ľ ave. THE Π *** isusse erer whole MODE ---stow wase -24-2-422

Low		
		N 18
Careless, hasty	worker versus careful, delik	berate worker.
	° ~ ~ 0	HIGA
Unhappy versus h	happy.	
wص	NAD	h i s
• .	achiever versus probable hig	
ow	HED	+ HIGH
Immature versus	mature.	
~	MED	wie A
Uncooperative, d	efiant versus cooperative, c	ompliant.
ow	мгр	HIGH
and a second		
~		Hich
₽₩	мго Sus attractive.	
ow Jnattractive ver: ow	мго Sus attractive.	нісн, нісн, tries hard, persis
	MFD Sus attractive. MFD	нісн Міся
ow Jnattractive ver ow Gives up easily, ow Jould like to hav	MFD Sus attractive. MFD needs to be prodded versus MFD ve removed from class versus	HIGH HIGH tries hard, persis Work HIGH
ow Jnattractive ver ow Sives up easily, ow Jould like to hav	MFD Sus attractive. MFD needs to be prodded versus MED	HIGH HIGH tries hard, persis Work HIGH
w Jnattractive ver w Gives up easily, w Jould like to hav another year for bw weasn't require a	MFD Sus attractive. MFD needs to be prodded versus MFD ve removed from class versus r the sheer joy of it. MFD special attention versus con-	HIGH HIGH tries hard, persis work Would like to kee HIGH cerns me a great d
w Jnattractive ver ow Sives up easily, w Jould like to hav another year for ow boesn't require a rould like to be	MFD Sus attractive. MFD needs to be prodded versus MFD ve removed from class versus r the sheer joy of it.	HIGH HIGH tries hard, persis work Would like to kee HIGH cerns me a great d
Just tractive ver Just tractive ver Sives up easily, Would like to hav another year for bow Hoesn't require a rould like to be	MFD Sus attractive. MFD needs to be prodded versus MFD ve removed from class versus r the sheer joy of it. MFD special attention versus con- able to devote much more at	HIGH HIGH tries hard, persis work Would like to kee HIGH cerns me a great d tention.
Just ractive ver Just ractive ver Sives up easily, W Jould like to hav another year for bow Doesn't require a rould like to be w Jot noticeable ver	MFD Sus attractive. MFD needs to be prodded versus MFD ve removed from class versus r the sheer joy of it. MED special attention versus con- able to devote much more at MED ersus stands out, very notice	HIGH HIGH tries hard, persis work Would like to kee HIGH tention. NIGH tention.
Jnattractive ver Jnattractive ver Sives up easily, Would like to hav another year for bow Hoesn't require a rould like to be Mot noticeable ve	MFD Sus attractive. MFD needs to be prodded versus MFD ve removed from class versus r the sheer joy of it. MFD special attention versus con able to devote much more at MED ersus stands out, very notice MED	HIGH HIGH tries hard, persis work Would like to kee HIGH cerns me a great d tention.
Just tractive ver Just tractive ver Sives up easily, We Hould like to hav another year for ow Hould like to be we hould like to be we hould like to be	MFD Sus attractive. MFD needs to be prodded versus MFD ve removed from class versus r the sheer joy of it. MED special attention versus con- able to devote much more at MED ersus stands out, very notice	HIGH HIGH tries hard, persis work Would like to kee HIGH tention. NIGH tention.
w Inattractive ver w ives up easily, w lould like to hav another year for w oesn't require to ould like to be ould like to be	MFD Sus attractive. MFD needs to be prodded versus MFD ve removed from class versus r the sheer joy of it. MFD special attention versus con able to devote much more at MED ersus stands out, very notice MED	HIGH HIGH tries hard, persis work Would like to kee HIGH tention. NIGH tention.

From The Student Attribute Study, by Brophy, Evertson, Anderson, Baum, and Crawford, 1976, Page 471.

ATTITUDE-TO-STUDENT QUESTIONNAIRE.

Directions given by Silberman, 1969, and by Willis and Brophy, 1974 were to nominate one child/ three children (respectively) to each of the four attitude to student groups. The researcher in the present study used the same definitions but asked the teacher to place all 26 students in his class into one of the four attitude to student groups. The descriptions below were the ones provided to the teacher in order to elicit the students' names. The teacher filled out the form himself in the presence of the researcher.

ATTACHMENT

If you could keep one student for another year for the sheer joy of it, whom would you pick?

INDIFFERENCE

If a parent were to drop in unannounced for a conference, whose child would you be least prepared to talk about?

CONCERN

1

If you could devote all your attention to a child who concerned you a great deal, whom would you pick?

REJECTION

If your class was to be reduced by one child, whom would you be relieved to have removed?

INSTRUCTIONS FOR FILLING OUT THE 7-POINT RANKING SCALE from

Brophy, Evertson, Anderson, Baum, and Crawford, 1976, <u>The Student Attribute Study</u>, Page 469.

"For each scale, select your three highest and three lowest students and put their names in the highest and lowest sections of the scale. Then " select your next four highest and lowest students and put their names in the next highest and lowest sections of the scale. Do the same thing for the next highest and lowest section. You may leave some blanks in this section if you feel that you do not have enough students whose names belong here. Put all remaining students' names in the middle section. The students in each section of the scale will be considered to have the <u>same</u> ranking, so do not spend too much time making fine discriminations between students."

For some analysis purposes the researcher in the present study combined scores of 1 and 2 into HIGH, scores of 3, 4, and 5 into MIDDLE, and scores of 6 and 7 into LOW rankings on the scales.

Six class ranking scales were completed in all, by the teacher, in the present study:

General Intellectual Ability Probable Achievement in Language Arts Probable Achievement in Mathematics Motivation to Do School work Maturity Cooperation/Compliance

LOW

HIGH

ł

PROJECT QUEST

A joint project of the Centre for Research in Teaching and the Edmonton Public School Board, Winter 1978-1979.

HIGH INFERENCE CODING SHEET

1. Teacher used a system of rules dealing with personal and procedural matters.
1_{1} 2_{1} 3_{1} 4_{1} 5_{1}
LOW MED HIGH NOT OBSERVED
2. Teacher prevented misbehaviors from continuing.
3. Teacher directed disciplinary action accurately.
4. Teacher moved around the room a lot (monitoring seatwork).
5. Teacher handled disruptive situations in a low-key manner (non-verbal, proximity, eye contact).
6. Teacher insured assignments were interesting and worthwhile while the children worked independently.
7. Teacher used a system of rules which allowed pupils to carry out learning tasks with a minimum of direction.
8. Teacher optimized academic learning time. Pupils were actively involved and productively engaged in learning tasks.
9. Teacher used a standard signal to get students' attention.
10. Teacher did not begin speaking to the group until all students were paying attention.
11. Teacher stopped speaking/instructing until all students were paying attention.
12. Teacher used a variety of instructional techniques adapting instruction to meet learning needs.
13. Teacher encouraged quantity and quality of work in Language Arts.
4. Teacher used a system of spot-checking assignments.
5. (Do not complete until last minute.) Teacher related math games and independent activities to the concepts being taught.

from Project Quest...continued

- 16. (Do not complete until last visit.) Teacher increased the amount of written work so that pupils did some writing every day.
- 17. Teacher used techniques that provide for the gradual transition from concrete to more abstract activities.
- 18. Teacher used an appropriate mixture of high and low order questions.
- 19. Teacher was aware of what was going on in the classroom.
- 20. Teacher was able to attend to more than one issue at a time.
- 21. Teacher facilitated the smooth flow of the lesson or a smooth transition from one activity to another.
- 22. Teacher's behavior maintained the pace of the lesson.
- 23. Teacher was clear in presentations to the class.
- 24. Teacher was able to motivate children.
- 25. Teacher provided evidence of "caring", "accepting", and "valuing" of the children.
- 26. Teacher responded accurately to both obvious and less obvious meanings, feelings, and experiences of the children.
- NOTE: For the following items, please use the five point scale to indicate how <u>frequently</u> the behaviors occurred during the period observed.
- 27. Many different pupils were selected by the teacher to respond to questions.
- 28. When pupil's answers were incorrect or only partially correct, the teacher used techniques such as rephrasing, giving clues, or asking a new question to help the pupil to give an improved response.
- 29. The teacher used praise to reward outstanding work as well as to encourage pupils who were not always able to do outstanding work.
- 30. The teacher used mild criticism on occasion to communicate expectations to more able pupils.
- 31. When pupils initiated interaction, the teacher accepted and integrated the pupil question, comment, or other contribution.